report. hw 734052, 1995-03. RD/RA work plan. pdf

_

WORK PLAN

Remedial Design/Remedial Action

Former Accurate Die Casting Site Fayetteville, New York

March 1995



Work Plan

Remedial Design/Remedial Action

Former Accurate Die Casting Site Fayetteville, New York



David G. Van Arnam Senior Vice President

March 1995



5000 Brittonfield Parkway East Syracuse, New York 13057 Contents

1. Introduction	1
1.1. General	1
1.2. Background	1
1.3. NYSDEC selected remedial alternative	4
1.4. Purpose of the RD/RA work plan	5
2. Pre-design activities	7
2.1. General	7
2.2. PCB/PAH/VOC soils area	7
2.3. Septic tank sludge	7
2.4. Bedrock ground water	8
2.4.1. Data review/recovery well design	8
2.4.2. Test bedrock recovery well installation	9
2.4.3. Bedrock hydrogeologic evaluation	9
2.4.4. Ground water treatment system evaluation	10
2.4.5. Basis of Design Report	10
2.5. TCE soils excavation verification sampling	10
2.6. Long-term ground water monitoring program	11
2.7. Health and safety plan	12
1 Desim estimation	12
5. Design activities	15
3.1. General	13
3.2. Pre-Final (90 percent) design	13
3.2.1. Technical specifications	14
3.2.2. Contract drawings	14
3.2.3. Construction Quality Assurance Plan (COAP)	14
3.2.4. Citizen participation plan	14
3.2.5. Air monitoring plan	15
3.3. Final (100 percent) design	15

i

4.	Remedial action construction	17
	4.1. General	17
	4.2. Requirements	17
	4.3. Performance of remedial action construction	18
5.	Progress monitoring and reporting	19
	5.1. General	19
	5.2. Project management and responsibilities	19
	5.2.1. Project officer	20
	5.2.2. Engineering project manager	20
	5.2.3. Engineering coordinator	20
	5.2.4 Project supervisor	20
	5.3. Progress monitoring and documentation	21
	5.4. Progress meetings	21
	5.5. Unforeseen and emergency condition notification	22
	5.6. Field changes	22
6.	Post-remedial construction activities	25
	6.1. General	25
	6.2. Notice of completion	25
	6.3. Operation and Maintenance Plan	25
	6.4. Record drawings and certification	26
	6.5. Remedial actions summary report	26
	6.6. Remedial actions contingency plan	27

Tables

1. Schedule

Ground water TCE concentrations
 Monitoring well specifications and ground water elevations

....

Figures

- 1. Site map
- 2. Site plan
- 3. TCE soils excavation boring plan

Appendixes

- A. Basis of Design Report (December 1994)
- B. PCB/PAH/VOC Soils Excavation Work Plan (February 1995 with revised pages dated March 3, 1995
- C. SPDES permit
- D. Soil boring procedures
- E. Ground water monitoring plan outline
- F. Progress report forms
- G. Operation and maintenance plan outline

Exhibits

A. Health and safety plan

1. Introduction

1.1. General

This document is the Work Plan for Remedial Design/Remedial Action (RD/RA) activities to be conducted at the former Accurate Die Casting Site located at 547 East Genesee Street in the Village of Fayetteville, New York (Figure 1). Based on the results of the Remedial Investigation and Feasibility Study completed in August 1994, the New York State Department of Environmental Conservation (NYSDEC) has issued a Record of Decision (ROD) directing that remedial design/remedial action be performed at the Site.

This Work Plan summarizes how the five elements of the selected remedy identified in Section 7 of the ROD issued by the NYSDEC in December 1994 (Site Number 7-34-052) will be implemented. This Work Plan includes a description of the design and anticipated construction activities necessary to implement the selected remedy at the site. These activities will be initiated according to the schedule presented as Table 1.

This section of the Work Plan presents background information pertaining to the site, a description of the selected remedy for the site, and purpose and scope of this Work Plan.

1.2. Background

The facility was constructed in 1950 as a die casting industry. Die Casting activities were conducted at the site until mid-1988 when the then owners, the former Accurate Die Casting Corporation and George and Theresa Slyman, abandoned the facility. In December 1988, ITT Commercial Finance Corporation (ITT), as mortgagee-inpossession of the property, became the successful bidder and finalized the foreclosure proceeding relative to the Site.

After the conclusion of the foreclosure process, a consultant for ITT conducted a Phase I (June 1989) and Phase II (September 1990) environmental assessment. During the Phase II environmental assessment:

- Approximately 70 drums of waste found inside the building after foreclosure were characterized and disposed;
- Sludge from an abandoned trichloroethylene (TCE) degreasing system was removed and the system was decontaminated; and
- A TCE free product pool, which was discovered adjacent to and outside the northeast corner of the building proximal to monitoring well MW-3 (Figure 2), was pumped and disposed until no TCE free product was detected in samples.

Upon reviewing the Phase II assessment of the former Accurate Die Casting site, the NYSDEC required that a Remedial Investigation be conducted. The investigation was completed in accordance with the Remedial Investigation (RI)/Feasibility Study (FS) Work Plan (Stearns & Wheler, May 1992). In the RI Report (Stearns & Wheler, December 1993), it was concluded that:

- A TCE plume extends in the overburden aquifer from outside the northeast corner of the facility north towards Bishop Brook;
- Bedrock ground water had been impacted by TCE;
- The highest TCE concentrations in soils were observed at about 20 to 25 feet below grade outside the northeast corner of the facility, at the interface between the sand/gravel and clayey till layers;
- Surface and subsurface soils contain residues of polychlorinated biphenyls (PCBs), polynuclear aromatic hydrocarbons (PAH), and volatile organic compounds (VOCs) in the location referred to as the PCB/PAH/VOC Soils Area shown on Figure 2; and

• The material inside the abandoned septic tank contains levels of zinc that could present a risk to the environment.

In May 1994, an Interim Remedial Measure (IRM) Work Plan was prepared to separately address the soils exhibiting TCE outside the northeast corner of the facility and the TCE impacted overburden aquifer. The IRM Work Plan was subsequently approved by the NYSDEC in a letter dated May 23, 1994. In accordance with the approved Work Plan, the soil remediation activities were initiated on May 24, 1994 and completed on June 22, 1994, as documented in the Summary Report dated October 1994. Activities to address the overburden aquifer were initiated in September 1994.

As part of the soil remediation activities conducted outside the northeast corner of the facility, approximately 3,000 cubic yards (CY) of soil were excavated. Of that amount, approximately 1,250 CY of soil exhibited TCE levels above the established Remedial Action Objective (RAO) of 0.7 mg/kg, but below 15 mg/kg. The soil exhibiting TCE above 0.7 mg/kg was processed on site through a mechanical volatilization system. Once processed, the average residual TCE concentration was 0.13 mg/kg. The treated soils, as well as the excavated soils that did not contain TCE above the RAO, were subsequently used to backfill the excavation. However, before backfilling, a 3 to 4 feet layer of clay was placed and compacted at the bottom of the excavation to replace clay that was removed from above the bedrock.

In association with the soil remediation activities conducted outside the northeast corner of the facility, a ground water collection sump was constructed on top of the clay layer at the base of the excavation, prior to backfilling. The sump is located approximately 20 feet from the north wall of the east building addition, and approximately 20 feet east of the main building, as shown in Figure 2. The purpose of the ground water collection sump is to monitor the quality and facilitate removal of overburden ground water in the previously excavated area.

In accordance with the approved IRM Work Plan (O'Brien & Gere, May 1994), an overburden aquifer test recovery well, RW-1, was also installed on site (Figure 2) for the purpose of conducting a 24-hour pump test and hydrogeologic evaluation. The results of the 24-hour pump test and hydrogeologic evaluation are presented in the Basis of Design Report (Appendix A) which also presents a description of the proposed overburden ground water recovery and treatment system. The Basis of Design Report was approved by the NYSDEC in a letter dated December 29, 1994.

A "constructable quality" design is presently being prepared for approval by the NYSDEC for the overburden ground water recovery and treatment system based on the Basis of Design Report. As discussed during the meeting attended by the NYSDEC, NYS Department of Health, and O'Brien & Gere on January 25, 1995, the sump installed outside the northeast corner of the building will also be utilized as a ground water recovery point as part of the overburden ground water recovery and treatment system.

As a result of the January 25, 1995 it was also agreed that:

- Based upon data collected during the first two quarters following start-up of the overburden ground water recovery system, ITT will assess the need for and, if necessary, recommend to the NYSDEC locations for additional piezometers to fill data gaps. Upon NYSDEC approval, the additional piezometers would be installed.
- Quarterly letter reports will be submitted to the NYSDEC during the first year of operation presenting the data obtained during that quarter. The letter reports will also include an assessment of the effectiveness of the recovery system, and discuss the need for system modifications or additional wells, if any.
- Piezometer PZ-2 will be utilized as a ground water sampling point to provide water quality data for the area between monitoring wells MW-6 and MW-17.

1.3. NYSDEC selected remedial alternative

Based upon the results of the RI/FS at the former Accurate Die Casting site and the criteria identified for the evaluation of alternatives, the NYSDEC has further directed that the contaminated soil and sludge be excavated and disposed off-site and the TCE impacted bedrock ground water be extracted and treated on-site. The elements of the selected remedy identified in Section 7 of the ROD are as follows:

- The PCB/PAH/VOC soils from the former oil spill area, located on the northwest portion of the site, will be excavated and disposed in a permitted landfill;
- The sludge from the septic tank, located on the north-east portion of the site, will be excavated and disposed in a permitted landfill;
- The TCE impacted bedrock ground water will be extracted and treated on-site. The treated ground water will be discharged to Bishop Brook;
- Soil samples will be obtained from the area northeast of the facility, where soils containing TCE were excavated as part of an Interim Remedial Measure, to verify that a potentially significant source of TCE is no longer present in the vadose zone soils; and
- A long-term ground water monitoring program will be implemented to monitor the effectiveness of the ground water (overburden and bedrock) and soil remediation program.

1.4. Purpose of the RD/RA work plan

The purpose of this work plan is to present a description of the remedial design and remedial action tasks to be completed to implement the five elements of the selected remedy for the site. The work plan also includes:

- A Progress Monitoring and Reporting Plan which:
 - Describes the actions that will be taken to apprise the NYSDEC of activities completed and those proposed.
 - Identifies the individuals responsible for providing project status updates.
 - Provides samples of the forms and format that will be utilized to document project progress.

- Establishes benchmark events soon after which a job progress meeting may be proposed with the NYSDEC.
- Provides the procedure by which the NYSDEC and ITT will be apprised of emergency or unforeseen conditions at the site;
- A draft outline of the Operations & Maintenance Manual that will be prepared following the installation of the overburden aquifer ground water recovery and treatment system;
- A draft outline of the Sampling & Analysis Plan that will describe the monitoring that will be performed;
- A Health & Safety Plan for the protection of people at and in the vicinity of the site during construction;

O'Brien & Gere Engineers, Inc.

2. Pre-design activities

2.1. General

This section presents a description of the pre-design tasks to be completed prior to initiating design activities appropriate to the selected remedy for the site. For convenience, each of the five components of the remedy presented in Section 1.3 is addressed separately. The first two components consist of characterization of the soils in the PCB/PAH/VOC soils area and the sludge in the septic tank. The other three components involve evaluation of the bedrock aquifer, verification sampling in the former TCE excavation area, and long-term ground water monitoring.

2.2. PCB/PAH/VOC soils area

The PCB/PAH/VOC soils area will be addressed in accordance with the Work Plan dated February 1995 that has been prepared and submitted to NYSDEC for approval. The PCB/PAH/VOC Soils Excavation Work Plan (Appendix B) presents the pre-design activities that will be undertaken to characterize the soils for disposal off-site and to delineate the approximate vertical and horizontal extent of PCB/PAH/VOC residues in the soil.

2.3. Septic tank sludge

The septic tank and the sludge contained within the tank will be addressed in accordance with the Work Plan dated February 1995

Final: March 8, 1995 Div71Y prepared for excavating the PCB/PAH/VOC soils area. The PCB/PAH/VOC Soils Excavation Work Plan presents the pre-design sampling and analyses that will be performed to characterize the septic tank sludge for disposal. Following pre-characterization, the septic tank and its contents will be removed and disposed off-site concurrently with the removal of soils containing PCB/PAH/VOCs above the RAOs.

2.4. Bedrock ground water

This section presents the pre-design activities for addressing the bedrock aquifer, including:

- Data review and recovery well design
- Test bedrock recovery well installation
- Hydrogeologic evaluation
- Ground water treatment system evaluation
- Preparation of a bedrock ground water Basis of Design Report

A description of the pre-design tasks associated with bedrock ground water is presented below.

2.4.1. Data review/recovery well design

Prior to design and installation of the test bedrock recovery well, available data pertaining to the bedrock geology, ground water flow and ground water quality will be reviewed. This information will be used to finalize the location and design of the recovery well. The well design will consider the depth to and competency of the underlying bedrock in addition to structural and hydrogeologic characteristics of the bedrock.

2.4.2. Test bedrock recovery well installation

Based on current site conditions, it is anticipated that a single bedrock recovery well (RW-2) will be installed near the northeast corner of the facility. The bedrock aquifer water quality data (Table 2) obtained during the Remedial Investigation indicates that this area exhibits the highest dissolved concentrations of TCE in the bedrock aquifer.

Installation will consist of drilling a nominal 16-inch diameter borehole approximately 5 feet into competent bedrock and securing a 12-inch casing into bedrock using a portland cement/bentonite grout. A 11-7/8-inch borehole would then be advanced to 75 feet.

Should field conditions indicate that the competency of the shallow bedrock will not support an open-hole design, an 8-inch diameter screen and casing would be installed into the borehole.

Subsequent to installation, the test bedrock recovery well will be developed using either a combination of mechanical surging and pumping or air lift methods. Well development will continue until the well yields relatively sediment-free water.

2.4.3. Bedrock hydrogeologic evaluation

Following installation of the test bedrock recovery well, a pump test will be performed to estimate the design flow rates and to provide a preliminary evaluation of the bedrock aquifer characteristics. The well will be pumped continuously for up to 24 hours at a constant flow rate. The flow rate for the test will be established based on well yields obtained following well installation and development. Water levels in the pumping well, nearby bedrock monitoring wells, and selected overburden monitoring wells will be measured throughout the test. Once the test is complete and the pump is turned off, water level monitoring will continue until the water levels approximate static conditions observed prior to the test.

In addition, one discharge water sample will be collected during the test and will be analyzed for VOCs utilizing EPA Method 8010/8020 to assess VOC influent concentrations. Ground water generated during the pump testing will be treated and discharged to Bishop Brook in accordance with the NYSDEC SPDES Permit included in Appendix C.

2.4.4. Ground water treatment system evaluation

The ground water treatment system, installed to treat overburden aquifer ground water as part of an IRM, will be evaluated utilizing data obtained as a result of the bedrock hydrogeologic evaluation. The evaluation will focus on the ground water treatment system capacity, the influent flow rate generated by the overburden ground water recovery system and that expected from the bedrock ground water recovery system, anticipated influent VOC concentrations, and the ability of the system to accommodate the additional influent flow while maintaining compliance with the treated water discharge permit.

2.4.5. Basis of Design Report

A bedrock ground water Basis of Design Report will be prepared and submitted to the NYSDEC prior to initiating design. The Basis of Design Report will present the results and conclusions of the bedrock hydrogeologic evaluation and ground water treatment system evaluation, and present a description of ground water treatment system modifications that may be necessary to accommodate bedrock ground water flows. The Basis of Design Report will also present details of the bedrock test recovery well, a piping and system component diagram for the ground water treatment system installed as part of the overburden aquifer IRM depicting modifications required, and preliminary specifications for major system components.

2.5. TCE soils excavation verification sampling

Pursuant to the NYSDEC letter dated October 28, 1994, six soil borings will be advanced in the former TCE excavation area outside the northeast corner of the facility. Three of the borings will be located within the limits of the backfilled excavation, two will be installed along the eastern edge of the backfilled excavation, and one will be installed along the northern edge of the backfilled excavation as shown on Figure 3. The borings will be completed to a depth

equal to the highest recorded static ground water level (Table 3) in this area but not deeper than 25 feet.

The soil borings will be completed using conventional hollow-stem auger drilling techniques and soil samples will be obtained at two foot intervals for field screening using a photoionization detector (PID) in accordance with the procedures provided as Appendix D. The PID, equipped with a 10.2 eV lamp, will be calibrated to correct for the photoionization sensitivity of 8.9 for TCE compared to an isobutylene standard of 7 ppm (v/v).

Based on the results of the field screening, two soil samples will be obtained from each of the borings installed and analyzed in a laboratory for the presence of VOCs utilizing EPA Method 8010/8020. The soil samples selected from each boring will be from the two, non-contiguous depth intervals exhibiting the highest PID readings during field screening. Should no PID readings be observed, then soil samples will be obtained from intervals corresponding to approximately 1/3 and 2/3 of the boring depth.

The results of the soil analyses and the field screening data will be submitted to the NYSDEC for review in a letter report within 20 business days of the receipt of the final laboratory report by O'Brien & Gere.

2.6. Long-term ground water monitoring program

Within 60 calendar days of completing construction and initiating operation of the overburden aquifer ground water recovery and treatment system, installed as part of an interim remedial measure, a long-term ground water monitoring program plan will be submitted to the NYSDEC for approval. The ground water monitoring program plan will include a summary of the long-term ground water treatment system performance monitoring requirements. A draft outline of a ground water monitoring program plan is provided as Appendix E for review.

In general, the monitoring plan will include a schedule for measuring water levels as well as collecting ground water samples for analysis. It is anticipated that, following start-up of the overburden and bedrock recovery systems, water levels will be measured weekly for the first month, biweekly for the following two months and monthly for a period of six months. The frequency may be modified depending upon information obtained. Ground water samples will be collected during 1995 from all the on-site monitoring wells on a quarterly basis beginning in January 1995. The number of wells sampled per event, and the frequency of the sampling events, may be modified in subsequent years as appropriate based on the information obtained in 1995.

As may be appropriate, the ground water monitoring program plan would be revised within 60 days after completing construction of the bedrock aquifer ground water recovery and treatment system.

2.7. Health and safety plan

A site specific Health and Safety Plan, provided as Exhibit A, was prepared to provide guidance for O'Brien & Gere Technical Services, Inc. personnel during the conduct of remedial activities. This document is adopted for O'Brien & Gere Engineers' on-site personnel during the installation of soil borings, and conduct of soil sampling and excavation activities. The Health and Safety Plan was prepared in accordance with the Federal Occupational Safety and Health Administration guidelines (29 CFR 1910.120) utilizing data generated during the RI, including site history and subsurface conditions.

O'Brien & Gere Technical Services will be responsible for the implementation and enforcement of the Health and Safety Plan (HASP) during the activities to be conducted on the site.

3. Design activities

3.1. General

This section presents the remedial design activities that are to be completed as part of design of a bedrock ground recovery system at the site. The design activities include preparation of a Pre-Final (90 Percent) Design package followed by a Final (100 Percent) Design package. Design activities associated with the removal and disposal of the PCB/PAH/VOC soils and septic tank are addressed in the PCB/PAH/VOC Soils Excavation Work Plan dated February 1995.

3.2. Pre-Final (90 percent) design

A Pre-Final (90 Percent) Design package of "constructable quality" documents for the bedrock ground water recovery and treatment system will be submitted to the NYSDEC for review. The pre-final (90 percent) Design will include:

- Special provisions specifying procedures for collection, destruction, treatment, and/or disposal of hazardous wastes and substances and their constituents and degradation products, and of any soil or other materials contaminated thereby.
- Special provisions for physical security of the site, as well as for monitoring the health and safety of persons living and/ or working at or in the vicinity of the site while undertaking remedial activities.

The Pre-Final (90 Percent) Design documents will consist of the elements described in the following sections.

3.2.1. Technical specifications

The Technical Specifications will define the qualitative requirements for products, materials, and workmanship for the major components associated with Remedial Action Construction activities. The Technical Specifications will be prepared using the Construction Specifications Institute (CSI) format.

3.2.2. Contract drawings

The Contract Drawings will be prepared using AutoCADD, and the drawings will be D-size (24 inch x 36 inch). The Contract Drawings will include a Site Plan showing the location of ground water recovery well, routing of pipe and the treatment facility, ground water treatment process schematic, and mechanical, electrical, and structural drawings providing necessary details for construction.

3.2.3. Construction Quality Assurance Plan (CQAP)

The CQAP will cover elements necessary to document that the completed Remedial Action Construction meets the requirements of the Remedial Design. The CQAP implementation begins at the start of construction and continues through the completion of construction. Operation of the ground water treatment system will be addressed separately in the Operation and Maintenance (O&M) Manual.

3.2.4. Citizen participation plan

The Citizen Participation Plan incorporates appropriate activities outlined in the NYSDEC publication, "New York State Inactive Hazardous Waste Citizen Participation Plan", dated August 30, 1988, and any subsequent revision thereto, and 6 NYCRR Part 375. It is assumed that the NYSDEC will continue to implement the Citizen Participation Plan activities for the site.

3.2.5. Air monitoring plan

Air monitoring procedures to be implemented during construction of the remedial action are described in the HASP (exhibit A). Air monitoring requirements and protocols to be implemented during operation of the ground water treatment system will be addressed in the operations and maintenance manual, as appropriate.

3.3. Final (100 percent) design

After receipt of one round of consolidated comments from the NYSDEC on the Pre-Final (90 percent) design, O'Brien & Gere will execute the required revisions and submit the Final (100 percent) Design documents to the NYSDEC for approval. The Final (100 percent) Design documents will be of "constructable quality" for use by O'Brien & Gere Technical Services in completing construction of the bedrock ground water recovery and treatment system, and will have the signature and seal of a professional engineer licensed to practice in the State of New York.

O'Brien & Gere Engineers, Inc.

4. Remedial action construction

4.1. General

Upon NYSDEC's approval of the final bedrock aquifer ground water recovery and treatment system design, the Remedial Action Construction phase of the project will be initiated. This phase will commence upon approval of the remedial design and it will be completed upon NYSDEC's certification of completion of construction work and the successful start-up of the bedrock ground water recovery and treatment system.

This section does not address construction activities associated with the removal and disposal of the PCB/PAH/VOC soils and septic tank since these activities are described in the PCB/PAH/VOC Soils Excavation Work Plan dated February 1995.

4.2. Requirements

Within 90 calendar days of NYSDEC's approval of the Final Design, construction of the bedrock aquifer recovery and treatment system will be initiated. Within 45 calendar days of award of the contract, the following items will be submitted to NYSDEC:

- Requests for modification of the approved Final Design based on construction methods identified by the Remedial Action Contractor, or modifications of the Construction Schedule, or other new information.
- A construction schedule of sufficient detail to monitor the progress of key project phases and tasks on the Contractor's critical path. The schedule will be maintained by the Contractor

and the Contractor will provide a letter detailing the circumstances of any changes or delays of schedule.

4.3. Performance of remedial action construction

Upon receipt of NYSDEC's written approval of the final design as well as the written approval or disapproval of any requests for modification of the final design and/or construction schedule, Remedial Action Construction will be initiated in accordance with this work plan and the approved remedial design.

5. Progress monitoring and reporting

5.1. General

This section identifies the individuals that are responsible for monitoring the progress of the remedial design and remedial actions, and for providing periodic status updates to NYSDEC. This section also presents the format and describes the contents of the monthly progress reports, identifies certain benchmark events after which progress meetings may be conducted, and establishes the procedures by which ITT and NYSDEC will be apprised of emergency or unforeseen conditions encountered at the site.

5.2. Project management and responsibilities

A project management team from both O'Brien & Gere Technical Services, Inc. and O'Brien & Gere Engineers, Inc. (O'Brien & Gere) has been assembled to plan, implement, and monitor the progress of remedial design and remedial activities at the site. O'Brien & Gere Technical Services, Inc. will be the lead organization for managing the implementation of the RD/RA activities. O'Brien & Gere has been retained by O'Brien & Gere Technical Services to prepare and implement the engineering activities defined in this RD/RA Work Plan. O'Brien & Gere will also be responsible for preparing the monthly progress reports incorporating data provided by O'Brien & Gere Technical Services for submittal to ITT and the NYSDEC.

The individuals assigned to this program, as well as a description of their respective roles and responsibilities, are identified below.

5.2.1. Project officer

Terry L. Brown, P.E. of O'Brien & Gere Technical Services will serve as project officer for the RD/RA program. Mr. Brown is responsible for coordinating the overall management of the RD/RA activities, directing field and office personnel, and managing the administrative aspects of the project. Mr. Brown will be the primary point of contact for all written correspondence from the NYSDEC and ITT to the Companies of O'Brien & Gere.

5.2.2. Engineering project manager

David S. Towers, P.E. of O'Brien & Gere will serve as the engineering project manager for the RD/RA program. Mr. Towers is responsible for coordinating the preparation of this and associated work plans, remedial design documents, and monthly monitoring reports. Mr. Towers will also serve as the secondary contact for the NYSDEC to address questions with regard to the progress of project and status of verification sampling efforts.

5.2.3. Engineering coordinator

Alfred R. Farrell, P.E. of O'Brien & Gere will serve as the engineering coordinator for the RD/RA program. Mr. Farrell is responsible for the day-to-day monitoring of the status of activities being conducted by O'Brien & Gere Technical Services.

5.2.4 Project supervisor

Mr. Anthony Geiss of O'Brien & Gere Technical Services will serve as the project supervisor during implementation of the remedial actions. Mr. Geiss' responsibilities will include managing the day-today activities in the field in accordance with this and associated work plans, and the remedial design documents. Mr. Geiss will also be responsible for providing Mr. Farrell weekly reports, while construction activities are underway, detailing the activities completed during the week and the activities proposed to be undertaken during the following two weeks. Mr. Geiss will also be responsible for apprising Mr. Brown and Mr. Farrell of conditions encountered in the field that either were unforeseen or may result in a delay of the task completion time.

5.3. Progress monitoring and documentation

Progress reports, using the format presented in Appendix F, will be prepared for submittal to NYSDEC each month, starting one month after the approval of this Work Plan and the effective date of the Consent Order, and continuing until the tasks described within this Work Plan are complete. These progress reports, which will be submitted to the NYSDEC by the 10th day of each month, will present a summary of the activities performed during the proceeding month and a summary of the activities proposed to be undertaken within the next 45 calendar days.

The progress reports will also include information regarding estimated task completion dates, and provide a description of unresolved delays encountered or anticipate that may affect the future schedule for implementation of the RD/RA activities and efforts made to mitigate those delays or anticipated delays.

5.4. Progress meetings

O'Brien & Gere Technical Services, Inc. will provide the NYSDEC written notice, at least 5 business days in advance, of scheduled project progress meetings, including the pre-construction and substantial completion meetings, and initiation of the following field activities:

- Soil boring installations;
- Ground water sampling events;
- Invasive activities such as soil excavation;

• Start-up and testing of ground water recovery and treatment system.

For planning purposes, it is anticipated that a scheduled project progress meeting may occur on or shortly after the following events:

- Submittal of the TCE soils excavation verification sampling letter report to the NYSDEC;
- Initiation of continuous operation of the overburden ground water recovery and treatment system; and
- Submittal of the first Annual Ground Water Monitoring Report.

This notification requirement is waived for activities performed in response to unforeseen and emergency conditions.

5.5. Unforeseen and emergency condition notification

In the event of a potentially significant unforeseen or emergency condition, O'Brien & Gere Technical Services, Inc. will immediately notify ITT and the NYSDEC by telephone, with follow-up notification provided in writing. The notice will apprise the recipient of the nature of the unforeseen or emergency condition, potential impact to the project schedule, and measures taken or initiated to remedy the situation or minimize impacts.

For no unforeseen or emergency condition shall the NYSDEC require advance notice, or penalize ITT for failure to provide advance notification.

5.6. Field changes

During performance of the Remedial Actions described in this Work Plan, conditions on-site may necessitate changes to the NYSDEC approved design and/or associated work plans. Such requests will be

O'Brien & Gere Engineers, Inc.

submitted in writing and include a description of the situation/conditions necessitating a field change, and a description of the modifications proposed and the potential impact to the completion of the remedial actions.

Remedial design/remedial action

Final: March 8, 1995 Div71Y

6. Post-remedial construction activities

6.1. General

This section identifies the documents that are to be submitted to the NYSDEC following completion of the remedial actions described within this Work Plan and the PCB/PAH/VOCs Soils Excavation Work Plan (O'Brien & Gere, February 1995).

6.2. Notice of completion

Within 30 calendar days of the completion of remedial construction activities on site, a Notice of Completion for Remedial Action Construction will be submitted to NYSDEC.

6.3. Operation and Maintenance Plan

Within 60 calendar days of completing construction and initiating operation of the overburden aquifer ground water recovery and treatment system, an Operation & Maintenance Plan will be submitted to the NYSDEC for approval. The Operation & Maintenance Plan will provide a summary of the ground water recovery and treatment system controls and functions, normal operation parameters, required system inspection frequency, maintenance requirements, and a trouble shooting guide. A draft outline of an Operation & Maintenance Plan is provided as Appendix E for review. The Operations & Maintenance Plan would be subsequently revised, if necessary, within sixty days of completing bedrock aquifer remedial actions.

6.4. Record drawings and certification

Within 60 calendar days of completing construction and initiating operation of the bedrock aquifer ground water recovery and treatment system, a set of Record Drawings depicting the ground water recovery and treatment system will be submitted to the NYSDEC. The Record Drawings will accompany a letter, signed and stamped by an engineer licensed to practice in New York State, certifying that the ground water recovery and treatment system was completed in accordance with the NYSDEC approved remedial design, and the NYSDEC approved modifications thereto.

6.5. Remedial actions summary report

Within 60 calendar days of completing all remedial actions described within this Work Plan and PCB/PAH/VOC Soils Excavation Work Plan (O'Brien & Gere, February 1995), a Remedial Actions Summary Report will be submitted to the NYSDEC. The Remedial Actions Summary Report will present an overview of the remedial actions completed at the former Accurate Die Casting site to implement the components, listed in section 1.3, of the remedy presented in the ROD. The summary report will also present a summary of the previously completed Interim Remedial Measures and provide a reference listing of pertinent Work Plans and Reports prepared previously for the site.

In accordance with the Consent Order, the Remedial Actions Summary Report will be signed and stamped by an engineer licensed to practice in New York State, and will include a statement certifying that the remedial actions were completed at the site in accordance with the NYSDEC approved work plans and the NYSDEC approved revisions thereto.

6.6. Remedial actions contingency plan

As part of the long-term ground water monitoring program, an annual ground water monitoring report will be submitted to the NYSDEC. If, based on the long-term ground water monitoring data, it becomes apparent that the remedial actions completed in accordance with this Work Plan are not effectively remediating the site, a Remedial Actions Contingency Plan will be prepared and submitted to the NYSDEC.

The Contingency Plan would include a description of the status of remedial efforts, present data obtained as a result of the monitoring efforts, and present practicable and feasible measures that may be implemented to facilitate remediation. In the event that practicable measures are not available, the Contingency Plan would present the data and conclusions that may warrant a revision of the RAOs for the site.

TABLES

• •

.

Table 1 Project Schedule

Former Accurate Die Casting Site

	Fa	yette	rille, N	lew Y	Ork								
~	and the second secon	1.1000		<u></u>		1995				0	200.23	1998	<u></u>
S.	Description	Mar	Apr	May	Jun		Aug	Sep	Oct	NOV	Dec	Jan	Feb
PH	E-DESIGN ACTIVITIES					1	· ·						
1.	PCB/PAH/VOC soils area									1			l
_	pre-characterization									 			<u> </u>
2.	Septic tank sludge						i						
	pre-characterization	ļ	L		L		L						
3.	Bedrock Ground Water												
	Evaluations		_					1		1			
	- Test bedrock recovery well												
	installation & pump test		ļ				L				L		·
	 Bedrock hydrogeologic 	Ì											ł –
	evaluation		1										<u> </u>
	- Ground water treatment				• •		ļ						
	system evaluation		<u> </u>				ļ				<u> </u>	<u> </u>	<u> </u>
	 Bedrock ground water 						ļ						
	Basis of Design Report		<u> </u>			<u> </u>				 	<u> </u>	ļ	<u> </u>
	- NYSDEC review of Basis		1								ļ	t '	
<u> </u>	or Design Report (1)	 	<u> </u>	L	L			i	L_	ļ	I		L
4.	TUE soils excavation clean-up												
	venincation					<u> </u>		r .					
	 Sampling & analysis 				1								
										Ļ		<u> </u>	<u> </u>
	- TCE soils verification				ļ								1
_	sample results letter report	ļ	ļ								ļ	<u> </u>	
5.	Long-term Ground Water							1			[
	Monitoring Plan					<u> </u>	<u> </u>	L			i.		
DE	SIGN ACTIVITIES	ļ				<u>,</u>				-			
٦.	PCB/PAH/VOC Excavation Plan			ļ						l			
~	Pedrack Cround Water Deseuse	ļ		1		L							
2.	Bedrock Ground Water Recovery												
	Bro final (20%) design		r	γ -		r	<u> </u>						
	submittal					•				-			
	Construction Quality Assurance												
	Plan Submittal			Ì				 -	i				Ι.
		-					┼╾┈╼╴	<u> </u>				<u>├</u> ──	<u> </u>
	- Final design submittar												
	 NYSDEC review of final 		<u> </u> · · ·				· · · · ·					┟╍╍╍╍	<u> </u>
	design (1)							1			ļ	1	
		•			-	1	1			L	L	<u>ا</u>	2
4	Overburden Greund Weter Beenverst and		1				<u> </u>						
١.	Treatment Sustem Construction (0)				_	1							
•	POD/DALIA/OO Daile Asso Europetica (0)									<u> </u>	 		—
2.	PCB/PAH/VOC Sous Area Excavation (2)												
_	De des sis Comme d'Marten De service		-										
З.	Bedrock Ground Water Recovery											1	
			1					<u> </u>	L	L			L
PO	ST-REMEDIAL CONSTRUCTION ACTIVITIES	<u> </u>						<u> </u>					<u> </u>
1.	Notice of Completion												l
	A	<u> </u>	<u> </u>			 				<u> </u>	[<u> </u>	
2.	Operation & Maintenance Plan	i	ł							•			
		<u> </u>			L	ļ				<u> </u>	L		L
3.	Record Drawings & Certification	1											
			l		L			ļ			L	L	
4.	Remedial Actions Summary Report												
L													
Not	es: (1) Schedule provides 30 calendar days for	revie	w by I	he N'	YSDE	C.			a	[\rdra	tab1		

Notes: (1) Schedule provides 30 calendar days for review by the NYSDEC.

(2) Actual duration of construction is not known. A detailed construction schedule will be provided prior to initiating construction activities on-site.

(3) Monthly progress reports will be submitted to the NYSDEC in accordance with the Consent Order.

(4) Shading indicates the anticipated period during which the identified task will start and be completed, unless otherwise noted.

.

Table 2

Ground Water TCE Concentrations

	Trichloroethylene Concentrations (1)									
Date Sampled:	8/30/89	12/4/89	5/20/90	5/28/92	7/22/94	10/18/94				
Monitoring Well (3)										
MW-1	112	ND	2	ND	NS	Dry				
MW-2	ND	ND	1	ND	NS	ND				
MW-3	Free Product	>55,000	440,000	340,000	Abandoned	Abandoned				
MW-4	NS	7	43	6	270	23				
MW-5	NI	340	344	110	330	410				
MW-6	NI	700	454	510	390	360				
MW-7 (B)	NI	ND	ND	ND	ND	ND				
MW-8	NI	ND	ND	ND	NA	ND				
MW-9	NI	109	106	60	72	74				
MW-10 (B)	NI	NI	NI	4,500	1,600	1300				
MW-11 (B)	NI	NI	NI	5,200	5,500	5300				
MW-12	NI	NI	NI	36	44	35				
MW-13	NI	NI	NI	110	740	510				
MW-14	NI	NI .	NI	67	150	120				
MW-15 (B)	NI	NI	NI	N	NS	14				
MW-16 (B)	NI	NI	NI	NI	NS	6				
MW-17	N	NI	Nt	NI	260	140				

Accurate Die Casting Facility Fayetteville, NY

Notes: ND - Not detected at concentrations greater than analytical detection limit.

NS - Not sampled.

NI - Well not installed at time of sampling.

NA - Not analyzed.

- (1) Concentrations reported in ug/L (ppb).
- (2) Sample collected 8/19/92 because MW-13 and MW-14 were dry on 5/28/92.
- (3) Monitoring wells MW-1 through MW-16 installed by Stearns & Wheler, monitoring well MW-17 installed by O'Brien & Gere Engineers, wells MW-1, MW-7, MW-10, MW-11, MW-15, and MW-16 are bedrock ground water monitoring wells.

(B) - Bedrock well.
TABLE 3		
MONITORING WELL SPECIFICATIONS AND GROUND WATER ELEY	VATIO	NS
ACCURATE DIE CASTING		•
FAYETTVILLE, NY		

	•	Well																
	Ground	Casing	Scree	ned I	Interval	GROUND	WATER EL	EVATIONS										
Well No.	Elevation	Elevation	Eli	evati	on	5/28/92	6/26/92	8/7/92	9/26/94	9/27/94	10/18/94	11/2/94	11/17/94	11/30/94	12/15/94	12/27/94	1/13/95	1/25/95
					~~ .	DOV	ODV	70.00			001/							
MW-1	99.36	101.11	75.4	-	85.4	DHT	DHT	79.69			UHT							
MW-2	91.8	94.68	76.6	•	86.6	83.21	82.81	84.32	83.10	83.28	80.12			-				_
MW-3	97.65	99.63	73.7	•	83.7	80.44	80.09	81.63	ABDN	ABDN	ABDN	Abandone	d	-				
MW-4	65.62	68.52	46.6	-	56.6	51.08	49.95	50.81	47.22	52.21	46.79						-	
MW-5	88.21	90.42	49.2		59.2	60.71	63.76	61.22	59.87	59.91	59.45			_				-
MW-6	77.46	79.38	46.4	•	56.4	60.50	60.49	60.46	59.51	59.52	59.05							
MW-7(B)	75.66	78.34	34.3		44.3	54.59	54.55	54.47	53.90	53.97	53.55							
MW-8	88.21	91.78	53.9	- 1	63.9	66.38	66.38	66.83	61.59	61.65	60.99			→				
MW-9	102.44	104.03	49.7	- !	59.7	60.46	60.51	61.83	59.57	59.59	59.08		—					
MW-10(B)	97.51	99.69	43.03		53.03	61.15	61.99	61.69			56.02	55.07	55.19	54.94	55.19	55.02	54.94	54.95
MW-11(B)	91.48	93.80	43.1	-	53.1	62.34	63.70	63.66	58.41	58.39	57.47	50.01	56.68	55.59	56.63	56.55	55.63	55.63
MW-12	93.62	94.14	51.9	- 1	61.9	62.24	60.74	62.77	59.77	59.79	59.31							
MW-13	98.73	100.92	77.7	•	87.7	DRY	80.62	80.92	•••		78.70	82.92	78.21	78.21	80.92	78.34	78.25	77.83
MW-14	98.76	100.62	74.6		84.6	75.11	79.07	81.54			66.18	60.12	80.54	80.54	80.20	80.54	80.62	60.45
MW-15(B)	96,1	98.90	32.7	•	42.7	NI	NI	NI		***	53.47			-				
MW-16(B)	98,5	100.85	50.8	- 1	60.8	NI	NI	NI			61.67							
MW-17	66,9	69.24	53.7	- 1	63.7	NI	NI	NI	54.61	54.61	54.08	-			_			-
PZ-1	81,8	83.95	49.8	- 3	59.8	NI	NI	NI	59.56	59.57	59 .10						-	
PZ-2	80.6	83.06	42.8	•	52.8	NI	NI	NE	59.35	59.36	58.89		-		· <u> </u>			
RW-1	78.4	80.28	29.4-39.4	-	45.4-50.4	NI	NI	NE	56.88	56.89	58.22							
Sump	-	98.50		-								76.04	74.83	75.00	75.17	74.83	75.00	75.00

Notes: (B) - Bedrock well.

NI - Well not installed at time of monitoring.

----- Water level not monitored.

ABDN - Abandoned.

MW-1 through MW-16 installed during Remedial Investigation (Stearns & Wheler).

TME:skf/ACC396.2

5

O'Brien & Gere Engineers, Inc.

10-Feb-

.











~

.

. .

• • .

• .

At.

-.

.

.

APPENDIX A

BASIS OF DESIGN REPORT (DECEMBER 1994)

BASIS OF DESIGN REPORT

Accurate Die Casting Site Ground Water Recovery and Treatment System

ITT Commercial Finance Corporation

December 1994



Basis of Design Report

Accurate Die Casting Site Ground Water Recovery and Treatment System

ITT Commercial Finance Corporation

December 1994

5000 Brittonfield Parkway P.O. Box 4873 Syracuse, New York 13221

Contents

Tables	iii
Figures	iv
Appendices	v
1. Introduction	1
1.1. General	1
1.2. Project background	1
1.2.1. TCE soils IRM summary	2
1.2.2. Ground water	4
1.3. Ground water recovery and treatment remedial	
action objectives	4
	ſ
2. Site hydrogeologic evaluation	5
2.1. General	5
2.2. IRM piezometer and monitoring well	
installations	5
2.3. Overburden ground water quality	6
2.4. Site hydrogeologic conditions	7
2.4.1. Geologic conditions	7
2.4.2. Overburden aquifer saturated	
thickness	8
2.5. Test recovery well installation	9
2.6. Aguifer performance test	11
2.7. Conceptual hydrogeologic model and ground	
water recovery scenario	14
2.7.1. Conceptual hydrogeologic model	14
2.7.2. Ground water recovery scenario	15
3. Ground water recovery and treatment system	16
3.1. General	16
3.2. Ground water recovery and conveyance system	16
3.3. Ground water treatment system	18
3.4. Instrumentation and controls	20
3.4.1. Ground water recovery well pump	20

i

	3.4.2.	Water flow meter, strip chart recorder	
		and flow totalizer	21
	3.4.3.	Pressure sensor	21
	3.4.4.	Alarm notification devices	21
3.5.	Treated	water discharge	22

O'Brien & Gere Engineers, Inc.

Final: December 1, 1994 TME:bdm/TEC396.26

ii

Tables

1 2 Monitoring Well Specifications Ground Water TCE Concentration

. `

iii

Figures

- 1 Site Map
- 2 Site Plan
- 3 Overburden Aquifer TCE Concentrations
- 4 Hydrogeologic Cross-Section A-A'
- 5 Hydrogeologic Cross-Section B-B'
- 6 Overburden Aquifer Saturated Thickness Isopach Map
- 7 Saturated Thickness of Sand and Gravel Overburden Material
- 8 Recovery Well Construction Diagram
- 9 Overburden Aquifer Area of Influence Isopach Map
- 10 Ground Water Recovery and Treatment Process Schematic
- 11 Ground Water Recovery and Treatment System Pipe Routing Plan

Appendices

- Α
- В
- С
- D
- IRM Piezometer and Monitoring Well Logs Ground Water Monitoring Laboratory Reports Aquifer Performance Test Data Background Water Quality Laboratory Report Effluent Limitations and Monitoring Requirements Ε

1. Introduction

1.1. General

The purpose of this Report is to present the Basis of Design of the ground water recovery and treatment system for the former Accurate Die Casting facility located at 547 East Genesee Street in the Village of Fayetteville, New York. As proposed in the Interim Remedial Measure (IRM) Work Plan dated May 1994 and approved by the New York State Department of Environmental Conservation (NYSDEC) in a letter dated May 23, 1994, a ground water recovery and treatment system will be installed to address overburden ground water containing volatile organic compounds (VOCs) on site.

This section presents background information about the site, and select Remedial Investigation (RI) data that are pertinent to the design of a ground water recovery and treatment system. This section also presents the remedial action objectives (RAOs) of a ground water recovery and treatment system.

Subsequent sections present the results of the aquifer performance test conducted to assess site hydrogeologic conditions, and present a description of the proposed ground water recovery and treatment system. Included within these sections are the data, calculations and assumptions used during evaluation of the hydrogeologic conditions and preparation of the design.

1.2. Project background

The Accurate Die Casting site is located on a 32-acre parcel at 547 East Genesee Street in the Village of Fayetteville, New York (Figure 1). Bordering properties include out-of-service farmland to the

Final: December 1, 1994 TME:bdm/TEC396.26

north, residential areas to the east and west, and commercial properties to the south along East Genesee Street. An out-of-service railroad siding extends along the western border of the site and Bishop Brook extends along the northern border of the site.

The facility was constructed in the early 1950s as a die casting industry. Die casting activities were performed at the site until mid-1988, at which time the then facility owners abandoned the facility. In December 1988, ITT Commercial Finance Corporation was the successful and sole bidder at a foreclosure sale. ITT Commercial Finance Corporation is the current owner of the property. In January 1990, the facility was included in the NYSDEC's Registry of Inactive Hazardous Waste sites as a Class 2 site.

While die casting operations were conducted at the site, a degreaser system was used to degrease castings inside the building. Through performance of the RI, it was concluded that releases of TCE from the degreaser system have resulted in an impact to the soil and ground water at the site. In the RI Report (Stearns & Wheeler, December 1993), it was concluded among other things that:

- Ground water quality had been impacted by TCE. An overburden TCE plume extends from near monitoring well MW-3 north towards Bishop Brook;
- The highest TCE concentrations in soils were observed at about 20 to 25 feet below grade around MW-3, at the interface between the sand/gravel and till layers;
- Soils beneath the floor of the building had not been impacted. Soil samples from two monitoring wells (MW-13 and MW-14) installed just inside the building did not exhibit TCE levels above the analytical detection limit (<0.04 mg/kg).

1.2.1. TCE soils IRM summary

The area of TCE-contaminated soil identified during the RI is located on the northern side of the east addition to the main building in the vicinity of the former location of MW-3, as shown on Figure 2. Soil remediation activities in this area were initiated on May 24, 1994 in accordance with the Interim Remedial Measure (IRM) Work Plan (May 1994) approved by the NYSDEC and were completed on June 22, 1994.

As part of the Soil Remediation IRM, approximately 3,000 cubic yards (CY) of soil were excavated. Of that amount, approximately 1,250 CY of soil exhibited TCE levels above the established RAO of 0.7 mg/kg, but below 15 mg/kg. The soil exhibiting TCE above 0.7 mg/kg was processed on site through a mechanical volatilization system. Once processed, the average residual TCE concentration was 0.13 mg/kg. The treated soils, as well as the excavated soils that did not contain TCE above the RAO, were subsequently used to backfill the excavation.

Prior to backfilling the excavation, samples of soil were obtained from the base of the excavation as well as from the side walls. Based on this sampling, the soils exhibiting TCE above the RAO were limited to a semi-confining layer. Additional information with regard to the Soil Remediation Activities IRM is presented in the Summary Report dated October 1994.

In association with the Soil Remediation Activities IRM, a ground water collection sump was constructed on top of the clay layer at the base of the excavation, prior to backfilling. The sump is located approximately 20 feet from the north wall of the east building addition, and approximately 20 feet east of the main building as shown in Figure 2. The sump consists of a 24-inch diameter slotted pipe installed vertically to a depth of approximately 22 feet. This pipe is coupled to a horizontal 12-inch diameter slotted pipe placed on top of the 3 feet of clay placed at the bottom of the excavation prior to backfilling.

The purpose of the ground water collection sump is to facilitate removal of shallow ground water in the area of the excavation, if necessary. In accordance with the recommendations presented in the Soil Remediation Activities IRM Summary Report, ground water elevations in monitoring wells MW-10, MW-11, MW-13 and MW-14 and the sump installed within the excavation will be measured on a bi-weekly basis between October 31, 1994 and January 1, 1995. Once ground water levels have been measured, the sump will be bailed and sampled. The ground water sample will be analyzed for the presence of VOCs using EPA Method 8010/8020. The purpose of the bailing and sampling is to confirm that overburden ground water in the

Final: December 1, 1994 TME:bdm/TEC396.26 location of the excavation is no longer impacted by soils, and to assess whether the need exists for additional soil remedial or sampling efforts, or ground water recovery in the area of the excavation. The data obtained as a result of this monitoring, as well as recommendations for future actions in this area, will be presented to the NYSDEC.

1.2.2. Ground water

Based on the data generated as a result of the RI and subsequent investigations, it has been concluded that TCE is present in both the shallow and bedrock aquifers on site. Ground water quality is discussed further in Section 2.3.

1.3. Ground water recovery and treatment remedial action objectives

The overall remedial action objective for the Accurate Die Casting site is to provide for the protection of human health and the environment by controlling the sources of VOCs at the site and/or minimizing the migration of the ground water containing VOCs. The specific remedial action objectives associated with the proposed ground water recovery and treatment system are:

- Minimize the migration via ground water of dissolved TCE from the site towards Bishop Brook (or other receptors);
- Pursuant to the NYSDEC's directive, reduce the amount of TCE in the aquifer to the maximum practical extent by appropriate and technically practicable methods in an effort to achieve VOC concentrations less than 5 $\mu g/l$;

 Eliminate or minimize the potential for significant releases to surface water; and

Minimize, to the extent practicable, the potential risks associated with implementing remedial actions.

O'Brien & Gere Engineers, Inc.

2. Site hydrogeologic evaluation

2.1. General

In accordance with the NYSDEC approved IRM Work Plan, a hydrogeologic evaluation was completed at the former Accurate Die Casting site to develop design parameters for the construction and operation of a shallow overburden ground water remediation system. Tasks conducted pursuant to the IRM Work Plan included the installation of two piezometers, one overburden ground water monitoring well, one ground water test recovery well, and the completion of a 24-hour aquifer performance test. This section presents the results of the hydrogeologic evaluation.

2.2. IRM piezometer and monitoring well installations

In accordance with the approved IRM Work Plan, two piezometers (PZ-1, PZ-2), and one monitoring well (MW-17) were installed to further characterize site hydrogeologic conditions and provide additional monitoring points for the aquifer performance test. PZ-1 and PZ-2 were installed approximately 50 feet west and 100 feet east of MW-6, respectively. MW-17 was installed approximately 240 feet west of MW-6 (Figure 2). In addition to acquiring hydrogeologic information, MW-17 was installed to assess ground water quality between MW-6 and MW-4.

PZ-1, PZ-2, and MW-17 were installed by OP-TECH Environmental Services, Inc. between August 1 and August 4, 1994. The piezometers and monitoring well were installed to the bedrock surface using conventional hollow stem auger drilling techniques. Bedrock was encountered at PZ-1, PZ-2, and MW-17 at depths of approximately 30, 36, and 12 feet below grade, respectively.

Continuous split spoon samples were collected at two foot intervals throughout PZ-1, and below the water table at PZ-2. Split spoon samples were collected at five foot intervals at MW-17. Each soil sample was described by the on-site hydrogeologist. Boring logs are contained in Appendix A.

PZ-1, PZ-2, and MW-17 were constructed with 2-inch diameter PVC. Ten feet of 0.010-inch slot PVC well screen were utilized at PZ-1 and PZ-2 which was flush threaded to appropriate lengths of PVC riser casing to bring each well head approximately 2.5 feet above grade. Eight feet of 0.010-inch slot screen was utilized at MW-17. A silica sandpack compatible with a 0.010-inch slot well screen was installed from the bottom of the screen to approximately 2 feet above the screened interval in each well. Approximately 2.5 feet of bentonite pellets were installed on top of the sandpack in each well. The remaining annular space was filled with cement bentonite grout to grade. Each piezometer/well was completed with a locking steel protective guard pipe. Monitoring well specifications are listed in Table 1.

2.3. Overburden ground water quality

Ground water samples were collected from site monitoring wells on two occasions; July 22 and October 18 and 19, 1994 to evaluate the concentration of VOCs in the ground water. The ground water samples were delivered to O'Brien & Gere Laboratories, Inc. subsequent to daily sample collection using chain of custody procedures. Ground water samples were analyzed for volatile organic compounds (VOCs) by EPA Method 8010/8020. Table 1 presents a summary of TCE concentrations. Figure 3 illustrates the overburden aquifer TCE concentrations based on samples collected in October 1994. Laboratory data sheets are contained in Appendix B.

As illustrated on Figure 3, TCE in the overburden ground water extends from MW-13 north to MW-6. MW-9 and MW-12 define the eastern extent and MW-17 defines the western extent.

TCE was detected in monitoring well MW-4 at a concentration of 23 $\mu g/l$. It should be noted that MW-4 is located in the vicinity of a prior waste oil spill discovered by the NYSDEC (*circa* 1986) cross-gradient to the overburden ground water flow.

2.4. Site hydrogeologic conditions

Site hydrogeologic conditions indicate that the extent of the overburden aquifer is limited to the south, east, and west of RW-1. The limited nature of the aquifer is evidenced by an analysis of geologic conditions, saturated thicknesses, and ground water conditions observed while excavating the source area soils.

2.4.1. Geologic conditions

Site geologic conditions were assessed using the geologic descriptions presented in the RI Report (Stearns & Wheler, December 1993) and through additional work completed by O'Brien & Gere as part of the IRM.

The available geologic information indicates that the site geology consists of glacially derived overburden material consisting of varying amounts of sand, silt, clay, gravel, and cobbles overlying Upper Silurian sedimentary bedrock composed of shale to shaley dolostone.

The overburden stratigraphy consists of two distinct units as follows:

- 1. A more permeable silty fine to coarse grained sand and gravel unit; overlying
- 2. A less permeable, dense till composed of poorly sorted silt, clay, and gravel with varying amounts of sand.

Figure 4 presents a hydrogeologic cross-section A-A' (Figure 2) that extends from south to north across the site. As illustrated on Figure 4, the thickness of the overburden sand and gravel unit ranges from approximately 22 feet at MW-1 to approximately 40 feet at MW-5, generally increasing from south to north. Till thickness ranges from approximately 1 foot at MW-1 to approximately 9 feet at MW-10 and 11. Till was not encountered at MW-5 based on boring log information

Figure 5 presents a hydrogeologic cross-section B-B' (Figure 2) that extends from east to west across the northern portion of the site. The site overburden aquifer is thickest along an axis extending between MW-6 and MW-9 (approximately 47 ft to 53 ft) in the vicinity of RW-1. From this point, the aquifer thins out to 24 ft to the east in the vicinity of MW-7 and 20 ft to the west in the vicinity of MW-4.

As illustrated on Figure 5, the thickness of the sand and gravel unit in the northern portion of the site ranges from approximately 10 feet at MW-17 and MW-9 to approximately 24 feet at MW-6. The thickness of the till material ranges from approximately 2 feet at MW-17 to approximately 16 feet at PZ-2. Till was not encountered at MW-4 based on boring log information.

2.4.2. Overburden aquifer saturated thickness

The overburden aquifer saturated thickness, as expected, correlates with overburden geology. Figure 6 illustrates the site overburden saturated conditions as a whole. Figure 7 illustrates the saturated thickness of the sand and gravel overburden material.

Saturated conditions are thickest (approximately 27 ft) at RW-1 and MW-6 (Figure 6). From this point, saturation decreases toward the southern portion of the site with saturation ranging from approximately 6 feet at MW-13 to approximately 11 feet at MW-14 (Figure 6). Wells MW-10, 11, 15, and 16 are screened in bedrock, however boring logs completed by Stearns & Wheler indicate that the overburden was dry at the time of installation.

The saturated thickness of the overburden aquifer in the northern portions of the site ranges from approximately 0.5 feet at MW-17 to approximately 27 feet at MW-6 (Figure 6). Saturated conditions decrease towards the east to approximately 21 ft at MW-9.

The sand and gravel materials, being more permeable than the till unit, provide a preferential migration pathway for VOCs in the overburden aquifer. As a result, the relatively permeable sand and gravel unit should also contain the higher concentrations of contaminants. Comparison of the TCE plume (Figure 3) and the sand and gravel saturated thickness map (Figure 7) indicates that the distribution of VOCs correlates to the saturated sand and gravel thickness. In addition, as indicated on Figure 7, the saturated sand and gravel material is limited in areal extent to the northeast portion of the site.

The saturated thickness of the sand and gravel material is also limited at the southern portion of the site. As evidenced by dewatering operations during the excavation of source area soils, the overburden aquifer in the southern portion of the site does not yield significant quantities of ground water. As described in the Soil Remediation Activities IRM Summary Report (O'Brien & Gere Engineers, October, 1994) dewatering was achieved by initially pumping approximately 3,000 gallons to a holding tank inside the former Accurate Die facility. Subsequent to removing the initial volume of stored ground water, it was noted that overburden aquifer yield in this vicinity was only approximately 2 gpm. As indicated in section 1.2.1., overburden aquifer characteristics in this area are continuing to be evaluated.

2.5. Test recovery well installation

An 8-inch test recovery well RW-1 was installed approximately 15 feet south of MW-6 as illustrated in Figure 2. This location meets two design criteria:

- 1. Located in the thickest portion of the overburden aquifer to provide the greatest possible drawdown; and
- 2. Screens approximately 10 ft of saturated sand and gravel that contains VOCs to provide hydraulic control of the preferential migration pathway at the site.

RW-1 was installed to evaluate overburden characteristics and develop the necessary data for the design of a ground water recovery

system. RW-1 was drilled, installed, and developed by A.W. Kincaid Well Drilling during September and October 1994, under supervision of the on-site hydrogeologist and NYSDEC personnel.

RW-1 was installed using cable tool drilling methods. Initially, a 12inch diameter temporary steel casing was advanced to the bedrock surface at a depth of approximately 50 feet below grade. The 12-inch diameter borehole was then advanced approximately four feet into the bedrock to allow the screen to be set at the bottom of the overburden aquifer.

RW-1 was constructed with two screen sections, as illustrated on Figure 8. The purpose of this well design was to maximize recovery of ground water from both the sand and gravel and till portions of the overburden aquifer thereby increasing well efficiency.

As illustrated, RW-1 was constructed with a five foot carbon steel sump section from a depth of approximately 49 to 54 feet below grade. A bentonite slurry was emplaced around the sump section to provide a seal between the overburden and bedrock. A "shale packer" was fastened approximately six inches from the top of the sump section to aid in supporting the weight of the overlying sandpack, as well as to provide a seal between the overburden and bedrock. A ten foot length of 0.030-inch slot continuous wire wound stainless steel well screen (lower screen) was welded to the top of the sump section. The lower screen was installed from approximately 39 to 49 feet below grade. A six foot carbon steel blank section was welded to the top of the ten foot lower screen section. The six foot blank was installed from approximately 33 to 39 feet below grade. A five foot length of 0.030-inch slot continuous wire wound stainless steel screen (upper screen) was welded to the top of the six foot blank section. The upper screen was installed from approximately 28 to 33 feet below grade. Carbon steel casing was welded to the top of the upper screen to bring the well head approximately 2.5 feet above grade.

A silica sandpack compatible with a 0.030-inch slot screen was installed around the lower screened section from approximately 35 to 49 feet below grade. A bentonite seal was installed around the blank section separating the upper and lower screens from approximately 33.5 to 35 feet below grade. A sandpack was installed around the upper screen from approximately 21 to 33.5 feet below grade. A bentonite seal was installed from approximately 19 to 21 feet below grade. The remaining annular space was filled with a cement/bentonite grout from approximately 19 feet to grade.

Subsequent to installation, RW-1 was developed to optimize well efficiency. Development, consisting of surging and pumping methods, was conducted on September 16 and 19, 1994.

Surging was conducted within both the upper and lower screened sections. Subsequent to surging, sediment and sandpack material which were pulled into the well were removed using a bailer and contained on plastic sheeting near the well. The amount of sediment being pulled into the well decreased with each subsequent surging and bailing period.

The pumping portion of development was initiated when the coarse grained sediments ceased to enter the well during surging. The well was pumped at various rates between 15 and 60 gallons per minute (gpm) to assess the effectiveness of development and to provide a preliminary aquifer performance test flow rate.

Development water was contained in the on-site 50,000 gallon storage tank and subsequently treated by granular activated carbon (GAC) filtration. Treated ground water was discharged to the bank of Bishop Brook in accordance with the NYSDEC discharge requirements.

2.6. Aquifer performance test

A 24-hour aquifer performance test was conducted on September 28 and 29, 1994 to evaluate and select an optimal remedial flow rate for the recovery well, evaluate aquifer characteristics, and assess the influence of pumping on the existing shallow overburden and bedrock aquifers.

A 5 horsepower submersible pump was used to conduct the test. The test was performed at a flow rate of 25 gpm. An in-line flow meter (totalizer) was used to monitor flow rate. Measurements were recorded hourly throughout the test. Discharge was directed to a 50,000 gallon holding tank in the former Accurate Die facility.

Final: December 1, 1994 TME:bdm/TEC396.26 The following thirteen wells were selected to be monitored during the aquifer performance test:

Overburden Aquifer:	RW-1, PZ-1, PZ-2, MW-2
-	MW-4, MW-5, MW-6, MW-8
	MW-9, MW-12, and MW-17

Bedrock Aquifer:

MW-7 and MW-11

Prior to the start of the aquifer performance test, two rounds of water levels were recorded from the wells on September 26 and 27, 1994, to assess background conditions.

During the aquifer performance test, drawdown measurements were obtained using electronic water levels probes. Water level data were collected in the wells RW-1, MW-6, PZ-1, and PZ-2 at the following frequency:

Elapsed Time

Measurement Frequency

0 to 10 minutes 10 to 60 minutes 1 to 2 hours 2 to 24 hours Every minute Every 5 minutes Every 15 minutes Every hour

Wells at greater distances from the recovery well were monitored at least every hour. Upon completion of the aquifer performance test, recovery rates were measured in all wells which were influenced by pumping. The aquifer performance test drawdown data are included in Appendix C.

After 24 hours of pumping, drawdown in the overburden aquifer ranged from 0.09 feet at MW-5 to 0.47 feet at MW-6. Drawdowns of 1.15 and 0.03 feet were recorded in bedrock wells MW-7 and MW-11, respectively.

A water level decrease of 0.85 feet was recorded at MW-4, however, declines in water levels in MW-4 appear to be more related to rain events which may have increased water levels prior to the pump test. The water level in MW-4 was continuing to drop more than 24 hours after the pump was turned off. Wells MW-2 and MW-8 water levels continually increased during the test. The water level increases are

most likely the result of periodic rain events occurring prior to and during the test.

Drawdown data obtained from the monitoring points installed in the overburden aquifer indicate the area of influence created by pumping at RW-1 extends approximately 400 feet to the east-southeast and approximately 300 feet to the west (Figure 9). Periodic precipitation, prior to and during the test, influenced water level data obtained in the western portion of the site (MW-2, MW-8, and MW-4), therefore making it difficult to assess whether the zone of influence extends to these areas.

The bedrock aquifer appears to be hydraulically connected to the overburden aquifer as indicated by the drawdown recorded at MW-7. Drawdown in bedrock well MW-11 varied between 0.01 and 0.03 feet.

Aquifer test time-drawdown data was analyzed using the Cooper-Jacob straight-line method (unconfined conditions) for each well under the influence of pumping. The test data was also analyzed using Theis solutions for unconfined aquifer conditions contained in the AQTESOLVTM Aquifer Test Solver program Version 10 Documentation (Geraghty and Miller, Inc., 1988, 1989). Aquifer performance test analysis plots are included in Appendix B.

Aquifer transmissivity values were calculated for early and later portions of the time-drawdown curves. The early time-drawdown data indicate an average transmissivity of approximately 66,000 gallons/day/foot (gpd/ft), which appears to represent ground water storage in the sand and gravel materials.

The later time-drawdown data indicate an average transmissivity of approximately 25,000 gpd/ft that results from breaks in the timedrawdown curves approximately 250 to 300 minutes into the test. The breaks in the Jacob plots indicate a boundary condition was encountered during the aquifer performance test. This boundary condition likely reflects the limited areal extent of the saturated sand and gravel.

The decrease in aquifer transmissivity as indicated by time-drawdown data likely represents initial dewatering of the limited sand and gravel portion of the overburden aquifer. As such, transmissivity values should continue to decrease in conjunction with long-term aquifer pumping. Similar to conditions observed in the source area excavation, aquifer yield should decrease as ground water is removed from storage.

2.7. Conceptual hydrogeologic model and ground water recovery scenario

2.7.1. Conceptual hydrogeologic model

Ground water flow in the overburden aquifer is generally south to north toward Bishop Brook under a hydraulic gradient of approximately 0.04 feet/foot (RI Report, Stearns & Wheler, December 1993). Review of the site hydrogeologic conditions indicates a small saturated thickness in the southern portion of the site. This condition suggests limited upgradient recharge to the northern portion of the overburden aquifer.

Site hydrogeologic conditions also indicate decreasing overburden saturated conditions to the east and west of RW-1 in the northern portion of site. This condition, combined with a moderately steep ground water flow gradient, indicates a limited potential for lateral site overburden aquifer recharge from these areas.

Review of Figures 7 and 9 indicates that the aquifer performance test area of influence corresponds to the thick sand and gravel portions of the saturated overburden aquifer. This correlation suggests that RW-1 pumping at 25 gpm is apparently recovering ground water along the axis extending from MW-9 to MW-6. In addition, this area of influence encompasses a significant portion of the area containing VOCs in the ground water as shown in Figure 3, and indicates that hydraulic control will be achieved through pumping of RW-1.

RW-1 will unlikely recover water from the thinner portion of the aquifer to the west in the vicinity of MW-17 and MW-4. In addition, the small aquifer saturated thickness in this area suggests an additional recovery well would be an inefficient method of ground water recovery in this area.

2.7.2. Ground water recovery scenario

The focus of ground water recovery should be in the thickest portion of the saturated overburden aquifer. In addition, recovery should focus on recovering and controlling ground water in the sand and gravel unit, the preferential migration pathway.

Given the limited extent of the site overburden aquifer and the poor upgradient and lateral potential for recharge, ground water recovery should consist of initial dewatering of the northern portion of the aquifer along the axis of extending from MW-6 to MW-9 (Figure 6) with RW-1 pumping at 25 gpm. However, it should be noted that RW-1 discharge should decrease over time as the aquifer is dewatered.

Subsequent to recovery system design and implementation, a ground water monitoring program would be undertaken to monitor aquifer response to long-term pumping. This program would track dewatering rates and assess aquifer response to seasonal variations.

Based on the data obtained as a result of the monitoring program, an assessment of the system performance and effectiveness would be made. If necessary, modifications to the design or operation of the proposed ground water recovery and treatment system described in the following section would be made.

3. Ground water recovery and treatment system

3.1. General

This section presents a description of the ground water recovery and treatment system that is proposed to remediate overburden ground water at the site. This section also includes the data, calculations, and assumptions used during development of the Basis of Design. A process schematic of the proposed ground water recovery and treatment is presented as Figure 10. For convenience, this section discusses the major system components separately, as listed below:

- Ground Water Recovery and Conveyance system;
- Ground Water Treatment system;
- Instrumentation and controls; and
- Treated Water discharge

3.2. Ground water recovery and conveyance system

Test recovery well RW-1 will be utilized for ground water recovery. The construction details of this recovery well are presented on Figure 8. As presented in Section 2, a 24-hour pump test was performed on this recovery well on September 26 and 27, 1994. Based on the results of the 24-hour pump test, it is expected that this recovery well will initially produce ground water at a rate of 25 gpm. The longterm continuous well ground water production rate, however, is expected to be less than 25 gpm, as presented to in the preceding section. The recovery well pump proposed for use in RW-1 has the following characteristics, and was selected based on the criteria presented below:

Operating flow rate: 18 to 32 gpm

Pump performance points:	Capacity	Head
	(gpm)	(ft)
	10	90
	20	70
	25	60
	30	50
(* - design flow rate)		
Motor characteristics:	1⁄2 Hp, 3450 230 volt) rpm, 1 pH
	•	

11/2"

The recovery well pump inlet and controls will be set in accordance with the recovery well schedule presented below in Table 3.2-1.

Pump discharge pipe size:

Pump Water Intake Elevation	Water "Pump On" Elevation	Water "Pump Off" Elevation	Pump Failure Alarm Water Elevation	Redundant Pump Off and Alarm Water Elevation
478 ft	498 ft	485 ft	502 ft	482 ft

Table 3.2-1. Recovery Well Schedule

Recovered ground water will be conveyed to a 500 gallon equalization tank, located inside the main building approximately 500 feet from the recovery well, via a 2-inch diameter high density polyethylene (HDPE) pipe, as shown on Figure 11. The HDPE pipe will have a dimension ratio of 17.0 and a pressure rating of 100 psi.

The 500 gallon equalization tank will be equipped with the following:

Removable cover;

- 2 inch top entry water inlet;
- 1¼ inch water discharge (suction) pipe with foot valve. Discharge pipe inlet will be set 1 foot above the bottom of the tank;
- 2 inch tank vent pipe. The vent pipe will be vented outside the facility;
- 3 level float switches. The lower two float switches will be used to turn the equalization tank discharge pump on and off. The third and uppermost float switch will be used to trigger a tank full alarm and turn the recovery well pump off to prevent overfilling.

The equalization tank discharge pump shall be a close-coupled, selfpriming centrifugal type having the following characteristics:

Suction connection: 11/4" NPT

Discharge connection: 11/2" NPT

Motor characteristics: % HP, 230 volt, single phase.

Pump performance points:

-	Capacity (gpm)		Head (psi)
	39		20
	30		.30
-	11	•	40

Notes: Based on a suction lift of 10 ft. Design flow rate is 25 gpm.

3.3. Ground water treatment system

The recovered ground water will be treated by granular activated carbon (GAC) filtration. The treatment system will consist of two 1,000 pound capacity GAC vessels connected in series, the first

operating as the primary treatment vessel and the second operating as a backup treatment vessel. However, the GAC vessels will be headered together such that either can serve as the primary unit.

The ground water treatment system will be equipped with three water sample collection taps: one located upstream of the first GAC vessel, one between the two GAC vessels, and one downstream of the second GAC vessel. The ground water treatment system will also be equipped with two pressure gauges and one electronic pressure sensor. A pressure gauge and the electronic pressure sensor will be located upstream of the first GAC vessel, and the second pressure gauge will be located between the two GAC vessels.

The ground water treatment system proposed has the following characteristics, and was designed based on the criteria presented below:

Average influent flow rate: 25 gpm

<u>Pressure drop through each 1,000 lb. GAC vessel:</u> 3 psig (approx.) (at start up with flow rate of 25 gpm)

Influent trichloroethylene concentration: $500 \mu g/L$ to 5,000 $\mu g/L$

Anticipated GAC use rate: 5 lbs/day to 30 lbs/day

<u>Anticipated Primary GAC unit break-through time:</u> (@ 25 gpm, continuous flow) 1 to 6 months depending on VOC concentrations

Additional background water quality data is presented below in Table 3.3-1.

Table 3.3-1.	Ground	water	characteristics
--------------	--------	-------	-----------------

Compound	Concentration	-
Iron (total)	0.29	mg/L
Iron (filtered)	0.15	mg/L
Sulfate	9	mg/L
Calcium	77	mg/L
Total dissolved solids	280	mg/L

Final: December 1, 1994 TME:bdm/TEC396.26

Compound	Concentration	
Total alkalinity	260	mg/L
рН	7.3	
Calcium carbonate	192	mg/L
Corrosivity	-0.2	
Sodium	8	mg/L
Magnesium	24	mg/L
Hardness	290	mg/L
Specific conductance	570	µmho/cm
Total suspended solids	5	mg/L
Total organic carbon	2	mg/L
Methylene Blue Act. substances	<0.1	
Total Kjeldahl nitrogen (as N)	0.92	mg/L

Table 3.3-1. Ground water characteristics

Note: Data based on sample obtained during pump test. Laboratory Report included as Appendix D.

3.4. Instrumentation and controls

The ground water recovery and treatment system will be equipped with instruments and controls that will allow it to operate continuously in an unattended mode with only occasional monitoring.

3.4.1. Ground water recovery well pump

The ground water recovery well will be equipped with a ground water level transducer connected to the pump controls and alarm. Four ground water elevation settings will be provided to provide the following described functions:

- High water setting to trigger an alarm to notify the operator that the recovery well pump may not be operating;
- Pump on setting;
- Pump off setting; and
- Low water setting to trigger an alarm to notify the operator that the recovery well pump may not have turned off. The setting will trip a relay opening the power circuit to the pump.

The power circuit to the recovery well pump will also be wired so that it is open whenever an alarm condition has been triggered by the tank full float switch placed inside the equalization tank.

3.4.2. Water flow meter, strip chart recorder and flow totalizer A flow meter with flow totalizer will be installed between the equalization tank and first GAC unit to monitor the quantity of ground water that has been recovered and treated.

3.4.3. Pressure sensor

A pressure sensor will be installed immediately upstream of the primary GAC vessel and will be set to trigger a high pressure alarm at a preset level. Such an alarm condition will disconnect power to the recovery well and equalization tank pumps.

3.4.4. Alarm notification devices

Since the ground water treatment facility will generally be unmanned, an audible alarm device will be installed to alert the operator of alarm conditions at the site. The control panel will also be equipped with indicator lights that will indicate the current operating status of the recovery well pump and other system components.

3.5. Treated water discharge

It is proposed that treated water be discharged, via a 2 inch diameter HDPE pipe, to either a nearby storm drain or to Bishop Brook, as shown on Figure 11. The treated water will be sampled twice per month and monitored for compliance with the Effluent Limitations and Monitoring Requirements established by the NYSDEC, under State Pollutant Discharge Elimination System (SPDES) permit #734052. A copy of the Effluent Limitations and Monitoring Requirements sheets are provided as Appendix E.

Respectfully submitted,

O'BRIEN & GERE ENGINEERS, INC.

David G. Van Arnam, P.E. Senior Vice President

Prepared by:

David J. Carnevale Hydrogeologist

Timothy M. Eddy Project Hydrogeologist

Alfred Farrell, P.E. Project Engineer

David S. Towers, P.E. Senior Project Engineer

Reviewed by:

James R. Heckathorne, P.E. Managing Engineer

Deborah Y. Wright, CPG Managing Hydrogeologist
TABLES

		11011									
	Ground	Casing	Screet	ned	Interval	GROUND	WATER EL	EVATIONS	3		
Well No.	Elevation	Elevation	Ele	eva	tion	5/28/92	6/26/92	8/7/92_	9/26/94	9/27/94	10/18/94
					·						
MW-1	99.36	101.11	75.4	-	85.4	DRY	DRY	79.69	NR	NR	DRY
MW-2	91.8	94.68	76.6	-	86.6	83.21	82.81	84.32	83.10	83.28	80.12
MW-3	97.65	99.63	73.7	-	83.7	80.44	80.09	81.63	ABDN	ABDN	ABDN
MW-4	65.62	68.52	46.6	-	56. 6	51.08	49.95	50.81	47.22	52.21	46.79
MW-5	88.21	90.42	49.2	÷	59.2	60.71	63.76	61.22	59.87	59.91	59.45
MW-6	77.46	79.38	46.4	-	56.4	60.50	60.49	60.46	59.51	59.52	59.05
MW-7(B)	75.66	78.34	34.3	-	44.3	54.59	54.55	54.47	53.90	53. 97	53.55
MW-8	88.21	91.78	53.9	-	63. 9	66.38	66.38	66.83	61.59	61.65	60.99
MW-9	102.44	104.03	49.7	•	59.7	60.46	60.51	61.83	59.57	59.59	59.08
MW-10(B)	97.51	99.69	43.03	-	53.03	61.15	61.99	61.69	NR	NR	56.02
MW-11(B)	91.48	93.80	43.1	-	53.1	62.34	63.70	63.66	58.41	58.39	57.47
MW-12	93.62	94.14	51.9	-	61.9	62.24	60.74	62.77	59.77	59. 79	59.31
MW-13	98.73	100.92	77.7	-	87.7	DRY	80.62	80.92	NR	NR	78.70
MW-14	98. 76	100.62	74.6	-	84.6	75.11	79.07	81.54	NR	NR	86.18
MW-15(B)	96.1	98.90	32.7	-	42.7	NI	NI	NI	NR	NR	53.47
MW-16(B)	98.5	100.85	50.8	-	60.8	NI	NI	NI	NR	NR	61.67
MW-17	66.9	69.24	53.7	-	63.7	NI	NI	NI	54.61	54.61	54.08
PZ-1	81.8	83.95	49.8	-	59.8	NE	NI	NI	59.56	59.57	59.10
PZ-2	80.6	83.06	42.8	-	52.8	NI	NI	NI	59.35	59.36	58.89
RW-1	78.4	80.28	29.4-39.4	1	45.4-50.4	NI	NI	NI	56.88	56.89	58.22

TABLE 1 MONITORING WELL SPECIFICATIONS ACCURATE DIE CASTING FAYETTVILLE, NY

Notes: (B) - Bedrock well.

NI - Well not installed at time of monitoring.

NR - Water level not monitored.

المالا

ABDN - Abandoned.

MW-1 through MW-16 installed during Remedial Investigation (Stearns & Wheler).

Table 2

Ground Water TCE Concentrations

		Trichloro	ethylene Conce	ntrations (1)		
Date Sampled:	8/30/89	12/4/89	5/20/90	5/28/92	7/22/94	10/18/94
Monitoring Well (3)						
MW-1	112	ND	2	ND	NS	Dry
MW-2	ND	ND	1	ND	NS	ND
MW-3	Free Product	>55,000	440,000	340,000	Abandoned	Abandoned
MW-4	NS	7	43	6	270	23
MW-5	NI	340	344	110	330	410
MW-6	NI	700	454	510	390	360
MW-7 (B)	NI	ND .	ND	ND	ND	ND
MW-8	NI	ND	ND	ND	NA	ND
MW-9	NI	109	106	60	72	74
MW-10 (B)	NI	NI	NI	4,500	1,600	1300
MW-11 (B)	NI	NI	NI	5,200	5,500	5300
MW-12	NI	NI	NI	36	44	35
MW-13	NI	NI	NI	110	740	510
MW-14	NI	NI	NI	67	150	120
MW-15 (B)	NI	NI	NI	NI	NS	14
MW-16 (B)	NI	<u>NI</u>	NI	NI ·	NS	6
MW-17	NI	NI	NI	NI	260	140
Excavation Sump	NI	NI	NI	NI	20,000	NS

Accurate Die Casting Facility Fayetteville, NY

Notes: ND - Not detected at concentrations greater than analytical detection limit.

NS - Not sampled.

NI - Well not installed at time of sampling.

NA - Not analyzed.

(1) - Concentrations reported in ug/L (ppb).

(2) - Sample collected 8/19/92 because MW-13 and MW-14 were dry on 5/28/92.

(3) - Monitoring wells MW-1 through MW-16 installed by Stearns & Wheler, monitoring well MW-17 installed by O'Brien & Gere Engineers, wells MW-1, MW-7, MW-10, MW-11, MW-15, and MW-16 are bedrock groundwater monitoring wells.

(B) - Bedrock well.

FIGURES

.









.















ENGINEERS, NC.





FIGURE 10



 \bigcirc

LEGEND

BALL VALVE

NORMALLY CLOSED

PRESSURE GAUGE

SAMPLE COLLECTION

PRESSURE SENSOR

FLOW METER W/ TOTALIZER

FLOW DIRECTION (NORMAL)

ACCURATE DIE CASTING FAYETTEVILLE, NEW YORK BASIS OF DESIGN

SYSTEM PROCESS SCHEMATIC

NOT TO SCALE

FILE NO. 2488.396-20F





APPENDIX A

.

O'BR	BRIEN & GERE ENGINEERS, INC			INEERS.	INC.	TEST BORING LOG	REPORT OF BORING					
Client:		OBG T	annical S	Services		Sampler: 2" Split Spoon	Page 1 of	f 2				
							Location	: 50'E o	f MW-6			
Proj. L	.oc:	Accura	te Die C	asting		Hammer: 140 lbs.	Start Dat					
File No	o.:	2488.3	96.523			Fall: 30"	End Date	: 8/2/94	•			
Boring	Com	pany:	OP-TEC	CHH		·····	Screen		Grout			
Forem	an:		Steve L	.aramee		• •	Riser		Sand Pac			
	ieolog	gist:	DJ Car	nevale		·····	Chatting		Bentonite			
Depth							Change		Testing			
Below		Depth	Blows	Penetr/	-N-	Sample Description	General	Equip.	HN			
Grade	No.	(feet)	/6"	Recovery	Value		Descript	Installed	(PF			
0	1	0-2	2-6	2/2	12	Dry to damp, brown, very fine to fine SAND,		V V	0			
		i	6-6	(0920)		little silt, 1 pc gravel in tip of shoe.						
2	2	2-4	2-5	2/1	18	Damp, dark brown SILT, little very fine		A K	1.			
			13-18	(0925)		sand, shaly rock in shoe.		N N				
3								1 1				
				64 7	<u></u> -							
	3	4-6	8-11 36/05	2/1.5		Dry, gray brown tine SAND and SiL1, little			13			
5			00/0.0	(0040)		shoe, cobbles in cutting.						
								A A				
6	4	6-8	5-7	2/2	15	Dry to damp, fine to coarse SAND, little silt		۱ ۱	10			
			8-7	(1010)		and fine to medium gravel, cobbles in cutting.						
					·							
8	5	8-10	4-7	2/2	14	Damp to moist, as above, trace very						
			7-7	(1015)		coarse GRAVEL, cobbles in cutting.		A A				
9								1 1				
- 10		10.10		0.40	10			N N				
10	•	10-12	7-16	(1100)	13	sand little coarse sand trace fine broken						
11				(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		gravel.		N N				
								Λ Λ				
12	7	12-14	6-9	2/2	15	Damp to moist, brown gray, medium to		N N	11			
13			6-6	(1120)		coarse SAND, some tine sand, little medium to						
14	8	14-16	4-8	2/2	14	Moist, brown gray, fine SAND, some medium		$\Lambda = \Lambda$	10			
			6-5	(1145)		sand, trace silt.		1 1				
15			 			4						
16	9	16-18	3-4	2/2	8	Moist to wet, grav brown fine SAND, some						
			4-4	(1255)	<u> </u>	medium sand, little to trace of coarse sand.		K K				
17												
18	10	18-20	5-5	2/2	9	woist, gray brown, fine to medium SAND,			2			
19			<u> </u>			at 19.5'						
			i —			j						
20	11	20-22	3-4	2/2	9	Wet, gray brown, fine to medium SAND,			9			
- 01			5-6	(1350)		little coarse sand and silt, some fine gravel						
21						21.5' to 22', brown clayey sand in tip of						
22	12	22-24	5-5	2/2	12	Wet, gray brown, fine to medium SAND. little			3			
			7/16	(1425)	<u> </u>	fine to medium gravel, coarse sand, silt,		-				
23						trace to little clay red to brown clay in		=				
			l		l	tip of shoe at 24'		i = i i i i i i i i i i i i i i i i i i				

.

						TEST BORING LOG	REPC	ORT OF	BORING
O'BR	IEN	& GER	E ENG	SINEERS,	INC.			PZ-1	
Client:		OBG Te	hnical S	Services		Sampler: 2" Split Spoon	Page 2 o	f 2	(MW-6
Proj. L	oc:	Accura	le Die C	asting		Hammer: 140 lbs.	Location	. JUE 0	U 141 14-0
File No	.:	2488.39	96.523			Fall: 30"	Start Date	e: 8/1/94 : 8/2/94	•
Boring	Con	pany:	OP-TEC	ж			Screen	=	Grout
Forem	an:	alet	Steve L	arance			Riser		Sand Pack
	0010	yısı.					Stratum		Field
Depth							Change		Testing
Below		Depth	Blows	Penetr/	"N"	Sample Description	General	Equip.	HNU
Grade	No.	(feet)	/6"	Recovery	Value	· · · · · · · · · · · · · · · · · · ·	Descript	Installed	(ppn)
24	13	24-26	3-1	2/2	6	Wet, gray brown fine to medium SAND, some		=	3.8
			5-7	(1450)		silt, little coarse sand, trace to little			
25						c:ay			
26	14	26-28	9-19	2/2		Wet, as above to 26.5' to red to brown CLAY			3.8
			26-60/.5	(1705)		till, very fine gravel to 26.8 ft., to gray		-	
27						fine sand, some medium sand, silt, clay and		=	
						broken gravel shale fragments.		=	
28	15	28-30	27-55	2/2		Wet, brown gray fine SAND, some medium sand,		- 🛛	4
			60/.5	(1705)		little coarse sand, to 28.7' to gray green			
29						weathered shale to 29.2' to gray brown tine			
					ļ	to medium broken faceted rocks fragments			
30	16	30-32	50/6	2/0		No recovery	l .		
		····		(0815)		1	1	=	
31]		=	
]	=	
32	17	32-34	64/0	2/0		No recovery as above	l		-
			ļ	<u> </u>		4 ¹ ·			
- 33						B O B = 32'			· _
34						1 − − − −			
						1			
37]			
						Į .			
38				·	l	4.			
30						4			
~~						4			
40						1			
						1	1		
41]	· ·		
						4			
42						4			
43						4			
				<u> </u>	<u> </u>	1			
44				······		1			
]			
45]	l		
		ļ				4	l		
46		 			· ·	4]		
47						4	· ·		
-11						1			
	in eer	een 32-22	2	Bentonite 19.	6- 17.0	A	· · · · · · ·	ι	<u>بر معرفة المحمد الم</u>

٠

· .

						TEST BORING LOG	REPC	RT OF	BORI	ING
O'BR	IEN	& GEF	RE ENC	GINEERS,	INC.		{	PZ-2		
Client:		OBG T	ehnical s	Services		Sampler: 2" Split Spoon	Page 1 of	12 NE024		N.C
Proj. L	.oc:	Accura	te Die C	asting		Hammer: 140 lbs.	Location	NOUM		0-11
- File Nr	. .	Fayette	ville 26.523	-		Fall: 30"	Start Dat	e: 8/3/94 : 8/4/94	1	
Boring	<u>, Con</u>	pany:	OP-TEC	ж			Screen		Grout	t
Forem OBG G	an: Seolo	aist:	Steve L	.aramee nevale			Risør		Sand Bento	Pa onit
							Stratum	·	Fie	eld
Depth Below		Denth	Blows	Penetr/	-N-	Sample Description	General	Equip.	Tes	stin I HN
Grade	No.	(feet)	/6"	Recovery	Value		Descript	Installed		(PI
0	1	0-2	2-3	2/2	6	Damp, medium brown, and fine SAND.				N
1			3-3	(0705)	<u> </u>					
								N N		
2										
3								N N		
							1.			
						1				
5	2	5-7	2-4	2/2	7	Moist, gray brown fine SAND, trace little silt.				0
6			3-3	(0720)						
								N N		
7						• · · ·				
8										
0						4				
					·	1			· ·	1
10	3	10-12	5-5	2/2	10	Moist to wet, gray brown, fine to coarse sand		N N		1.
11			5-7	(0735)		and very tine to medium gravel.				;
								N N		ļ
12			<u> </u>]
13								A A		
14			 							
								N N		
15	4	15-17	6-7	2/2	16	As above				0.
16				(0000)		1				
17						•				
						1				1
18						}				
19	<u> </u>		<u> </u>		<u> </u>	1				
		00.00								
20	5	20-22	12-10	2/2 (0830)	22	ver, reddish brown clayey silt matrix till, some fine to medium sand and very fine gravel.				0.
21						.		N R		1
22	6	22-24	8-22	2/2	53	Wet, grav brown till of claving SII T and fine				
	Ŀ		31-30	(0840)		sand, some medium sand broken shale				
23						fragments to medium gravel.				1

.

•

- ,

,

						TEST BORING LOG	REPC	ORT OF I	BORING
O'BRI	IEN	& GEF	E ENG	SINEERS,	INC.			PZ-2	
Client:		OBG T	ehnical s	Services		Sampler: 2" Split Spoon	Page 2 o Location	f 2 : N50'W	of MW-6
Proj. L	oc:	Accura Favette	te Die C wille	asting		Hammer: 140 lbs.	Start Dat	e: 8/3/94	
File No	».:	2488.39	96.523	•		Fall: 30"	End Date	: 8/4/94	
Boring	Соп	npany:	OP-TEC	CH C			Screen		Grout
Forema OBG G	an: ieolo	aist:	Steve L DJ Carl	.aramee nevale			Riser		Sand Pack Bentonite
							Stratum		Field
Depth Below		Depth	Blows	Penetr/	"N"	Sample Description	General	Equip.	Testing HNU
Grade	<u>No.</u>	(feet)	/6*	Recovery	Value		Descript	Installed	(ppn)
24	7	24-26	25-60/.2	0.5/0.5		As of previous page			3.8
				(90920)				55555 355 58	
25									
28	8	26-28	52-85/3	05/05		Wet molted reddish brown to greenish gray			78
	v	20-20	52-00/.0	(1040)		brown claving SiLT till with fine cand and fine	(
27				(1040)		rounded gravel, some little to medium sand and			
- <u></u>	·	I			<u>├</u>	broken shale fragments with medium gravel.		—	
28	9	28-30	55/0.2	0.5/0.5		wet, as above.	l	=	5.8
		····	<u> </u>	(1100)			ľ		
29		i		· · ·				=	
								- III - III	
30	10	30-32	98-135	1/1.0		Wet, brown, fine sandy till with fine rounded		=	3.6
				(1145)		gravel, little medium sand, some sitt.			
31]		
								=	
32	11	32-34	21-30	2/	63	Wet, gray, very fine sandy till, fine rounded		=	28
			33-19	(1305)		GRAVEL, solvent odor.		SS = SS	
33			<u> </u>					=	
			- 10					=	
- 34	12	34-35	3-10	1.5/1.5	90	wet, brown very tine SAND and SILT with		S _ S	22
- 35				(1330)		shale fragments in fine andy silt with fine		፟	
				<u>├</u>		rounded gravel, solvent odor, black, moist to			1
36	13	36-38	63-80/0.	0.7/0.7		wet, weathered shale.			3.8
				1550					
37				1					
						· · ·			
38	14	38-40	53-36	2/	55	Black, weathered SHALE wet			1.8
			19-16	(1615)					
39									
		l			ļ	B.Q.B = 40'	1		
40		<u> </u>	┝──	<u> </u>	ļ				
- <u></u>					ļ				
		•••		·			ľ		
42				·					
43									
1									
44							1		
			ļ						
45		<u> </u>		<u> </u>	ļ				
		l	 						,
+0					└ <u>·</u>				
47		<u> </u>	i	1	<u> </u>				
		<u> </u>	<u> </u>						
								-	

.

-

:

'BR	IEN	& GEF			INC.			MW-17	
lent:		OBG Te	ennical S	Services		Sampler: 2" Split Spoon	Page 1 of	f 1	
ol. L	oc:	Accura	te Die C	asting		Hammer: 140 lbs.	Location	:	
		Fayette	ville, N.	Υ.		Ealta 201	Start Dat	e: 8/4/9	4
	J.: J Com	2400.3	OP-TEC	хн			Screen		Grout
orem	an:		Steve L	.aramee			Riser		Sand Pack
8G (16010	gist:	DJ Carl	nevale		·····	Stratum		Field
epth		-					Change	-	Testing
elow rade	No.	Depth (feet)	Blows /6"	Penetr/ Recoverv	"N" Value	Sample Description	Descript	Equip.	(ppn)
0	1	0-2	1-1	2/2		Moist, brown, fine sand, little silt.			7.2
1			2-1	(1245)					
<u> </u>									
2									
3			<u> </u>						
4									
5	2	5-7	3-5	2/2 ·	10	Moist, gray brown, fine to medium SAND,			11.4
8		i	5-11	(1315)		coarse sand, little to trace of very fine		=	
<u> </u>								=	
7								=	
8								=	
0								=	
8								=	
10	3	10-12	8-19	2/2	47	Wet, grayish purple till (SILT) with fine to		=	9.8
11			28-27	(1340)		11.2', to similar till but brown, moist with some		=	
10						coarse GRAVEL and shale fragments to 12.		=	
12	4	12-14	8-19 31-70	(1415)	64	Moist to wet, gray green weathered SHALE.		-	38
13]		=	
14	5	14-16	5-45	2/	102	Moist, gray brown weathered SHALE.			56
			57-26	(1450)					
15				•					
16			•						
17		· ·	[
			·						
18									
19				Í					
20						4			
						· · · ·			
21			<u> </u>						
22									
22									
E J		╞────	ł	<u> </u> _					

APPENDIX B

.



Volatile Organics Method 8010/8020

LABORATORIES, INC.

OBG TECHNICAL SERVICES, INC.

2488.396.517 JOB NO.

CLIENT.

Att in the case

Accurate Die Casting

DESCRIPTIONACCUTATE					1.7	
7.00.04				MATRIX:	Water	<u> </u>
DATE COLLECTED	DATE RI	ECEIVED	-21-94	DATE ANAL	YZED7-25,	26-94
DESCRIPTION:	MW-7	MW-12	MW-6	MW-5	MW-9	MW-13
SAMPLE NO.:	U0138	U0139	· U0140	U0141	U0142	U0143
BENZENE	<1.	<1.	<10.	<10.	<1.	<10.
BROMODICHLOROMETHANE	<1.	<1.	<10.	<10.	<1.	<10.
BROMOFORM	<10.	<10.	<100.	<100.	<10.	<100.
BROMOMETHANE	<10.	<10.	<100.	<100.	<10.	<100.
CARBON TETRACHLORIDE	<1.	<1.	<10.	<10.	<1.	<10.
CHLOROBENZENE	<1.	<1.	<10.	<10.	<1.	<10.
CHLOROETHANE	<1.	<1.	<10.	<10.	<1.	<10.
2-CHLOROETHYLVINYL ETHER	<10.	<10.	<100.	<100.	<10.	<100.
CHLOROFORM	<1.	<1.	<10.	<10.	<1.	<10.
CHLOROMETHANE	<10.	<10.	<100.	<100.	<10.	<100.
DIBROMOCHLOROMETHANE	<1.	<1.	<10.	<10.	<1.	<10.
1,2-DICHLOROBENZENE	<5.	<5.	<50.	<50.	<5.	<50.
1,3-DICHLOROBENZENE	<5.	<5.	<50.	<50.	<5.	<50.
1,4-DICHLOROBENZENE	<5.	<5.	<50.	<50.	<5.	<50.
DICHLORODIFLUOROMETHANE	<10.	<10.	<100.	<100.	<10.	<100.
1,1-DICHLOROETHANE	<1.	<1.	<10.	<10.	<1.	<10.
1,2-DICHLOROETHANE	<1.	<1.	<10.	<10.	<1.	<10.
1,1-DICHLOROETHYLENE	<1.	<1.	<10.	<10.	<1.	<10.
1,2-DICHLOROETHENE (total)	<1.	<1.	<10.	<10.	<1.	<10.
DICHLOROMETHANE	<1.	<1.	<10.	<10.	<1.	<10.
1,2-DICHLOROPROPANE	<1.	<1.	<10.	<10.	<1.	<10.
cis-1,3-DICHLOROPROPYLENE	<1.	<1.	<10.	<10.	<1.	<10.
trans-1,3-DICHLOROPROPYLENE	X1 .	<1.	<10.	(10.	a.	<10.

Page 1 of 2

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

Authorized: August 4, 1994 Date:



Volatile Organics Method 8010/8020

OBG TECHNICAL SERVICES, INC.

JOB NO.

2488.396.517

DESCRIPTION

CLIENT.

Accurate Die Casting

				MATRIX:	Water	
DATE COLLECTED7-20-94	DATE RE	CEIVED7-2	21-94	DATE ANALY	YZED7-25,2	26-94
DESCRIPTION:	MW-7	MW-12	M₩-6	MW-5	MW-9	MW-13
SAMPLE NO.:	U0138	U0139	U0140	U0141	U0142	U0143
ETHYLBENZENE	<1.	<1.	<10.	·<10.	<1.	<10.
1,1,2,2-TETRACHLOROETHANE	<1.	<1.	<10.	<10.	<1.	<10.
TETRACHLOROETHYLENE	<1.	<1.	<10.	<10.	<1.	<10
TOLUENE	<1.	<1.	<10.	<10.	<1.	<10.
1,1,1-TRICHLOROETHANE	<1.	<1.	<10.	<10.	<1.	<10.
1,1,2-TRICHLOROETHANE	<1.	<1.	<10.	<10.	<1.	<10.
TRICHLOROETHYLENE	<1.	44.	390.	330.	72.	740.
TRICHLOROFLUOROMETHANE	<1.	<1.	<10.	<10.	<1.	<10.
VINYL CHLORIDE	<1.	<1.	<10.	<10.	<1.	<10.
XYLENE (total)	<3.	<3.	<30.	<30.	<3.	<30.
 A substant of the second system of the			· -	• •	···• • · ·	•
المالية المالية المالية المتلقة المتحاطية المستعملية والمرابع		, .		·		
• • • • • • • • • • • • • • • • • • • •			-		•	
		· .		-		
			•			-
needer all an anna anna anna anna anna anna a						· ·
ատուտաց առուլ իրկրվում է որդքները է է է է է է է է է է է է է է է է է է է			• •			ì
		· · · · · · · · · · · · ·				•
				-	,	
a and the state of		and the second second			1 m 1 m 1 m	

Comments:

.....

._____.

Methodology: USEPA.SW - 846, November 1986, 3rd Edition Certification No.: 10155 $\mu g/1$ Units:

Page 2 of 2 Authorized: August 4, 1994 Date:_

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300. Box 4942 / Syracuse, NY 13221 / (315) 437-0200



CLIENT_

Volatile Organics Method 8010/8020

JOB NO.

2488.396.517

Accurate Die Casting

OBG TECHNICAL SERVICES, INC.

DESCRIPTION				MATRIX:	Water	
DATE COLLECTED 7-20,21-94	DATE RE		-21-94	DATE ANAL	YZED 7-2	5,26,27-94
DESCRIPTION:	MW-14	MW-4	MW-11	MW-10	Sump	QC Trip
SAMPLE NO.:					Pump	
	U0144	U0145	U0146	U0147	U0148	U0149
BENZENE	<10.	<10.	<100.	<50.	<500.	<1.
BROMODICHLOROMETHANE	<10.	<10.	<100.	<50.	<500.	<1.
BROMOFORM	<100.	<100.	<1000.	<500.	<5000.	<10.
BROMOMETHANE	<100.	<100.	<1000.	<500.	<5000.	<10.
CARBON TETRACHLORIDE	<10.	<10.	<100.	<50.	<500.	<1.
CHLOROBENZENE	<10.	<10.	<100.	<50.	<500.	<1.
CHLOROETHANE	<10.	<10.	<100.	<50.	<500.	<1.
2-CHLOROETHYLVINYL ETHER	<100.	<100.	<1000.	<500.	<5000.	<10.
CHLOROFORM	<10.	<10.	<100.	<50.	<500.	<1.
CHLOROMETHANE	<100.	<100.	<1000.	<500.	<5000.	<10.
DIBROMOCHLOROMETHANE	<10.	<10.	<100.	<50.	<500.	<1.
1,2-DICHLOROBENZENE	<50.	<50.	<500.	<250.	<2500.	<5.
1,3-DICHLOROBENZENE	<50.	<50.	<500.	<250.	<2500.	<5.
1,4-DICHLOROBENZENE	<50.	<50.	<500.	<250.	<2500.	<5.
DICHLORODIFLUOROMETHANE	<100.	<100.	<1000.	<500.	<5000.	<10.
1,1-DICHLOROETHANE	<10.	<10.	<100.	<50.	<500.	<1.
1,2-DICHLOROETHANE	<10.	<10.	<100.	<50.	<500.	<1.
1,1-DICHLOROETHYLENE	<10.	<10.	<100.	<50.	<500.	<1.
1,2-DICHLOROETHENE (total)	<10.	42.	<100.	<50.	<500.	<1.
DICHLOROMETHANE	<10.	<10.	<100.	<50.	<500.	<1.
1,2-DICHLOROPROPANE	<10.	<10.	<100,	<50.	₹500.	<1.
cis-1,3-DICHLOROPROPYLENE	<10.	<10.	<100.	<50.	<500.	<1.
trans-1,3-DICHLOROPROPYLENE	×10	11	<100	C-0	<i>₹</i> ; ₹500.	<1.

Page 1 of 2

Authorized: August 4, 1994 Date: _

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

.

.



Volatile Organics Method 8010/8020

OBG TECHNICAL SERVICES, INC. CLIENT_

2488.396.517 JOB NO.

	MATRIX: Water										
DATE COLLECTED	DATE RE	ECEIVED 7-2	21-94	DATE ANAL	YZED 7-25	,26,27-94					
DESCRIPTION:	MW-14	MW-4	MW-11	MW-10	Sump Pump	QC Trip Blank					
SAMPLE NO.:					F						
ETUVI DENZENE	U0144	U0145	U0146	U0147	U0148	U0149					
	<10.	<10.	· <100.	<50.	<500.	<1.					
1,1,2,2-TETRACHLOROETHANE	<10.	<10.	<100.	<50.	<500.	<1.					
TETRACHLOROETHYLENE	<10.	<10.	<100.	<50.	<500.	<1.					
TOLUENE	<10.	<10.	<100.	<50.	<500.	<1.					
1,1,1-TRICHLOROETHANE	<10.	<10.	<100.	<50.	<500.	<1.					
1,1,2-TRICHLOROETHANE	<10.	<10.	<100.	<50.	<500.	<1.					
TRICHLOROETHYLENE	150.	270.	5500.	1600.	20,000.	<1.					
TRICHLOROFLUOROMETHANE	<10.	<10.	<100.	<50.	<500.	<1.					
VINYL CHLORIDE	<10.	<10.	<100.	<50.	<500.	<1.					
XYLENE (total)	<30.	<30.	<300.	<150.	<1500.	<3.					
				· · · ·							
an a											
الي الي المانية المحمد محمد محمد المعموم موجوع المحمد المعموم المراكز المراكز المحمد المحمد المحمد المحمد المح المراكز المراكز المحمد المحم المراكز المراكز المحمد المح				• • •	· · · · · · · · · · · · · · · · · · ·						
					· · ·	· · ·					
						· · · · ·					
					معرف المعرف الم	· · · · · · · · · · · · · · · · · · ·					

Comments:

Methodology: USEPA.SW-846, November 1986, 3rd Edition 10155 **Certification No.:** µg/1 Units:

Page 2 of 2 Authorized: August 4, 1994 Date:

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

Job No. <u>2488 - 396 - 53</u>1 Sheet _____ of ____



Office:	SVRACUSE, NY
Address:	5000 Britton Field PKWy
Phone:	437-6100

CHAIN OF CUSTODY

e Cast Ile, N	ing	COLLECT (Signature)	ED BY: A.	ete La	retto
Date	Time	Sample Matrix ¹	Sample Type ²	No. of Containers	ANALYSIS REQUESTED
7/20	0950	water	GMB	2.	
7/20	1030			2	
7/20	1050			2	
7/20	1115			. 7	
2/20	R35			2	
7/20	1255			2	
7/20	1315			2	5 8010/8020
7/21	1030			2	<u> \</u>
7/21	1125			2	
1/21	11 55			2	
7/20					80/0/8020
7/20	1345	unter	912-3	2	8010 / 8020
					· · · ·
	e CASH ile, N Date 7/20	$\begin{array}{c c} CASFing\\ Ile, NY\\ \hline \\ Date \\ \hline \\ Time\\ \hline \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/20 \\ 7/21 \\ 7/20 \\ 7/21 \\ 7/20 \\ 7/21 \\ 7/20 \\ 7/21 \\ 7/20 \\ 7/21 \\ 7/20 \\ 7/20 \\ 7/21 \\ 7/20 \\$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	e $CAshing$ COLLECTED BY: D_{12} $1 e_1 NY$ (Signature) D_{11} Date Time Sample Sample Date Time Matrix ¹ Type ² $7/20$ 0950 WALE $GraB$ $7/20$ 1050 1050 1050 $7/20$ 1050 1050 1050 $7/20$ 1050 1050 1050 $7/20$ 1050 1050 1050 $7/20$ 1050 1050 1050 $7/20$ 1255 1050 1050 $7/20$ 1255 1050 1050 $7/20$ 1255 1050 1050 $7/20$ 1315 1050 1050 $7/21$ 1125 1125 1125 1125 $7/20$ 1345 4042 972 $7/20$ 1345 4042 972 $7/20$ 1345 4042 972	e CASHING COLLECTED BY: Defe Law $1 e_1 NY$ (Signature) $11e_1 NY$ $11e_1 NY$ $11e_1 NY$ Date Time Sample Sample No. of Date Time Matrix ¹ Type ² Containers $7/20$ 0950 UNLer GAB 2 $7/20$ 1050 2 2 $7/20$ 1050 2 2 $7/20$ 1050 2 2 $7/20$ 1255 2 2 $7/20$ 1255 2 2 $7/20$ 1315 2 2 $7/21$ 1/25 2 2 $7/21$ 1/25 2 2 $7/20$ 1315 2 2 $7/20$ 1345 $4Mer$ $5mr$ 2 $7/20$ 1345 $4Mer$ $5mr$ 2

² Type = grab, composite

Relinquished by: Dut fart	Date	Time	Received by: Alldy Smith	Date	Time
of OBG Engineers	7/21	1340	* OBG Laboratories the.	1.4.94	1340
Relinquished by:	Date	Time	Received by:	Date	Time
of:			ot:		
Relinquished by:	Date	Time	Received by:	Date	Time
of:			of:		
Use this space if shipped via courier (e.g., Fed Ex)	Date	Time	Courier Name:	Date	Time
ot.			*Attack delivery/courier receipt to Chain of Castody	50. %: X 	
Relinquished by:	Date	Time	Received by:	Date	Time
of:			of:		

Bark Coller Hongs. C 3°C+5°C; rec'd. 2 Mals OC. The separate Blark One vial multo not labelled - matched w/ Hime Callected.

, September 17, 1990



Volatile Organics Method 8010/8020

LABORATORIES, INC.

ACCURATE DIE CASTING CLIENT_

3435.021.517 JOB NO.

Fayetteville, NY DESCRIPTION

	<u> </u>		- <u></u>	MATRIX:	Water	
DATE COLLECTED 10-18-94	DATE R	ECEIVED	10-18-94	DATE ANALY	YZED	1,25-94
DESCRIPTION:	MW-8	MW - 2	MW-6	MW-17	MW-5	MW-9
SAMPLE NO .:	U4028	U4029	U4030	U4031	U4032	U4033
BENZENE	<1.	<1.	<5.	<2.	<5.	<1.
BROMODICHLOROMETHANE	<1.	<1.	<5.	<2.	<5.	<1.
BROMOFORM	<10.	<10.	<50.	<20.	<50.	<10.
BROMOMETHANE	<10.	<10	<50.	<20.	<50.	<10.
CARBON TETRACHLORIDE	<1.	<1.	<5.	<2.	<5.	<1.
CHLOROBENZENE	<1.	<1.	<5.	<2.	<5.	<1.
CHLOROETHANE	<1.	<1.	<5.	<2.	<5.	<1.
2-CHLOROETHYLVINYL ETHER	<10.	<10.	<50.	<20.	<50.	<10.
CHLOROFORM	<1.	A.	<5.	<2.	< 5 .	<1.
CHLOROMETHANE	<10.	<10.	<50.	<20.	<50.	<10.
DIBROMOCHLOROMETHANE	<1.	<1.	<5.	<2.	<5.	<1.
1,2-DICHLOROBENZENE	<5.	<5.	<25.	<10.	<25.	<5.
1,3-DICHLOROBENZENE	<5.	<5.	<25.	<10.	<25.	<5.
1,4-DICHLOROBENZENE	<5.	<5.	<25.	<10.	<25.	<5.
DICHLORODIFLUOROMETHANE	<10.	<10.	<50.	<20.	<50.	<10.
1,1-DICHLOROETHANE	<1.	<1.	<5.	<2.	<5.	<1.
1,2-DICHLOROETHANE	<1.	<1.	<5.	···· <2.	<5.	<1.
1,1-DICHLOROETHYLENE	<1.	<1.	<5.	<2.	<5.	<1.
1,2-DICHLOROETHENE (total)	<1.	<1.	6.	<2.	<5.	<1.
DICHLOROMETHANE	<1.	<1.	<5.	<2.	<5.	<1
1,2-DICHLOROPROPANE	K1 .	<u></u> {1.	<5.	<2.	<5.	<1.
cis-1,3-DICHLOROPROPYLENE	<1.	<1.	<5.	<2.	<5.	<1.
trans-1,3-DICHLOROPROPYLENE	4	<u></u> {1.5	Κ5.	<2 .	₹5	<1

Page 1 of 2

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Srittonfield Parkway / Suite 300. Box 4942 / Syracuse. NY 13221 / (315) 437-0200

Kny Authorized: November 3, 1994 Date:

	_	
_	 	
=		
\equiv		
=		_=
	 	-

ACCURATE DIE CASTING CLIENT_

Volatile Organics Method 8010/8020

3435.021.517 • JOB NO.

Fayetteville, NY DESCRIPTION

	· · · · · · · · · · · · · · · · · · ·

<u> </u>				MATRIX:	Water	
DATE COLLECTED 10-18-94	DATE RI		10-18-94	DATE ANAL	YZED 10-21	,25-94
DESCRIPTION:	MW-8	MW-2	M₩-6	MW-17	MW-5	MW-9
SAMPLE NO .:	U4028	U4029	U4030	U4031	U4032	U4033
ETHYLBENZENE	<1.	<1.	<5.	<2.	<5.	<1.
1,1,2,2-TETRACHLOROETHANE	<1.	<1.	<5.	<2.	<5.	<1.
TETRACHLOROETHYLENE	<1.	<1.	<5.	10.	5.	<1.
TOLUENE	<1.	<1.	<5.	<2.	<5.	<1.
1,1,1-TRICHLOROETHANE	<1.	<1.	<5.	<2.	<5.	<1.
1,1,2-TRICHLOROETHANE	<1.	<1.	<5.	<2.	<5.	<1.
TRICHLOROETHYLENE	<1.	<1.	360.	140.	410.	74.
TRICHLOROFLUOROMETHANE	<1.	<1.	<5.	<2.	<5.	<1.
VINYL CHLORIDE	<1.	<1.	Հ5.	2.	<5.	<1.
XYLENE (total)	<3.	<3.	<15.	<6.	<15.	<3.
			an a			
		· · · · · · · · · · · · · · · · · · ·	<u>م معرف محمد محمد محمد محمد محمد محمد محمد محم</u>		· · · · · · · · · · · · · · ·	
				ودر بنه» محتسب بر مع التي ودر بنه» محتسب بر مع التي		and the second
			ала андар алараа алараа алараа 1975 - 1976			
	با با می رو بند است. از با می رو بند است.	a a a a a a a a a a a a a a a a a a a		الاین الا الاین الیون الیونیون الاین الاین الاین الیونیو (الاین الیونیو) الیو (ایوریویی الیونیو)	and the second secon	
	رد و العربي الداريج. المراقع المراجع المراجع المراجع				a a construction of the second s	

Comments:

Methodology: USEPA,SW - 846, November 1986, 3rd Edition Certification No.: 10155

 $\mu g/1$ Units:

Page 2 of 2

ż

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

Authorized:	Non-ka) la Juce
Date:	November 3, 1994



Volatile Organics Method 8010/8020

LABORATORIES, INC.

CLIENT	ACCURATE	DIE	CASTING	
				_

JOB NO. 3435,021,517

DESCRIPTION _____ Fayetteville, NY

DATE COLLECTED _

DESCRIPTION:

				MATRIX: Wate	er
10-18-94	DATE RE	CEIVED 10-1	18-94	DATE ANALYZED _	<u>10-21-94</u>
	MW-11	MW-16	MW-13		
	P10 - 1 1	MM-10	-15		

SAMPLE NO.:	U4034	U4035	U4036			
BENZENE	<100.	<1.	· <5.			
BROMODICHLOROMETHANE	<100.	<1.	<5.			
BROMOFORM	<1000.	<10.	<50.			
BROMOMETHANE	<1000.	. <10.	<50.			
CARBON TETRACHLORIDE	<100.	<1.	<5.			G
CHLOROBENZENE	<100.	<1.	<5.			
CHLOROETHANE	<100.	<1.	<5.			
2-CHLOROETHYLVINYL ETHER	<1000.	<10.	<50.			
CHLOROFORM	<100.	<1.	<5.	· · · · · · · · · · · · · · · · · · ·	a sanah ing pananang inak i	
CHLOROMETHANE	<1000.	<10.	<50.			
DIBROMOCHLOROMETHANE	<100.	<1.	<u>ج</u> .			
1,2-DICHLOROBENZENE	<500.	<5.	<25.			
1,3-DICHLOROBENZENE	<500.	<5.	<25.			
1,4-DICHLOROBENZENE	<500.	<5.	<25.			-
DICHLORODIFLUOROMETHANE	K1000 .	<10.	<50.		a an	an a
1,1-DICHLOROETHANE	<100.	<1.	<5.			
1,2-DICHLOROETHANE	<100.	<1.	<5.			
1,1-DICHLOROETHYLENE	<100.	<1.	<5.	و و در و و و و و و و	والمراجع والمراجع والمراجع	an gan ganan saga ng ang ang ang ang ang ang ang ang a
1,2-DICHLOROETHENE (total)	<100.	<u></u> {1.	<u>۲</u> 5.		میں ایک	
DICHLOROMETHANE	<100.	<1.	<5.	·	-	- e option et sie over
1,2-DICHLOROPROPANE	- <u>~</u>	in side of the	×5.			المحمد المعمد
cis-1,3-DICHLOROPROPYLENE	<100.	<1.	<5.	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		
trans-1,3-DICHLOROPROPYLENE	×100 "		45			

Page 1 of 2

lon

November 3, 1994

Authorized:

Date:

				Vol Met	atile On hod 801	rganics 10/8020
CLIENTACCURATE DIE CAST	ING	<u></u>	<u> </u>	JOB NO	3435.021.5	17
DESCRIPTION Fayetteville	•, <u>NI</u>			MATRIX -	Water	
DATE COLLECTED 10-18-94	DATE RI	ECEIVED_10-1	8-94	DATE ANALY	ZED	-94
DESCRIPTION:	MW-11	MW-16	MW-13	'.		[
SAMPLE NO.:	U4034	U4035	U4036			
ETHYLBENZENE	<100.	<1.	<5.			
1,1,2,2-TETRACHLOROETHANE	<100.	<1.	<5.			
TETRACHLOROETHYLENE	<100.	<1.	<5.	• .		
TOLUENE	<100.	<1.	<5.	-		
1,1,1-TRICHLOROETHANE	<100.	<1.	<5.			
1,1,2-TRICHLOROETHANE	<100.	<1.	<5.	· · ·		
TRICHLOROETHYLENE	5300.	· · · · 6.	510.		· · · · · · · · · · · · · · · · · · ·	
TRICHLOROFLUOROMETHANE	<100.	<1.	<5.	an an a' sheer sa 201 a sa	i - Lander Maar witten -	
VINYL CHLORIDE	<100.	<1.	ζ5.			
XYLENE (total)	<300.	<3.	<75.	: Andrew M. (Bernach - M. (استینام <u>کند کردان م</u> ادید. ا	l saga di ens ta n' <u>f</u>
	، معربة مرسمية يحر د. 	، معيدين ميترينين مي معين ال المارينين ويتشاطر معاد ال	איז			
		,		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	• • • •
	المربقة المعادية بريام من المعادية . المربقة المعادية المعادية المعادية . المربقة المعادية المعادية المعادية .	مورید وی در این این در میکنور دارند.	میدرجند و این میدیند. این ا افتار مخترک را مدینا سا	میرید. وریو در بالاست بالا از این این این این بالا تخصیف بالاست ا		· · · · · · · · · · · · · · · · · · ·
	المنظر مواجعة معينية. المراجع المراجع المنظر المنظر المراجع المراجع المراجع المراجع المراجع المراجع المراجع الم المراجع المراجع	10 - 2000 - 2000 - 2000 - 2000 - 2000 10 - 2010 - 2010 - 2010 - 2010 - 2010 - 2010			a and an and a second	· · · · · · · · · · · · · · · · · · · ·
	and a second		يېږ کې دو ده ده منځه. 1 1			
		a an				·····

Comments:

Methodology:	SEPA,SW-846, November 1986, 3rd Edition
Certification No.	10155
Units:	μg/l

Page 2 of 2

N ភ U. Authorized: November 3, 1994 Date:____

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

1

u diricit o geire		
ENGINEERS, INC.	•	

JUU	110.	α	10	UN	170	

Sheet / of

CHAIN OF CUSTODY

Office: <u>SYRACUSE</u> Address:<u>SYRACUSE</u>, <u>M.Y.</u> Phone: <u>(3/5)437-6/00</u>

CLIENT: ACCURATE DIE CASTING LOCATION: FAYETTVILLE, N.Y.			COLLECT (Signature)	ED BY:	aun C	int
SAMPLE DESCRIPTION	Date	Time	Sample Matrix ¹	Sample Type ²	No. of Containers	ANALYSIS REQUESTED
MW-B	10/15/44	1150	water	grab	2	8010/8020
Mw-2	whield	1215	water	arah	2	8014 8020
MW-6	10/1994	1245	water	orab	2	8010/2020
Mw-17	10/12/04	1305	water	ara b	2	8010/9020
MW-5	10/19/2-1	1350	Water .	arah	2	8010/ 2020
$M_{W}-9$	10/19/04	1425	water	arah.	2	8010/ 9020
MW-11	10/19/24	1510	water	arah	2	8010/9020
MW-16	10/10/04	1550	water	arab	2	8010/ 8020
$M\omega - 13$	10/29/	1615	water	arah	2	8010/8020
		•		<u> </u>		
,			•			
					·	

¹ Matrix = water, wastewater, air, sludge, sediment, etc.

¹ Type = grab, composite

Relinquished by: Chaun Old	Date	Time	Received by:	Date	Time
& OBrien & Gero Eng. Inc.	10/18/94	5:olaph	o£		
Relinquished by:	Date	Time	Received by:	Date	Time
۵¢			ot		
Relinquished by.	Date	Time	Received by:	Date	Time
of:			.et	<u> </u>	
Use this space if shipped we counter (e.g., Fed. Es)	Date	Time	Courier Name	Date	Time
of:			*Attack delivery/courier receipt to Chain of Custody		
Relinquished by:	Date	Time	Received by Mo Oliver	Date	Time
of			of NBG/Uhboratnie she	10/19/41	Siobon

djb/woC



Volatile Organics Method 8010/8020

CLIENTACCORATE DIE CA	JOB NO	_ JOB NO3435.021.5								
DESCRIPTION Fayettevi	DESCRIPTION Fayetteville, NY									
·		<u></u>		MATRIX:	Water					
DATE COLLECTED 10-19-94	DATE R	ECEIVED 10-	19-94	DATE ANALY	7ZED10	-21-94				
DESCRIPTION:	MW - 7	MW-12	MW-15	MW-10	MW-14	QC Trip Blank				
SAMPLE NO .:										
	U4074	U4075	U4076	U4077	U4078	U4079				
BENZENE	<1.	<1,	<1.	<10.	<1.	<1.				
BROMODICHLOROMETHANE	<1.	<1.	<1.	<10.	<1.	<1.				
BROMOFORM	<10.	<10.	<10.	<100.	<10.	<10.				
BROMOMETHANE	<10.	<10.	<10.	<100.	<10.	<10.				
CARBON TETRACHLORIDE	<1.	<1.	<1.	<10.	<1.	<1.				
CHLOROBENZENE	<1:	<1.	<1.	<10.	<1.	<1.				
CHLOROETHANE	<1.	<1.	<1.	<10.	<1.	<1.				
2-CHLOROETHYLVINYL ETHER	<10.	<10.	<10.	<100.	<10.	<10.				
CHLOROFORM	<1.	(1)	<1. ¹	<10.	<1.	<1.				
CHLOROMETHANE	<10.	<10.	<10.	<100.	<10.	<10.				
DIBROMOCHLOROMETHANE	<1.	<1.	<1.	<100.	<1.	<1.				
1,2-DICHLOROBENZENE	<5.	<5.	<5.	<50.	<5.	<5.				
1,3-DICHLOROBENZENE	<5.	<5.	<5.	<50.	<5.	<5.				
1,4-DICHLOROBENZENE	<5.	<5.	<5.	<50.	<5.	<5.				
DICHLORODIFLUOROMETHANE	<10.	<10.	<10.	<100.	<10.					
1,1-DICHLOROETHANE	<1.	<1.	<1.	<10.	<1.	<1.				
1,2-DICHLOROETHANE	<1.	<1.	<1	<10.	<1.	<pre></pre>				
1,1-DICHLOROETHYLENE	<1.		<1.	<10.	<1. <1.	<1.				
1,2-DICHLOROETHENE (total)	<pre></pre>	(1	<1	<10	<1.	<pre></pre>				
DICHLOROMETHANE	<pre></pre>	.د. معرب شد و منه مرجود . . 1 .	<1.	<10	<pre></pre>	<1. <1.				
1,2-DICHLOROPROPANE		ંગ્ર	21 × 1	2100	<10 ·	· · · · ·				
cis-1,3-DICHLOROPROPYLENE	<1.	<1.		منتش ع 2 000 میں ا						
trans-1,3-DICHLOROPROPYLENE			A	100	210	~~~~~				

Page 1 of 2

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

متكاملة فللترجيل والتدارك

(i $k\omega$ Ω Authorized: November 3, 1994 Date:

-			
	-	and the second se	
	_		\equiv
_	_		
-			

Volatile Organics Method 8010/8020

3435.021.517

JOB NO.

ACCURATE DIE CASTING CLIENT

DESCRIPTION Fayettevill	e, NY							
				MATRIX:	Water	•		
DATE COLLECTED 10-19-94	DATE RE	CEIVED 10-1	9-94	DATE ANALY	DATE ANALYZED 10-21-94			
DESCRIPTION:	M₩-7	MW-12	MW-15	MW-10	MW-14	QC Trip Blank		
SAMPLE NO .:								
	U4074	U4075	U4076	U4077	U4078	U4079		
ETHYLBENZENE	<1.	· <1.	<1.	<10.	<1.	<1.		
1,1,2,2-TETRACHLOROETHANE	<1.	<1.	<1.	<10.	<1.	<1.		
TETRACHLOROETHYLENE	<1.	<1.	<1.	<10.	<1.	<1.		
TOLUENE	<1.	<1.	<1.	<10.	<1.	<1.		
1,1,1-TRICHLOROETHANE	<1.	<1.	<1.	<10.	<1.	<1.		
1,1,2-TRICHLOROETHANE	<1.	<1.	. <1.	<100.	<10.	<1.		
TRICHLOROETHYLENE	<1.	35.	14.	1300.	120.	<1.		
TRICHLOROFLUOROMETHANE	<1.	<1.	<1.	<10.	<1.	<1.		
VINYL CHLORIDE	<1.	<1.	<1.	<10.	< 1	(1.		
XYLENE (total)	<3.	<3.	<3.	<30.	<3.	<3.		
	·····	می می این این این این این این این این این ای	n and a series of the series o	and the second sec				
	t to come	a a sangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrangerrang		م موجد مربق مربق مربق مربق مربق . مربق مربق مربق مربق مربق مربق مربق .	and a state of the			
en er samelingen van de steren en er serveren en en serveren en en serveren en en serveren en serveren en en s In serveren en s	••• •• •	1	· · · · · · · ·	i tan ing binan j	· · · · · · · · · · ·	··· ·		
	مىيەر بار بار مىلىسى - مىلىپ بارىيا بار مىلىسى -	er og en generationen er er og en generationen er og en generation						
	· · · · ·							
			مان المرجم متور معروم معروم مع الا		، موجد الديني الدريق هر المحمد المحمد المحم	· · · · · · · · · · · · · · · · · · ·		
	· · · · · · · · · · · · · · · · · · ·	A Land			n an an an an Sina - Sina an			

Comments:

I

7

Methodology: USEPA,SW-846, November 1986, 3rd Edition **Certification No.:** 10155

 $\mu g/1$

Units:

Page 2 of 2

Ka lon Authorized: November 3, 1994 Date

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

Office: SyRAcuse NY									
Address: Soo Brittonfield	<u> </u>					CHAIN	OF CUSTO	DY	
Phone: 437-6100									
	. /						lah		·
CLIENI: ACCORTE Die CA	String	, i		ECÚ	ED BY: Pe	A Loeno	/Crawn	$\gamma c \epsilon$	1
LOCATION: FAYE MUITE,	, Wr		(Signa)	.ure)	pait	and	Chaun Or	Ŵ	
			Samp	le	Sample	No. of			
SAMPLE DESCRIPTION	Date	Time	Matri	r,	Type ²	Containers	ANALYSIS	REQUE	STED
444.2	1.1.9	1016	1. 1			~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	0.11		•
mw-	1 10	0015	write		9172	<u> </u>	<u>ec 101</u>	8020	
MW-12	10/19	0830	wahe		grab	_ 7	8010/8	020	
Mw-15	10/19	0930	whe		Grab	?	8010 8	020	
mw-10	10/19	100	wishe	~	émb	7	801018	070	
mw. 19	IO/A	i130	4AU	e	grab	2	800 180	050	
					· · ·		i		
Trip BlAnk	iolia		und-e	<i>r</i>	-	1	8010 180	2.0	
						· ·			
						•			
			····		· . ·				
			·						
			<u> </u>				<u> </u>	<u> </u>	<u> </u>
· · · · · · · · · · · · · · · · · · ·							`		
						1- // - /-		11.00 1	
Collentenp . 3 c	(als)	Mw-1	o pro	atrix	= water, waste	water, air, s	ludge, sedimer	it, etc.	æ_
abels-matched by time Cu	lert	ed (1)	رت (ک	<u>pc</u> =	grab, compo				
Relinquished by: Maun V. VIII		_ Date	Time	Rea	cived by:	<u>1148</u>	WHL_	Date	Time
a Buin & Gon Eng. Inc	, 	10/11/4	# 1330	ot j	AB61	<u>165</u>		10/19/25	1330
Relinquished by:	••••	Date	Time	Reco	cived by:			Date	Time
at					• • •			_	
			+						
remainmen a):			lime	Rece	aved by:			Date	Time
ot			6	œ					
Use this space if shipped via contier (e.g. For Reliminated here	LEr)	Date	Time	Cou	cier Name:			Date:	Tine
				8 7 /10	ett delivery/couri	er receipt to C	hais of Costody-		
Relinquished by:	· · - ·	- Date	Time	Rece	ived by:			Date	Time
of:		<u> </u>		of:					

djb/wpC

•

CERIEN S GERE ENGINEERS, INC.

•• ,

Job No. <u>2488 396</u> Sheet <u>1</u> of <u>1</u>

				-	
		-			
		_	-	_	_
	-	-			
			_		_
					_
	-		Circle in the local division of the local di	-	
		_	-	_	_
_		-			
-				ł	

Volatile Organics Method 8010/8020

ACCURATE DIE CASTING CLIENT_

3435.021.517 JOB NO. _

DESCRIPTION Fayetteville	, NY						
		MATRIX: Water					
DATE COLLECTED 10-20-94	DATE RI	ECEIVED 1	0-20-94	DATE ANALY	YZED10-	10-21-94	
DESCRIPTION:	MW - 4					ļ	
SAMPLE NO .:	U4172						
BENZENE	<1.						
BROMODICHLOROMETHANE	<1.		ĺ				
BROMOFORM	<10.						
BROMOMETHANE	<10.	. ·					
CARBON TETRACHLORIDE	<1.		· · · · · · ·				
CHLOROBENZENE	<1.						
CHLOROETHANE	<1.	م موجوع المرجع المرجع الم					
2-CHLOROETHYLVINYL ETHER	<10.	و رواند و در دو مو			. Na mana pa ritugu ya ku ku ku ku	من بد ای و مند مورد د	
CHLOROFORM	<1.	معاقبا والمحدود والم				· · · · · · · · ·	
CHLOROMETHANE	<10.				ىدەرمەرمەردەر ئار 140مىس		
DIBROMOCHLOROMETHANE	<u><1.</u>)	دي ۽ ڪري ۽ پير ڪري ۽ پيري ۽		ی در این میں در معمد در ا		
1,3-DICHLOROBENZENE	<5.	· ·· · · · · · ·	-	· · · · · · · · · · · · · · · · · · ·	···· ···		
1,4-DICHLOROBENZENE	<5.						
DICHLORODIFLUOROMETHANE	<10.	• 6	· · · · · · · · · · · · · · · · · · ·	میں بین بین میں بین بین ہیں۔ میکھانی ہے۔ بین بین میں بی	م میں بیروں دیور میں میں ہیں۔ 		
1,1-DICHLOROETHANE	<1.					,	
1,2-DICHLOROETHANE	<1.	· · · · · ·	· · · · · · · · · · · · · · · · · · ·		and a second at	· · ·	
1,1-DICHLOROETHYLENE	<1.						
1,2-DICHLOROETHENE (total)	38						

DICHLOROMETHANE <1. ----1.2-DICHLOROPROPANE 2-17 50.00 $\langle 1 \rangle$ - e., ۰. et Carta State Contraction cis-1.3-DICHLOROPROPYLENE <1. trans-1,3-DICHLOROPROPYLENE à

Page 1 of 2

ふい Authorized: November 3, 1994 Date:

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

38
	Ň	
_	-	 _
_		
		- Andrews
		-
Contract of the local division of the local		-
		_

Volatile Organics Method 8010/8020

LABORATORIES, INC.

CLIENT____ACCURATE DIE CASTING

JOB NO. 3435.021.517

DESCRIPTION ____

Fayetteville, NY

			<u></u>	MATRIX:	Water	
DATE COLLECTED10-20-9	4 DATE R	ECEIVED 10-	-20-94	DATE ANAL	rzed10-	21-94
DESCRIPTION:	MW-4					
SAMPLE NO.:	U4172					
ETHYLBENZENE	<1.					
1,1,2,2-TETRACHLOROETHANE	<1.					
TETRACHLOROETHYLENE	<1.				· ·	
TOLUENE	<1.					
1,1,1-TRICHLOROETHANE	<1.					
1,1,2-TRICHLOROETHANE	<1.					
TRICHLOROETHYLENE	23.		and the second	and the second secon	م المعدم من معريق ال	
TRICHLOROFLUOROMETHANE	<1.	•				
VINYL CHLORIDE	<1.					
XYLENE (total)	<3.					
	NNT Construction of the second s		· · · · · · · · · · · · · · · · · · ·			· · · · · · · · · · · · · · · · · · ·
	. Printing and a set		ارمین میرون به مار و شده میروند. رامین میرون به مار و شد قالتمو از ا	میں ور میں اور		ی: مر بین (الاسینین بین (
	and graphics and g		in the second		میں ایک	
			······		امین رسیم در ایرمان سر ۱۹۹۵ - ۲۰۰۱ ۱۹۹۵ - ۲۰۰۱ - محمد کار	
			مى بىرىمەر يىلى يورىمە مىلى بارد خىلى بىرىمەر مىلى مەر			

Comments:

Methodology: USEPA.SW-846, November 1986, 3rd Edition Certification No.: 10155 Units: µg/1

Page 2 of 2 Τx k Authorized: November 3, 1994 Date:

OBG Laboratories, Inc., an O'Brien & Gere Limited Company 5000 Brittonfield Parkway / Suite 300, Box 4942 / Syracuse, NY 13221 / (315) 437-0200

	1				1001	NO. <u>2700</u> J/h
Office: <u>SYRACHSE N.Y.</u> Address: <u>5000 Brittan Fic</u>	- <u>1d Pk</u>	'wy			CHAIN	OF CUSTODY
Phone: (315)437-6100					- <u> </u>	
CLIENT: ACCURATE DIE LOCATION: FAYETTLELL	CRST / E. N.,	w∈ <u>ŀ.</u>	COLLECI (Signature)	ED BY: C.	HAWN OU O <u>DO</u> U	DELL
SAMPLE DESCRIPTION	Date	Time	Sample Matrix ¹	Sample Type ²	No. of Containers	ANALYSIS REQUESTED
MW-4	10/20/0	<u>0930</u>	Water	grab	/	8010/8020
		 '		<u> </u>		
	'	<u> </u>)	┼───-		
					1	
				ļ		
	}			<u> </u>	<u> </u>	· · · · · · · · · · · · · · · · · · ·
	1]		 		
		·				
	+	 				
	+		· · · · · · · · · · · · · · · · · · ·		┢╍╍╍╍╂	

¹ Matrix = water, wastewater, air, sludge, sediment, etc. ² Type = grab, composite

Relinquished by: Claum O'Dell	Date	Time	Received by:	Date	Time
* O'Buen & Gere Engineers, Inc	10/20/94	1620	ot		
Relinquished by:	Date	Time	Received by:	Date	Time
			œ		
Relinquished by:	Date	Time	Received by:	Date	Time
of:		- ·	ot		
Use this space if shipped via counter (e.g. Fed Er) Relignmented here	Dister	Time	Counce Name	Date	Time
of			Street Alter and a street and and		
Relinquished by:	Dant	Time	Received to HAMS & UNIRAL.	Dem	77
of:			* Min Eler hou town he	UTrated	11:21

•

.

• • •

A

. . .

. .

APPENDIX C

.

. .

Accurate Die-Cesting Facility Aquifer Performance Test

Weil Number: CRW-1 Sintic: 2.7.58 Weil Depth: 56.26 Aquiter Thickness: 29.35

· 3

I. I

÷.,

		-					
1	Elapsed	Depth	Drawdown	Concileo	Elapsed	Depth	Drawdown
	Тітте	То		00	Time	То	
	(Mist)	Water	(Feet)		(Min)	Water	(Feet)
	1	25.67	2.91	2.66	1140	31.46	8.58
- 1	2	28-58	5.70	5.10	1200	31.47	8.59
	<u>,</u>	29.36	1.119	5.70	1760	31, 50	8.62
	4.	29.74	6	5.99	1320	31.54	8.66
L	~	29.98	7.10	6.17	380	31.52	8.68
	1	20.12	7.2-	6.27_			
	7	30.26	7.32	6.37	0	31.59	8.71
	9	30.34	7,14			25.78	2.90
. •	<u> </u>	30.42	7.54	649	2	24.42	1.54
-	10	30.48	7.5.	1.53	1	24.12	1.24
	12.	30.72	7.34	6.70	T V	23.95	1.07
i	20	34.73	7.55	1.71		12.84	0.96
	ac	30.74	7.85	1.72	6	23.77	0.39
•.•	9.47	2078	790	/ วน	ار	23.72	0.84
E	36	30.95	1.97	1. 29		23.69	0.11
3	V.c.	30.88	7.50	6.81	9	23.66	0.78
1	he	1.01	3 3	1. 44	10	24.63	0.75
•	57)	30.03	8.25	6.85		23.54	0.66
	A	30.94	8.56	L.RL	20	23.42	2.60
	1.	30.96.	8.08	6 87	25	23,44	0.56
ř.	75	34.60	7.17	6.90		27.4.	0.52
•.	- 0 -)	31.03	8.15	L.92 F	15	73.38	0.50
5	115	31.05	7.17	6.93	45	23.34	0.46
	120	31.06	8.17	6.94	55	23.32	0.44
	180	31.11	8.23	6.98	228	Z2.6Z	0.21
	740	31.18	8.30	7.02	394	22.58	0.30
	300	31.21	8,33	17.05	480	22.58	0.50
٠.	2.60	31.24	8.36 7.	Þ 8/30	1485	22.57	0.31
	120	31,27	8,39 7.	9 8:49		·	
	481	31.29	8.41	7.10	1		
	540	31,26	8.38	7.08			1.
	600	31.28	8.40	7.09			•
	660	31.30	8.42	2.11			· ,
	1270	31.31	8.43	7.11 /			
•	180	31, 32	8.44	7.12 /			<u> </u>
	840	31,34	8.46	5.13			3
:	900	31.36	8.48	1.15			,
	460	31.38	8.50	7.16			
	1020	31.40	8.52	7.18			
	1080	31.42	8.54	7.19			

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
,		• •
	4	
	} .	
	1	• •t ·`
	T	1
	1	
	1	
	1	
· · · · · · · · · · · · · · · · · · ·	<u> </u>	· · ·
	<u> </u>	
<u> </u>	†	
		1
		•
		,

90% Rec = 23.75

سيمان والأرزاد

7.22

7.22 7.24 7.24 7.27 7.28

 $r_{\rm eff}$

۰. 1' '

i q

14

Ň.,

1.85

1273- 9399

6 1198









Accurate Die-Cesting Facility Aquitar Performance Test

Weil Number: Mw - 6Static: (9.79 Well Depth: 32.98 Actuiter Thickness: 27

Elepeed	Depth	Drawdown		Elapsed	Depth	Drawdown
Time	То			Time	То	
(Min)	Water	(Feet)		(Min)	Water	(Feet)
1.5	19.81	0.03		140	20.20	0.42
2.5	19.43	0.05		200	20.22	0.44
3.6	19.94	0.06		1260	20.22	0.44
4.5	9.84	0.06		1300	20.23	0.45
5.5	19.85	0.07		13%	20,25	1.47
615	9.86	0.08	Rec	0		
7.5	19.86	0.08	i j	<u> </u>	20,20	6.42
8.5	19.86	n.08		2.5	20.18	0.40
9.5	19.07	0.09		25	20.18	0,40
10.5	19.87	0.09		4.5	ao.11	0.39
15.5	19.88	0.10		5.5	80.16	0.38
20.5	19.90	0.12		6.5	10.16	0,38
25.5	19.92	0.14		7.5	20.15	0.37
12.5	19.93	2.15		8.5	10.12	0.37
75.5	19.94	0.10	•	9.3	20 14	0.36
40.5	19.95	0.17		10.5	20.14	0.36
+15 5	19.96	2,18		12:5	20.12	0.34
50.5	19.97	0.19		20.	20.11	0.33
<u>_ج ۾ ج</u>	<u></u>	0.20		30.5	20,70	0.31
50.5	11.17	0.17	. A	80.5	00.01	0.51
75.5	19.98	05.6		33.5 /	20.08	0.30
-2-5-	14.98	0.20	5 7 1	40.5	20.01	0.30
2.2	14.44	0-2		- 29	20.0	0.49
	70.00	0.72		104	26.07	0.01
740	2000	0,04	-		20.00	0.17.
<u>~10</u>	70.00	7 26 9 21		K34-	20.02	
	20.07	0 70	ممم آ	222	liaan.	0.00
	2000	0.29	2352	291170	10 9 3	
	20,04	0.31	· ·	7771	1 <u>77777</u>	- 010
1-200	20.10	1 2 2 2	105	1401	10 95	0.17
1220	3017	1 24	8.4/-	100	<u> </u>	
610	2014	1 31	- 16 AM		<u> </u>	<u> </u>
220	20.15	0.37	1			†
780	20.15	0.57	1	<u> </u>	<u>† </u>	†
840	20.12	0.39	1			†
900	20.17	10.39	1		<u> </u>	T
960	20-18	0.40	1			
020	20.19	0.41	1			
1080	20.19	0.41]			

Eapeed	Depth	Drawdown
Time	То	
(Mirt)	Water	(Feet)
		_
	<u>.</u>	
L	1	
L		
<u>ا</u>		
<u> </u>	ļ	
	_	
<u> </u>	· ·	
	<u> </u>	
<u> </u>		
<u> </u>		.:
<u> </u>	- <u> </u>	·
·	+	
[· ·	
		
ļ		
 		
┝───		
 	+	
┣───	+	
┝───	+	
<u> </u>		-{
		- <u> </u>
		
⊢—		+
 		
	+	
	1	
	<u>† </u>	+
— —		

ो :> ÷

90°2- 19.87

÷

Y







Accurate Die-Casting Facility Aquifer Performance Test

Well Number: PZ-1 Static: 24.29 Well Depth: 34.22 Aquiter Thickness: 6

Elapsed	Depth	Drawdown	
Time	То		
(Min)	Water	(Feet)	
	24.29	0.00	
2	24.29	0,00	
3	24.29	0.00	
4	24.29	0.00	
_ 5	24.30	0.01	
6	24.30	10,0	
_ ۲	24.30	0.01	
8	24.30	0.01	
9	24.30	0.01	
10	24.30	0.01	
15	24.31	0.02	
20	24.31	0.02	
<u>v</u> ,	24.32	D.03	
_30	24.32	0.03	
35	24.33	0.04	
40	21.33	0.04	
45	24.34	20.0	
50	24 34	0.05	
55	24.34	0.05	
60	24.34	0.05	
75	24.35	0.06	
90	24.36	0.07	
105	24.36	0.07	
120	21.37	D.08	
187	24.37	D.D8	
241	24.38	<u>٥،٥٩</u>	
304	24.40	يار ه	
364	24.41	0.12	
427	24.42	0.13	
485	24.44	0.15	
545	24.45	0.16	
685	24.45	0.16	
668	24,48	0.19	
727	24.49	0.20	
785	21.48	0.19	7 130
841	24.41	0.20	44
705	27.50	0.U	
<u> </u>	24.1	0.11	
	24	0.20	
1 INXV		I 7.7.14	

	Elapsed	Depth	Drawdown
	Time	То	
	(Min)	Water	(Feet)
	1145	24.53	0.24
	1205	24.53	0.24
	1265	24.54	0-25
	1325	24.56	0.27
	1385	24.57	0.28
	0	24.57	0.28
	2.1	24.57	0.28
	25	24.56	0.27
	35	24.56	0.27
	4.5	14.56	0.27
į	5.5	24.56	0.27
	65	24.56	かわ
i	7.5	24.56	0.27
	8.5	24.56	0.27
	9.5	24.56	0.27
	105.	24.56	0.27
	16	24.56	0.27
	21	24,56	0.27
	26.5	24.55	0.26
	30.5	24.55	0.26
	37	24.55	0.26
	43	24.54	0.25
	49	24.54	0.25
ĺ	56	24.54	0.25
	62	24,53	0.24
	69	24,53	0.24
	77	24.53	0.24
	109	24.52	0.23
	155	24.51	0.22
	229	24.50	0.21
	400	24,49	0.20
	488	24.48	0.19
-			
,	1495	24.48	0.19
,			
i			

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
<u></u>		
	<u> </u>	
	+	
·	+	-
0.0		
		_
	┼───	+
	<u> </u>	
	<u> </u>	
	┼───	1
	+	
	╆╍╌╍╍	
	<u> </u>	
	 -	
	<u> </u>	-
	<u> </u>	
	<u>+</u>	
	<u> </u>	
	┼───	
	+	
	<u> </u>	
	-	
	+	
	<u>+</u>	
	┣───	
	 -	
	 -	
	<u> </u>	
	<u> </u>	
	 	
	+	-
	┼───	
	L	1

a a start a st







-Casting Facility Aquiliar Performance Accurate Die

Well Number: PZ-2

Static: 2.3.6/ Weil Depth: 40, yo Aquifer Thickness: 12.5

Depth

Water 23.6

23.61

23.61

23.61 23,60

23.6 23.62 23.63

23,63

23,70

23.72

23,74

7 23,72

73

23.

23

23,

23 75

25 23

23 2

23

23

23. 23,9

2

23

කු ටු.

23 23 P

24.00

24,00

23A 23

9 Z\$.

63

68 23.69

23

To

Elapsed

Time

(Min)

12

3

Ŧ

5

64

q

10 15 20

251

30

35

45

10

50

60

75

90

75 70

0.2

ራ

182

0

8

84

02

108

23

· · ·

5

	,			
Drawdown		Elapsed	Depth	Drawdown
		Time	То	
Feet)		(Min)	Water	(Feet)
0.00		1141	24.02	0.41
0,00		120	24.03	0.42
0,00		1261	24.04	0.43
0.00		1321_	24,04	0.43
0.01		1381	24,06	0.45
0.01	Rev	0	Z4 <u>06</u>	0.45
0.01	~]		24.05	0.44
0.02		2	24.05	0.44
0.02		3	24,04	0.43
0.02		Ψ	24.03	0.42
0.04		5	24.03	0.42
0.06		<u> </u>	24.03	0.42
0.07		7	2%.23	0.42
0.08		8	24.02	0.41
0.09		9	24,02	0.41
0.10		10	24.02	0.41
12.11		15	24.01	0.40
Dell	İ	20	24.00	0.39
0.12		255	23.99	0.38
0.13		300	73 9R	0.37
0.14		35.5	23.98	0.37
2.16		40.0	29.97	0.36
0.12		UK O	23.96	0.35
1 10		55 0	29.94	0.25
1.20		630	27.05	0.34
1.72		50	23.04	0.22
n <u>24</u>		To Z	220177	
<u>UAT</u>	•	18 3	72 20	0.20
0 10		775	72 0/	0.25
0,29		201	177 81	0.43
0.30		1270	23,01	0.00
01J/ 1 27		780	23.01	- U.L.
0.54	9/50	JUNA	07 70	
0,57	8:50,00	770	23.77	╡╸[┍]┊╏<mark>╳</mark>╶╺ ╡
0.54	1		<u> </u>	
0.36	ł			<u> </u>
0. 36 A 217	4	L		╂────┤
0.3+	ł			.┼────┤
0.58	{			╋────┥
0.39	ł	<u> </u>	}	╉╾╍╌╸┥
<u>0.79</u>	J	L	.i	<u>i </u>

Elapsed	Depth	Drawdown
Time	Το	
(Min)	Water	(Feet)
	1	
	1	
	1	1
<u> </u>	<u>† </u>	11
	1	<u> </u> −−−−−
}		
	1	†l
	1	1{
	<u> </u>	
	<u>; </u>	
-		
	†	
	1	
<u> </u>		
	<u> </u>	<u> </u>
	1	<u> </u>
	+	<u> </u>
<u> </u>	<u>+</u>	<u>├</u> {
	† – – –	<u>†</u> 1
	<u> </u>	<u>+</u> {
<u> </u>	+	╉━╍┯┥
<u> </u>		╁╼╍╍╍╌┤
	1	╂───┤
	1	╂────┤
<u> </u>	1	
 	+	╉━╍╼╍┤
	+	╁╼╼╼─┤
<u> </u>		╉╼╼──┤
		╂────┤
<u> </u>	+	
L		

, Bayes

₹7₽ .

. ÷.







Accurate Die-Casting Facility Aquifer Performance Test

Weil Number: MW-17 Static: 14.63 Weil Depth: 15.58 Aquifer Thickness: 0.5

Bapeed	Depth	Drawdown		Elapsed	Depth	Drawdown	
Time	To			Time	То		
(Min)	Weter	(Feet)		(Min)	Water	(Feet)	•
21	14.63	0,00		72_	14.78	0.15	
26	14.63	0.00		107	14.81	0.18	
31	14.63	0.00		154	14,81	0.18	
36	14.63	0.00		225	V4. 81_	0.18	
4/	14.64	0.01		398	14.81	0.18	
46	14.65	0.02	4:03pm	483	14,81	0.18	
51	14.65	0.02	a 120				
56	14.67	0.04	7/30	1492	14.8	.18	
Ы	14.67	0.04	8.36 /1				
76	/4.68	0.05					
91	14.68	0.05					
06	14,70	0.07					
121	14.70	0.07					
183	1471	0.08					
244	14.71	0.00					
えのう	14.70	0.07					
362	14.71	0.08					
424	14,71	0.08	}				
483	14.68	0.05	l				
543	14.71	0.98					
603	14.90	0.07]		1. A		
666	14.72	0.09]				
925	14.75	0.12]				
784	11.72	0.09] .				
843	14.2/	0.08					•
903	14:05	0.12					
963	14.25	0.12			·		
1023	14.76	0.13				·	
1083	14.78	0,15	1				
1143	14.78	10.15	ł				
1203	14.78	21.0	J				
263	14.76	D.B	1				
1523	19.75	0.12		•	<u> </u>		
1383	14.77	0,14		L	<u> </u>		l
<u> </u>	ļ	ļ	1	L	I	<u> </u>	
		[4	L	ļ		
<u>⊢\$\$</u> _	<u> 17・76</u>	0.13	1	L	<u> </u>		
⊢7.}	14.76	0.13	ļ	L	ļ		
120-	17:39	1013	ł	J			
<u>_54</u>	14.76	1 0.13	J	L	L		

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
	ļ	
	•	
	ļ	
	ļ	
	<u> </u>	
	<u> </u>	
	•	
		
	<u> </u>	
	<u> </u>	┠┉╼╶───┤
		 }
·		
		
	 	
<u> </u>	<u>†</u>	
		
	<u>†</u>	
	<u> </u>	
	t	<u>}</u>
	t	

÷.

~

1.51

.

-61

. مرجد





Accurate Die-Casting Facility Aquiter Performance Test

Weil Number: Mw-4 Static: 14.60 Weil Depth: 23.03

Aquifer Thickness: g

·	Elapsed	Depth	Drawdown
	Time	То	
	(Min)	Water	(Feet)
.	199	14.70	0,10
	259	14.75	0,15
	319	14.79	0.19
	378	14.82	0.22
	446	74.85	0.25
	502	14.85	0.25
	562	74.88	0,28
	620	14.93	0.33
	686	14.96	0.36
	748	14.99	0.39
	803	15.03	0.43
	862	15.06	0.46
-	910	15.11	0.51
	990	16.16	0.56
	_1°40	<i>`is.t</i> ¶	0.59
	1/00	15.22	0.62
	1160	15.26	0.66
	1220	15,30	0.70
	280	15.35	0.75
	1340	15.39	0.79
	400	15.45	0.85
•			
- Aec	73	15.55	0.95
	128	15.58	0.98
-	175	15.63	1.03
	245	15.69	1.09
	420	15.89	1.29
4:50 pm	510	15.99	1.39
- - - - - - - - - - - - - -	<u> </u>		ļ
19 A: 19 AM	1519	17.01	2-41
		<u>{</u>	
	ļ	<u> </u>	
-		· 	
			
		<u> </u>	
	 	<u> </u>	↓
		╉────	<u> </u>
	<u> </u>	+	

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
		_
		-
L		
ļ	<u> </u>	
L	<u> </u>	
	<u> </u>	
 		
		- · ·
. 		
		
 		
		<u>· </u>
}		
	1	
	1	···
	_	
	- 	
		
 		
	<u> </u>	
. <u> </u>		

Elapsed	Depth	Drawdown
Тіте	То	
(Min)	Water	(Feet)
_		
	<u> </u>	
		•
	1	_
	L	
	<u> </u>	
	<u> </u>	_ <u> </u>
		•
		<u></u>
	<u> </u>	
	-	<u></u>
		
	<u> </u>	
		
	+	
	+	
	+	
	+	
 		
	+	
 		- <u> </u>
 	+	
	1	

n.

Job Rec = 14.69





15.5 288

. 4.,

Accurate Die-Cesting Facility Aquitier Performance Test

<u>-</u> . • - +

Weii Number: 🗰 Mu - 9
Static: 44.34
Well Depth: 54.20
Aquiter Thickness:
d l

1	Elapsed	Depth	Drawdown	
	Time	То		
	(Min)	Water	(Feet)	
	80	44.25	0.01	
	95	44.35	0.01	
	109	44,36	0.02	
	124	44.36	0,02	
	191	44.35	0.01	
	251	44.36	0.02	
i	311	44.37	0.03	
	370	44.38	0.04	
	436	44.40	0.06	
	492	44.40	0.06	
	552	44.41	0,07	
	610	44.4Z	0.08	
	674	44.42	0.08	
	736	44.45	0.11	
	194	44.45	0.11	
	851	44.46	0.12	
į	911	44.47	0.13	j
	971	44.47	0.13	
	1031	44.48	0.14	
	1091	44,49	0.15	
	115	44.49	0.15	
	211	44.50	0.16	
	1271	44.52	0.14	
	131	44.54	0.10	
	1391	44.55	0.21]
Rec	51	44.54	0.20	r –
1	62	44.55	0.2	
V	117	44.55	0.21]
•	162	44.56	0.22]
	236	44,55	0.21	
	HOB	44.55	0.21	
4:180	498	44.55	0.21_]
- 1/30]
9; 06AM	1500	44.54	0.20	ľ
		<u> </u>	<u> </u>	1
		ļ	ļ	1
	ļ	<u> </u>	ļ	1
	J		<u> </u>	1
	J	<u> </u>	 	ł
	L	I	f	J

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
<u> </u>		
		
<u></u> <u></u> _		
		
		(
-		
. <u> </u>	<u> ~</u>	
	<u> </u>	
	<u> </u>	
		
	f	
	<u> </u>	
	†	
_		
	† · · · ·	1
	1	1
	1	
	Ţ	
	1	<u> </u>
_		
	<u> </u>	<u> </u>
		ł.



90% Rec 2 44.36





Accurate Die ting Facility Aquiter Performance Test

Well Number: Mw-12 Statio: 34.26 Well Depth: 42.20 Aquiter Thickness: 11

1300 49.7

e c J

Elapeed	Depth	Drawdown		Elapsed	Depth	Drawdown
Time	То			Time	То	
(Min)	Water	(Feet)	al-	(Min)	Water	(Feet)
72.5	24.27	0.01	7/80	1501	34,46	0.20
23.0	24.27	0.01	9201 11		17. <u>1</u>	
42	24.28	0.02				
53	24.21	0.01				
63	34.28	0.02				
19	34.29	0.03	i			
93	24.21	0.03				
108	34.30	0.04				
122	34.30	0.04				
188	34.30	0.04				
247	34.31	0.05	· ·			
307	34.33	OP7				
367	34.34	0,08				· · ·
<u>432</u>	34.36	0.10				
<u>488</u>	3437	0.11				
548	34.37	0.11				
<u>607</u>	34, 38	0.12	·			
<u>67/</u>	39.39	0,13		<u> </u>	<u> </u>	
731	34.40	0.14			<u> </u>	
790	34,41	0.15				
847	37.42	0. (6				
<u>907</u>	34.41	0.15			<u> </u>	
401	124,47	0.17		· · · ·		
1027	34.94	0.18				
1087	21.44	0.13		ļ	{	├ ───┤
	124.25		1		╆╼───	
67-	14.16		ł			
<u>, 60 [</u>	134 5	0.74	1	┣]
1381	24/44	017	1		<u> </u>	<u>├</u> ────┤
	╽╱┈╙ ╾		1			
38	34.49	0.23	1			i
48	34.48	0.22	1	<u> </u>		t
58	124.50	0.24	1	⊢ −−−−	<u> </u>	<u> </u>
712	3449	0.23	1			t{
159	34.48	0.22	1		1	t{
232	3449	0.23	1			
404	34.47	0.21	1		-	[
493	134,47	0.21	j .			
			1 .			T

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
	·	
	<u> </u>	
	<u> </u>	
·	<u> </u>	
ļ	ļ	
<u> </u>	· · ·	ļ
	└───	
ļ	<u> </u>	[
	ļ	
	┣_──	
	<u> </u>	
	 	
	<u> </u>	
	<u> </u>	
	<u></u>	
	┣━━━━	
	<u> </u>	
		· · ·
		
	1	
	1	
	1	
	i	
1	1	

90% Rec = 34.28

: 14







-

Accurate Die-Casting Facility Aquifer Performance Test

1 14 P

Weil Number: Mw -7 Statio: 24.32 Weil Depth: 44.0 Aquifer Thickness: 3' in overburden

Eapeed	Depth	Drawdown	•	Elapsed	Depth	Drawdown
Time	To			Time	То	
(Min)	Water	(Feet)	. · · · · ·	(Min)	Water	(Feet)
21	24.60	0.28		111	24.55	0.23
25	24.70	0.38		157	24.45	0:13
22	24,76	0,44		231	2440	0.08
36	24.52	0.50		402	a4.36	0.04
41	24.88	0,56		791	24.36	0.04
51	24.96	0,64	9/30			
62	155.04	0,72	8:58 m	1498	24,38	0.06
17	25,12	0,80			<u> </u>	
91	15.16	0.84			L	
106	25.20	0,88		-	· ·	
131	25,24	0,92			<u> </u>	
186	125.30	0,98			L	
246	25.32	1.00				
306	2535	1,03			L	
365	1:5.36	1.04	ļ		<u> </u>	
9.29	25.37	1.05	!		L	
487	25.39	1.07	Į		L	
547	25,39	1.09	ļ		<u> </u>	
606	25.39	1.07			┟	
670	25.40	1.08		· ·	┟	
724	25.41	1.09	1	_		
1 <u>787</u>	2541	1.09	1		┣	
846	25.4	<u>[.94</u>	ł	<u> </u>	<u> </u>	
1904	LS. YA	1.10	{		╂╍─────	
406	7	Par 110	ł	<u> </u>	<u> </u>	
1020	25.43	<u></u>	ł		╂	┝────┤
1006	25.23		{		┢	<u>├───</u> ┤
1140		1.16	ł		┟	<u>॑</u>
100	25.14		ł		╁	├────┤
1200	107.7	1.1.2	ł		┼───	┼╌╌╌╌┥
127-	115.70 he.un		· ·	-	┼──	╂─────┤
h12.	1 1 1	1.19	-		 	<u> </u> {
20	2493	A.L.				╂──────┨
	174 25	A CC	1	 	+	│ ┥.
51	194 8A	1 0.50	1	 	+	†
62	2420	0.41	1		†	<u> </u>
175	54 25	0.45	1	— —	†	<u>┼─────</u> ┥
1-25	24.21	0.39	1	}	┼╴───	<u>├</u> ┤
175	74.1.7	0.25	1		+	╁────┤
	<u> </u>		3	Ľ	<u> </u>	

Depth Elapsed Drawdown Time То Water (Min) (Feet)

n an an an Arainmean Arainmean A Arainmean an Arainmean
2.1

ъ

Note: some type of obstruction in well of 24.65' BTOC






Accurate Die-Casting Facility Aquiter Performance Test

Well Number: Mu-11 Static: 35-38 Well Depth: 50,67 Aquifer Thickness:

eed	Depth	Drawdown		Elapsed	Depth	Drawdown		Elapsed	Depth	Drawdown
	То			Time	То			Time	То	
	Water	(Feet)		(Min)	Water	(Feet)		(Min)	Water	(Feet)
$\overline{\boldsymbol{\Sigma}}$	35,30	0.01								
5	35.18	0.00]			
0	35-38	0.00]			
5	35.38	0.00					}	[]		
3	35.40	0.02]			
52	35.39	0,01								
13	35.40	0,02]	[
72	35.39	0.01]]			
38	35.40	0,02]]			
75	35.40	0,02								
55	35.39	0.01			1		J			
3	35.39	0.01			ļ]			
-6	35,39	0.01	ł	L			1		<u> </u>	<u> </u>
8	35.40	0.02								
16	35.41	0.03	1			-		L	<u> </u>	<u> </u>
58	35.40	0.02					1			<u> </u>
1-	35.40	0.02								<u> </u>
2	25.40	0.02			<u> </u>		· ·	L	<u> </u>	
<u>52</u>	35.41	0.03			<u> </u>		1		ļ,	<u> </u>
2 <u>92</u>	35 39	0-01	1			<u> </u>	1.]		<u> </u>
52	35.39	0.00					1	J	<u> </u>	
212	35.28	0.00	4				4			_ _
<u> 11</u>	35.71	0.01	4				4	ļ	<u> </u>	·
2.20	35,10	0,02	4		+		- ·			
57 L	175.11	0.03			+		I		┽────	
		+	4		+		4	}		- <u> </u>
24	35.41	0.03	4				4	}	+	
40	35.71	0.03	4			- <u> </u>	4		+	+
20	132.70	0.06	4				-		+	
	32 20	0.01	-		+	- _ ` 	-			
<u>11</u>	74.20		-	 			4		+	+
<u> </u>	00,00		1	 				}	+	
500	35-42	+	-1	 		-{	1		┥╴──	+
<u>, , ,</u>	1221.12		d ·	 	-1	· †	1			
	+	+	1	h	···				-	
		1	1				1 ·		-	1
	†	1	1		1	1	1		1	1
	1	1	1		1	1	4	·		1
	1	1	1				1		1	1
	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	360 100 Water 35.34 5 35.36 5 35.37 2 35.37 2 35.37 2 35.37 2 35.37 2 35.37 2 35.37 2 35.37 2 35.37 3 35.40 72 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 35.39 $31.35.39$ 35.39 $21.35.38$ $21.2.35.38$ $21.2.35.38$ $21.2.35.38$ $21.2.35.38$ $21.2.35.38$ $21.2.35.39$ 35.39 $21.2.35.39$ 35.39 $21.2.35.36$ 35.39 $21.2.35.36$ 35.39 $21.2.35.36$ 35.39 35.39	Soci Deptity Drawdown To To Water (Feet) 2 35.28 0.01 5 35.28 0.02 5 35.37 0.01 5 35.37 0.01 2 35.40 0.02 5 35.37 0.01 3 35.40 0.02 72 35.37 0.01 38 35.40 0.02 72 35.37 0.01 38 35.40 0.02 72 35.37 0.01 38 35.40 0.02 74 35.40 0.02 75 35.40 0.02 74 35.40 0.02 74 35.40 0.02 75 37.40 0.02 74 35.40 0.02 75 37.40 0.02 72 35.40 0.02 74 35.40 0.02 74	Soc Deptin Drawown To T_0 Water (Feet) D $35, 38$ 0.00 S $35, 37$ 0.01 S $35, 37$ 0.01 S $35, 37$ 0.01 S $35, 40$ 0.02 S $35, 37$ 0.01 S $35, 39$ 0.01 S $35, 39$ 0.01 S $35, 39$ 0.01 S $35, 39$ 0.01 S $35, 40$ 0.02 S $4, 54$ 0.02 S $4, 54$ 0.02 S $35, 40$	To Time To Time To (Min) 2 $35, 30$ 0.01 5 $35, 37$ 0.00 2 $35, 37$ 0.00 2 $35, 37$ 0.01 5 $35, 37$ 0.00 2 $35, 37$ 0.01 5 $35, 37$ 0.01 2 $35, 37$ 0.01 3 $35, 40$ 0.02 72 $35, 37$ 0.01 $35, 37$ 0.01 $35, 37$ $35, 37$ 0.01 $35, 37$ $35, 37$ 0.01 $35, 37$ $35, 37$ 0.01 $35, 37$ $4, 35, 40$ 0.02 $35, 37$ $1, 35, 40$ 0.02 $35, 77$ $1, 35, 40$ 0.02 $35, 77$ $1, 35, 40$ 0.02 $35, 77$ $1, 35, 40$ 0.02 $35, 77$ $1, 35, 38$ 0.00 $35, 77$ $272, 35, 57$ 0.00 $35, 37$	380 10000 100000 100000 100000 351000 0.01 (Mir) Water 2 35100 0.01 0.02 0.01 52 35100 0.02 0.01 0.02 52 35.37 0.01 0.02 0.02 52 35.37 0.01 0.02 0.01 38 35.40 0.02 0.01 0.02 55 35.9000 0.01 0.02 0.01 11 35.40 0.02 0.01 0.02 11 35.40 0.02 0.02 0.02 0.02 11 35.40 0.02 0.02 0.02 0.02 0.02 212 35.70 0.000 0.02 0.02 0.02 0.02	Boold Depth Depth Depth Depth Display (feet) 1 To To To To To 0 $35, 38$ 0.01 To To To 2 $35, 38$ 0.00 To To To 2 $35, 37$ 0.01 To To To To 1 3 $35, 40$ 0.02 To To To To 1 $35, 37$ 0.01 To To	Deptin Deptin <thdeptin< th=""> <thdeptin< th=""> <thdeptin< t<="" td=""><td>Description Description Direction The To To Time Time Time Water (Feet) (Min) Water (Feet) (Min) 2: 3_{5}, 3_{6} 0.01 (Min) Water (Feet) (Min) 2: 3_{5}, 3_{7} 0.01 (Min) (Min) (Min) (Min) 2: 3_{5}, 3_{7} 0.01 (Min) (Min) (Min) (Min) 2: 3_{5}, 3_{7} 0.01 (Min) (Min) (Min) (Min) (Min) 2: 3_{5}, 3_{7} 0.01 (Min) (Min) (Min) (Min) (Min) 3: 3_{5}, 4_{10} 0.02 (Min) (Min) (Min) (Min) (Min) 3: 3_{5}, 4_{10} 0.02 (Min) <td< td=""><td>Cost Cost Cost Cost Cost 10 Water (Feet) (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 38$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Mi</td></td<></td></thdeptin<></thdeptin<></thdeptin<>	Description Description Direction The To To Time Time Time Water (Feet) (Min) Water (Feet) (Min) 2: 3_{5} , 3_{6} 0.01 (Min) Water (Feet) (Min) 2: 3_{5} , 3_{7} 0.01 (Min) (Min) (Min) (Min) 2: 3_{5} , 3_{7} 0.01 (Min) (Min) (Min) (Min) 2: 3_{5} , 3_{7} 0.01 (Min) (Min) (Min) (Min) (Min) 2: 3_{5} , 3_{7} 0.01 (Min) (Min) (Min) (Min) (Min) 3: 3_{5} , 4_{10} 0.02 (Min) (Min) (Min) (Min) (Min) 3: 3_{5} , 4_{10} 0.02 (Min) (Min) <td< td=""><td>Cost Cost Cost Cost Cost 10 Water (Feet) (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 38$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Mi</td></td<>	Cost Cost Cost Cost Cost 10 Water (Feet) (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 28$ $a. a a$ (Min) Water (Feet) 2 $35, 38$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Min) (Min) (Min) (Min) (Min) 2 $35, 37$ $a. a a$ (Mi

v.



Accurate Die-Casting Facility Aquifer Performance Test

Weil Number: $\mathcal{M} \mathcal{W} \rightarrow \mathcal{T}$ Static: 30.4-8 Well Depth: 41. 20 Aquifer Thickness: $\{3\}$

Elapsed	Depth	Drawdown		Elapsed	Depth	Drawdown		Elapsed	Depth	Drawdown
Time	То			Time	То		}	Time	То	
(Min)	Water	(Feet)		(Min)	Water	(Feet)	}	(Min)	Water	(Feet)
22	30 48	0.00	4/30	1503	30,59	D.11				
23	30.48	000	9:03AA							
37	30.48	000								
43	30.48	3.00								
53	30.48	3.00					ļ			
63	30.48	0.00							ļ	!
78	30.45	0.00							<u> </u>	<u> </u>
93	30.48	1.00		L			1			<u> </u>
103	30.48	0.00				· · · ·	1	·		
192123	30.48	0,00					1			<u> </u>
204	<u>1.30.48</u>	0.00		<u> </u>	<u>}</u> _		1		1	<u> </u>
<u>249</u>	80.46	0.00		L	<u> </u>	ļ	4			
309	32.49	GHODI								
368	30.49	GAD OPI			ļ	<u> </u>				
124	30.50	0.02	!		ļ				<u> </u>	
490	30.50	0.02	1				4			ļ
1220	30.50	0.02	4		<u> </u>	<u> </u>	-			- <u> </u>
1960	30.51	0.03	1		<u></u> -	ļ	4			
673	30.52	0.04	ł		<u> </u>		4			
233_	130.52	0.04	{			<u> </u>	4			
<u>792</u>	30.53	0.05		 		╂	4		+	<u> </u>
849	30.50	0.04	4		<u>}</u>	┨─────	4	 		- <u>{</u>
109	30153	<u> n.05</u>	-	_			-			-
969	130.51	0.05	-			<u> </u>	4		+	<u> </u>
1029	30.54	<i>a.up</i>	4	 		{	4		+	
1	1 30.54	1-0.06	4		╂─────	<u> </u>	4	.	+	
	130171	0.00	4	 	╋───╸		4	}	+	
1 troy	130.54	0.00	4				4	Ì		+
1204	120.35	0.01	4				-			
1301	20122		4		+		-		+	
++++2%7-	30.71	10.0-1	4				-			
-	12.57	A 06	-			+	-			
1 To	20.57	10.00	1	}		1	4			· · · · ·
	40.51	0.09	1		1	1	1			
The second	30.50	0 12	1		1	1	1		1	
160	30.58	0.989	1		1		1			1
2341	30.58	0.40			T		1			
406	130.59	0.11	1				1			
n 495	30.50	7 0.11	7							

90% Rec = 30.49

Accurate Die-Casting Facility Aquifer Performance Test

Well Number: MW-2. Static: ||.0| Well Depth: (8.03 Aquiter Thickness:

Ĩ	Elapsed	Depth	Drawdown
	Time	То	
	(Min)	Water	(Feet)
	83	10,98	-0,03
	96	10,98	-0,03
	111	10.98	-0,03
	126	10.97	-0,04
	196	10.95	-0.06
	256	10.94	-0.07
	316	70.92	-0.09
	375	10.91	F0.10
	443	10.89	- 0.12
	499	10.89	-0.12
	559	10.87	- 0.14
	616	10.85	-0.16
	683	10.85	1-0.16
	245	10.87	-018
	799	10,83	-0.18
	859	10.81	- 0-20
	916	10.80	-0-21
	976	10.79	-0.22
	1036	10.79	-0.22
	096	10.77	-0.24
	156	10,77	-0,24
10.81	1216	10.76	-0.25
10	1276	10.76	-0.25
	1336	10.76	-0.25
	1396	10,75	-0.26
1	<u> </u>	<u> </u>	<u> </u>
	68	10.74	-0.27
	<u>1 a 4</u>	1/0.73	-0.28
¥.	1/+1	10.74	
in the second se	KHX_	$\frac{10, +2}{10, -2}$	<u>-0.21</u>
	17/0	1/0.+2	-0.61
	1205	10,72	1-0.01
<u>-9/3</u> 0	15-111	10 711	+ <u> </u>
14.80	<u></u>	10.44	1-0.21
	` } -	┽────	┼───┤
	 	+	╂────┤
	 	+	┼───┤
	}	+	┼──┤
	 	╉━━━━━	┼───┤

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
	ļ 	
	 	
,	<u></u>	<u> </u>
		{
		<u>├</u> ────────
		ļ
	<u>├</u> ──	├
	<u> </u>	
	 	
		<u> </u>
	1	<u>†</u>
· ·		
	ļ	
	<u> </u>	ļ
ļ	┢────	ļ
	<u> </u>	<u> </u>
		ļ
	┼────	<u> </u>
}		<u> </u>
<u> </u>		<u> </u>
┣────	+	<u> </u>
	<u> </u>	╋╼╼╼╼
	<u> </u>	<u> </u>
L	 	t
	<u>† – – – – – – – – – – – – – – – – – – –</u>	+
	<u> </u>	+

Elapsed	Depth	Drawdown
Time	To	
(Min)	Water	(Feet)
	·	
	<u> </u>	
		ļ
	L	
	L	L
	┣_────	
<u> </u>	 	· · · · · ·
	<u> </u>	<u> </u>
	 	
		 -
	 	<u> </u>
	Ļ	
	 	
 	}	·
}	<u> </u>	<u> </u>
	<u> </u>	<u> </u>
<u> </u>	<u> </u>	╂─────
	┝────	
	┢━━━	· · · ·
		1
	<u> </u>	· ·
	T	
	Ī	
		ļ
	L	L
	Ļ	<u> </u>
	<u> </u>	<u> </u>
	┢────	
	<u> </u>	<u> </u>
	<u></u>	<u>}</u>
	4	

Accurate Die-Casting Facility Aquiter Performance Test

Well Number: MW=8 Static: 29.89 Well Dept 1: 37.83 Aquifer Thickness: 11

	Elapsed	Depth	Drawdown
	Time	To	
	(Min)	Water	(Feet)
	R1.0	29.89	0.00
	96	29.89	0.00
	79111	29.89	0.00
	125	29,89	0.00
	195	29.86	-0.03
	254	29.86	-0.03
	315	29.86	-0.03
	373	29.85	-0.04
	441	29.84	-0.05
	497	29.83	-0.06
	557	29.84	-0.05
	615	29.82	-0.07
	681	29.87	-0.07
	743	29.82	- 0.07
	798	29.83	-0.06
	852	29.8Z	-0.07
	915	19,81	-0.08
•	als	129.51	-0.08
•	1033	29.91	-0.08
	1093	19.81	-0.08
	1153	29.81	-0.08
	17.13	29.80	-0.09
	1233	29.80	- 0.04
	133 3	21.81	-0.08
•	1393	29,8	-0.08
Rec	67	89.81	-0.08
1	Taa	29,80	-0.09
11	17-0	29.81	-0,08
$\boldsymbol{\nu}$	240	29.80	-0.09
	41.3	24.80	-0.09
1:230m	503	29.80	-0.09
		1	
9/30	1512	29.81	-0.08
9:12 nm		T T	1
		<u> </u>	
	· · · · · ·	1	
		1	
		1	
		<u> </u>	11

Elapsed	Depth	Drawdown
Time	То	
(Min)	Water	(Feet)
	ļ	
		
 	<u> </u>	
		
	<u> </u>	
	- 	
L		
		
	- <u></u>	
		
		+
		- <u></u> ┤-·
	+	-{
	+	
	- 	<u> </u>
	+	
\	1	- <u> </u>
	1	-1
	1	
	1	

Elapsed	Depth	Drawdown
Тіле	То	
(Min)	Water	(Feet)
		
·	!	
	· · · ·	├───
	<u> </u>	i
	[
	<u> </u>	·
┝╾──────		
	·	
· · ·	1	
-	<u> </u>	
	 	
<u></u>	<u> </u>	
<u> </u>	<u> </u>	
<u> </u>	1	
	· · · ·	
		1
		1

2

t:230

.

•

APPENDIX D

.

_		and the second se	
		· · · · · · · · · · · · · · · · · · ·	
_		الكت الفسندي ا	
			_
	_		_
-			

LABORATORIES, INC.

Safe Drinking Water Act

192.

CLIENT	O'BRIEN & G	ERE ENGINEERS, INC.		JOB NO. 3435.0	21.076	
DESCRIPTION	Accurate	Die Casting				
	Recovery	Well		MATRIX: Wate	er	
SAMPLE NO	U3214	DATE COLLECTED	9-29-94	DATE RECEIVED	9-29-94	

Primary Inorganic Chemicals	p pm	Secondary Inorganic Chemicals	ppm
Arsenic	<0.005	Chloride	• –
Barium	0.2	Copper	<0.01
Cadmium	<0.01	Iron	0.29
Chromium	<0.01	Manganese	-
Fluoride	-	Sulfate	9.
Lead	<0.005	Zinc	0.03
Mercury	<0.0002	Corrosivity	-0.2
Nitrate	-	Calcium	77.
Selenium	<0.005	Total dissolved solids	-280.
Silver	<0.01	Total alkalinity	260.
Sodium	8.	рH	7.3
		Temperature	10.°C

Organic Chemicals	ррь
Endrin	-
Lindane	-
Methoxychlor	-
Toxaphene	-
2,4-D	-
2,4,5-TP (Silvex)	-

Comments:

•		

Calcium carbonate

Certification No.: 10155

/ Kon Authorized: _ October 24, 1994 Date:_

-CBG-Laboratoriase, Inc. 17 Cliffind Gare Limited Company - CCC Entranol: 2011 (CCC) - CLiffind Care 2942, Symputes, 17 (3221) (CCC) 457-0200



LABORATORIES, INC.

Laboratory Repor

CLIENT ____ O'BRIEN & GERE ENGINEERS, INC.

JOE NO. 3435.021.076

DESCRIPTION Accurate Die Casting

MATRIX: Water DATE COLLECTED ____9-29-94 9-29-94 _ DATE RECEIVED ___ Description: Recovery Well Sample # U3214 Total Metals: ANTIMONY <0.06 MAGNESIUM 24. NICKEL <0.05 Other Analyses: 290. HARDNESS AMMONIA NITROGEN <0.05 5 DAY BIOCHEMICAL OXYGEN DEMAND <5. CHEMICAL OXYGEN DEMAND <10. HEXAVALENT CHROMIUM <0.01 NITRITE NITROGEN <0.05 SPECIFIC CONDUCTANCE umho/cm 570. TOTAL CYANIDE <0.01 TOTAL PHOSPHORUS <0.05 TOTAL SUSPENDED SOLIDS 5. TOTAL ORGANIC CARBON 2. METHYLENE BLUE ACT. SUBSTANCES <0.1

Comments:

Certification No.: 10155

Units:

and the

mg/l unless otherwise noted

CBG Laboratories, Inc., an O'Brien & Gere, Company CCD Birthomard Persovav, Buite 200, Box (1942), Nicolash (1

October 24, 1994

-			
_			_
_			
		and the second se	_
		the second division in which the second division is not the second division of the second division is not the second division of the seco	
	the second s		
-			

Laboratory Report

LABORATORIES, INC.

O'BRIEN & GERE ENGINEERS, INC. CLIENT__

_____ JOB NO. ____3435.021.076

DESCRIPTION Accurate Die Casting

	·	MATRIX: Water		
	DATE COLLECTED 9-29-94	DATE RECEIVED	9-29-94	
Description:	Recovery Well Filtered			
Sample #	U3215			
LAB FILTERED IRON	0.15			
· · · · · · · · · · · · · · · · · · ·				
Comments:	Certi	fication No.: 10155	,	

Units: mg/1

sunorized. October 24, 1994

.

183 Laboratories, Inc., an O'Brien & Gare, Company 1911 Contont ald Paraway (1994a, 1901 Box 4942, Syrabuby, 1971 (1922) (1997 - 1997) 1929 - Box Alder Box (1997) .

2488.396.51

11



SAMPLE ANALYSIS REPORT

Reviewed By

1/91

Life Science Laboratories. Inc. warrants, to the best of its knowledge and belief, the accuracy of the analytical test results contained in this report, but makes no other warranty, expressed or implied, especially no warranties of merchantability or fitness for a particular purpose. By Client's acceptance and/or use of this report. Client agrees that LSL is hereby released from any and all liabilities, claims, damages or causes of action affecting or which may affect Client as regards to the results contained in this report. Client further agrees that the only remedy available to Client in the event of proven non-conformity with the above warranty shall be for LSL to re-perform the analytical test(s) at no charge to Client.

The data contained in this report are for the exclusive use of the Client to whom it is addressed, and the release of these data to any other party, or the use of the name, trademark or service mark of Life Science Laboratories, Inc. especially for the use of advertising to the general public, is strictly prohibited without the express prior written consent of Life Science Laboratories, Inc.



Office: Starting and the second s	:_ <u>/_</u> of	No. $\underline{\sim 950.57}$ Sheet	JODI					
CLIENT: ACCURATE DIE LOCATION: FAIETTEVILLE, M.Y. SAMPLE DESCRIPTION Date Time Sample No. of SAMPLE DESCRIPTION Date Time Sample No. of ANALYSIS REDL Matrix Type ² Consisters ANALYSIS REDL Matrix Type ² Consisters ANALYSIS REDL Sheets and/old Contact Date 7 Contact Contact Contact 7 Contact Date 7 Contact Date 7 Contact Contact 7 Contact Contact 7 Contact Contact 7 Contact 7 Contac	Y	OF CUSTODY	CHAIN			-	,) Office: <u>542ACUSE</u> Address: <u>542ACUSE</u> , <u>N.Y.</u> Phone: <u>(315)437-6100</u>
SAMPLE DESCRIPTION Date Time Sample Sample Sample Date Time Matrix Type ² Containers ANALYSIS REDL Recovered* Weble 4/26400750 Water 9726 Genetioners ANALYSIS REDL Recovered* Weble -	<u> </u>	U	un Áld	ED BY:	COLLE (Signatur	NU	LE	CLIENT: ACCURRTE DIE LOCATION: FRIFTENT
Recover if well glight 0750 Water grab 6 See the attac Sheets and/ok contact Dake 7 for analy 515 Feauired, Delite client and a client and client and y 515 required, required, Delite client and client and client and y 515 required, required, Delite client and client and client and y 515 required, required, Delite client and client and client and y 515 required, required, Delite client and client and client and client and y 515 required, required, Delite client and clie	QUESTEI	ANALYSIS REQU	No. of Containers	Sample Type ²	Sample Matrix ¹	Time	Date	SAMPLE DESCRIPTION
Skeets and/ek Cartact Daile 7 for analy 515 Fe awired Politic cleal and Politic cleal and Politic cleal and Politic cleal and Politic clear and Polit	ached	see the attack	6	9rab	Water	0750	9/29/94	RECOVERY Well
Relinquished by: Date Time Received by: Date Relinquished by: Date Time <	lor	sheets and lor						
	Towers	Contact Dave To						
Politic clend a d etc. and yes Politic clend a d etc. and yes Print thurs and yes print thurs P	/5	for analysis					<u> </u>	
Deltti plinel a di etti pri c'hl pri andyn per Dent touriei Den touriei Imaria = water, wastewater, air, sludge, sediment, etc. ¹ Matrix = water, wastewater, air, sludge, sediment, etc. ² Type = grab, composite Retinquisbed by: Imaria Date Time Retinquisbed by: Imaria Date Time Received by: Imaria Date Time Retinquished by: Imaria <td></td> <td>required.</td> <td></td> <td></td> <td>· · · ·</td> <td></td> <td>╞╌╌┞</td> <td>·</td>		required.			· · · ·		╞╌╌┞	·
Deletion limit and and end on chlorine analyzin per Dere there and an analyzin per Dere there are analyzin per 1 Matrix = water, wastewater, air, sludge, sediment, etc. 2 Type = grab, composite Retinquished by: Date Date Date Time Received by: Date Od Time Received by: Od Time Received by:								
ADDITAL (1) Date (1) OF CARLES (1) CALLES (1)			1		0-00			Delite alcolo
1 Matrix = water, wastewater, air, sludge, sediment, etc. 2 Type = grab, composite Relinquished by: Date 0ct Date 7 Date 7 Date 7 Date 0ct Date 7 Date 0ct Date 7 Date 0ct Date 7 Date 0ct Courier Name: 0ct Plate 0ct Plate 0ate Time 0ate Oat		fac	angus !	NUNE (LM			in ex	Dry Tower
1 Matrix = water, wastewater, air, sludge, sediment, etc. 2 Type = grab, composite Relinquished by: Jewn Olell Date Time Received by: Water, wastewater, air, sludge, sediment, etc. 2 Type = grab, composite Relinquished by: Jewn Olell Date Time Received by: Water, Water, Water, Water, Water, Water, etc. 3 Type = grab, composite Received by: Water, Jew								
1 Matrix = water, wastewater, air, sludge, sediment, etc. 2 Type = grab, composite Relinquished by: Use OWL Date Time Received by: Wset Lip Stript Y to Date ot DBuen & Gene Eng., Inc., 9/20/04/377 or Date Optimized Stript Content of the Stript								······································
1 Matrix = water, wastewater, air, sludge, sediment, etc. 2 Type = grab, composite Relinquished by: Jawn Ollell Date Time Received by: With Ministry Ministry Ministry Date Time Received by: With Ministry Date Time Received by: With Ministry Date Time Received by: Date Date Date Time Received by: Date Date Content of the space of states space of states Date Time Received by: Date Date Content of the space of states space of states space of states Content of the space of states Date Time Relinquished by:								
¹ Matrix = water, wastewater, air, sludge, sediment, etc. ² Type = grab, composite Retinquished by: <u>Juwn OMU</u> Date Date Time Received by: <u>Juwn OMU</u> Date oct <u>Janhy</u> / 377 Oct <u>Janhy</u> / 377 Retinquished by: Date Time Received by: Date Time Received by: Date Oct								
¹ Matrix = water, wastewater, air, sludge, sediment, etc. ² Type = grab, composite Relinquished by:	.	<u></u>						
Retinquished by: Use time Received by: Use time Received by: Use time Date Retinquished by:	stc.	ludge, sediment, etc.	water, air, s site	= water, waste	¹ Matr			
cet Date Time Received by:	Jate Tiel	Mith Date	1115	ined by: W.		Date		Relinquished by Clawn AMIL
Relinquished by:	4-511220	YSNAC 9-4-5	nyatar	RAG Lat	4/377 0	9/20/2	<u>,</u>	A Brien & Gere End, INC
of:	Date Tir	Date		ived by:	Time F	Date		Relinquished by:
Relinquished by: Date Time Received by: Date Date oć oć oć oć oć Date Date </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>ot</td>								ot
of Of Of Date Time Courier receipt to Chain of Custody Relinquished by: Date Time Received by: Date of: Of)ate Tin	Date		ived by:	Time F	Date		Relinquished by:
Use this space if shipped via counier (e.g. Fed Er) Date Time: Counier Name: Date relinquished by:					•			oć
Relinquished by: Attack delivery/courier receipt to Chain of Custody Relinquished by: Date Time Received by: Date of: of:	ine Tin	Dine		ier Name	Time: C	Date	d.Ex)	Use this space if shipped via courier (e.g., Fe
Relinquished by:								Relinquished by:
Date Time Received by: Date of: of:		hain of Custody	er receipt to C	els delivery/cours				
סר: ל לל:	ate Time	Date		wed by:	Time R	Date	<u>~</u>	Kelinquished by:
(nology) len o 10 6 4 1.53		5- (200 10 A	n n Ann VIP		<u> </u>		

OBG LABORATORIES, INC. CASE FILE FORM

	•				
PROGRAM INFORMA	TION			· · · · · · · · · · · · · · · · · · ·	
Client: 46		Div.	R	ef. No	
Program:	<u>Iccurate Di</u>	ie			
Location: City	Town FALL	etterdu	State	NU	
Date Rec'd: 9-	29-94 Time	Regid: /	327 s	ampler: (. D'hell
Custody Seal:	Intact	t	Not Inta	act 🖉	<u></u> NA
AFTER HOURS CUS	TODY	·			
RELINQUISHED BY:		DATE TIME RE	CEIVED BY SECURITY	GUARD :	DATE T
		.			
RELINQUISHED BY SECURITY O	UARD TO COOLER:	DATE TIME RE	CEIVED BY SAMPLE CU	ISTODIAN:	DATE T
					<u></u>
COMMENTS (DISCPRI	ANCY	·			·
AND AT N	105°P	A SI M	NPO dE1	Jiltoled	1601
IIIAA not A	upper ttor	T An A	en Main	TAIRA	<u> </u>
LIDUILA (A	trai in	0.20 C	$\frac{\alpha}{2} + \frac{\alpha}{2} + \frac{\alpha}$	stiar D	$\frac{\pi}{1}$
MUMAR JU	lucer on	9-50-		ALA MA	<u>TRC</u>
MARINE	<u>ann</u>	<u>n ni</u>	<u>onuneo</u>	phaces	<u>CpC</u>
fittles as	nd preses	ved is	n the	lab ur	<u>a</u>
Ta alizza	<u>_761 ji</u>	ttered 1	Lon.	·	
			·		
					•
RESOLUTION/CLIENT	r comment:				
· · · · · · · · · · · · · · · · · · ·					
				· · · · · · · · · · · · · · · · · · ·	
,				-	
	~ 6	· · ·			
igned: 11/1/1	1 SMAL	Project	Manager An	proval: /	hi
igned. <u>Archan</u>	10.1		nunuger np		<u> </u>
U-20		/ / /	71177178817		
Date: <u>9-30-</u>	<u> </u>	_ QA/QC A			- <u></u>
Pate: <u>9-30-</u>					
AMPLE DISPOSAL	94 				· · · · · · · · · · · · · · · · · · ·
Date: <u>9-30-</u> <u>AMPLE DISPOSAL</u> isposal Procedur	e: <u>Riuti</u>				· · · · · · · · · · · · · · · · · · ·
ate: <u>9-30-</u> <u>AMPLE DISPOSAL</u> isposal Procedur	e: <u>Riut</u>				
ate: <u>9-30-</u> <u>AMPLE DISPOSAL</u> isposal Procedur	e: <u>Rint</u>	QA/QC A]	Signed:	Dax, K.)	Jantu:

ļ

APPENDIX E

. .

ŧ

91-20-2a (1/89) SPDES No.: #734052 Part 1, Page 1 of 2 ACCURATE DIE CASTING SITE, ONONDAGA COUNTY EFFLUENT LIMITATIONS AND MONITORING REQUIREMENTS During the period beginning <u>December 1, 1993</u> <u>December 1, 1998</u> and lasting until the discharges from the permitted facility shall be limited and monitored by the permittee as specified below: Minimum Monitoring Requirements Outfall Number & **Discharge Limitations** Measurement Sample Effluent Parameter Daily Avg. Daily Max. Units Frequency Type Remedial Discharge to Bishop Brook (0-66-11-P26-37-6-2-4), Class C (TS) Flow Monitor 150.000 -GPD Continuous Meter pН (6.5 - 8.5)SU 2/week Grab Solids, Total Suspended Monitor 20 ma/l Weekly 3-hr. comp. Solids, Total Dissolved Monitor 500 mg/l Weekly 3-hr. comp. CBOD₅ Monitor Monitor Monthly mg/i 3-hr. comp. TKN Monitor Monitor Monthly mg/l 3-hr. comp. TOD* Monitor 15 mg/l Monthly Calculated **Dissolved Oxygen** Monitor 7 min. Monthly Grab mg/l . . Aluminum, Dissolved Monitor 200 2/month 3-hr. comp. μg/l Monitor Antimony, Total 100 2/month μg/l 3-hr. comp. Chromium, Total Monitor 500 2/month 3-hr. comp. µg/l Cobalt, Total Monitor 10 μg/l 2/month 3-hr. comp. Copper, Total Monitor 100 μg/l 2/month 3-hr. comp. Iron, Total Monitor 300 2/month 3-hr. comp. μg/i . Lead, Total Monitor 20 2/month μg/l 3-hr. comp. Mercury, Total Monitor 0.8 μg/l 2/month 3-hr. comp. Nickel, Total Monitor 200 2/month 3-hr. comp. μg/l Silver, Total Monitor 100 2/month 3-hr. comp. μg/l Vanadium, Total Monitor 30 μg/ŀ 2/month 3-hr. comp. Zinc, Total Monitor 300 2/month μg/l 3-hr. comp. Monitor Cis-1,2-Dichloroethylene 10 μg/l 2/month Grab Monitor Trans-1,2-Dichloroethylene 10 2/month Grab μg/l Methylene Chloride Monitor Grab 50 2/month μg/L Monitor 1,1,2,2-Tetrachloroethane 30 μg/l 2/month Grab Tetrachloroethyiene Monitor 20 μg/F 2/month Grab Toluene Monitor 20 2/month Grab μg/l Trichloroethylene Monitor 10 Grab µg/i 2/month Acetone Monitor 1000 μg/l 2/month Grab -Hexanone Monitor 1000 2/month Grab <u>µд/</u>Г 4-Methyl-2-Pentanone Monitor 1000 2/month Grab μg/l

Part 1, Page 2 of 2

Special Conditions:

Authorization is valid only for the period noted above but may be renewed if appropriate. A request for renewal must be received 6 months prior to the expiration date to allow for a review of nonitoring data and reassessment of monitoring requirements.

Only site generated wastewater is authorized for treatment and discharge.

Discharge is not authorized until such time as an engineering report, plans and specifications are submitted detailing the proposed method of treatment and approval is granted by the Department.

 $*TOD = 1.5 \times CBOD_5 + 4.5 \times TKN.$

SPDES PERMIT FACT SHEET

Prepared by:	Shavne Mitchell Date: <u>12/02/93</u>
Company: <u>Accurate Die Casting</u>	Permit No.: <u>734052</u>
Location: <u>Onondaga County, Fayetteville</u>	Industrial Code No.:
Industrial Segment: <u>NA</u>	Part No.:
Type of Processing & Production Rate:	
Site Remediation.	
Basis for Technology Effluent Limitations:	
Best Professional Judgement (BPJ).	
PARAMETER	BASIS FOR PERMIT CONDITION
Outfall No.: 001 :Remedial	Discharge: Nominal Flow: <u>0.15 MGD</u>
Flow pH TSS	BPJ WQ BPJ
TDS CBOD TKN	WQ Monitor Monitor
TOD DO Aluminum	WQ WQ WQ
Antimony Chromium Cobalt	BPJ BPJ WQ
Copper Iron Lead	BPJ WQ WQ
Mercury Lickel Lilver	WQ/Detection Limit BPJ BPJ
Vanadium Chloride Zinc Cis-1.2-DCE	WQ WQ BPJ
Frans-1.2-DCE Methylene Chloride 1.1.2.2-PCA	BPJ BPJ BPJ

SPDES PERMIT FACT SHEET (contd)

Page 2

PARAMETER BASIS FOR PERMIT CONDITION Outfall No.: 001 :Remedial (contd) Discharge: Nominal Flow: 0.15 MGD PCE WQ BPJ TOLuene BPJ TCE BPJ Acetone BPJ 2-Hexanone BPJ 4-Methyl-2-Pentanone BPJ

.

APPENDIX B

PCB/PAH/VOC SOILS EXCAVATION WORK PLAN (FEBRUARY 1995 WITH REVISED PAGES DATED MARCH 3, 1995)

h

PCB/PAH/VOC Soils Excavation

Former Accurate Die Casting Facility Fayetteville, New York

February 1995



Work Plan

PCB/PAH/VOC Soils Excavation

Former Accurate Die Casting Facility Fayetteville, New York



and

David G. VanArnam, P.É. Senior Vice President

February 1995



5000 Brittonfield Parkway Syracuse, NY 13221 Contents

1.	Introduction	1
	1.1. General	1
	1.2. Background	1
	1.3. Purpose	3
2.	Summary of available information	5
	2.1. General	5
	2.2. Precharacterization data	5
	2.2.1. PCB/PAH/VOC soils area	5
	2.2.2. Septic tank	6
3.	Health and safety plan	7
	3.1. General	7
	3.2. Project organization	7
	3.3. Hazard identification	8
	3.4. Work zones	8
	3.5. Site control	9
	3.6. Employee training	9
	3.7. Medical surveillance	9
	3.8. Personnel protection	10
	3.9. Monitoring	10
	3.10. Decontamination	11
	3.11. Emergency response plan	11
	3.12. Community air monitoring plan	12
4.	Plan for addressing PCB/PAH/VOC soils and septic tank	13
	4.1. General	13
	4.2. Task 1 - project planning	13
	4.3. Task 2 - community relations	14
	4.4. Task 3 - field investigation and implementation of	- •
	remedial measures	14
	4.4.1. Characterization of soils/areal delineation	14
	4.4.2. Characterization of septic tank contents	15

۰.

.

i

PCB/PAH/VOC Soils

4.	Plan for addressing PCB/PAH/VOC soils and septic tank (Continued)	Ξ.
	4.4.3. Excavation of PCB/PAH/VOC soils	16
	4.4.4. Removal of septic tank	17
	4.4.5. Restoration	18
	4.4.6. Equipment decontamination	18
	4.5. Task 4 - data evaluation	19
	4.6. Task 5 - construction management and reporting	19
5.	Project management	21
	5.1. Project organization and responsibilities	21
	5.1.1. Project officer	21
	5.1.2. Engineering manager	21
	5.1.3. Field operations manager	21
	5.2. Project schedule	22

Tables

1. PCB/PAH/VOC soils area precharacterization analysis

2. Septic tank sludges precharacterization analysis

3. Project schedule

Figures

1. Site map

2. Site plan

3. Soil sampling locations/analytical results

ü

Appendices

- A. Laboratory reports
- B. Soil boring protocols
- C. Immunoassay sampling protocol, EPA Method SW846-4020
- D. Chain of custody form

Exhibits

A. Health and safety plan

iii

1. Introduction

1.1. General

This work plan presents the activities that are to be conducted, as a phase of the Remedial Design/Remedial Action (RD/RA) actions required by the Record of Decision (ROD) dated December 1994, to address soils containing polychlorinated biphenyls (PCBs), polynuclear aromatic hydrocarbons (PAHs), and volatile organic compounds (VOCs) at the Accurate Die Casting Site in Fayetteville, New York. This work plan also describes those activities that are to be conducted to address the septic tank and its contents.

1.2. Background

The former Accurate Die Casting site is situated on a 32-acre parcel at 547 East Genesee Street in Fayetteville, New York as shown on Figure 1. The facility, as shown on Figure 2, was constructed in the 1950's and had been used for die casting operations until 1988, when it was abandoned. ITT Commercial Finance Corporation (ITT), which possessed a mortgage on the real property, foreclosed on the premises and was the successful bidder at the auction. Shortly thereafter, an environmental assessment for property transfer was conducted by Stearns & Wheler for ITT, the asset lender. As a result of the site assessment, it was concluded that the potential for environmental contamination existed at the site and that further investigation would be necessary. Stearns & Wheler was subsequently retained to perform a Phase II assessment of the site, completing it in September 1990.

Upon reviewing the Phase II assessment of the former Accurate Die Casting facility, the New York State Department of Environmental Conservation (NYSDEC) required that a Remedial Investigation be conducted. The investigation was completed in accordance with the Remedial Investigation (RI)/Feasibility Study (FS) Work Plan (Stearns & Wheler, May 1992). As presented in the Final Report -Remedial Investigation (Stearns & Wheler, December 1993), the data obtained during the investigation indicates that:

- Ground water quality has been impacted by trichloroethene (TCE). An overburden TCE plume extends from near the former location of MW-3 north towards Bishop Brook. Bedrock ground water also has been impacted as evidenced by levels of TCE detected in samples obtained from bedrock monitoring wells.
- The highest TCE concentrations were observed at about 20 to 25 feet below grade in MW-3, at the interface between the sand/gravel and till layers.
- A portion of the site, referred as the PCB/PAH/VOC soils area, contains polychlorinated biphenyls (PCBs), polynuclear aromatic hydrocarbons (PAHs), and volatile organic compounds (VOCs) above Technical Administrative Guidance Memorandum (TAGM) soil cleanup objectives for New York State.
- The septic tank on site contains materials that could potentially result in VOC and metal concentrations above TAGM soil cleanup levels.

The area containing TCE in the subsurface soil proximal to MW-3 was remediated in June 1994 as part of an Interim Remedial Measure (IRM). The objective of that IRM was to remediate the subsurface soils in the vicinity of MW-3 to reduce the potential for further migration of TCE from the suspected source area to the ground water. As part of that IRM, a ground water recovery and treatment system will also be installed to address the impacted overburden ground water.

2

1.3. Purpose

The purpose of this work plan is to describe the activities that are to be completed to address the soils located in the PCB/PAH/VOC soils area and sludges contained in a septic tank on-site. These activities are intended to specifically address two of the five components of the selected remedy identified in the ROD. The activities include development and implementation of an excavation plan for the PCB/PAH/VOC soils area, removal and disposal of septic tank sludges, and restoration of the two areas in an environmentally sound manner.

The objectives of this work phase are:

- 1. Remediate the surface and subsurface soil in the PCBs/PAH/VOC area near MW-4 to reduce the potential for migration of contaminants from the area.
- 2. Prevent incidental ingestion of surface soils containing PCBs/PAH/VOC, and prevent dermal adsorption of residues.
- 3. Remove the residues of PCBs/PAH/VOCs in surface and subsurface soil to achieve the Remedial Action Objectives (RAOs) established by the ROD.
- 4. Remove the septic tank and its contents to prevent potential release of the tank contents to the surrounding soil.

PCB/PAH/VOC Soils

O'Brien & Gere Engineers, Inc.

Final: February 6, 1995 Div71Y

4

2. Summary of available information

2.1. General

This section presents select remedial investigation (RI) data that are pertinent to the development of a plan to address the soils containing PCB/PAH/VOC residues and the septic tank located on site.

2.2. Precharacterization data

2.2.1. PCB/PAH/VOC soils area

The PCB/PAH/VOC soils area is located in the northern portion of the site immediately south of MW-4, as shown on Figure 2. Preliminary investigations were conducted in August 1993 by Stearns & Wheler to evaluate the extent of PCB/PAH residuals within this area.

The August 1993 investigation included the collection of grab samples from a depth of 0 to 6 inches below the ground surface at seventeen randomly selected locations shown on Figure 3. From these locations, a total of 20 samples were collected and analyzed in a laboratory for PCBs and PAHs. In addition, the three NYSDEC split-samples out of these 20 samples were analyzed for VOCs. The analytical results are summarized in Table 1 and the laboratory data reports are included in Appendix A. In 14 of the 20 samples obtained, PAH analytes were present in the soils at concentrations ranging from 0.26 to 113 mg/kg. PCB concentrations in 16 of the 20 samples ranged between 0.01 and 2.6 mg/kg, and VOCs (Dichloroethylene) were present in the three split samples collected by NYSDEC at concentrations between .018 and 190 ppm.

O'Brien & Gere Engineers, Inc.

2.2.2. Septic tank

During the RI, the sludge from a decommissioned septic tank was also sampled to assess if it was a potential source of ground water contamination on the site. The tank had been decommissioned by filling it with gravel, however, the gravel did not completely fill the tank. Near the top of the tank, a white substance with a soil-like consistency was found. A sample of the white substance was obtained and analyzed for volatile organics and metals. A summary of the results of the analyses is presented in Table 2. No volatile organics were detected in the sample but several metals were detected in the material. Of the metals detected in the material, only zinc was found to be at a concentration significantly above levels exhibited in other soil samples from the site. During implementation of this work plan, additional samples of the septic tank contents will be collected and analyzed to characterize the contents for off-site disposal.

6

3. Health and safety plan

3.1. General

A site specific Health and Safety Plan, provided as Exhibit A, was prepared to provide guidance for O'Brien & Gere Technical Services, Inc. personnel during the conduct of remedial activities in these two areas. This document is adopted for O'Brien & Gere Engineers onsite personnel during the installation of soil borings, and conduct of soil sampling and excavation activities. The Health and Safety Plan was prepared in accordance with the Federal Occupational Safety and Health Administration guidelines (29 CFR 1910.120) utilizing data generated during the RI, including site history and subsurface conditions.

O'Brien & Gere Technical Services will be responsible for the implementation and enforcement of the Health and Safety Plan (HASP) while activities are conducted on the site.

3.2. Project organization

The HASP includes an organization listing identifying key personnel involved in the proposed activities. The listing identifies the following personnel, and describes their duties as related to health and safety issues:

- Project Officer
- Project Supervisor
- Corporate Safety Coordinator

- Health and Safety Specialist
- Field Supervisor/Site Health and Safety Coordinator

The on-site Health and Safety Coordinator (HSC) will have the following responsibilities:

- The HSC will be available at all times during site operations;
- The HSC will have the authority to enforce the HASP and stop operations if the safety or health of site personnel is jeopardized;
- The HSC will evaluate all monitoring data and make necessary field decisions regarding site health and safety procedures;
- The HSC may order evacuation of the site if necessary to protect the safety and health of workers.

3.3. Hazard identification

Potential health and safety hazards which may be encountered during the execution of the work are identified in the HASP. Examples of anticipated hazards include, but are not limited to, exposure to soil containing polychlorinated biphenyl (PCB), polynuclear aromatic hydrocarbons (PAH), and/or volatile organic compounds (VOCs), and safety hazards inherent in working around excavations and machinery. The HASP includes descriptions of methods to be employed to reduce the risks associated with the identified hazards.

3.4. Work zones

The HASP describes work zones in which specific operations or tasks will take place, and describes specific site entry and decontamination procedures at designated control points in accordance with the provisions of 29 CFR 1910. At a minimum, three work zones (exclusion zone, contamination reduction zone, and clean zone) will be established to perform this work. A map showing the work zones will be provided to the NYSDEC and included in the HASP once the excavation plan, described later in Section 4.4.3, has been prepared.

3.5. Site control

The contractor has a site control program established as part of the HASP. The following information is included in the site control program:

- Site map, with work zones;
- Site entry procedures;
- Description of the use of "buddy" system;
- Site communications, including means of alerting workers to emergencies;
- Locations of nearest medical assistance, and emergency phone numbers.

3.6. Employee training

Employees performing on-site activities and the supervisors responsible for the site are trained to the level required by their assigned function as specified by 29 CFR 1910.120(e). Written certification of the successful completion of the necessary training is required by those regulations.

3.7. Medical surveillance

The contractor will establish and implement a Medical Surveillance Program (MSP) for employees engaged in on-site operations,

Final: February 6, 1995 Div71Y 9

consistent with 29 CFR 1910.120(f). The MSP includes physical examinations performed by or under the supervision of a licensed physician. The written opinion of the attending physician on the employee's ability to perform the required work is made available to the employee.

The contractor will retain a record of the required medical surveillance information for the appropriate period as specified in 29 CFR 1910.120.

3.8. Personnel protection

Engineering controls, work practices, the use of personal protection equipment (PPE), or a combination of these will be implemented during site operations to protect employees from exposure to hazardous substances and safety hazards as required by 29 CFR 1910.120(g). A written PPE program including the following elements will be incorporated into the HASP:

- PPE inspection prior to, during and after use;
- PPE selection (based on site hazards), use, and limitations, including heat stress and cold injury protection;
- PPE maintenance, storage, decontamination, and disposal;
- PPE training, proper fit, and procedures for donning and doffing PPE;
- Evaluation of the effectiveness of the PPE program.

3.9. Monitoring

The HASP defines a monitoring program in accordance with 29 CFR 1910.120(h) to select and maintain proper engineering controls, work practices, and PPE. Breathing zone air monitoring will be performed

O'Brien & Gere Engineers, Inc.

10

to identify levels of airborne particulates and VOCs to determine the necessary level of employee protection.

3.10. Decontamination

The contractor will develop and implement decontamination procedures as required by 29 CFR 1910.120(k) which will minimize employee contact with hazardous substances or equipment and materials that have contacted hazardous substances.

3.11. Emergency response plan

The HASP includes an emergency response plan section meeting the requirements of 29 CFR 1910.120(l). The emergency response plan addresses the following elements:

- Pre-planning of site operations to prevent emergencies;
- Personnel roles, lines of authority, and communications;
- Emergency recognition and prevention;
- Safe distance and places of refuge;
- Evacuation routes and procedures;
- Emergency first aid, medical treatment, alerting, and response procedures;
- Emergency and personnel equipment maintained at the site for emergencies.

3.12. Community air monitoring plan

The HASP includes a Community Air Monitoring Plan prepared in accordance with New York State Department of Health guidance. The plan includes monitoring for VOCs and particulates at the perimeter of the work area, and describes removal actions to be implemented if target levels (5 ppm above background for VOCs, 100 μ g/m³ for particulates) are exceeded.

O'Brien & Gere Engineers, Inc.

Final: February 6, 1995 Div71Y
4. Plan for addressing PCB/PAH/VOC soils and septic tank

4.1. General

This section presents the plan for addressing the PCB/PAH/VOC soils area and septic tank on-site. The plan consists of the following tasks.

Task 1 - Project Planning

Task 2 - Community Relations

- Task 3 Field Investigation and Implementation of Remedial Measures
- Task 4 Data Evaluation
- Task 5 Construction Management and Reporting

The order in which these tasks are presented is the general order in which they will be performed. However, since some tasks (e.g., Data Evaluation) occur throughout conduct of the work, the order of presentation gives only a general indication of the order the tasks will follow.

4.2. Task 1 - project planning

This task involves the performance of several subtasks to produce the project planning documents and project schedule necessary to execute the work described herein. These subtasks include: field reconnaissance, project meetings, evaluation of existing data, and development of project plans including draft and final revisions of this Work Plan and a Health and Safety Plan.

4.3. Task 2 - community relations

O'Brien & Gere Engineers, Inc., on behalf of O'Brien & Gere Technical Services will provide copies of necessary documents to the local repository, and the NYSDEC offices. O'Brien & Gere will also be available for at least one public informational meeting, as scheduled and coordinated by NYSDEC, to provide technical support if required. O'Brien & Gere will be prepared to present pertinent data and recommendations at the meeting and respond to questions.

4.4. Task 3 - field investigation and implementation of remedial measures

The soil sample analytical data that was collected in August 1993 enabled a preliminary assessment of the PCB/PAH/VOC soils area. However, additional soil characterization is needed to further define the horizontal and vertical extent of the PCB/PAH/VOC compounds in the soils. Further soil characterization will also be conducted to characterize the soils for disposal.

4.4.1. Characterization of soils/areal delineation

Six soil borings will be advanced to a depth of approximately six feet in the PCB/PAH/VOC soils area, at the locations shown on Figure 3, to further delineate the extent of PCBs/PAH/VOCs. Soil samples will be collected from each boring at approximately two foot depth intervals in accordance with the protocols enclosed as Appendix B and will be screened in the field for VOCs using a photoionization detector (PID). The PID, equipped with a 10.2 eV lamp, will be calibrated to correct for the photoionization sensitivity of 8.9 for TCE compared to an isobutylene standard of 7 ppm (v/v). The samples will also be screened for PCBs using an immunoassay sampling kit in accordance with EPA approved analytical method SW846-4020. To provide a correlation between the field screening and "actual" concentrations, one of the samples screened for VOCs using the PID and one of the samples screened for PCBs using the immunoassay test will be analyzed by EPA Method 8010/8020 and EPA Method 8080, respectively. No screening in the field is proposed for PAHs. If the results of the field screening indicate PCBs or VOC levels above the established RAOs in the samples collected from the bottom of the boring, the depth of the boring will be increased, as

deemed appropriate by the hydrogeologist on-site, to more accurately delineate the vertical extent of the residues.

Following collection, soil samples will be transported to O'Brien & Gere Laboratories, Inc. using a chain-of-custody form (Appendix D) where, based on field screening results, two samples from each boring will be selected for PCB analyses using EPA Method 8080, PAHs using EPA Method 8270, and VOCs using EPA Method 8010/8020. One sample selected from each boring will be from the interval that exhibited the highest PID reading, and the other sample selected will be from the interval that exhibited the highest PCB concentration. However, if the same interval exhibits both the highest PID reading and PCB concentration, a sample from a second, non-contiguous interval will be selected for laboratory analyses. Also, if several intervals exhibit PID or PCB levels close to the highest values obtained during field screening, the sample selected will be from the interval having the greatest depth. A composite soil sample, mixed in the laboratory using portions of the twelve soil samples selected for laboratory analyses, will also be analyzed for toxicity characteristic leachate procedure (TCLP) metals, volatiles, semi-volatiles, pesticides and herbicides as well as for flashpoint, pH. and cyanide and sulfide reactivity to precharacterize the soils for disposal off-site.

Based on the concentration of PCBs/PAH/VOCs exhibited in the soil samples, and the results of the field screening performed during these and previous soil sampling activities within the area, a plan depicting the approximate limits of the proposed excavation will be prepared for submittal to the NYSDEC. Using the laboratory results of the composite soil sample, a waste characteristic profile sheet also will be completed and submitted to an appropriate off-site disposal facility for acceptance of the soils to be excavated.

4.4.2. Characterization of septic tank contents

In order to characterize the contents of the septic tank for disposal off-site, a grab sample will be obtained and submitted to O'Brien & Gere Laboratories, Inc. for analyses. The sample will be analyzed for TCLP metals, volatiles, semi-volatiles, pesticides and herbicides as well as for flashpoint, pH, and cyanide and sulfide reactivity. Using the laboratory results of the septic tank contents, a waste characteristic profile sheet will then be completed and submitted to an appropriate off-site disposal facility for acceptance of the tank contents.

4.4.3. Excavation of PCB/PAH/VOC soils

A "constructable quality" excavation plan will be prepared and submitted for NYSDEC approval. The excavation plan will include one or more figures showing the approximate extent of the excavation, based on the results of the proposed soil sampling and analyses described in Section 4.4.1; the location of material staging areas; erosion and sediment controls; and the proposed exclusion and decontamination zones. The excavation area will encompass the sample locations containing PCB, PAH, and VOC residuals above the RAOs. The excavation plan submittal also will include performance specifications appropriate to the type of work proposed such as earthwork, backfill and compaction, sedimentation and erosion control, surface water diversion, excavation dewatering, material handling and disposal, and sampling analyses.

Prior to initiating excavation activities, acceptance of the material by an off-site disposal facility will be obtained so that the material need not be staged on-site prior to transport off-site. Also, a temporary construction barrier will be erected on the perimeter of the exclusion zone, established around the area to be excavated, to reduce the potential for unauthorized personnel from inadvertently entering the area. Additional work area controls, such as silt fences and storm water runoff diversions, also will be installed to minimize the amount of storm water runoff entering or leaving the exclusion and contamination reduction zones.

Excavation of the PCB/PAH/VOC soil area will be performed using conventional excavating equipment (backhoe, dump truck, etc.). The extent of excavation will be directed by the on-site environmental technician based upon field screening results which will be compared to soil precharacterization data. Field screening will consist of collecting soil samples from the base and walls of the excavation on a 20 foot grid pattern and testing the samples for VOCs using a portable PID and for PCBs using an immunoassay sampling kit in accordance with EPA approved analytical method SW846-4020. The results of the field screening will be recorded within the field log book.

If necessary, as the soil is removed it will be transported to a designated temporary staging area located on site. In this staging area, the soils will be temporarily stored on top of and be covered by polyethylene to minimize contact with the surface and minimize erosion or contact with precipitation prior to transportation off-site to an appropriate disposal facility.

Upon completion of excavation, based upon field screening, not less than ten confirmatory soil samples will be collected on a 20 foot grid pattern from the floor and walls of the excavation. The samples will be transported to O'Brien & Gere Laboratories, Inc. using appropriate chain-of-custody procedures where they will be analyzed for PCBs using EPA Method 8080, PAHs using EPA Method 8270, and VOCs using EPA Method 8010/8020. The data package submitted to the NYSDEC shall consist of the validated results of these analyses, as well as the results of the:

- Trip blank (1 per container)
- Matrix spike (1 per 20 samples obtained)
- Matrix spike duplicate (1 per 20 samples obtained)
- Equipment blank (1 per 20 samples obtained)
- Method quality assurance/quality control data

Once residual levels in soils remaining in place meet the RAOs presented in the Record of Decision (ROD) the excavated area will be restored.

4.4.4. Removal of septic tank

Prior to removing the septic tank contents, acceptance of the material by an off-site disposal facility will be obtained so that the material need not be staged on-site prior to transport off-site. Removal of the septic tank contents will be performed using conventional equipment (backhoe, vacuum truck, dump truck, etc.). If it is necessary to temporarily stockpile the material, it will be transported to a temporary staging area located on-site. In this staging area, the material will be placed on top of and be covered by polyethylene sheeting to minimize contact with the surface and prevent erosion or contact with precipitation prior to transportation off-site to an appropriate disposal facility.

Once the contents of the septic tank are removed the tank will also be removed. If the tank is constructed of concrete, it will be disposed off-site at an appropriate landfill facility, but if it is a metal tank, it will be cleaned, have holes drilled/cut in its side, and be transported to a metal scrapyard.

4.4.5. Restoration

Once excavation activities have been completed, as concluded based on the results of verification sampling and approved by the NYSDEC, the silt fences will be removed and disposed of off-site, as appropriate. The two excavated areas will then be backfilled with clean granular material and/or graded to prevent ponding of surface water and restored with a grass cover to minimize erosion.

4.4.6. Equipment decontamination

O'Brien & Gere Technical Services shall construct, operate and maintain a temporary equipment decontamination area on-site at a location approved by the Owner. Equipment which has been in contact with the exclusion zone shall be stored at a location approved by Owner or shall be thoroughly decontaminated to the satisfaction of the Owner's on-site representative prior to leaving the contamination reduction area.

Materials taken into the Exclusion Zone will be assumed to be contaminated and therefore be decontaminated before the item leaves the Contamination Reduction Area. Contaminated vehicles, equipment, and materials shall be cleaned prior to leaving the site. Decontamination shall consist of thorough brushing to remove solid materials and pressure steam cleaning or through other equivalent methods. The method of decontamination shall meet the requirements of federal, state, and local authorities and shall be subject to approval by the Owner.

The equipment decontamination area will consist of an impermeable material (sloped to a sump area) with appropriate height curbing. O'Brien & Gere Technical Services shall collect liquid waste material in a temporary tanker or similar device for treatment and or disposal off-site.

Personnel engaged in vehicle decontamination shall wear protective equipment including disposable clothing and respiratory protection as necessary in accordance with the Health and Safety Plan.

4.5. Task 4 - data evaluation

This task includes the data reduction and evaluation effort. O'Brien & Gere Engineers will organize, analyze, interpret, and present the data obtained during implementation of the activities described in this Work Plan, including:

- Evaluation of the soil samples obtained from the five proposed borings to assess the vertical and horizontal extent of PCBs/PAH/VOC residues.
- Evaluation of PCB/PAH/VOC verification sampling and analytical results to document the levels of PCBs/PAH/VOC residuals remaining following completion of this the remedial measures.
- Evaluation and documentation of soil sampling activities and incorporation of the data into a risk assessment, as part of a contingency plan, if necessary.

This information will be submitted to the NYSDEC for review as it becomes available, and will be included, summarized, and interpreted in an PCB/PAH/VOC soils excavation Summary Report.

4.6. Task 5 - construction management and reporting

This task includes the coordination and direction of field activities and the reporting of findings to the NYSDEC. A description of the scheduling and reporting requirements are presented in Section 5 of this Work Plan.

5. Project management

5.1. Project organization and responsibilities

A project management team has been assembled to plan, coordinate, and implement the activities described in this Work Plan. The lead organization in the process is O'Brien & Gere Technical Services, Inc., a full service environmental contracting firm. O'Brien & Gere Engineers, Inc. has been retained by O'Brien & Gere Technical Services to prepare this Work Plan and provide engineering inspection. The roles and responsibilities of the individuals assigned to this program are described below.

5.1.1. Project officer

Terry L. Brown, P.E. of O'Brien & Gere Technical Services will serve as project officer for this program. Mr. Brown is responsible for coordinating the overall management of the project, directing field and office personnel, and managing the administrative aspects of the project.

5.1.2. Engineering manager

David S. Towers, P.E. of O'Brien & Gere Engineers will act as the engineering manager. Mr. Towers will develop the soil excavation plan, review deliverables prior to submittal to NYSDEC, monitor the project progress, and oversee the remedial efforts.

5.1.3. Field operations manager

Mr. Anthony Geiss of O'Brien & Gere Technical Services will serve as field operation manager. Mr. Geiss' responsibility will include overseeing the day-to-day activities necessary in the field to implement the work described in this Work Plan.

5.2. Project schedule

The project schedule for the activities described in this Work Plan is presented as Table 3. As shown on the table, it is anticipated that NYSDEC review and approval of this work plan will be completed within four weeks of its submittal. Following NYSDEC review and approval, field investigation activities will be initiated within three weeks (pending availability of drilling subcontractors) and completed within six weeks.

Upon completion of the field activities and evaluation of the data, an excavation plan will be developed. It is estimated that excavation activities including restoration of the area will take approximately six weeks to complete once initiated. Within 30 business days following completion of excavation backfilling and receipt of validated laboratory reports, a Summary Report documenting the work completed will be prepared and submitted to NYSDEC.

Prepared by:

Alfred R. Farrell, P.E. Project Engineer

David S. Towers, P.E. Senior Project Engineer

Reviewed by:

James R. Heckathorne, P.E. Managing Engineer

1

TABLES

Table 1	
PAH, VOC and Total PCB Concentrations in Soil from Forme	er Waste Oil Spill Area

Former Accurate Die Ca	asting Site
Fayetteville, New '	York

Analyte	S-1	S-2	S-3	S-4	S-5	S-5s	S6	S-8	S-9	S-10	S-11	S-12	S-13	S-13s	S-14	S-14s	S-15	S-16	S-17	S-18
Naphthalene			-			N/A								N/A		N/A				1.3
Acenaphthylene						N/A								N/A		N/A		•		
Acenaphtene		0.12				N/A						2	3.5	N/A	2.8	N/A		0.18		3.6
Fluorene		0.11				N/A								N/A	2.6	N/A		0.15		3.3
Phenanthrene	0.19	1.6		1.7		N/A	3.8	0.27	0.82	6.1	0.27	15	10	N/A	24	N/A	1.7	1.8	0.92	25
Anthracene	0.19	0.19				N/A	1.2		0.79	4.8		6		N/A	3.3	N/A	1.3	0.31	0.13	4.4
Fluoranthene	0.16	1.7			0.82	N/A	3.2	0.26	0.68	7.3		17	9.3	N/A	20	N/A	1.3	1.9	1.3	24
Pyréne	0.43	1.1	0.044	1.5	0.89	N/A	. 3.5	0.68	0.78	7.3	0.22	11	7.7	N/A	13	N/Ă	1.4	4.2	1.3	15
Benzo(a)anthracene		0.38				N/A	2.1	0.1	0.74	7.1		13	7.4	N/A	12	N/A	2.5	0.71	0.34	5.7
Chrysene	0.11	0.66			4.8	N/A	5.1	0.16	1.2	14		17	16	N/A	17	N/A	10	1.2	0.66	8.4
Benzo(b)fluoranthene		0.46				N/A				6.4		8.9	5.7	N/A	7.7	N/A		0.73	0.47	4.6
Benzo(k)fluoranthene		0.47				N/A			0.27	5.4		8.4	5.1	N/A	6.7	N/Ā		0.74	0.51	5.1
Benzo(a)pyrene		. 0.38			1.7	N/A				6.2	 	9.6	5.4	N/A	7.7	N/A	1.3	0.81	0.49	5.3
Indeno(1,2,3-cd)pyrene						N/A								N/A		N/A		0.61	•	3.1
Dibenzo(a,h)anthracene						N/A								N/A		N/A				í
Benzo(g,h,i)perylene						N/A								N/A		N/A				3
Total PAH compounds	1.08	7.17	0.044	3.2	8.21	N/A	18.9	1.47	5.28	64.6	0.49	107.9	70.1	N/A	116.8	<u>_N/Ā</u>	19.5	13.34	6.12	111.8
Total PCBs	· ·			0.32	1.9	N/A	0.7		0.37	2.6	0.24	1.7	1.43	N/A	1.5	N/A	1.5			
Methylene chloride	N/A	N/A	N/A	N/A	N/A	0.052B	N/A	N/A	N/A	N/A	N/A	N/A	N/A		N/A		N/A	N/A	N/A	N/A
Dichloroethylene	N/A	N/A	N/A	N/A	N/A	0.018B	N/A	N/A	N/A	N/A	N/A	N/A	N/A	190	N/A	19	N/A	N/A	N/A	N/A
Trichloroethylene	N/A	N/A	N/A	N/A	N/A		N/A	N/A	N/A	N/A	N/A	N/A	N/A		N/A		N/A	N/A	N/A	N/A
· · ·																			al\p	ahtab1

Notes:

(1) Blank spaces indicate that the analyte was not detected.

(2) Samples designated with an "s" represent split sampling data performed by the NYSDEC in July 1993.

(3) N/A indicates that the analyte was not analyzed for.

Table 2

Septic Tank Sludge Precharacterization Analysis

Former Accurate Die Casting Facility Fayetteville, New York

<u>'Analyte (ppm)</u>	NYSDEC TAGM Level	Septic Tank Sludge
Aluminum	30 or SB	15,100
Antimony	30 or SB	, 19 J
Arsenic	7.5 or SB	6.7
Barium	300 or SB	83.5
Beryllium	0.14	0.87
Cadmium	1 or SB	ND
Calcium	SB	55,500
Chromium	10 or SB	27
Cobalt	30 or SB	12.3
Copper	25 or SB	65.7 J
Iron	2,000 or SB	24,600
Lead	30 or SB	23.3
Magnesium	SB	28,000
Manganese	SB	891
Mercury	0.10	ND
Nickel	13 or SB	28.2
Potassium	4,000 or SB	2,200
Selenium	2 or SB	ND
Silver	200 or SB	ND
Sodium	3,000 or SB	126
Thallium	20 or SB	ND
Vanadium	150 or SB	31.9
Zinc	20 or SB	644

Notes:

SB means Site Background ND indicates the analyte was not detected

Table 3

,۰۰۰ ...

...

. '

:

. 7

Project Schedule

Former Accurate Die Casting Site Fayetteville, New York

的复数形式 使某人 计正式公式	1.			·	<u>,</u>									۷	Neel	k 🦾			•						ž.			_	
Description	1	2	3	4	5	6.	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22	23	24	25	26	27	28	29
Project Planning						-		£	•																				
- Work Plan approved by the NYSDEC			1		<u> </u>			T																					
Precharacterization Investigations																													
- Additional soll boring/sampling								T	—	I .																1			
and septic tank sampling																													
- Laboratory analyses																							L						
- Prepare waste profile sheets and			1		[!		ł .				
obtain disposal facility acceptance	<u> </u>							I	l																	·			
Excavation Plan Development				•												_												——	<u> </u>
- Excavation area survey			<u> </u>			<u> </u>																	_	ļ					
- Submittal of excavation plan																					ľ								
to the NYSDEC								I															L	.	L			\vdash	
- NYSDEC review of the excavation plan								I															 						
- Comment response			1			<u> </u>		1	<u> </u>									<u> </u>	·						 '			┝╍╌┦	
 Excavation Plan approved 	·							<u> </u>		L_							ŀ						L_	Į	L		<u>ц</u>		
Material Excavation, Removal, and Dispos	al					· 			·															r –			·	——	
- mobilization and site controls installed			1	1	L	<u> </u>		1			L							L				I	<u> </u>	<u> </u>	Ļ				
 Excavation/transportation/disposal 								1		L				•				L							I				
- Verification soil sampling and analyses			1		L			<u> </u>											·		ļ		<u> </u>					\vdash	
- Laboratory analyses	_		1					I											<u> </u>				ļ	 					
- Preliminary verification results to DEC			1					1		ļ				. .								.	 	<u> </u>	 				
- Backfilling & Restoration			1				L	<u> </u>						l				L_{-}			I		<u> </u>				<u> </u>		
Summary Report														-	.	<u> </u>		·											
- Verification data validation			1_					L		_			L		<u> </u>			L						I ·					
- Report preparation	1_		1			1	_	1	 						 	<u>ا</u>	 	L		 		 	 	<u> </u>					
- Report submittal to the NYSDEC				<u> </u>	L			L	<u> </u>				L_		L		[L_{-}				I	L	L					

al\irmtab3

· · · ·

• •

FIGURES









APPENDIX A

LABORATORY REPORTS

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. •

.

SAMPLE MATRIX	SOIL		SAMPLE ID:	S- 1
CONC. LEVEL	: LOW		LAB ID:	1765104
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	2.00
ANALYSIS DATE	: 8/14/93		X HOISTURE:	17
	• • • •		UG/KG	
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	800,0 U.]	
. 2	208-96-8	Acenaphthylene	800.0 U.	
3	83-32-9	Acenaphthene	800.0 U.	
4	86-73-7	Fluorene	800.0 U.	
5	185-01-8	Phenanthrene	180.0 J.	
6	120-12-7	Anthracene	800.0 U	
7	206-44-0	Fluoranthene	150.0 J.	
8	129-00-0	Pyrene	430.0 J.	
9	56-55-3	Benzo(a)Anthracene	800.0 U	
10	218-01-9	Chrysene	800.0 U.	
⁻ 11	205-99-2	Benzo(b)Fluoranthene	800.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	800.0 U.	
13	50-32-8	Benzo(a)Pyrene	800.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	800.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	800.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	800.0 U.	
	!		!	

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

.

SAMPLE MATRIX	soil		SAMPLE ID:	\$-1RE
CONC. LEVEL	.: LOW	· ·	LAB 1D:	1765104
EXTRACTION DATE	: 8/2/93	•	DIL FACTOR:	2.00
ANALYSIS DATE	: 8/18/93		X HOISTURE:	17
			UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	800.0 U.	
2	208-96-8	Acenaphthylene	800.0 U.	
3	83-32-9	Acenaphthene	800.0 U.	
4	86-73-7	Fluorene	800.0 U.	
5	85-01-8	Phenanthrene	190.0 J.	
. 6	120-12-7	Anthracene	190.0 J.	,
7	206-44-0	Fluoranthene	160.0 J.	•
8	129-00-0	Pyrene	340.0 J.	
9	56-55-3	Benzo(a)Anthracene	800.0 U.	
10	218-01-9	Chrysene	110.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	800.0 U.	
· 12	207-08-9	Benzo(k)Fluoranthene	800.0 U.	
13	50-32-8	Benzo(a)Pyrene	800.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	800.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	800.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	800.0 U.	
	1	.1		

1 D-T NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

.SAMPLE MATRIX: SOIL SAMPLE ID: s-1 CONC. LEVEL: LOW LAB SAMPLE ID: 1765104 EXTRACTION DATE: 8/02/93 DIL FACTOR: 1.00 ANALYSIS DATE: 8/15/93 X MOISTURE: 17 UG/KG CMPD # CAS Number PESTICIDE/PC8 COMPOUND (DRY BASIS)

.

- t

. •

-			
1	319-84-6	alpha-BHC	AK I
2	319-85-7	beta-BHC	i na
3	319-86-8	delta-BHC	j na
4	58-89-9	gamma-BHC(Lindane)	I NA
5	76-44-8	Keptachlor	i na
6	309-00-2	Aldrin	AN I
7	1024-57-3	Keptachlor Epoxide	Í NA
8	959-98-8	Endosulfan 1	} KA
9	60-57-1	Dieldrin	NA NA
10	72-55-9	4,4'-DDE	I. NA
11	70-20-8	Endrin	NA NA
12	33213-65-9	Endosulfan II	A NA
13	72-54-8	4,4-DDD	KA KA
14	1031-07-8	Endosulfan Sulfate	I NA
15	50-29-3	4,4'-DDT	NA T
16	72-43-5	Methoxychlor	NA NA
17	53494-70-5	Endrin Ketone	L NA
18	7421-36-3	Endrin Aldehyde	j na
19	57-74-9	Chiordane	NA I
20	8001-35-2	Toxaphene	NA NA
21	12674-11-2	Aroclor-1016	100.000 U.
22	11104-28-2	Aroclor-1221	100.000 U.
23	11141-16-5	Aroclor-1232	100.000 U.
24	53469-21-9	Aroclor-1242	100.000 U.
25	12672-29-6	Aroclor-1248	j 100.000 U.
26	11097-69-1	Aroclor-1254	190.000 U
27	11096-82-5	Aroclor-1260	190.000 U.
			Í

SEMIVOLATILE ORGANICS AMALYSIS DATA SHEET

. 1

SAMPLE MATRIX	SOIL		SAMPLE ID:	\$-2
CONC. LEVEL	: LOV		LAB ID:	1765101
TEXTRACTION DATE	: 8/2/93	•	DIL FACTOR:	2,00
ANALYSIS DATE	: 8/13/93		X MOISTURE:	22
			UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	800.0 U. 1	
2	208-96-8	Acenaphthylene	800.0 u. j	
3	83-32-9	Acenaphthene	į 120.0 J. į	
4	86-73-7	Fluorene	j 110.0 J. j	
5	85-01-8	Phenanthrene	1600.0	•
6	120-12-7	Anthracene	190.0 J.	
7	206-44-0	Fluoranthene	1700.0	
	129-00-0	Pyrene	1100.0	
9	56-55-3	Benzo(a)Anthracene	380.0 J.	
10	218-01-9	Chrysene	660.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	460.0 J.	
. 12	207-08-9	Benzo(k)Fluoranthene	470.0 J.	
13	50-32-8	Benzo(a)Pyrene	380.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	800.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	800.0 U.	
16	191-24-2	Benzo(g,h,i)Perviene	800.0 U.	
		·	!	

0000034

1 D-T

.

÷

. :

,::<mark>:</mark> . - . . .

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	•			
	SAMPLE MATRIX:	SOIL	SAMPLE ID:	S-2
	CONC. LEVEL:	LOW	LAS SAMPLE ID:	1765101
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	1.00
	ANALYSIS DATE:	8/14/93	% HOISTURE:	22
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPO	UND	(DRY BASIS)
י י	317-04-0 310-85-7	alpha-onu bete-BVC	l	NA NA
. 7	319-0J-7 310-04-0	delte.RVC	1	NA NA
د ر	<u>5</u> 17-00-0 59-90-0		1	NA NA
4	70°07°7 76-1/-9	gamma-BRU(Lindane)	1	NA
2	70° 44° 0	Neptachior	1	NA
•				KA
7	1024-57-3 ·	Reptachior Epoxide		NA NA
8	0-57-98-8	Encosultan I		NA
9	60-57-1	Dieldrin		. NA
10	72-55-9	4,4'-DDE		AK
11	70-20-8	Endrin	1	NA
12	33213-65-9	Endosulfan II		NA
13	72-54-8	4,4-DDD	. [NA
14	1031-07-8	Endosulfan Sulfate	·	NA
15	50-29-3	4,4'-DDT	1	AA
16	72-43-5	Methoxychlor	1	NA
17	53494-70-5	Endrin Ketone		NA
18	7421-36-3	Endrin Aldehyde	1	NA
19	57-74-9	Chlordane	ł	NA
20	8001-35-2	Toxaphene		NA
21	12674-11-2	Aroclor-1016		100.000 U.
Z2	11104-28-2	Aroclor-1221	1	100.000 U.
23	11141-16-5	Aroclor-1232	1	100.000 U.
24	53469-21-9	Aroclor-1242	1	100.000 U.
25	12672-29-6	Aroclor-1248		100.000 U.
26	11097-69-1	Aroclor-1254	Ĩ	210.000 U.
27	11096-82-5	Aroclor-1260	Ì	210.000 U.
1	i i		. i	

-

:

...

. .

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. 2

. •

S-3	SAMPLE ID:		IX: SOIL	SAMPLE MATRIX
1765102	LAB ID:		EL: LOW	CONC. LEVEL
1.00	DIL FACTOR:		re: 8/2/93	EXTRACTION DATE
19	% MOISTURE:		TE: 8/13/93	ANALYSIS DATE
	UG/KG			
	(DRY BASIS)	PAH COMPOUNDS	# CAS Number	CHPD #
) 400.0 U.]	Naphthalene	1 91-20-3	1
	400.0 U.	Acenaphthylene	2 208-96-8	2
	400.0 U.	Acenaphthene	83-32-9	3
	400.0 U. j	Fluorene	86-73-7	. 4
	400.0 U.	Phenanthrene	85-01-8	5
	400.0 U.	Anthracene	5 120-12-7	6
	400.0 U.	Fluoranthene	206-44-0	7
	400.0 U.	Pyrene	3 129-00-0	8
	400.0 U. [Benzo(a)Anthracene	56-55-3	9
	400.0 U. [Chrysene	218-01-9	. 10
	400.0 U.	Benzo(b)Fluoranthene	205-99-2	11
	400.0 U. j	Benzo(k)Fluoranthene	2 207-08-9	12
	400.0 U.	Benzo(a)Pyrene	50-32-8	13
• •	400.0 U.	Indeno(1,2,3-cd)Pyrene	193-39-5	. 14
	400.0 U. I	Dibenz(a,h)Anthracene	53-70-3	- 15
	400.0 U.	Benzo(g,h,i)Perylene	[191-24-2	16
		.!		

18 PAH Nytest environmental inc.

.

÷

.

.

É,

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. •

SAMPLE MATRIX: SOI	L	SAMPLE ID:	S-3RE
CONC. LEVEL: LC	N	LAB ID:	1765102
EXTRACTION DATE: 8/2/9	23	· DIL FACTOR:	1.00
ANALYSIS DATE: 8/16/9	3	X HOISTURE:	19
		UG/KG	
CHPD # CAS Numbe	PAH COMPOUNDS	(DRY BASIS)	
1 91-20-3		400.0 U.	
2 208-96-8	Acenaphthylene	400.0 U.	
3 83-32-9	Acenaphthene	400.0 U. [
4 86-73-7	Fluorene	400.0 U.]	
5 85-01-8	Phenanthrene	400_0 U. j	
6 120-12-7	Anthracene	400.0 U.	
7 206-44-0	Fluoranthene	400.0 U.	
8 129-00-0	Pyrene	44.0 J.	
9 56-55-3	Benzo(a)Anthracene	j 400.0 U. j	
10 218-01-9	Chrysene	400.0 U.	
11 205-99-2	Benzo(b)Fluoranthene	400.0 U.]	
12 207-08-9	Benzo(k)Fluoranthene	400.0 U.	
13 50-32-8	Benzo(a)Pyrene	400.0 U. [· ·
14 193-39-5	Indeno(1,2,3-cd)Pyrene	400.0 U.	
15 53-70-3	Dibenz(a,h)Anthracene	400.0 U.	
16 191-24-2	Benzo(g,h,i)Perylene	1 400.0 U. [
	<u></u>		

1 D-T NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRI	X: SOIL SA	MPLE ID:	5-3
	· CONC. LEVE	L: LOW LAB SA	MPLE ID:	1765102
•	EXTRACTION DAT	E: 8/02/93 DIL	FACTOR:	1.00
	ANALYSIS DAT	E: 8/14/93 X H	OISTURE:	19
			UG/I	KG
CMPD #	CAS Number	PESTICIDE/PCB COMPOUND	(DR	Y BASIS)
1	319-84-6	alpha-BHC	Į	NA
2	319-85-7	beta-BHC	i	NA
3	319-86-8	delta-BHC	i	NA
4	58-89-9	gamma-8HC(Lindane)	· i	NA
5	76-44-8	Heptachlor	i	NA
· 6	309-00-2	Aldrin	. 1	NA
7	1024-57-3	Neptachlor Epoxide		NA
8	959-98-8	Endosulfan I	. 1	NA NA
9	60-57-1	Dieldrin		NA NA
10	72-55-9	4.4'-DDE	· •	
11	70-20-8	Endrin		RA MA
12	33213-65-9	Endosulfan II		NA I
13	72-54-8	4.4-000		NA
14	1031-07-8	Endosul fan Sul fare		NA I
. 15	50-29-3	4.41-00T	-	XA
16	72-43-5	Methorychion		NA
17	53494-70-5	Fodrin Ketone	E E	RA
18	7421-36-3	Endrin Aldebude		NA
19	57-74-9	Chlordane		. NA I
20	8001-35-2		·	NA J
21	12674-11-2			NA
22	11104-28-2			100.000 U. J
	11141-16-5		1 1	00.000 U.
26	53640-21-0		1	00.000 U.
25	17677-20-4	AFOCLOP-1242	1	00.000 U. ļ
26	11007-40-4	AFOCIOF-1245	1	00.000 U.
20	11077-07-1	Aroclor-1254	2	00.000 U.
er j	11030-05-2	Aroclor-1260	2	00.000 u. j
			· 1	Í

: :

÷÷

: :

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

	. •	. ŧ		
SAMPLE MATRI)	SOIL		SAMPLE 1D:	\$-4
CONC. LEVEL	LOW		LAB ID:	1765120
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	20.00
ANALYSIS DATE	£: 8/13/93		X MOISTURE:	44
			UG/KG	
CHPD #	CAS Number	PAH CONPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	12000.0 U. J	
2	208-96-8	Acenaphthylene	12000.0 U.	
. 3	83-32-9	Acenaphthene	12000.0 U.	
4	86-73-7	Fluorene	12000.0 U.	
5	85-01-8	Phenanthrene	1700.0 J.	
6	120-12-7	Anthracene	12000.0 U.]	•
7	206-44-0	Fluoranthene	່ 12000.0 ບ. j	
. 8	129-00-0	Pyrene	[1500.0 J. [
9	56-55-3	Benzo(a)Anthracene	12000.0	·
10	218-01-9	Chrysene	12000.0 U.	
11	205-99-2	Benzo(b)fluoranthene	12000.0 U. (
12	207-08-9	Benzo(k)Fluoranthene	12000.0 U.	
13	50-32-8	Benzo(a)Pyrene	12000.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	12000.0 U. [
15	53-70-3	Dibenz(a,h)Anthracene	12000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	12000.0 U.	
	I			

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

1

. ٠

٠.

	•			
SAMPLE MATRIX	soil		SAMPLE ID:	S-4RE
CONC. LEVEL	.: LOW		LAB 1D:	1765120
EXTRACTION DATE	8/2/93		DIL FACTOR:	20.00
ANALYSIS DATE	: 8/16/93		X MOISTURE:	44
			UG/KG	
СМРО #	CAS Number	PAR COMPOUNDS	(DRY BASIS)	
· 1	191-20-3	Naphthalene	12000.0 U.	
2	208-96-8	Acenaphthylene	12000.0 U.	
3	83-32-9	Acenaphthene	12000.0 U.	
4	86-73-7	fluorene	12000.0 U.	
5	85-01-8	Phenanthrene	1500.0 J.	·
6	120-12-7	Anthracene	12000.0 U.	• •
7	206-44-0	Fluoranthene	12000.0 U.	
8	129-00-0	Pyrene	12000.0 U.	
9	56-55-3	Benzo(a)Anthracene	12000.0 U.	
10	218-01-9	Chrysene	12000.0 U.	
· 11	205-99-2	Benzo(b)Fluoranthene	12000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	12000.0 U.	
13	50-32-8	Benzo(a)Pyrene	12000.0 0.	
- 14	193-39-5	Indeno(1,2,3-cd)Pyrene	12000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	12000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	12000.0 U.	
	1			

1 D-T

- NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	SAMPLE ID:	\$-4
	CONC. LEVEL:	LOW	LAB SAMPLE ID:	1765120
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	5.00
	ANALYSIS DATE:	8/14/93	X HOISTURE:	: 44
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPO	LIND	(DRY BASIS)
1	319-84-6	alpha-BHC		KA
2	319-85-7	beta-BKC	· ·	NA
3	319-86-8	delta-BHC	ĺ	HA
4	58-89-9	gamma-BHC(Lindane)		HA
5	76-44-8	Heptachior	·	NA
6	309-00-2	Aldrin		HA
7	1024-57-3	Heptachlor Epoxide		NA
8	959-98-8	Endosulfan I		NA
9	60-57-1	Dieldrin		NA
10	72-55-9	4,41-DDE		AK
- 11	70-20-8	Endrin		NA
12	33213-65-9	Endosulfan II		NA
13	72-54-8	4,4-DDÒ		NA
14	1031-07-B	Endosulfan Sulfate	í l	NA
15	50-29-3	4,4'-DOT	ĺ	NA
16	72-43-5	Methoxychlor	Í	NA
17	53494-70-5	Endrin Ketone	1	NA
18	7421-36-3	Endrin Aldehyde		NA
Ü 19	57-74-9 [Chlordane	ł	NA
20	8001-35-2	Toxaphene	1	NA
21	12674-11-2	Aroclor-1016	ł	710.000 U.
22	11104-28-2	Aroclor-1221	· •	710.000 U.
23	11141-16-5	Aroclor-1232		710.000 U.
24	53469-21-9	Aroclor-1242	· ·	710.000 U.
25	12672-29-6	Aroclor-1248	· · · · ·	710.000 U.
26	11097-69-1	Aroclor-1254		1400.000 U.
27	11096-82-5	Aroclor-1260	l	320.000 J.
	I			

• • .•

i tet i

÷

1B PAH

NYTEST ENVIRONMENTAL INC.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

	• •	• *		
SAMPLE MATRI	X: SOIL		SAMPLE ID:	DUPDL
CONC. LEVE	L: LOW		LAB ID:	1765118
EXTRACTION DATE	E: 8/2/93		DIL FACTOR:	24.00
ANALYSIS DATE	E: 8/18/93		X HOISTURE:	30
	÷		UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	2700.0 DJ 1	
2	208-96-8	AcenaphthyLene	11000.0 U. J	
	83-32-9	Acenaphthene	11000.0 DJ	•
4	86-73-7	Fluorene	8800.0 DJ	
5	85-01-8	Phenanthrene	47000.0 D.	
6	120-12-7	Anthracene	15000.0 D.	
7	206-44-0	Fluoranthene	49000.0 D.	
8	129-00-0	Ругере	49000.0 0.	
9	56-55-3	Benzo(a)Anthracene	1 19000.0 D.	
10	218-01-9	Chrysene	24000.0 D.	
11	205-99-2	Benzo(b)Fluoranthene	12000.0 0.	·
12	207-08-9	Benzo(k)Fluoranthene	13000.0 D.	
13	50-32-8	Benzo(a)Pyrene	15000.0 D.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	11000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	11000.0 0.	
16	191-24-2	Benzo(g,h,i)Perylene	11000.0 U.	
		· · · · · · · · · · · · · · · · · · ·		

0000014

•: •

i . .

.:

. . - -

.

:

.

:

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. •

SAMPLE MATRIX	: SOIL		SAMPLE ID:	DUP
CONC. LEVEL	: LOW		LAB ID:	1765118
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	5.00
ANALYSIS DATE	: 8/17/93		X MOISTURE:	30
			UG/KG	
CHPC #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	·
1	91-20-3	Naphthalene	2500.0	
. 2	208-96-8	Acenaphthylene	2400.0 U.	
·· 3	83-32-9	Acenaphthene	9500.0	
4	86-73-7	Fluorene	6700.0	
. 5	85-01-8	Phenanthrene	18000.0	
6	120-12-7	Anthracene	11000.0	
7	206-44-0	Fluoranthene	21000.0	
8	129-00-0	Pyrene	60000.0 E. (
9	56-55-3	Benzo(a)Anthracene	18000.0	
.10	218-01-9	Chrysene	24000.0 E.	
11	205-99-2	Benzo(b)Fluoranthene	10000.0	
12	207-08-9	Benzo(k)Fluoranthene	12000.0	
. 13	50-32-8	Benzo(a)Pyrene	15000.0	
. 14	193-39-5	Indeno(1,2,3-cd)Pyrene	8800.0	
15	53-70-3	Dibenz(a,h)Anthracene	2400.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	8000.0	
	I	.[

1 D-T NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

SAMPLE WATRIX: WATER CONC. LEVEL: LOW EXTRACTION DATE: 8/02/93 ANALYSIS DATE: 8/14/93

÷

SAMPLE ID:	WASH BLK
LAS SAMPLE ID:	1765121
DIL FACTOR:	1.43
X HOISTURE:NA	

UG/L

CAS Number CHPD #

• ;.

. . .

1.1

.

1 .

ŗ

PESTICIDE/PCB COMPOUND

_		• • • • • • • • • • • • • • • • • • •	····	
1	319-84-6	alpha-BHC		MA NA
2	319-85-7	beta-BKC		L KA
3	319-86-8	delta-BHC		NA I
4	58-89-9	gaama-BHC(Lindane)		NA
	76-44-8	Heptachlor		NA .
· · 6	309-00-2	Aldrin		XA
1 7	1024-57-3	Heptachlor Epoxide		NA NA
. 8	959-98-8	Endosulfan I		NA NA
· 9	60-57-1	Dieldrin ^{s p} jaiù ,		AK I
a 10	72-55-9	4,41-DDE 100-14,4 j		- NA
: 11	70-20-8	Endrin		NA
·· 12	, 33213-65-9	Endosulfan II	4 1	NA
÷′ 13	72-54-8	4.4-DDD		NA -
. 14	1031-07-8	Endosulfan Sulfate	· · · ·	NA NA
²⁰ 15	50-29-3	4.4'-DDT	• • •	NA NA
<u>≷</u> + 16	72-43-5	Methoxychlor		
A* 17	53494-70-5	Endrin Ketone		NA
5- 18	7421-36-3	Endrin Aldehyde		NA
19	57-74-9	Chlordane	0.	NA
20	8001-35-2	Toxaphene		. NA
21	12674-11-2	Aroclor-1016		0.700 u
22	11104-28-2	Aroclor-1221		0.700 U.
÷ 23	11141-16-5	Aroclor-1232		0.700 U.
24	53469-21-9	Aroclor-1242		. 0.700 U.
25	12672-29-6	Aroclor-1248 TOUNDIA		0.700 U.
26	11097-69-1	Aroclor-1254		1.000 U.
	11096-82-5	Aroctor-1260	1	1.000 U.
		······································	l	

18 PAH Nytest Environmental Inc.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

:::

	• •	. 1		
SAMPLE MATRIX	SOIL		SAMPLE ID:	S-!
CONC. LEVEL	: LOW		LAB ID:	1765107
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	10.0
ANALYSIS DATE	: 8/13/93		X MOISTURE:	34
			UG/KG	
CHPO #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	5000.0 U.	
2	208-96-8	Acenaphthylene	5000.0 U.	
3	83-32-9	Acenaphthene	5000.0 U.	
4	86-73-7	Fluorene	5000.0 U.	
5	85-01-8	Phenanthrene	5000.0 U.	
6	120-12-7	Anthracene	5000.0 U.	•
7	206-44-0	Fluoranthene	5000.0 U.	
8	129-00-0	ругене	890.0 J.	
9	56-55-3	Benzo(a)Anthracene	5000.0 U.	
10	218-01-9	Chrysene	4800.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	5000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	5000.0 U.	
['] 13	50-32-8	Benzo(a)Pyrene	5000.0 U.	
14	193-39-5	[Indeno(1,2,3-cd)Pyrene	5000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	5000.0 U.	•
16	191-24-2	Benzo(g,h,i)Perylene	5000.0 U.	
	I	·		

18 PAH Nytest environmental inc.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

.

1

:•• ۰.

3 ÷. ..

-

SAMPLE MATRIX:		, T	SAMPLE ID:	
CONC LEVEL	104	· ·	LAR TO:	1
EVTRACTION DATE-	8/2/01		DII FACTOR-	•
ANALYSTE DATE.	8/17/07		Y WOTCTUDE.	
ARALISIS DATE:	6/11/93		A HUISIUKE:	
CAPO #	CAS NUMBER	PAR CORPOURUS	(DKT BASIS)	
1	91-20-3	Kaphthalene	5000.0 U.	
2	208-96-8	Acenaphthylene	5000.0 U.	
3	83-32-9	Acenaphthene	5000.0 U.	
	86-73-7	Fluorene	5000.0 U.	
5	85-01-8	Phenanthrene	5000.0 U.	
6	120-12-7	Anthracene	5000.0 U.	
7	206-44-0	Fluoranthene	820.0 J.	
. 8	129-00-0	Pyrene	630.0 J.	
9	56-55-3	Benzo(a)Anthracene	5000.0 U.	
10	218-01-9	Chrysene	2800.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	5000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	5000.0 U.	
13	50-32-8	Benzo(a)Pyrene	1700.0 J. 1	
14	193-39-5	Indeng(1.2.3-cd)Pyrene	5000.0 U. I	
15 I	53-70-3	Dibenz(a,h)Anthracene	5000.0 u. l	
1. 1.				

1 D-T NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

SAMPLE ID:

DIL FACTOR:

LAB SAMPLE ID:

S-5

5.00

1765107

SAMPLE MATRIX: SOIL CONC. LEVEL: LOW EXTRACTION DATE: 8/02/93 ANALYSIS DATE: 8/25/93

CMPD #

1 | 2 |

-1

3

5

CAS Number

319-84-6

319-85-7

319-86-8

58-89-9 76-44-8

309-00-2

ł

: 8/25/93	% MOISTURE:		34
PESTICIDE/PCB COM	2011 ועח	UG/X	G .
		CDRY	BASIS)
elpha-BHC			
beta-BHC		1	KA
delta-BHC		1	NA
gamma-BHC(Lindane)			KA
Heptachlor -		-	NA
Aldrin	11 A. A. A.		NA
Neptachlor Epoxide	in and the	·. I	XA
Endosul fan t	Bernard and a	ļ	HA
Dieldrin	1966 (197	1	NA
	/ 19 <u>.</u>	1	NA
*,*** ₩₽E	and the second	1	NA
Endrin	ntroats (

1024-57-3	Heptachion Securida	, NA
959-98-8	Fordered from the second secon	NA NA
60-57-1	I statist	l na
72-55-0		NA NA
70-70 0	4,4'-DDE 3.00 A.	l NA
1 7720-8	Endrin ormania (
33213-65-9	Endosulfan II Compto Star	1 KA
72-54-8	4,4-DOD 500	I NA
1031-07-8	Endosulfan Sulfate	I NA in
50-29-3	4.4'-DDT	NA I
72-43-5	Hethorychion	NA
53494-70-5	Fodrin Ketore	I NA I
7421-36-3		1 - NA 🖓
57-74-9	Chican Aldenyde	1 84
8001-35-2		NA -
1267/-11-5	Toxaphene	I NA I
1110/ 20 5	Aroclor-1016	610,000,0
11104-28-2	Aroclor-1221	410.000 U. J
11141-16-5	Aroclor-1232	
53469-21-9	Aroclor-1242	610.000 U.
12672-29-6	Aroclor-1248	610.000 U_
11097-69-1	Aroclor-1254	1 1900.000
11096-82-5	Arector-1260	1200.000 U. 1
		1 1200.000 U. j

-
94: 15:12 FROM NYS. ENVIR. CONSERVATION

та

P.08

NEW YORK STATE DEPARTMENT OF ENVIRONMENTAL CONSERVATION

MOBILE LABORATORY VOLATILE ANALYSIS

SITE NAME: ACCURATE DIE CASTING

FIELD ID:S-5

SITE CODE: 734052 PERCENT SOLIDS: 69.0 SAMPLE NUMBER: 793-211-01 MATRIX: SOIL

SUBMISSION DATE: 07/30/93 ARCHIVE NO.: V21101

ANALYSIS DATE: 08/03/93 DATA FILE NO.: 9304C24A.D

	CONC (PPB)	NON TARGET COMPOUNDS:
Chloromethane	ND 1	· · · · · · · · · · · · · · · · · · ·
Bromomethane	ND	I
Vinyl chloride	ND	1 .
Chloroethane	ND 1	t i i i i i i i i i i i i i i i i i i i
Methylenc chloride	52B . 1	1
Acstone	ND	1
Carbon disulfide	ND	i
1,1-Dichloroethene	ND	1
1,1-Dichloroethane	ND	1
trans-1,2-Dichloroethene	ND	1
Chloreform	ND I	1
1,2-Dichloroethane	NÐ I	1
2-Butanone	ND I	1
1,1,1-Trichloroethane	ND	1
Carbontstrachloride	ND	
Vinyl acetate	ND I	1
Bromedichloromethane	ND I	1
1,1,2,2-Tetrachlorosthame	ND I	1
1,2-Dichloropropane	ND	1
trans-1,3-Dichloropropene	ND I	l · · · · ·
Trichloroethens	18B I	
Dibromachloromethane	ND I	l .
1,1,2-Trichloroethane	ND	I
benzene	ND I	
cis-1,3-Dichloropropene	ND I	
2-Chloroethylvinylether	ND I	۲. ۱
bromoform D. Hannak	ND	8
A-MEXADONG	ND	6
4-Hethyl-2-pentanone	ND I	. · ·
Tetrachloroethene	ND I	1
FOLDERE Children	ND I	
	ND I	ND = LESS THAN 5 PPB
CinyiDenzene		
27.97.978 Taka 1. Mataka	ND I	ALL CONCENTRATIONS LESS THAN
Total Aylenes	ND I	> PPB ARE ESTIMATES
-0101 UNIOPO(0)0000 Tatal O)=51-5-5	NU I	1 、
Fatal utonjorobenzene	ND I	1

. :

- ---

•

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

SAMPLE MATRI)	soil			SAMPLE ID:	S-6
CONC. LEVEL	.: LOW			LAB 10:	1765109
EXTRACTION DATE	8/2/93			DIL FACTOR:	20.00
ANALYSIS DATE	8/14/93			X HOISTURE:	31
			UG/	KG	
CHPD #	CAS Number	PAH CONPOUNDS	(DR	Y BASIS)	
· 1	91-20-3	Naphthalene	1	10000.0 U.	
2	208-96-8	Acenaphthylene	i	10000.0 0.	
3	83-32-9	Acenaphthene	i	10000.0 U. j	
4	86-73-7	Fluorene	i	10000.0 U.	
5	85-01-8	Phenanthrene	∵i –	3800.0 J.	
6	120-12-7	Anthracene	1	1100.0 J.	
7	206-44-0	Fluoranthene	i	3200.0 J.	
8	129-00-0	Pyrene	i	3500.0 J.	
9	56-55-3	Benzo(a)Anthracene	, j	10000.0 U. j	
10	218-01-9	Chrysene and the	: 1	10000.0 U.	
. 11	205-99-2	Benzo(b)Fluoranthene	ej I	10000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene		10000.0 U. j	
13	50-32-8	Benzo(a)Pyrene 10. 25	1	10000.0 U. I	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	si –	10000.0 U.	•
15	53-70-3	Dibenz(a,h)Anthracene	зİ	10000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	:	10000.0 U.	
	\	.1 <u></u>	<u></u>	l	-

. .

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

SAMPLE MATRIX	soft		SAMPLE ID:	S-óRE
CONC. LEVEL	.: Low		LAB ID:	1765109
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	20.00
ANALYSIS DATE	: 8/18/93		X HOISTURE:	31
			UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Nephthalene	1 10000.0 U.]	
2	208-96-8	Acenaphthylene	10000.0 U.	
3	83-32-9	Acenaphthene	10000.0 U.	
4	86-73-7	Fluorene	10000.0 U.	•
5	85-01-8	Phenanthrene /s ==	3600.0 J.	
6	120-12-7	Anthracene	1200.0 J.	
7	206-44-0	Fluoranthene	3000.0 J.	
: 8	129-00-0	Pyrene	2900.0 J	
9	56-55-3	Benzo(a)Anthracene	2100.0 J.	
. 10	218-01-9	Chrysene	5100.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	10000.0 U.	
12	207-08-9	Benzo(k)fluoranthene	10000.0 U.	
13	50-32-8	Benzo(a)Pyrene	10000.0 U.	
- 14	193-39-5	Indeno(1,2,3-cd)Pyrene	10000.0 U.	
. 15	53-70-3	Dibenz(a,h)Anthracene	10000.0 U.	
. 16	191-24-2	Benzo(g,h,i)Perylene	10000.0 U	÷
<u>t</u>	Í		·]	

•

.

:::.

1 D-T

NYTEST ENVIRONMENTAL INC.

¦.

. .

÷.,

1999 A.

い

:.

.

.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SANPLE MATRIX:	SOIL	SAMPLE ID:	s-6
	CONC. LEVEL:	LOV	LAB SAMPLE ID:	1765109
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	5.00
	ANALYSIS DATE:	8/25/93	# HOISTURE:	31
		-,		UG/KG
CMPD #	CAS Number	PESTICIDE/PCB CONPO	UKO	(DRY BASIS)
, - 1	319-84-6	slpha-BHC	 !	HA I
2	319-85-7	beta-BHC		. KA
3	319-86-8	delta-BHC	i	NA I
4	58-89-9	gamma-BHC(Lindane)	i	NA
. 5	76-44-8	Heptachlor		XA S
6	309-00-2	Aldrin	i	, AA
7	1024-57-3	Heptachlor Epoxide	÷ .	NA I
8	959-98-8	Endosulfan 1		XA J
.9	60-57-1	Dieldrin	· · ·	KA
10	72-55-9	4,4'-DDE		KA
11	70-20-8	Endrin	- : 1	- XA
12	33213-65-9	Endosulfan II	·	NA I
13	72-54-8	4,4-000		, NA
14	1031-07-8	Endosulfan Sulfare	· · · j	NA I
15	50-29-3	4,41-DDT		AK
16	72-43-5	Methoxychlor	· · ·	NA
17	53494-70-5	Endrin Ketone		NA
18	7421-36-3	Endrin Aldehyde	1	NA .
19	57-74-9	Chlordane '		HA [
20	8001-35-Z	Toxaphene	1	XX
21	12674-11-2	Aroclor-1016	. 1	580.000 U.
22	11104-28-2	Aroclor-1221	• 1	580.000 U. }
23	11141-16-5	Aroclor-1232		580.000 U.
24	53469-21-9	Aroclor-1242	1	580.000 U.
25	12672-29-6	Aroclor-1248	1	580.000 U.
26	11097-69-1	Aroclor-1254	· · · · · ·	700.000 J.
27	11096-82-5	Aroclor-1260	. 1	1200.000 U.]

•••

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

SAMPLE MATRIX	soil		SAMPLE ID:	5-8
CONC. LEVEL	: LOW		LAB ID:	1765103
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	2.00
ANALYSIS DATE	: 8/14/93		X HOISTURE:	17
			UG/KG	•
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	1000.0 U.	
2	208-96-8	Acenaphthylene	1000.0 U.	
- 3	83-32-9	Acenaphthene	1000.0 U.	
-4	86-73-7	Fluorene	1000_0 U.	
5	85-01-8	Phenanthrene	270.0 J.	
6	120-12-7	Anthracene	1000.0 U.	
7	206-44-0	Fluoranthene	260.0 J.	
8	129-00-0	Ругеле	680.0 J.	
9	56-55-3	Benzo(a)Anthracene	1000.0 U.	
10	218-01-9	Chrysene	1000.0 U.	
, 11	205-99-2	Benzo(b)Fluoranthene	1000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	1000.0 U.	
13	50-32-8	Benzo(a)Pyrene	1000.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	1000.0 0.	
15	53-70-3	Dibenz(a,h)Anthracene	1000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	1000.0 U.	
	i	- <u> </u>	!	

 \sim

18 PAH

NYTEST ENVIRONMENTAL INC.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

CANDIE MATDIN			5401 E 10.	e-905
SAMPLE MAIRIA	SUIL		SAUPLE ID:	3-OKE
CONC. LEVEL	LOW		LAB ID:	1765103
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	2.00
ANALYSIS DATE	: 8/18/93		X MOISTURE:	17
			UG/KG	
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	1000.0 U.	
2	208-96-8	Acenaphthylene	, 1000.0 U.	
5	83-32-9	Acenaphthene	1000.0 U.	
4	86-73-7	Fluorene	1000.0 U.	
5	85-01-8	Phenanthrene	260.0 J.	
6	120-12-7	Anthracene	1000.0 U.	
7	206-44-0	Fluoranthene	260.0 J.	
8	129-00-0	Ругеле	600.0 J.	
9	56-55-3	Benzo(a)Anthracene	100.0 J.	
10	218-01-9	Chrysene	160.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	1000.0 U	
12	207-08-9	Benzo(k)Fluoranthene	1000.0 U.	
13	50-32-8	Benzo(a)Pyrene	1000.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	1000.0 U	
15	53-70-3	Dibenz(a,h)Anthracene	1000.0 U.	
16	191-24-2	Benza(g,h,i)Perylene	1000.0 U.	
	!	.	!	

÷.

1 D-T

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

NYTEST ENVIRONMENTAL INC.

	SAMPLE MATRIX:	SOL	SAMPLE ID:	s-8
	CONC. LEVEL:	LOW	LAB SAMPLE ID:	1765103
	EXTRACTION DATE:	8/02/93	DIL FACTOR	1.00
	ANALYSIS DATE:	8/14/93	% HOISTURE	: 17
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPO	UND	(DRY BASIS)
1	319-84-6	elpha-BHC		KA
. 2	319-85-7	beta-BHC		NA
3	319-86-8	delta-BHC		NA 1
4	58-89-9	gamma-BHC(Lindane)	¢ 14	NA
5	76-44-8	Heptachlor		NA NA
6	309-00-2	Aldrin	- 34 č	NA
7	1024-57-3	Heptachior Epoxide		NA `
8	959-98-8	Endosulfan I	à cocar	NA
9	60-57-1	Dieldrin	•	NA
. 10	72-55-9	4,4'-DDE	. ·	, NA
11	70-20-8	Endrin	• :•	, NA
12	33213-65-9	Endosulfan II		NA
13	72-54-8	4,4-000	· · ·	NA
14	1031-07-8	Endosulfan Sulfate	•••••	NA
15	50-29-3	4,4*-DDT	1	NA
16	72-43-5	Methoxychlor	the start for	XA
17	53494-70-5	Endrin Ketone		NA
18	7421-36-3	Endrin Aldehyde	د استر ۲۱ م	NA
19	57-74-9	Chlordane	M 80 (2)	· NA .
20	8001-35-2	Toxaphene		NA
21	12674-11-2	Aroclor-1016		100.000 U.
22	11104-28-2	Aroclor-1221	•	100.000 U.
23	11141-16-5	Aroclor-1232	•	100.000 U.
24	53469-21-9	Aroclor-1242	,	100.000 U.
25	12672-29-6	Aroclor-1248	1 1 1 1 1 1	100.000 U.
26	11097-69-1	Aroclor-1254	··· ··	190.000 U.
27	11096-82-5	Aroclor-1260		190.000 U.

18 PAN Nytest environmental INC.

. . .

÷. -

ل

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

SAMPLE MATRIX	: són		SAMPLE ID:	S- 9
CONC. LEVEL	: LOW		LAB 1D:	1765106
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	4.00
ANALYSIS DATE	: 8/14/93		X NOISTURE:	42
			UG/KG	
CKP0 #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	2000.0 U.	
2	208-96-8	Acenaphthylene	2000.0 U.	
3	83-32-9	Acenaphthene	2000.0 U.	
4	86-73-7	Fluorene	2000.0 U.	
5	85-01-8	Phenanthrene	780.0 J.	
6	120-12-7	Anthracene	280.0 J.	• .
7	206-44-0	Fluoranthene	660.0 J.	
8	129-00-0	Pyrene	780.0 J. -	•
9	56-55-3	Benzo(a)Anthracene	2000.0 U.	
10	1218-01-9	Chrysene	800.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	2000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	2000.0 U.	
13	150-32-8	Benzo(a)Pyrene	2000.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	2000.0 U.	
15	53-70-3	0 (benz(a,h)Anthracene	2000.0 U.	•
16	191-24-2	Benzo(g,h,i)Perylene	2000.0 U.	
		I	.]	

18 PAH

NYTEST ENVIRONMENTAL INC.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

CANDIE MATORY					SAMPLE ID.	e.005
						47/510/
CUNC. LEVEL					LAB IU:	1/05100
EXTRACTION DATE	8/2/93			1	DIL FACTOR:	4.00
AHALYSIS DATE	E: 8/18/93				X MOISTURE:	42
				UG/KG		
Chipd #	CAS Number	PAH COMPOUNDS	-	(DRY BA	SIS)	
1	91-20-3	Naphthalene			2000.0 U.	
2	208-96-8	Acenaphthylene		1 :	2000.0 U. j	•
3	83-32-9	Acenaphthene		i	2000.0 U. (
. 4	86-73-7	Fluorene	,	į :	2000.0 0.]	
. 5	85-01-8	Phenanthrene		İ	820.0 J.	•
- 6	120-12-7	Anthracene	. * '	İ	790.0 J. j	· .
. 7	206-44-0	Fluoranthene		i	680.0 J.	
. 8	129-00-0	Ругепе	4	1	760.0 J.	•
9	56-55-3	Benzo(a)Anthracene	۰.	i	740.0 J.	
10	218-01-9	Chrysene		1	1200.0 J.	
11	205-99-2	Benzo(b)Fluoranthene		i i	2000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	• • •	į ·	270.0 J. [
13	50-32-8	Benzo(a)Pyrene	۰.	i :	2000.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrer	10	i :	2000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	•	j 7 3	2000.0 U.	
. 16	191-24-2	Benzo(g,h,i)Perylene	۰ ۲ بر	į :	2000.0 U.	
	l	·		.]	<u> </u>	

.

1 D-T

NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

SAMPLE ID:

s-9

	CONC. LEVEL:	LOW L	AB SAMPLE ID:	1765106
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	5.00
	ANALYSIS DATE:	8/25/93	% MOISTURE:	42
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPOUN	D	(DRY BASIS)
1	319-84-6	alpha-BHC	 .]	KA
Z	319-85-7	beta-BHC		NA
3	319-86-8	delta-BXC		NA
4	58-89-9	gamma-BHC(Lindane)		NA
5	76-44-8	Heptachlor	5. ge	. NA
6	309-00-2	Aldrin	1.1	XA
7	1024-57-3	Heptachior Epoxide		· KA
8	959-98-8	Endosulfan I - Nover		KA
9	60-57-1	Dieldrin	i	NA
10	72-55-9	4,41-DDE		NA
11	70-20-8	Endrin .:	ا	KA -
12	33213-65-9	Endosulfan II - Patry	1.17 1	NA ·
13	72-54-8	4,4-DDD	ا دىغان	NA
14	1031-07-8	Endosulfan Sulfate		NA
15	50-29-3	4,41-00T The	•*•••	NA ¹
16	72-43-5	Nethoxychior older	en sa j	- NA
17	53494-70-5	Endrin Ketone	E ISIA	NA
18	7421-36-3	Endrin Aldehyde 238 -	Paras i	KA
19	57-74-9	Chlordane aven	:-D [NA 11
20	8001-35-2	Toxaphene 2000	or, i	NA
21	12674-11-2	Aroclor-1016	I	690.000 U.
22	11104-28-2	Aroclor-1221	j	690.000 U.
23	11141-16-5	Aroclar-1232	i	690.000 U.
24	53469-21-9	Aroclar-1242	i	690.000 U.
25	12672-29-6	Aroclar-1248	i	690.000 U.
26	11097-69-1	Aroclor-1254		1400.000 U.
27	11096-82-5	Aroclor-1260	4	370.000 J.

SAMPLE MATRIX: SOIL

დ⊴ღე

4

.

1

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

	· ·		
SAMPLE MATRIX:	WATER	SAMPLE ID:	WASH BLK
CONC. LEVEL:	LOW	LAB ID:	1765121
EXTRACTION DATE:	8/2/93	DIL FACTOR:	1.00
ANALYSIS DATE:	8/14/93	% MOISTURE:	NA
		UG/L	

CMPD a	#	CAS	Number
--------	---	-----	--------

PAH COMPOUNDS

1	91-20-3	Naphthalene	10.0 U.
2	208-96-8	Acenaphthylene	10.0 U.
3	83-32-9	Acenaphthene	10.0 U.
4	86-73-7	Fluorene	10.0 U.
5	85-01-8	Phenanthrene	10.0 U.
6	120-12-7	Anthracene	10.0 U.
7	206-44-0	Fluoranthene	ຳ 10.0 ບ.
8	129-00-0	Pyrene	10.0 U.
9	56-55-3	Benzo(a)Anthracene	10.0 U.
D	218-01-9	Chrysene) 10.0 U.
1	205-99-2	[Senzo(b)Fluoranthene	l 10.0 U.
2	207-08-9	Benzo(k)Fluoranthene	10.0 U.
3	50-32-8	Benzo(a)Pyrene) 10.0 U.
4	193-39-5	Indeno(1,2,3-cd)Pyrene	10.0 U.
5	53-70-3	Dibenz(a,h)Anthracene	10.0 U.
6	191-24-2	Benzo(g,h,i)Perylene	10.0 U.
	l	_!	

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

÷

SAMPLE MATRIX	soil		SAMPLE ID:	S-10
CONC. LEVEL	: LOW		LAB 1D:	1765110
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	: 8/17/93		X MOISTURE:	22
			UG/KG	
CKPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	13000.0 U.	,
. 2	208-96-8	Acenaphthylene	13000.0 U.	
. 3	83-32-9	Acenaphthene	13000.0 U.	
- 4	86-73-7	Fluorene	13000.0 U.	
. 5	85-01-8	Phenanthrene	5900.0 J.	
6	120-12-7	Anthracene	4000.0 J.	
7	206-44-0	Fluoranthene	7300.0 J.	
8	129-00-0]Pyrene	6700.0 J.	
9	56-55-3	Benzo(a)Anthracene	7100.0 J.	
10	218-01-9	Chrysene	10000.0 J.	,
11	205-99-2	Benzo(b)Fluoranthene	6100.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	5400.0 J.	
13	j50-32-8	Benzo(a)Pyrene	6000.0 J.	
14	193-39-5	[Indeno(1,2,3-cd)Pyrene	13000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	13000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	13000.0 U.	•
		_ <u> ·</u>	[]	

0000017

.

•

....

. . .

.

Ξ.

. . .

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

.

. ÷

SAMPLE MATRIX	soil		SAMPLE ID:	S-10RE
CONC. LEVEL	.: LOW		LAB ID:	1765110
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	8/18/93		X HOISTURE:	22
-			UG/KG	
CHP0 #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	13000.0 U.	
2	208-96-8	Acenaphthylene	13000.0 U.	
. ** 3	83-32-9	Acenaphthene	13000.0 U. j	
4	86-73-7	Fluorene	13000.0 U.	
5	85-01-8	Phenanthrene	6100.0 J.	
6	120-12-7	Anthracene	4800.0 J.	
7	206-44-0	Fluoranthene	7300.0 J.	
8	129-00-0	Pyrene	7300.0 J.	
9	56-55-3	Benzo(a)Anthracene	· 4400.0 J.	
10	218-01-9	Chrysene	14000.0	
. 11	205-99-2	Benzo(b)Fluoranthene	6400.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	4400.0 J. J	
13	50-32-8	Benzo(a)Pyrene	6200.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	13000.0 U. j	
15	53+70-3	Dibenz(a,h)Anthracene	13000.0 U.	
. 16	191-24-2	Benzo(g,h,i)Perylene	13000.0 U.	
	I	, <u></u>		

1 D-T NYTEST ENVIRONMENTAL INC.

•

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	SAMP	LE ID:	S-10
	CONC. LEVEL:	LON	LAB SAMP	LE ID:	1765110
	EXTRACTION DATE:	8/02/93	DIL F	ACTOR:	5.00
	ANALYSIS DATE:	8/25/93	· 2 HOI	STURE:	22
					UG/KG
CHPD #	CAS Number	PESTICIDE/PCB	COMPOUND		(DRY BASIS)
1	319-84-6	alpha-SHC	·		NA
2	319-85-7	beta-BXC		ĺ	• NA - 1
3	319-86-8	delta-BHC		Í	NA
4	58-89-9	gamma-BHC(Lin	dane)	j	NA NA
5	76-44-8	Heptachlor	1.47225	İ	T E NA
6	309-00-2	Aldrin	amula	ĺ	A A A A A A A A A A A A A A A A A A A
7	1024-57-3	Heptachlor Ep	oxide. ,	· j	NA
8	959-98-8	Endosulfan I	الأحسر الم	1	
9	60-57-1	Dieldrin	-	İ	NA I
10	72-55-9	4,4'-DDE		Í	NA
11	70-20-8	Endrin	at 1243	i	, NA
12	33213-65-9	Endosulfan II	31. 2000	· · ·]	NA"
13	72-54-8	4,4-000	્ય્યન્ક્યને	i	NA -
14	1031-07-8	Endosulfan Su	lfate the	i	• • • •
15	50-29-3	4,4+-DDT		ì	KA
16	72-43-5	Hethoxychior		. j	· • • • • • • • • • • • • • • • • • • •
17	53494-70-5	Endrin Ketone	194 /6 (PB)	1.0	NA ANA
18	7421-36-3	Endrin Aldehyd	le : unitation en el constante		i i na
19	57-74-9	Chlordane	-nerrelation	. j	
. 20	8001-35-2	Toxaphene	an transfer	i i	NA
21	12674-11-2	Aroclor-1016	oltană E	- : İ	
22	11104-28-2	Aroclor-1221	1	ं रहत	510.000 U.
23	11141-16-5	Aroclor-1232	t, att.	-	510.000 U.
24	53469-21-9	Aroclor-1242	12.00	Ì	510.000 U.
25	12672-29-6	Aroclor-1248	APTA	· · İ	2600.000
26	11097-69-1	Aroclor-1254		Í	1000.000 U.
27	11096-82-5	Aroclar-1260		· i	1000.000 U.
				i	

.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. -

.

SAMPLE ID: LAB ID: 17 DIL FACTOR: X MOISTURE: UG/KG (DRY BASIS) 2000.0 U. 2000.0 U.		SOIL	SAMPLE MATRIX
LAB ID: 17 DIL FACTOR: X WOISTURE: UG/KG (DRY BASIS) 2000.0 U. 2000.0 U.			OF THE STATE
DIL FACTOR: % MOISTURE: UG/KG (DRY BASIS) 2000.0 U. 2000.0 U.		.: LOW	CONC. LEVEL
X WOISTURE: UG/KG (DRY BASIS) 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U.		8/2/93	EXTRACTION DATE
UG/KG (DRY BASIS) 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U.		8/17/93	ANALYSIS DATE
(DRY BASIS) 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U.			
2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 J. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 J.	PAH COMPOUNDS	CAS Number	CHPD #
2000.0 U. 2000.0 U. 2000.0 U. 2770.0 J. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 J.	Naphthalene	 91-20-3	1
2000.0 U. 2000.0 U. 270.0 J. 2000.0 U. 2000.0 U. 2000.0 U. 2000.0 J.	Acenaphthylene	208-96-8	. 2
2000.0 U. 270.0 J. 2000.0 U. 2000.0 U. 2000.0 U. 220.0 J.	Acenaphthene	83-32-9	3
Z70.0 J. 2000.0 U. 2000.0 U. 220.0 J.	Fluorene	86-73-7	4
2000.0 U. 2000.0 U. 220.0 J.	Phenanthrene	85-01-8	5
2000.0 U. (220.0 J. (Anthracene	120-12-7	6
220.0 J.	Fluoranthene	206-44-0	7
	Pyrene	129-00-0	8
[2000.0 U.]	Benzo(a)Anthracene	56+55-3	9
2000.0 U.	Chrysene	218-01-9	10
2000.0 U.	Benzo(b)Fluoranthene	205-99-2	11
2000.0 U.	Benzo(k)Fluoranthene	207-08-9	12
2000.0 U.	Benzo(a)Pyrene	50-32-8	13
2000.0 U.	Indeno(1,2,3-cd)Pyrene	193-39-5	. 14
2000.0 U.]	Dibenz(a,h)Anthracene	53-70-3	15
2000.0 U.	Benzo(g,h,i)Perylene	191-24-2	16

18 PAH

NYTEST ENVIRONMENTAL INC.

....

42

:

ţ

. .

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

.

.

	•		
SAMPLE MATRIX	X: 501L		SAMPLE ID:
CONC. LEVE	L: LOW		LAB ID:
EXTRACTION DATE	E: 8/2/93		DIL FACTOR:
ANALYSIS DAT	E: 8/18/93		% HOISTURE:
			UG/KG
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)
1	91-20-3	Naphthalene	2000.0 U. j
. 2	208-96-8	Acenaphthylene	2000.0 U.
	83-32-9	Acenaphthene	2000.0 0.
4	86-73-7	Fluorene	2000.0 U. (
5	85-01-8	Phenanthrene	260.0 J. j
6	120-12-7	Anthracene	2000.0 U.
7	1206-44-0	Fluoranthene	2000.0 U. I
8	129-00-0	Pyrene	2000.0 U. 1
9	56-55-3	Benzo(a)Anthracene	2000.0 0.1
10	218-01-9	Chrysene	2000.0 U. I
11	205-99-2	Benzo(b)Fluoranthene	2000.0 U. I
12	1207-08-9	Benzo(k)Fluoranthene	2000.0 U. I
. 13	150-32-8	Benzo(a)Pyrene	2000.0 U. 1
14	193-39-5	Indena(1.2.3-cd)Pyrene	2000.0 U. L
15	53-70-3	Dibenz(a,h)Anthracene	2000.0 U. 1
	1101 1/ 3	IRenanda h iNRendere	

1 D-T NYTEST ENVIRONMENTAL INC.

•

5

:

•...*

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

CONC. LEVEL: LOW LAB SAMPLE ID: 1765119 EXTRACTION DATE: 8/02/93 DIL FACTOR: 5.00 ANALYSIS DATE: 8/25/93 X MOISTURE: 28 UG/KG UG/KG UG/KG CMPD # CAS Number PESTICIDE/PCB COMPOUND (DRY BASIS) 1 319-86-6 alpha-8HC NA 2 319-85-7 beta-8HC NA 3 319-86-8 delta-8HC NA 4 58-89-9 gamma-8HC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endosulfan II NA 12 33213-65-9 Endosulfan Sulfate NA 13 72-54-8 4,4'-DDT NA 14 1031-07-8 Endosulfan Sulfate <th></th> <th>SAMPLE MATRIX:</th> <th>SOIL</th> <th>SAMPLE ID:</th> <th>s-11</th>		SAMPLE MATRIX:	SOIL	SAMPLE ID:	s-11
EXTRACTION DATE: 8/02/93 DIL FACTOR: 5.00 ANALYSIS DATE: 8/25/93 X MOISTURE: 28 UG/KG UG/KG UG/KG CHPD # CAS Number PESTICIDE/PCB COMPOUND (DRY BASIS) 1 319-86-6 alpha-BHC NA 2 319-86-8 delta-BHC NA 3 319-86-8 delta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4*-DDE NA 11 70-20-8 Endosulfan Sulfate NA 12 33213-65-9 Endosulfan Sulfate NA 13 72-54-8 4,4*-DDT NA 14 1031-07-8 Endrin Katone <td></td> <td>CONC. LEVEL:</td> <td>LOW</td> <td>AB SAMPLE ID:</td> <td>1765119</td>		CONC. LEVEL:	LOW	AB SAMPLE ID:	1765119
AHALYSIS DATE: 8/25/93 X MOISTURE: 28 UG/KG UG/KG UG/KG UG/KG 1 319-84-6 alpha-BHC NA 2 319-85-7 beta-BHC NA 3 319-86-8 delta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 10 72-55-9 4,4*-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4*-DDT NA 16 72-43-5 Methoxychlor NA 17 53694-70-5 Endrin Aldehyde NA		EXTRACTION DATE:	8/02/93	DIL FACTOR:	5.00
UG/KG UG/KG 1 319-84-6 alpha-BHC NA 2 319-85-7 beta-BHC NA 3 319-85-7 beta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4*-0DE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-0DD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-0DT NA NA 16 72-43-5 Methoxychlor NA NA 18 7621-43-3 Endrin Aldehyde NA		ANALYSIS DATE:	8/25/93	% HOISTURE:	: 28
CHPD # CAS Number PESTICIDE/PCB COMPOUND (DRY BASIS) 1 319-84-6 alpha-BHC NA 2 319-85-7 beta-BHC NA 3 319-86-8 delta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4*-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4*-DDT NA 14 1031-07-8 Endrin Aldehyde NA 15 50-29-3 4,4*-DDT NA 16 72-43-5 Methoxychlor NA					UG/KG
1 319-84-6 alpha-BHC NA 2 319-85-7 beta-BHC NA 3 319-85-7 beta-BHC NA 3 319-85-7 beta-BHC NA 3 319-85-7 beta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4*-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4*-DDE NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4*-DDT NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Aldehyde NA 18 7421	CMPD #	CAS Number	PESTICIDE/PCB COMPOUN	D	(DRY BASIS)
2 319-85-7 beta-BHC NA 3 319-86-8 delta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Endrin Aldehyde NA 17 53494-70-5 Endrin Aldehyde NA 18 7421-36-3 Endrin Aldehyde NA	1	319-84-6	alcha-BKC		NA
3 319-86-8 delta-BHC NA 4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endosulfan II NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4'-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Findrin Aldehyde NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Arccior-1016 560.000 U	2	319-85-7	beta-BXC		NA NA
4 58-89-9 gamma-BHC(Lindane) NA 5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endosulfan II NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Endrin Aldehyde NA 17 53494-70-5 Endrin Aldehyde NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA <tr< td=""><td>3</td><td>319-86-8</td><td>delta-BHC</td><td></td><td>NA NA</td></tr<>	3	319-86-8	delta-BHC		NA NA
5 76-44-8 Heptachlor NA 6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4'-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Endrin Ketone NA 17 53494-70-5 Endrin Aldehyde NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1232 560.000 U 22 11104-28-2 Aroclor-1232 560.000 U <	- 4	58-89-9	camma-BHC(Lindane)		NA '
6 309-00-2 Aldrin NA 7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Aldehyde NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1232 560.000 U 23 1114-16-5 Aroclor-1232 560.000 U 24 53469-21-9 Aroclor-1248 560.000 U <	5	76-44-8	Heptachlor	i	NA
7 1024-57-3 Heptachlor Epoxide NA 8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4*-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4*-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4*-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Endrin Aldehyde NA 17 53494-70-5 Endrin Aldehyde NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1232 560.000 U. 22 11104-28-2 Aroclor-1232 560.000 U. 23 11141-16-5 Aroclor-1248 560.000 U. 24 53469-21-9 Aroclor-1254	6	309-00-2	Aldrin		NA
8 959-98-8 Endosulfan I NA 9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Hethoxychlor NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-121 560.000 U. 22 11104-28-2 Aroclor-1232 560.000 U. 23 11141-16-5 Aroclor-1248 560.000 U. 24 53469-21-9 Aroclor-1248 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1260	7	1024-57-3	Heptachlor Epoxide		NA -
9 60-57-1 Dieldrin NA 10 72-55-9 4,4'-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Hethoxychlor NA 16 72-43-5 Hethoxychlor NA 18 7421-36-3 Endrin Xetone NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-121 560.000 U. 22 11104-28-2 Aroclor-1232 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1	8	959-98-8	Endosulfan I		NA
10 72-55-9 4,4'-DDE NA 11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 16 72-43-5 Methoxychlor NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Arocior-1016 560.000 U. 22 11104-28-2 Arocior-1232 560.000 U. 23 11141-16-5 Arocior-1242 560.000 U. 24 53469-21-9 Arocior-1248 560.000 U. 25 12672-29-6 Arocior-1254 1100.000 U. 26 11097-69-1 Arocior-1260 240.000	9	60-57-1	Dieldrin		NA
11 70-20-8 Endrin NA 12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-0DD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-0DT NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1232 560.000 U. 23 11141-16-5 Aroclor-1242 560.000 U. 24 53469-21-9 Aroclor-1248 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1250 240.000	10	72-55-9	4,4'-DDE	. 1	NA
12 33213-65-9 Endosulfan II NA 13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1232 560.000 U. 23 11141-16-5 Aroclor-1242 560.000 U. 24 53469-21-9 Aroclor-1248 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 240.000	· 11	70-20-8	Endrin	1	NA
13 72-54-8 4,4-DDD NA 14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1232 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 240.000	12	33213-65-9	Endosulfan II	1	NA
14 1031-07-8 Endosulfan Sulfate NA 15 50-29-3 4,4'-0DT NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Arocior-1016 \$60.000 U. 22 11104-28-2 Arocior-1221 \$60.000 U. 23 11141-16-5 Arocior-1232 \$60.000 U. 24 53469-21-9 Arocior-1242 \$60.000 U. 25 12672-29-6 Arocior-1254 1100.000 U. 26 11097-69-1 Arocior-1260 240.000	13	72-54-8	4,4-DDD	、 I	KA
15 50-29-3 4,4'-DDT NA 16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1221 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 240.000	14	1031-07-8	Endosulfan Sulfate]	- NA
16 72-43-5 Methoxychlor NA 17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1221 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 240.000	15	50-29-3	4,4'-DDT		. NA
17 53494-70-5 Endrin Ketone NA 18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1231 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 240.000	16	72-43-5	Methoxychlor	1	NA
18 7421-36-3 Endrin Aldehyde NA 19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1221 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 1100.000 U. 26 11097-69-1 Aroclor-1254 240.000	17	53494-70-5	Endrin Ketone		, NA
19 57-74-9 Chlordane NA 20 8001-35-2 Toxaphene NA 21 12674-11-2 Arocior-1016 560.000 U. 22 11104-28-2 Arocior-1221 560.000 U. 23 11141-16-5 Arocior-1232 560.000 U. 24 53469-21-9 Arocior-1242 560.000 U. 25 12672-29-6 Arocior-1254 1100.000 U. 26 11097-69-1 Arocior-1260 240.000	18	7421-36-3	Endrin Aldehyde		NA
20 8001-35-2 Toxaphene NA 21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1221 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 560.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1260 240.000	19	57-74-9	Chlordane	1	· NA
21 12674-11-2 Aroclor-1016 560.000 U. 22 11104-28-2 Aroclor-1221 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 560.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1260 240.000	20	8001-35-2	Toxaphene	1	NA
22 11104-28-2 Aroclor-1221 560.000 U. 23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1254 560.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1260 240.000	21	12674-11-2	Aroclor-1016	· · · · · · · · · · · · · · · · · · ·	560.000 [°] U.
23 11141-16-5 Aroclor-1232 560.000 U. 24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1248 560.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1260 240.000	22	11104-28-2	Aroclor-1221	ļ	560.000 U.
24 53469-21-9 Aroclor-1242 560.000 U. 25 12672-29-6 Aroclor-1248 560.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1260 240.000	23	11141-16-5	Aroclor-1232	1	560.000 U.
25 12672-29-6 Aroclor-1248 560.000 U. 26 11097-69-1 Aroclor-1254 1100.000 U. 27 11096-82-5 Aroclor-1260 240.000	24	53469-21-9	Aroclor-1242	1	560.000 U.
26 11097-69-1 Aroclar-1254 1100.000 U. 27 11096-82-5 Aroclar-1260 240.000	25	12672-29-6	Aroclor-1248	· _]	560.000 U.
27 11096-82-5 Aroclor-1260 240.000	26	11097-69-1	Aroclor-1254	1	1100.000 U.
	27	11096-82-5	Aroclor-1260		240.000

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

	•			
SAMPLE MATRIX	soil		SAMPLE ID:	s-12
CONC. LEVEL	LOW		LAB ID:	1765114
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	: 8/17/93		% MOISTURE:	17
			UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	12000.0 U.]	
2	208-96-8	Acenaphthylene	12000.0 U.	
3	83-32-9	Acenaphthene	12000.0 U.	
. 4	86-73-7	Fluorene	12000.0 U.	
5	85-01-8	Phenanthrene	14000.0	
6	120-12-7	Anthracene	2300.0 J.	
7	206-44-0	Fluoranthene	15000.0	
8	129-00-0	Pyrene	11000.0 J.	
· 9	56-55-3	Benzo(a)Anthracene	9900.0 J.	
10	218-01-9	Chrysene	13000.0	
11	205-99-2	Benzo(b)Fluoranthene	8400.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	7900.0 J.	•
13	50-32-8	Benzo(a)Pyrene) 9600.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	12000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	12000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	12000.0 U.	
	·	_ <u></u>		

. .

•

SEHIVQLATILE ORGANICS ANALYSIS DATA SHEET

.

. .

:•

• •

.

:

:

.

100

.

			SANDI F. ID.	S-179F
CONC LEVEL				47/511/
CUNC. LEVEL			LAB ID:	1/02114
EXTRACTION DATE	E: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	: 8/18/93		X MOISTURE:	17
			UG/KG	
СНРО #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	12000.0 U.	
· 2	208-96-8	Acenaphthylene	12000.0 U.	
	83-32-9	Acenaphthene	2000.0 J.	
4	86-73-7	Fluorene	12000.0 U.	
5	85-01-8	Phenanthrene	15000.0	
6	120-12-7	Anthracene	6000.0 J.	
7	206-44-0	Fluoranthene	17000.0	
8	129-00-0	Pyrene	11000.0 J.	
9	56-55-3	Benzo(a)Anthracene	13000.0	
10	218-01-9	Chrysene	17000.0	
11	205-99-2	Benzo(b)Fluoranthene	8900.0 J.	
12	207-08-9	Benzo(k)fluoranthene	8400.0 J.	
13	50-32-8	Benzo(a)Pyrene	9000.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	12000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	12000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	12000.0 U.	
	I	.!		

1 D-T NYTEST ENVIRONMENTAL INC.

. [..

:----

2

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	SAMPLE ID:	s-12
	CONC. LEVEL:	LOW	LAB SAMPLE ID:	1765114
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	5.00
	ANALYSIS DATE:	8/25/93	X MOISTURE:	: 17
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPOU	ND	(DRY BASIS)
1	319-84-6	alpha-BHC		NA
2	319-85-7	beta-8KC	j	. NA
3	319-86-8	delta-BHC	1	NA
4	58-89-9	gamma-BHC(Lindane)		NA
5	76-44-8	Heptachlor	1	NA
6	309-00-2	Aldrin	ĺ	NA
7	1024-57-3	Heptachlor Epoxide	Í	NA
8	959-98-8	Endosulfan 1	· · ·	NA
9	60-57-1	Dieldrin	.	HA.
10	72-55-9	4,4'-DDE	ļ	NA
11	70-20-8	Endrin	Í	NA
12	33213-65-9	Endosulfan II	ļ	KA
13	72-54-8	4,4-000	Ì	NA
14	1031-07-8	Endosulfan Sulfate	1	NA
15	50-29-3	4,4°-DDT	. 1	NA
16	72-43-5	Methoxychlor	1	NA
17	53494-70-5	Endrin Ketone.	· •	NA
18	7421-36-3	Endrin Aldehyde	1	HA
19	57-74-9	Chlordane	1	NA
20	8001-35-2	Toxaphene	· · · · · ·	NA
21	12674-11-2	Aroclor-1016	. 1	480.0CO U.
22	11104-28-2	Aroclor-1221	1	480.000 U.
23	11141-16-5	Aroclor-1232	1	480.000 U.
24	53469-21-9	Aroclor-1242	{	480.000 U.
25	12672-29-6	Aroclor-1248	1	1700.000
26	11097-69-1	Aroclor-1254	. [960.000 U.
27	11096-82-5	Aroclor-1260	ľ	960.000 U.
	I I		1	

.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. 1

· •

۰.

÷.,

:: ;;

SAMPLE MATRIX	SOIL		SAMPLE ID:	S-13
CONC. LEVEL	.: LOW		LAB ID:	1765111
EXTRACTION DATE	8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	E: 8/16/93		X MOISTURE:	20
			UG/KG	
Смрр` #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
. 1	91-20-3	Naphthalene	12000.0 U. (
2	208-96-8	Acenaphthylene	12000.0 U.	
	83-32-9	Acenaphthene	12000.0 U.	
4	86-73-7	Fluorene	12000.0 0.]	
5	85-01-8	Phenanthrene	12000.0 U. j	
6	120-12-7	Anthracene	12000.0 U. [
7	206-44-0	Fluoranthene	8100.0 J.	•
8	129-00-0	Pyrene	3900.0 J.	
9	56-55-3	Benzo(a)Anthracene	4100.0 J.	
10	218-01-9	Chrysene	14000.0	
11	205-99-2	Benzo(b)Fluoranthene	3100.0 J.	
12	207-08-9	Benzo(k)fluoranthene) 2400.0 J.)	
13	50-32-8	Benzo(a)Pyrena	1800.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	12000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	j 12000.0 U. j	
16	191-24-2	Benzo(g,h,i)Perylene	12000.0 U.	
-	1	-1		-

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

	. :			
SAMPLE MATRIX	SOIL		SAMPLE ID:	S-13RE
CONC. LEVEL	.: LOW		LAB ID:	176511
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	E: 8/18/93		X MOISTURE:	. 2
	•		UG/KG	
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	12000.0 U. [
2	208-96-8	Acenaphthylene	12000.0 U.	
	83-32-9	Acenaphthene	3500.0 J.	
4	86-73-7	Fluorene	12000.0 U.	
5	85-01-8	Phenanthrene	10000.0 J.	
6	120-12-7	Anthracene	12000.0 U.	
7	206-44-0	Fluoranthene	9300.0 J.	
8	129-00-0	Pyrene	7700.0 J.	
9	56-55-3	Benzo(a)Anthracene	7400.0 J.	
10	218-01-9	Chrysene	16000.0	
11	205-99-2	Benzo(b)Fluoranthene	5700.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	5100.0 J.	
13	50-32-8	Benzo(a)Pyrene	5400.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	12000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	12000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	12000.0 U.	
	l	·I	!	

•••:

3

1 D-T

NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	- SAMPLE ID:	: S-13
	CONC. LEVEL:	LOW	LAB SAMPLE ID	1765111
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	5.00
	ANALYSIS DATE:	8/25/93	X HOISTURE:	: 20
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPO	UND	(DRY BASIS)
. 1	319-84-6	alpha-8HC	•	RA
2	319-85-7	beta-BHC		AK .
3	.319-86-8	delta-BHC	-	· NA
4	58-89-9	gamma-BHC(Lindane)		HA
5	76-44-8	Heptachlor	·	NA
6	309-00-2	Aldrin		' NA
7	1024-57-3	Heptachlor Epoxide	· • • • • •	NA NA
8	959-98-8	Endosulfan I	- 11	NA
9	60-57-1	Dieldrin	• •	NA
10	72-55-9	4,41-DDE	· · ·	- NA
11	70-20-8	Endrin	abria - Carl	· XA
12	33213-65-9	Endosulfan II	<u>ن</u> ا	- NA
13	72-54-8	4,4-000	* <u>2</u>]	NĂ I
14	1031-07-8	Endosulfan Sulfate	1.14	XA.
15	50-29-3	4,4*-DDT		NA -
16	72-43-5	Methoxychlor	• .	AK .
17	53494-70-5	Endrin Ketone	<u>.</u>	NA
18	7421-36-3	Endrin Aldehyde	· • • •	KA
19	57-74-9	Chlordane	- ' - -	NA T
20	8001-35-2	Toxaphene		NA
21	12674-11-2	Aroclor-1016		430.000 J.
22	11104-28-2	Aroclor-1221		500.000 U.
23	11141-16-5	Arocior-1232		500.000 U.
24	53469-21-9	Aroclor-1242	:	1000.000
25	12672-29-6	Aroclor-1248		500.000 U.
26	11097-69-1	Aroclor-1254		1000.000 U.
27	11096-82-5	Aroclar-1260	2	1000.000 U.
j	Í	•	:	

:

5.

.0000058

1994 15:12 FROM NYS.ENVIR.CONSERVATION

85926873154637554

н.ыт

NEW YORK STATE DEPARTMENT OF ENVIRONMENTAL CONSERVATION

MOBILE LABORATORY VOLATILE ANALYSIS

SITE NAME: ACCURATE DIE CASTING

FIELD ID:S-13

SITE CODE: 734052 PERCENT SOLIDS: 83.0

MATRIX: SOIL

π

SAMPLE NUMBER: 793-211-02

SUBMISSION DATE: 07/30/93 ARCHIVE NO.: V21102

ANALYSIS DATE: 08/04/93

DATA FILE NO.:9304C31A.D

COMPOUND	CONC (PPB)	NON TARGET COMPOUNDS:
Chloromethane	ND	
Bromomethane	ND	
Vinul chloride	ND	
Chloroethane	ND	
Methylene chlorido	ND	
Acetone	ND	-
Carbon disulfide	ND I	
1,1-Dichloroethene	ND I	
1.1-Dichloroethane	ND I	l · · ·
trans-1,2-Dichloroethene	190,000 3	
Chloroform	ND	l'anna anna anna anna anna anna anna ann
1,2-Dichloroethane	ND I	
2-Butanone	ND I	
1,1,1-Trichloroethane	ND I	
Carbontstrachloride	ND !	
Vinyl acetate	ND	
Sromodichleromethene	ND I	
1,1,2,2-Tetrachioroethane	ND I	ļ.
1,2-Dichloropropane	ND I	
trans-1,3-Dichloropropene	. ND I	
Trichloroethene	ND I	
Dibromochloromethane	ND I	
1,1,2-Trichloroethane	ND i	· ·
Benzene	ND I	
cis-1,3-Dichloropropene	ND I	
2-Chloroethylvinylether	ND I	
Bromsform	ND I	
2-Hexanone	I GN	
4-Methyl-2-pontanone	ND I	
Tetrachlorgethene	ND I	
loluene	ND I	
Chlorobenzene	ND I	ND = LESS THAN 5 PPE
Ethylbenzeng	ND 1	
Styrene	ND	ALL CONCENTRATIONS LESS THAN
içtal Xylenes	ND	5 PPB ARE ESTIMATES
lotal Uhlorotoluene	ND	•
Fotal Dichlorobenzene	ND I	

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

. +

SAMPLE MATRIX	: SOIL		SAMPLE ID:	s-14
CONC. LEVEL	.: LOW		LAB ID:	1765105
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	: 8/14/93		X HOISTURE:	29
			UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	14000.0 U.	
2	208-96-8	Acenaphthylene	14000.0 U.	
3	83-32-9	Acenaphthene	14000.0 U.	
4	86-73-7	Fluorene	14000.0 U.	
5	85-01-8	Phenanthrene	24000.0	
6	120-12-7	Anthracene	3300.0 J.	,
7	206-44-0	Fluoranthene	20000.0	
8	129-00-0	Pyrene	13000.0 J.	
9	56-55-3	Benzo(a)Anthracene	8300.0 J.	
10	218-01-9	Chrysene	17000.0	
11	205-99-2	Benzo(b)Fluoranthene	6300.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	5400.0 J.	
13	50-32-8	Benzo(a)Pyrene	6400.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	14000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	14000.0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	14000.0 U.	
	I	·!		•

nytest environmental Inc.

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

. (

SAMPLE MATRIX	: \$01L		SAMPLE ID:	S-14RE
CONC. LEVEL	: LOW		LAB ID:	1765105
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	30.00
ANALYSIS DATE	: 8/18/93	• .	X MOISTURE:	25
	-		UG/KG	
CMPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	14000.0 U.	
2	208-96-8	Acenaphthylene	14000.0 U.	
. 3	83-32-9	Acenaphthene	2800.0 J.	
4	86-73-7	Fluorene	2600.0 J.	
5	85-01-8	Phenanthrene	22000.0	
. 6	120-12-7	Anthracene	2300.0 J.	·
7	206-44-0	Fluoranthene	17000.0	
8	129-00-0	Pyrene	12000.0 J.	
9	56-55-3	Benzo(a)Anthracene	12000.0 J.	
10	218-01-9	Chrysene at a star	14000.0	
- 11	205-99-2	Benzo(b)Fluoranthene	7700.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	6700.0 J.	
13	50-32-8	Benzo(a)Pyrene	7700.0 J.	
. 14	193-39-5	Indeno(1,2,3-cd)Pyrene	14000.0 U.	
· · 15	53-70-3	Dibenz(a,h)Anthracene	14000.0 U.	
. 16	191-24-2	Benzo(g,h,i)Perylene	14000.0 U.	
	l		· !	

NEW YORK STATE DEPARTMENT OF ENVIRONMENTAL CONSERVATION

MOBILE LABORATORY VOLATILE ANALYSIS

SITE NAME: ACCURATE DIE CASTING

15:13 FROM NYS.ENVIR.CONSERVATION

FIELD ID:S-14

SITE CODE: 734052 SAMPLE NUMBER: 793-211-03 MATRIX: SOIL

SUBMISSION DATE: 07/30/93 ARCHIVE ND.: V21103

PERCENT SOLIDS:

то

ANALYSIS DATE: 08/05/93 DATA FILE NO.: 9304036A.D

85926873154637554

69.0

P.09

COMPOUND	CONC	(PPE	3)	I NON	TARGET	COMPOL	INDS:	YES
Chloromethane	-	ND	1 1	UNKNOWN	HYDROCA	REON 1	45,00	0 3
Bromomethene	÷	ND		E	· ·	•		
Vinyl chloride		ND	1.1.2.2		No.			
Chloroethane	· · · ·	ND		1		 		
Mathylene chloride		ND		I Contraction and the second		۰.		
Acetone		ND		C GARA				
Carbon disulfide		NØ		1		• '		
1,1-Dichloroethene		ND	. 800	tan Section S		•		
1,1-Dichlorosthane	:-	ЮM			•		•	
trans-1,2-Dichlordethene	19	,000	J					
Chloroform		 ND 		1		、 ·		
1,2-Dichlorosthane		ND		La parte de				
2-Butanone		ND	ته . ا بب ب	1.			•	4
1,1,1-Trichloroethune	•	ND	· .	•			· ·	
Uarbontetrachloride	•	NÐ	•	:				(
Vinyl acetate		ND		l	· ,			
Srowedichloromethane	- ·	ND		i				
1,1,2,2-Tetrachioroethane	د •	ND		1		•		4
1,2-Dichlorepropane	· ·	ND	Ļ	1				
1cons-1,3-Dichlaropropens	· · · .	ND	5 JN					. •
richieroethene	. ,	ND	i	I and the	. ·	• •		· •
Dibremochleromethane		ND	. 1	ł .				·]
1,1,2-Trichloroethane		ND					•	. 1
benzene	•	ND		the second second	:	· .		
cis-1,3-Dichloropropens		ND	1	1			· ,	1
2-Chloroethylvinylether		ND	1	Ì	-	•		
SPONOTORM O Vision	• •	ND	1	1				
2-Hexanone	•	ND						
4-Metnyl-2-pentangne	•	· ND	••• 1	1 · .			-	
(etrachiorosthens)	•	ND	1	5		· ·	۰.	ı
Philes		ND		i · .	·			
Laioropenzene Caluta		ND	ł	ND ND	= LESS	THAN 2	500 P	-6
		ND	l			×		
atyrene -		ND	l	ALL CC	INCENTRA	TIONS	LESS	THAN
Takal Chisers		ND	1	2500	PPB ARE	ESTIM	IATES	1
Fotal Uniorotoluone - Tatal Dimblerit		ND	· ·					- 1
otal Ulchlorobenzene		ND.	l					

1 D-T .NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

· · ·

SAMPLE MATRIX:	SOIL	SAMPLE 1D:
CONC. LEVEL:	LOW	LAB SAMPLE 10:
EXTRACTION DATE:	8/02/93	DIL FACTOR:
ANALYSIS DATE:	8/25/93	X HOISTURE:

CMPD # -- CAS Number

Γ

.

ŧ.

PESTICIDE/PCB COMPOUND

UG/KG (DRY BASIS)

5-14

5.00

29

1765105

	and the second second second second second second second second second second second second second second second		
1	319-84-6	alpha-BHC	NA NA
2	319-85-7	beta-BHC	NA NA
3	319-86-8	delta-BXC	NA NA
4	58-89-9	gamma-BHC(Lindane)	NA NA
5	76-44-8	Heptachlor	NA NA
6	309-00-2	Aldrin	NA
7	1024-57-3	Heptachlor Epoxide	NA NA
8	959-98-8	Endosulfan I	NA
9	60-57-1	Dieldrin	NA NA
10	72-55-9	4,41-DDE	NA
11	70-20-8	Endrin	NA
12	33213-65-9	Endosulfan II	NA
13	72-54-8	4,4-DDD	NA NA
14	1031-07-8	Endosulfan Sulfate	HA NA
15	50-29-3	4,4'-DDT	NA
16	72-43-5	Hethoxychlor	NA NA
17	53494-70-5	Endrin Ketone	NA
18	7421-36-3	Endrin Aldehyde	NA
19	57-74-9	Chlordane	NA NA
20	8001-35-2	Toxaphene	1 NA
21	12674-11-2	Aroclor-1016	560.000 U.
22	11104-28-2	Aroclor-1221	560.000 U.
23	11141-16-5	Aroclor-1232	560.000 U.
24	53469-21-9	Aroclor-1242	560.000 U.
25	12672-29-6	Aroclor-1248	1500.000
26	11097-69-1	Aroclor-1254	1100.000 U.
27	11096-82-5	Aroclor-1260	1100.000 U.
			1

.

.....

F

.dr.

•

÷.

; .:

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

۰.

ιŧ

SAMPLE MATRIX	sotL		SAMPLE ID:	·s-15
CONC. LEVEL	.: Low		LAB ID:	1765108
EXTRACTION DATE	: 8/2/93		DIL FACTOR:	10.00
ANALYSIS DATE	: 8/13/93		% HOISTURE:	68
•			UG/KG	
CNPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
. 1	91-20-3	Naphthalene	10000.0 U.	
2	208-96-8	Acenaphthylene	10000.0 U.	
3	83-32-9	Acenaphthene	10000.0 U.	
4	86-73-7	Fluorene	10000.0 0.	
5	85-01-8	Phenanthrene	1700.0 J.	
6	120-12-7	Anthracene	1300.0 J.	
7	206-44-0	Fluoranthene	1300.0 J.	
8	129-00-0	Ругене	1400.0 J.	•
9]56-55-3 °	Benzo(a)Anthracene	10000.0 U. j	
. 10	218-01-9	Chrysene	10000.0	
11	205-99-2	Benzo(b)Fluoranthene	10000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	10000.0 U.	
. 13	50-32-8	Benzo(a)Pyrene	1300.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	10000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	10000.0 U.	•
16	191-24-2	Benzo(g,h,i)Perylene	10000.0 U.	
]	.]]	

0000027

t

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

, .:

:

.

:

SAMPLE MATRIX	soil		SAMPLE ID:	S-15RE
CONC. LEVEL	.: LOW		LAB ID:	1765108
EXTRACTION DATE	8/2/93		DIL FACTOR:	10.00
ANALYSIS DATE	8/17/93		X HOISTURE:	68
			UG/KG	
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3		10000.0 U. [
2	208-96-8	Acenaphthylene	10000.0 U.	
3	83-32-9	Acenaphthene	10000.0 U.	
4	86-73-7	Fluorene	10000.0 U.	
5	85-01-8	Phenanthrene	10000.0 U.	
6	120-12-7	Anthracene	10000.0 U.	-
7	206-44-0	Fluoranthene	10000.0 U.	
8	129-00-0	Pyrene	1400.0 J.	
9	56-55-3	Benzo(a)Anthracene	2500.0 J.	
10	218-01-9	Chrysene	9600.0 J	
11	205-99-2	Benzo(b)Fluoranthene	10000.0 U.	
12	207-08-9	Benzo(k)Fluoranthene	10000.0 U.	
13	50-32-8	Benzo(a)Pyrene	10000.0 U.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	10000.0 U.	
. 15	53-70-3	Dibenz(a,h)Anthracene	, 10000_0 U.	
16	191-24-2	Benzo(g,h,i)Perylene	10000.0 U.	
	1			

1 D-T • NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	SAMPLE ID:	: S-15
	CONC. LEVEL:	LOW	LAB SAMPLE ID:	: 1765108
	EXTRACTION DATE:	8/02/93	DIL FACTOR:	: 5.00
	ANALYSIS DATE:	8/25/93	X HOISTURE:	: 68
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPO	UND	(DRY BASIS)
1	319-84-6	alpha-BHC		NA NA
2	319-85-7	beta-BHC		на
3	319-86-8	delta-BHC		NA
4	58-89-9	gamma-BHC(Lindane)		АК
5	76-44-8	Heptachlor		HA .
6	309-00-2	Aldrin		L NA i
7	1024-57-3	Heptachlor Epoxide		NA
8	959-98-8	Endosulfan 1	1	NA
9	60-57-1	Dieldrin		NA I
10	72-55-9	4,4'-DDE	· .	NA
11	70-20-8	Endrin		NA :
12	33213-65-9	Endosulfan II	•	NA
13	72-54-8	4,4-DDD	. 1	NA
14	1031-07-8	Endosulfan Sulfate	1	NA
15	50-29-3	4,43-DDT		NA
16	72-43-5	Methoxychior	I	NA
17	53494-70-5	Endrin Ketone	· · · · ·	NA -
18	7421-36-3	Endrin Aldehyde		NA,
19	57-74-9	Chlordane	ļ	NA
20	8001-35-2	Toxaphene	ļ	NA
21	12674-11-2	Aroclor-1016	ļ	1300.000 U.
22	11104-28-2	Aroclor-1221	·	1300.000 U.
23	11141-16-5	Aroclor-1232		1300.000 U.
24	53469-21-9	Araclor-1242		1300.000 U.
25	12672-29-6	Aroclor-1248	1	1300.000 U.
26	11097-69-1	Aroclor-1254		2500.000 U.
27	11096-82-5	Aroclor-1260	1	1500.000 J.
•	1 1		. 1	

.

-

: :

0000055

18 PAH Nytest environmental inc.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET . •

CMPD # CAS	Number	PAH COMPOUNDS	UG/KG (DRY BASIS)	
			X HOISTURE:	- 20
ANALYSIS DATE:	8/17/03		UIL FACTOR:	2.00
EXTRACTION DATE:	8/2/93			1103115
	LOW		LAR ID-	1765114
CONC. LEVEL:	100		SAMPLE ID:	\$-16
SAMPLE MATRIX: '	SOIL			
	•			

CARD & CAS HUMDER PAH CO	7	H	l	l		1	ļ		l	l	l		I	l	1	2			l	l	l	l																	l	l	l	l	ł	ł						1								l	ļ	ŀ	ł	l			1	ļ	ļ		ļ	ļ	ļ	ļ	ļ	ļ	į	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	ł	į	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ	ļ			١	١	١	١	١
--------------------------	---	---	---	---	--	---	---	--	---	---	---	--	---	---	---	---	--	--	---	---	---	---	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	---	---	---	---	---	---	--	--	--	--	--	---	--	--	--	--	--	--	--	---	---	---	---	---	--	--	---	---	---	--	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	---	--	--	---	---	---	---	---

Ì

. :

(DRY BASIS)

1 91-20-3	Naphthalene	1000 0 11
2 208-96-8	Acenaphthylene i	1000.0 0.
3 83-32-9	Acapachehana	.U 0.0001
4 86-73-7	[stude:	180.0jJ.
5 185-01 0		150.0 J.
- 103-10-8	Phenanthrene	1800.0
0 1120-12-7	Anthracene	310.0 J.
7 206-44-0	Fluoranthene	1900_0
8 129-00-0	Pyrene	4200.0
9 56-55-3	Benzo(a)Anthracene	7200.0
10 218-01-9	Chrysene	/10.0 J.
11 205-99-2	[Benzo(b)Elunganebang	1200.0
12 207-08-9	Penadovorruorantnene	650.0 J.
13 150-32-0	Joenzo(K): Luoranthene	740.0 J.
16 107-70 6	[Benzo(a)Pyrene	780.0 .
173-34-5	Indeno(1,2,3-cd)Pyrene	610.0 J. j
15 155-70-3	Dibenz(a,h)Anthracene	1000.0 u. 1
16 191-24-2	Benzo(g,h,i)Perylene	1000 0 11

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

, 7

.

-

SAMPLE MATRIX	soil		SAMPLE ID:	S-16RE
CONC. LEVEL	: LON		LAB ID:	1765116
EXTRACTION DATE	: 8/2/93		DIL FACTOR: -	2.00
ANALYSIS DATE	: 8/18/93		X HOISTURE:	20
			UG/KG	
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
· 1	91-20-3	Naphthalene	1000.0 U.]	
2	208-96-8	Acenaphthylene	1000.0 U	
.3	83-32-9	Acenaphthene	170.0 J.	
- 4	86-73-7	Fluorene	140.0 J	
5	85-01-8	Phenanthrene	1700.0	
6	120-12-7	Anthracene	280.0 J.	
7	206-44-0	Fluoranthene	1600.0	
8	129-00-0	Pyrene	3800.0	
9	56-55-3	Benzo(a)Anthracene	670.0 J.	
10	218-01-9	Chrysene	1200.0	
. 11	205-99-2	Benzo(b)Fluoranthene	730.0 J.	
12	207-08-9	Benzo(k)Fluoranthene	740.0 J.	
13	50-32-8	Benzo(a)Pyrene	810.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	1000.0 U.	
. 15	53-70-3	Dibenz(a,h)Anthracene	1000.0 U.	
16	191-24-2	Benzo(g,h,i)Perviene	1000.0 0.	
	I	· 		

1 D-T NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

SAMPLE MATRIX:	SOIL	SAMPLE	ID:	S-16
CONC. LEVEL:	LOW	LAB SAMPLE	ID:	1765116
EXTRACTION DATE:	8/02/93	DIL FACT	OR:	5.00
ANALYSIS DATE:	8/18/93	X NOISTU	RE:	20
			UG/K	(G
CAS Number	PESTICIDE/PCB	COMPOUND	(DRY	BASIS)

CMPD #

1	319-84-6	alpha-BHC	KA
2	319-85-7	beta-BHC	HA I
3	319-86-8	delta-BHC	XA
4	58-89-9	gamma-BHC(Lindane)	KA.]
5	76-44-8	Heptachlor	NA I
6	309-00-2	Aldrin	KA
7	1024-57-3	Heptachlor Epoxide	NA I
8	959-98-8	Endosulfan I	ј на ј
9	60-57-1	Dieldrin	j ka j
10	72-55-9	4,4'-DDE	KA I
11	70-20-8	Endrin	} KA -]
12	33213-65-9	Endosulfan II	<u>і , жа</u>
13	72-54-8	4,4-000	AA
14	1031-07-8	Endosulfan Sulfate	L NA
15	50-29-3	4,41-DDT	XX
16	72-43-5	Hethoxychlor	AA
17	53494-70-5	Endrin Ketone	AA
18	7421-36-3	Endrin Aldehyde	NA 1
19	57-74-9	Chlordane	XA
20	8001-35-2	Toxaphene	KA j
21	12674-11-2	Aroclor-1016	500.000 U.
22	11104-28-2	Aroclor-1221	500.000 U.
23	11141-16-5	Aroclor-1232	500.000 U.
24	53469-21-9	Aroclor-1242	500.000 U.
25	12672-29-6	Aroclor-1248	500.000 U.
26	11097-69-1	Aroclor-1254	1000.000 u. j
27	11096-82-5	Aroclor-1260	1000.000 U.
			İİ

SENIVOLATILE ORGANICS ANALYSIS DATA SHEET

••••

: :

Ĺ

•

....

.

3

	•	· t		
SAMPLE MATRIX	SOIL		SAMPLE ID:	
CONC. LEVEL	.: LOW		LAB ID:	176
EXTRACTION DATE	8/2/93		DIL FACTOR:	
ANALYSIS DATE	: 8/17/93		X MOISTURE:	
			UG/KG	
CHPD #	CAS Number	PAH COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	[1000.0 U.]	
2	208-96-8	Acenaphthylene	1000.0 U.	
-3	83-32-9	Acenaphthene	1000.0 U.	
4	86-73-7	Fluorene	1000.0 U.	
5	85-01-8	Phenanthrene	920.0 J.	
6	120-12-7	Anthracene	130.0 J.	
7	206-44-0	Fluoranthene	1300.0	
8	129-00-0	Pyrene	1300.0	
9	56-55-3	Benzo(a)Anthracene	340.0 J.	
10	218-01-9	Chrysene	660.0 J.	
11	205-99-2	Benzo(b)Fluoranthene	470.0 J.	
12	207-08-9	(Benzo(k)Fluoranthene	510.0 J.	
13	50-32-8	Benzo(a)Pyrene	490.0 J.	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	1000.0 U.	
15	53-70-3	Dibenz(a,h)Anthracene	1000.0 0.	
16	191-24-2	Benzo(g,h,i)Perviene	1000.0 U. [
	!	.[

1 D-T "NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	SAMPLE ID	: DUP
	CONC. LEVEL:	LOW	LAB SAMPLE ID	: 1765118
	EXTRACTION DATE:	8/02/93	DIL FACTOR	: 5.00
	ANALYSIS DATE:	8/25/93	X NOISTURE	: 30
				UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPO	UND .	(DRY BASIS)
1	319-84-6	alpha-BXC		ЖА
2	319-85-7	beta-BHC) NA (
3	319-86-8	delta-8KC		KA
4	58-89-9	gamma-BHC(Lindane)		NA
5	76-44-8	Heptachlor		KA
6	309-00-2	Aldrin		KA
7	1024-57-3	Heptachlor Epoxide		NA
8	959-98-8	Endosulfan I		NA
·9	60-57-1	Dieldrin	i	NA
10	72-55-9	4,41-DDE		NA
11	70-20-8	Endrin		NA
12	33213-65-9	Endosulfan II		NA
13	72-54-8	4,4-000		· NA
14	1031-07-8	Endosulfan Sulfate	-	NA
15	50-29-3	4,4"-DDT		NA
16	72-43-5	Methoxychlor		NA
17	[`53494-70-5	Endrin Ketone		NA
18	7421-36-3	Endrin Aldehyde		NA
19	57-74-9	Chlordane		NA
20	8001-35-2	Toxaphene	i	NA
21	12674-11-2	Aroclor-1016		570.000 U.
22	11104-28-2	Aroclor-1221	1	570.000 U.
23	11141-16-5	Aroclor-1232	ļ	570.000 U.
24	53469-21-9	Aroclor-1242		570.000 U.
25	12672-29-6	Aroclor-1248	(570.000 U.
26	11097-69-1	Aroclor-1254		1100.000 U.
27	11096-82-5	Aroclor-1260	·	1100.000 U.
	<u> </u>		·	I

·••.;
1 D-T NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB CREANICS ANALYSIS DATA SHEET

	CANDIE MATRIY	soft	SANDIE ID	• \$ •17
				- 1765117
	LUNC. LEVEL			- 1.00
	EXTRACTION DATE:		T WOTCTINE	
	ANALISIS DATE	: 0/1//93	A NOISIGKE	• 60 Uc/Ye
CMPD 1	f CAS Number	PESTICIDE/PCB COMPOL	лю	(DRT BASIS)
4	319-84-6	alpha-BHC	,	NA
	2 319-85-7	beta-BHC	•	[XA]
3	5 319-86-8	delta-BHC	•	XA
	58-89-9	gamma-BHC(Lindane)	• •	AK
	5 76-44-8	Xeptachlor		NA
	5 309-00-2	Aldrin		. NA .
7	7 1024-57-3	Heptachlor Epoxide	· ·	NA 🖕
8	3 959-98-8	Endosulfan l		, KA [
Ċ,	60-57-1	Dieldrin		НА.
10	72-55-9	4,4'-DDE		KA []
1	70-20-8	Endrin	Ż	NA 🔂
12	2 33213-65-9	Endosulfan II		NA -
1	5 72-54-8	4,4-000		NA 📜
· 14	1031-07-8	Endosulfan Sulfate		NA .
15	5 50-29-3	4,4'-DDT	·	NA .
16	5 72-43-5	Methoxychlor	. '	NA 📜
17	7 53494-70-5	Endrin Ketone		NA .,
18	3 7421-36-3	Endrin Aldehyde		NA
15	57-74-9	Chiordane	• * *	NA J
. 20	8001-35-2	Toxaphene	• 1	NA ·
21	12674-11-2	Aroclor-1016		110.000 U. j
22	11104-28-2	Aroclor-1221	1	110.000 U.
23	11141-16-5	Aroclor-1232	. [110.000 ป.
24	53469-21-9	Aroclor-1242	· · · · · · · · · · · · · · · · · · ·	110.000 U.
25	12672-29-6	Aroclor-1248	1	110.000 U.]
26	11097-69-1	Aroclor-1254	-	210.000 U.
27	11096-82-5	Aroclor-1260	1	210.000 0.

1B PAH

NYTEST ENVIRONMENTAL INC.

٢

.....

•.•.

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

S-18	SAMPLE ID:		X: SOIL	SAMPLE MATRI
176511	LAB ID:		L: LOW	CONC. LEVE
4.00	DIL FACTOR:		'E: 8/2/93	EXTRACTION DATE
27	X MOISTURE:		E: 8/17/93	ANALYSIS DATE
	UG/KG			
	(DRY BASIS)	PAH COMPOUNDS	CAS Number	СМРО #
	1200.0 J.]	Naphthalene	91-20-3	1
	2000.0 U. [Acenaphthylene	208-96-8	2
	3300.0	Acenaphthene	83-32-9	3
	3000.0	Fluorene	86-73-7	4
	12000.0	Phenanthrene	85-01-8	5
• .	3600.0	Anthracene	120-12-7	6
	12000.0	Fluoranthene	206-44-0	7
	24000_0 E	Pyrene	129-00-0	8
	5600.0	Benzo(a)Anthracene	56-55-3	9
	8200.0	Chrysene	218-01-9	10
	3900.0	Benzo(b)Fluoranthene	205-99-2	11
	4300.0	Benzo(k)Fluoranthene	207-08-9	12
		Benzo(a)Pyrene	50-32-8	13
	3100.0	Indeno(1,2,3-cd)Pyrene	193-39-5	14
	2000.0.11	Dibenz(a,h)Anthracene	53-70-3	15
	3000 0 1	Benzo(g,h,i)Perviene	191-24-2	16

. 18 PAH NYTEST ENVIRONMENTAL INC.

٢.

•

SEMIVOLATILE ORGANICS ANALYSIS DATA SHEET

SAMPLE MATRI	X:		SANDIE TO.	
CONC. LEVE	L: LOW			5*160L
EXTRACTION DAT	E: 8/2/93		LAB ID:	1765115
ANALYSIS DAT	E: 8/18/93		DIL FACTOR:	16.00
			X HOISTURE:	23
CHPD #			UG/KG	
		PAR COMPOUNDS	(DRY BASIS)	
1	91-20-3	Naphthalene	1 1300 0 0 1 1	•
. 2	208-96-8	Acenaphthylene	1 1000.0 U3 1	
3	83-32-9	Acenaphthene	7600.00.1	
4	86-73-7	Fluorene		
5	85-01-8	Phenenthrene	3300.0 01	
. 6	1120-12-7	Anthenese	25000.0 0	
7	1206-44-0		4400.0 DJ	•••
· 8	120-00-0	In cuorantnene	24000.0 p	
	127-00-0	jryrene	15000.0 D	
· 10	120-22-2	Benzo(a)Anthracene	5700.0 DJ	
10	218-01-9	Chrysene	8400.0 p	
11	205-99-2	Benzo(b)Fluoranthene	4600.0 pJ 1	
. 12	207-08-9	Benzo(k)Fluoranthene	5100.0 pJ 1	
13	50-32-8	Benzo(a)Pyrene	5300.0 pJ	
14	193-39-5	Indeno(1,2,3-cd)Pyrene	6900.0.11	
15	53-70-3	Dibenz(a,h)Anthracene	4000 g u 1	
16	191-24-2	Benzo(g,h,f)Perviene	4000 0 U	
	•		0700.0 0.	•
		·		

1 D-T . NYTEST ENVIRONMENTAL INC.

TCL PESTICIDE/PCB ORGANICS ANALYSIS DATA SHEET

	SAMPLE MATRIX:	SOIL	SAMPL	E ID:	: S-18
	CONC. LEVEL:		SAMPL	E ID:	: 1765115
	EXTRACTION DATE:	8/02/93	DIL FA	CTOR	5.00
	ANALYSIS DATE:	8/25/93	% MOIS	TURE	: 23
					UG/KG
CMPD #	CAS Number	PESTICIDE/PCB COMPOUND			(DRY BASIS)
1	319-84-6	alpha-BHC			NA NA
2	319-85-7	beta-BKC			AK
3	319-86-8	delta-BHC			NA
4	58-89-9	gamma-BHC(Lindane)			NA
່ 5	76-44-8	Heptachlor		!	. NA
6	309-00-2	Aldrin			NA
7	1024-57-3	Heptachlor Epoxide		1	I NA
8	959-98-8	Endosulfan I	•		NA /
9	60-57-1	Dieldrin		1	NA
10	72-55-9	4,4'-DDE		1	. NA
11	70-20-8	Endrin			NA
12	33213-65-9	Endosulfan II		1	XA
13	72-54-8	4,4-000			NA
14	1031-07-8	Endosulfan Sulfate			·
15	50-29-3	4,4'-DDT		Í	NA NA
16	72-43-5	Methoxychlor			NA
17	53494-70-5	Endrin Ketone		1	NA T
18	7421-36-3	Endrin Aldehyde		I	NA
19	57-74-9	Chlordane	•	ĺ	NA
20	8001-35-2	Toxaphene		Ì	NA
21	12674-11-2	Aroclor-1016		1	520.000 U.
22	11104-28-2	Aroclor-1221		1	520.000 U.
23	11141-16-5	Aroclor-1232		1	520.000 U.
24	53469-21-9	Aroclar-1242		1	520.000 U.
25	12672-29-6	Aroclor-1248		· · I	520.000 U.
· 26	11097-69-1	Aroclor-1254		Í	1000.000 U
27	11096-82-5	Aroclor-1260		1	1000.000 U.
	1 1			1	

لا چ

÷

•

1

٤.,

riç. . :...

•••

APPENDIX B

SOIL BORING PROTOCOLS

SOIL BORING PROCEDURES

Soil borings shall be completed using hollow stem augers and/or other applicable drilling methods (i.e. air rotary, spun casing) to a depth of approximately 6 feet. If the hollow stem auger drilling method is utilized for monitoring well completion, the minimum inside diameter of the augers shall be 4-1/4 inches.

Samples of the encountered subsurface materials will be collected continuously to provide physical descriptions of the material at each location. The sampling method employed will be the ASTM Method D-1586-84 using either a standard 2-foot long, 2-inch outside diameter split-barrel sampler with a 140 lb. hammer or a 3-inch outside diameter sampler with a 300 lb. hammer. Upon retrieval of the sampling barrel, the collected sample shall be described, labelled, and placed in a glass jar.

A hydrogeologist will be on-site during the drilling and sampling operations to fully describe each soil sample including but not limited to: 1) soil type, 2) color, 3) percent recovery, 4) moisture content, 5) odor and other observations, such as organic content and cohesiveness. The Wentworth Soil Classification System will be used to describe the soil samples. The hydrogeologist will be responsible for retaining a representative portion of each sample in a glass jar labeled at a minimum with: 1) site name; 2) boring number; 3) sample interval; 4) date; and, 5) time of sample collection. Split-spoon soil samples will be field screened for volatile organic compounds (VOC) for Polychlorinated Biphenyls (PCBs) using an immunoassay PCB sampling kit as described within Appendix C. Field screening readings will be recorded in the field log.

The VOC screening procedure will be performed by placing a portion of each soil samples into a plastic self-sealing baggie or a glass drilling jar sealed with aluminum foil. The sample will be retained until it equilibrates with ambient temperature. The sample will then be gently agitated to volatilize potential VOCs and the headspace will be monitored with a PID.

The drilling contractor will be responsible for obtaining accurate and representative samples, informing the supervising hydrogeologist of changes in drilling pressure, and keeping a separate general log of soils encountered. Included in this log will be a record of blow counts (i.e. the number of blows from a 140 or 300 pound soil sampling drive weight required to drive the split barrel sampler 6 inches).

Soil sampling equipment, including split spoon samplers, will be decontaminated prior to the initial boring and following advancement of each successive boring in accordance with the decontamination protocol provided below:

Equipment Decontamination

Decontamination procedures will be followed for drilling and sampling activities. The split barrel sampling equipment will be decontaminated after each use. Drilling equipment mobilized to the site will receive an initial decontamination. Decontamination will consist of steam cleaning of the entire rig and associated equipment to the satisfaction of the hydrogeologist. The rear portion of the drill rig will be decontaminated as necessary by steam cleaning between soil boring installations. In addition, equipment used for sample collection will be decontaminated using a high pressure steam cleaner to remove soil and volatilize organics. Drilling equipment will be decontaminated prior to removing the equipment from the site.

The field sampling equipment cleaning and decontamination procedures will be as follows:

- 1. Non-phosphate detergent wash.
- 2. Tap water rinse.
- 3. Distilled water rinse

If necessary, sampling equipment should be numbered in a manner that will not affect their integrity and wrapped in a material that will prevent them from becoming contaminated. Information concerning decontamination methodology, date, time, and personnel will be recorded in the field log book. Field decontamination wastes will be contained in 55-gallon drums for subsequent treatment or disposal off-site.

APPENDIX C

IMMUNOASSAY SAMPLING PROTOCOL EPA METHOD SW846-4020

METHOD 4020

SOIL SCREENING FOR POLYCHLORINATED BIPHENYLS BY IMMUNOASSAY

1.0 SCOPE AND APPLICATION

DRAFT

1.1 Method 4020 is a procedure for screening soils to determine when total polychlorinated biphenyls (PCBs) are present at concentrations above 5 mg/Kg. Method 4020 provides an estimate for the concentration of PCBs by comparison with a standard.

1.2 Using the test kit from which this method was developed, 95 % of samples containing 0.625 ppm or less of PCBs will produce a negative result in the 5 ppm test configuration.

1.3 In cases where the exact concentrations of PCBs are required, quantitative techniques (i.e., Methods 8080/8081) should be used.

2.0 SUMMARY OF METHOD

2.1 Test kits are commercially available for this method. The manufacturer's directions should be followed. In general, the method is performed using an extract of a soil sample. Sample and an enzyme conjugate reagent are added to immobilized antibody. The enzyme conjugate "competes" with PCBs present in the sample for binding to immobilized anti-PCB antibody. The test is interpreted by comparing the response produced by testing a sample to the response produced by testing standard(s) simultaneously.

3.0 INTERFERENCES

3.1 Chemically similar compounds and compounds which might be expected to be found in conjunction with PCB contamination were tested to determine the concentration required to produce a positive test result. These data are shown in Table 1.

4.0 APPARATUS AND MATERIALS

4.1 PCB RIS<u>c</u> Test Kits (EnSys, Inc.), or equivalent. Each commercially available test kit will supply or specify the apparatus and materials necessary for successful completion of the test.

5.0 REAGENTS

5.1 Each commercially available test kit will supply or specify the reagents necessary for successful completion of the test.

4020-1

Revision Draft 1 October 1992

6.0 SAMPLE COLLECTION, PRESERVATION, AND HANDLING

6.1 See the introductory material to this chapter, Organic Analytes, Section 4.1.

6.2 Soils samples may be contaminated, and should therefore be considered hazardous and handled accordingly.

7.0 PROCEDURE

7.1 Follow the manufacturer's instructions for the test kit being used. Those test kits used must meet or exceed the performance indicated in Tables 2-4.

8.0 QUALITY CONTROL

8.1 Follow the manufacturer's instructions for the test kit being used for quality control procedures specific to the test kit used. Additionally, guidance provided in Chapter One should be followed.

8.2 Use of replicate analyses, particularly when results indicate concentrations near the action level, is recommended to refine information gathered with the kit.

8.3 Do not use test kits past their expiration date.

8.4 Do not use tubes or reagents designated for use with other kits.

8.5 Use the test kits within their specified storage temperature and operating temperature limits.

8.6 Method 4020 is intended for field or laboratory use. The appropriate level of quality assurance should accompany the application of this method to document data quality.

9.0 METHOD PERFORMANCE

9.1 A study was conducted using fourteen standard soils and three soil samples whose PCB concentration had been established by Method 8080. Replicates were performed on seven of the standard soils and on one of the soil samples for a total of 25 separate analyses. Each of two different analysts ran the 25 analyses. Results indicated that "<" assignments are accurate with almost 99% certainty at the 50 ppm level while ">" assignments are accurate with almost 99% certainty at the 50 ppm level while ">" assignments can be up to about 96% inaccurate as the sample concentration approaches that of the testing level. Corresponding certainties at the 5 ppm level are 92% and 82% respectively. Tables 2 and 3 summarize these results.

Revision Draft 1 October 1992 C.

10.0 REFERENCES

- J.P. Mapes, T.N. Stewart, K.D. McKenzie, L.R. McClelland, R.L. Mudd, W.B. Manning, W.B. Studabaker, and S.B. Friedman, "PCB-RIScTM - An On-Site Immunoassay for Detecting PCB in Soil", Ensys Inc., Research Triangle Park, NC 27709
- 2. PCB RISc[™] Users Guide, Ensys Inc.
- 3. R.W. Counts, R.R. Smith, J.H. Stewart, and R.A. Jenkins, "Evaluation of PCB Rapid Immunoassay Screen Test System", Oak Ridge National Laboratory, Oak Ridge, TN 37831, April 1992, unpublished

4020-3

Revision Draft 1 October 1992

Ì

Tab Possible Soil	le l Interferences [*]
Compound	Soil Equivalent Concentration (ppm) Required to Yield a Positive Result
1-Chloronaphthalene	10,000
1,2,4-Trichlorobenzene	10,000
2,4-Dichlorophenyl-benzenesulfonate	1000
2,4-Dichloro-1-naphthol	>10,000
Bifenox	500
Diesel fuel	>10,000
Pentachlorobenzene	>10,000
2,5-Dichloroaniline	>10,000
Hexachlorobenzene	>10,000
Gasoline	>10,000
Dichlorofenthion	10,000
Tetradifon	125

* Ensys, Inc. publication

Revision Draft 1 October 1992

Ï

E

1

4020-4

·		Es	timate	ed Erro	Tabl r Rates	e 2 s for 5	ppm Di	ilution				
True Value (ppm)	0	1	2	3	4	5	6	7	· <mark>8</mark>	9	10	20
Estimated Rate of False Positives (%)	1.3	13. 2	39. 2	65.2	82.3	-	-		-	-	-	-
Estimated Rate of False Negatives (%)	-	-	-	-	•	8.5	4.1	2.0	1.0	0.5	0.3	<0. 1

<u>∽</u>.. ⊥ .

		<u>Ést</u>	timate	d Error	Tabl Rates	e 3 for 5() ppm D	ilution	l			
True Value (ppm)	Ó	5	10	15	20	30	40	.50	60	70	80	100
Estimated Rate of False Positives (%)	1.0	7.9	24. 5	46.0	65.0	87.3	95.6	-		-	-	-
Estimated Rate of False Negatives (%)	-	-	-	-	•	-	•	1.7	0.7	0.3	0.2	<0. 1

Revision Draft 1 October 1992

·	Comparison	Table 4 of Immunoassay*	with GC
Sample ID	Screening Test Results	GC Results	Does screening test agree with GC/MS determination?
101	<5 ppm	<0.5 ppm	yes
284	<5 ppm	<0.5 ppm	yes
292	<5 ppm	<0.5 ppm	yes
199	<5 ppm	0.5 ppm	yes
264	<5 ppm	1 ppm	yes
257	<5 ppm	1.8 ppm	yes
259	<5 ppm	4 ppm	yes
265	<5 ppm	4.5 ppm	yes
200	<5 ppm	5 ppm	yes
170	5-50	5.8 ppm_	yes
198	<5 ppm	2.2-5.8 ppm	yes
172	5-50	6.2 ррт	no
169	5-50	7.2 ppm	yes
171	5-50	7.2 ppm	yes
202	<5 ppm, 5-50	1.3-7.2 ppm	yes
163	5-50	8.7 ppm	yes
165	5-50	9 ppm	yes
168	5-50	9_ppm	yes
166	5-50	9.3 ppm	yes
164	5-50	11.9 ppm	yes
204	5-50	12.8 ppm	yes
253	5-50	13 ppm	yes
203	5-50	13.5 ppm	yes
258	5-50	15 ppm	yes
106	5-50	15-19 ppm	yes
161	5-50	15.3 ppm	yes
167	5-50	16.2 ppm	yes
247	5-50	18_ppm	yes

• :

4020-6

Revision Draft 1 October 1992

1

۳.

1

ľ

ļ

ļ

Ê

		Table 4 Continued	
Sample ID	Screening Test Results	GC Results	Does screening test agree with GC/MS determination?
148	>50	18-34 ppm	no
205	5-50	20 ppm	yes
162	5-50	20.4 ppm	yes
175	5-50	21.2 ppm	yes
176	5-50	21.6 ppm	yes
197	5-50	32 ppm	yes
243	5-50	32 ppm	yes
252	5-50	32 ppm	yes
178	5-50	43.7 ppm	yes
201	5-50	43 ppm	yes
254	5-50, >50	56 ppm	yes
238	>50	46-60 ppm	yes
248	5-50	44-60 ppm	yes
250	>50	68 ppm	yes
242	5-50	30-69 ppm	yes
256	>50	73 ppm	yes
249	>50	96 ppm	yes
245	>50	102 ppm	yes
241	5-50	154 ppm	no
246	>50	154 ppm	yes
261	>50	204 ppm	yes
240	>50	251 ppm	yes
267	>50	339 ppm	yes
239	>50	460 ppm	yes
104	>50	200-3772 ppm	yes
108	>50	531-1450 ppm	yes

* for PENTA RISc Test Kit (EnSys, Inc.)

;

• •

. . .

4020-7

Revision Draft 1 October 1992

ï

•

NC

PCB RISC® SOIL TEST SYSTEM

RAPID IMMUNOASSAY SCREEN

User's Guide

IMPORTANT NOTICE

This method correctly identifies 95% of samples that are PCB-free and those containing 1 ppm or greater of PCBs. A sample that develops less color than the standard is interpreted as positive. It contains PCBs. A sample that develops more color than the standard is interpreted as negative. It contains less than 1 ppm PCBs.

This test system should be used only under the supervision of a technically qualified individual who is capable of understanding any potential health and environmental risks of this product as identified in the product literature. The components must only be used for the analysis of soil samples for the presence of polychlorinated biphenyls. After use, the kits must be disposed of in accordance with applicable federal and local regulations.

TROUBLE SHOOTER GUIDE

Wash Step- lack of vigorous washing may result in false positives or negatives depending on whether the wash error was committed on standard or sample tubes. *Solution*: make sure that the operator washes four times vigorously.

Pipette Calibration- an out-of-calibration pipette may result in false positives or negatives depending on whether the amount is greater or less than the specified transfer volume. Solution: check the calibration at least daily and after any extreme mechanical shock (such as dropping). An indication that the pipette is out of calibration is if the gold barrel is loose and will turn. (When set on 30 ml there should be about 1/4 of an inch between the white plunger and the end of the clear pipette tip.)

Air bubbles in the pipette or diluter- the presence of air bubbles in the pipette tip when transfering extracts may result in false positives or negatives depending on whether the error was committed on standard or sample tubes. *Solution*: quickly examine the pipette tip each time an aliquot is withdrawn and go back to the source and take another aliquot to displace the bubble if neccessary. Bubbles in the diluter can be in the tip or plunger assembly.

Mixing- lack of thorough mixing, when instructed, can cause inconsistent results. Solution: observe the mixing times in the instructions and to mix with sufficient force to ensure that the liquid is mixed.

Timing- it important to follow the timing steps in the instructions carefully. The incubation step in the antibody tubes can vary a bit without harm to the test (\pm 5 minutes). The color development step timing is critical and should be no less than 2 minutes and no greater than 3 minutes.

Addition of Drops- it is important to carefully count the drops added in the color development steps. The addition of ±1 drop to the instructed 5 drops can cause variability in the results RIGHT AROUND THE DETECTION LEVELS OF INTEREST.

One drop less would result in a darker color (a less dilute solution) which could result in a false negative. One drop more could result in a lighter color (a more dilute solution) and result in a false positive.

Wiping the Tubes- wiping of the tubes should be done before they are read in the spectrophotometer because smudges and fingerprints on the tubes can give potentially false negative readings.

Mixing Lot #'s- never mix lots! Each kit's components are QC'd together for optimal performance and may give inaccurate results with the components from other kits, that are not of the same lot #. Also, the user must NEVER mix components from different types of kits (ex: Petro kit buffer tubes can't be used with a PAH kit).

Storage and Operating Temperatures- temperature requirements are very important and should be strictly adhered to. This information can be found in the kit User's Guide. Shelf-life- each kit label contains the kit expiration date. To achieve accurate results, kits must be used prior to expiration.

WORKSTATION SET-UP

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

READ BEFORE PROCEEDING

- Follow diagram below to setup workstation.
- Items that you will need that are not provided in the test kit include: a permanent marking pen, laboratory tissue (or paper towels), a liquid waste container, disposable gloves.
- Do not expose reagents to direct sunlight.
- Do not attempt to run more that 12 tubes, two of which must be Standard tubes.
- Operate test at temperatures greater than 4°C / 40°F and less than 32°C /90°F.
- See table on page 9 for sensitivity to various aroclors.

TEST PREPARATION

 Label amber vial "PCB Standard", and the current date, Standard is usable for up to 2 weeks from this date. Open PCB Standard ampule by slipping ampule cracker over top, and then breaking tip at scored neck. Transfer to empty amber vial with bulb pipette. Always cap tightly when finished using Standard.



PCB Standard Ampule Cracker Bulb Pipette

Amber Vial



PHASE 1 EXTRACTION & PREPARATION OF THE SAMPLE

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

WEIGH SAMPLE



- 1a Place unused weigh boat on pan balance.
- 1b Press ON/MEMORY button on pan balance. Balance will beep and display 0.0.
- 1c Weigh out 10 % 0.1 grams of soil.
- 1d If balance turns off prior to completing weighing, use empty weigh boat to retare, then continue.







 $\alpha \alpha$

Extraction jar

Wooden spatula

EXTRACT PCBS



- 2a Uncap extraction jar and place on a flat surface. Without contacting solvent puncture foil seal with ampule cracker or sharp object. Peel the remainder of the seal off extraction jar.
- 2b Using wooden spatula, transfer 10 grams of soil from weigh boat into extraction jar.
- 2c Recap extraction jar tightly and shake vigorously for one minute.
- 2d Allow to settle for one minute. Repeat steps 1a - 2c for each sample to be tested.

FILTER SAMPLE



- 3a Disassemble filtration plunger from filtration barrel.
- 3b Insert bulb pipette into top (liquid) layer in extraction jar and draw up sample. Transfer at least ½ bulb capacity into filtration barrel. Do not use more than one full bulb.
- 3c Press plunger firmly into barrel until adequate filtered sample is available (place on table and press if necessary). Repeat steps 3a - 3c for each sample to be tested.



READ TO AVOID COSTLY MISTAKES

SAMPLE DILUTION DIAGRAM

1. The sample dilution procedure on the next page is for standard detection levels. The following diagram represents the sample dilution procedure for all other detection levels

2. Your kit may include extra dilution ampules to reach high detection levels.

3. EVERY AMPULE PROVIDED MUST BE USED!

If there are any questions concerning the dilution procedure please call Technical Services before running the samples to help avoid costly mistakes.

1-800-242-7472 or 919-941-5509 (X144, 148 or 149).



NOTE: YOUR ORDER MAY INCLUDE ADDITIONAL AMPULES IN ORDER TO ACHIEVE YOUR TEST LEVELS. ALWAYS TRANSFER FILTERED SAMPLE TO THE DILUTION AMPULE LABELLED WITH THE LOWEST PPM LEVEL AND THEN TRANSFER FROM IT TO THE NEXT HIGHER LEVEL DILUTION AMPULE.

PHASE 2 SAMPLE & STANDARD PREPARATION

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

READ BEFORE PROCEEDING

- Tap glass buffer tubes vigorously on hard surface to release buffer trapped in cap.
- Label the glass buffer and plastic antibody coated tubes with a permament marking pen. Uncap glass buffer tubes.
- When using the mechanical pipette always withdraw and dispense below the liquid level.
- "Shake tubes" means to thoroughly mix the contents with special care not to spill or splash.

DILUTE SAMPLES AND STANDARDS







t ppm

10 000





PC8 Standard

- 4a Open 1 and 10 ppm* dilution ampules by slipping ampule cracker over top, and then breaking top at scored neck.
- 4b Withdraw 30 μL of filtered sample using mechanical pipette and dispense below the liquid level in "1 ppm" dilution ampule. Repeat to transfer a total of 60 μL; gently shake ampule from side to side for 5 seconds to mix thoroughly.
- 4c Withdraw 30 μL from the "1 ppm" dilution ampule using mechanical pipette and dispense below the liquid level in "10 ppm" dilution ampule. Repeat to transfer a total of 60 μL; gently shake ampule from side to side for 5 seconds to mix thoroughly.
- 4d Transfer 30 μL from each dilution ampule into a glass buffer tube. Always wipe tip after dispensing into buffer tube.
- 4e Assemble new pipette tip on mechanical pipette and transfer 30 μL from Standard vial into two glass buffer tubes. Immediately replace cap on PCB Standard vial.
- 4f Shake all glass buffer tubes for 5 seconds.



Glass buffer tubes

PC8 Standard vial

PHASE 3 THE IMMUNOASSAY

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

READ BEFORE PROCEEDING

 This phase of the procedure requires critical timing and care in handling the antibody coated tubes.

INCUBATION 1



- 5a Set timer for exactly 10 minutes.
- 5b Start timing and immediately pour solution from each glass buffer tube into appropriate antibody coated tube. Tap glass tube on antibody coated tube to remove solution.
- 5c Shake all tubes for 5 seconds.



PREPARE ENZYME DROPPER



- 6a Crush glass ampule contained within enzyme dropper by pressing tube against hard edge.
- 6b Mix enzyme by turning dropper end-over-end 5 times. Do not shake.
- 6c Remove seal from enzyme dropper.

Repeat steps 6a - 6c to prepare one enzyme dropper for every 5 antibody coated tubes.



Enzyme dropper



INCUBATION II

7a Dispense first drop from enzyme dropper into liquid waste container.

Note: before dispensing drops, tap capped tip on hard surface to avoid dispensing air bubbles.

- 7b After the 10 minute incubation, set timer for 5 minutes.
- 7c Immediately dispense 3 drops of enzyme into each antibody coated tube by squeezing the dropper.
- 7d Shake antibody coated tubes for 5 seconds.



PHASE 3 THE IMMUNOASSAY

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

READ BEFORE PROCEEDING WASH PROCEDURE

- An accurate test requires a virgorous wash accomplished by directing a strong stream into the antibody coated tubes.
- The wash solution is a harmless, dilute solution of detergent.

WASH



- 8a After the 5 minute incubation (a total of 15 minutes), empty antibody coated tubes into liquid waste container.
- 8b Wash antibody coated tubes by vigorously filling and emptying a total of 4 times.
- 8c Tap antibody coated tubes upside down on paper towels to remove excess liquid. Residual foam in the tubes will not interfere with test results.
- Note: When running up to 12 antibody coated tubes, tubes can be washed in two groups one group immediately following the other group.

READ BEFORE PROCEEDING

- Keep Substrate dropper bottles vertical and direct each drop to bottom of antibody coated tubes. Addition of more or less than 5 drops may give inaccurate results.
- This phase requires accurate timing.



Wash bottle

PHASE 3 THE IMMUNOASSAY READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

- 9a Add 5 drops of Substrate A (yellow cap) to each antibody coated tube.
- 9b Set timer for exactly 2 ½ minutes.
- 9c Start timer and immediately add 5 drops of Substrate B (green cap) to each antibody coated tube.
- 9d Shake all tubes for 5 seconds. Solution will turn blue in some or all antibody coated tubes.
- 9e Stop reaction at end of 2 ½ minutes by adding 5 drops of Stop Solution (red cap).

Note: Blue solution will turn yellow when Stop Solution is added.

STU

SELECT STANDARD

10b

9e

- 10a Wipe outside of all antibody coated tubes.
- 10b Place both Standard tubes in photometer.
- 10c Switch tubes until the photometer reading is negative or zero. Record reading.
 - If reading is greater than 0.3 in magnitude (+ or -), results are outside QC limits. Retest the sample(s).
- 10d Remove and discard tube in right well. The tube in the left well is the darker standard.



Substrate bottles (A, B, & Stop Solution)

i je s

PHASE 3 THE IMMUNOASSAY

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

MEASURE SAMPLE



11a Place 1 ppm tube in right well of photometer and record reading.If photometer reading is negative or zero, PCBs are present.

If photometer reading is positive, concentration of PCBs is less than 1 ppm.

11b Place 10 ppm tube in right well of photometer and record reading.

If photometer reading is negative or zero, PCBs are present.

If photometer reading is positive, concentration of PCBs is less than 10 ppm.

AROCLOR SENSITIVITY

Aroclor 👘	Lowest Detection Level
1248	1.0 ppm
1254	0.4 ppm
1260	0.4 ppm
1242	2.0 ppm
1232	4.0 ppm
1016	4.0 ppm

QUALITY CONTROL

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

System Description

Each PCB RISc Soil Test System contains enough material to perform four complete tests, each at two detection levels, if desired.

The PCB RISc Soil Test is divided into three phases. The instructions and notes should be reviewed before proceeding with each phase.

Hotline Assistance

If you need assistance or are missing necessary Test System materials, call toll free: 1-800-242-RISC (7472).

Validation and Warranty Information

Product claims are based on validation studies carried out under controlled conditions. Data has been collected in accordance with valid statistical methods and the product has undergone quality control tests of each manufactured lot.

PCB-free soil and soil containing 1 ppm or greater of PCBs were tested with the EnSys PCB RISc analytical method. The method correctly identified 95% of these samples. A sample that has developed less color than the standard is interpreted as positive. It contains PCBs. A sample that has developed more color than the standard is interpreted as negative. It contains less than 1 ppm PCBs.

The company does not guarantee that the results with the PCB RISc Soil Test System will always agree with instrument-based analytical laboratory methods. All analytical methods, both field and laboratory, need to be subject to the appropriate quality control procedures.

EnSys, Inc. warrants that this product conforms to the descriptions contained herein. No other warranties, whether expressed or implied, including warranties of merchantability and of fitness for a particular purpose shall apply to this product.

EnSys, Inc. neither assumes nor authorizes any representative... or other person to assume for it any obligation or liability other than such as is expressly set forth herein.

Under no circumstances shall EnSys, Inc. be liable for incidental or consequential damages resulting from the use or handling of this product.

How It Works

Standards, Samples, and color-change reagents are added to test tubes, coated with a chemical specific to PCBs. The concentration of PCBs in an unknown Sample is determined by comparing its color intensity with that of a Standard.

Note: PCB concentration is inversely proportional to color intensity; the lighter the color development of the sample, the higher the concentration of PCBs.

Quality Control

Standard precautions for maintaining quality control:

- Do not use reagents or test tubes from one Test System with reagents or test tubes from another Test System.
- Do not use the Test System after any portion has passed its expiration date.
- Do not attempt the test using more than 12 antibody coated tubes (two of which are Standards) at the same time.
- Do not exceed incubation periods prescribed by the specific steps.
- Always dispense correct number of drops and wash the number of times indicated in this guide.
- Use EPA Method 8080 or Code of Federal Regulations Title 40, Part 136, Appendix A, Method 680 to confirm results.

Storage and Handling Precautions

Wear protective gloves and eyewear.

- Store kit at room temperature and out of direct sunlight (less than 80°F).
- Keep aluminized pouch (containing unused antibody coated tubes) sealed when not in use.
- If Stop Solution or liquid from the extraction jar comes into contact with eyes, wash thoroughly with cold water and seek immediate medical attention.
- Standard Solution contains PCBs. Test samples may contain PCBs. Handle with care.

MECHANICAL PIPETTE

READ ALL INSTRUCTIONS BEFORE PROCEEDING WITH THE TEST

HOW TO OPERATE THE MECHANICAL PIPETTE

To Set Or Adjust Volume

Remove push-button cap and use it to loosen volume lock screw. Turn lower part of push-button to adjust volume up or down. Meter should read "030". Tighten volume lock screw and replace push-button cap.

To Assemble Pipette Tip

Slide larger mounting end of pipette tip onto end of pipette. Holding tip in place, press push-button until plunger rod enters pipette tip. Ensure no gap exists between piston and plunger rod (see illustration).

To Withdraw Sample

With tip mounted in position on pipette, press push-button to first stop and hold it.

Place tip at bottom of liquid sample and slowly release push-button to withdraw measured sample. Ensure that no bubbles exist in liquid portion of sample. If bubbles exist, dispense sample and re-withdraw sample.

To Dispense Sample '

Place tip into dispensing vessel (immersing end of the tip if vessel contains liquid) and slowly press push-button to first stop. (Do not push to second stop or tip will eject).

Remove tip from vessel and release push-button.

To Eject Tip

Press push-button to second stop. Tip is ejected.

For additional information regarding operation and use of pipette, please refer to your pipette manual.



On-Site Quality Control/Quality Assurance Recommendations EnSys RIS^{©®} Test System

Please read the following before proceeding with field testing!

Sampling

The result of your screening test is only as valid as the sample that was analyzed. Samples should be homogenized thoroughly to ensure that the 10 grams you remove for field testing is representative of the sample as a whole. All other applicable sample handling procedures should be followed, as well.

Prior to Testing Samples

Carefully follow the instructions in the User's Guide included with every test kit. This is the key element in obtaining accurate results. In addition, store your unused test kits at room temperature and do not use them past their expiration date (see label on each test kit).

Internal Test QC

Two standards are analyzed with each sample to provide internal test system quality control. With both standards inserted in the photometer, a valid test is indicated when the magnitude of the displayed number (irrespective of the sign, + or -) is less than the value given in the User's Guide. Test runs resulting in a greater number should be repeated to ensure valid conclusions.

QA/QC

The validity of field test results can be substantially enhanced by employing a modest, but effective QA/QC plan. EnSys recommends that you structure your QA/QC plan with the elements detailed below. These have been developed based on the data quality principles established by the US Environmental Protection Agency.

- A. Sample documentation
 - 1. Location, depth
 - 2. Time and date of collection and field analysis

B. Field analysis documentation - provide raw data, calibration, any calculations, and final results of field analysis for all samples screened (including QC samples)

C. Method calibration - this is an integral part of EnSys RIS[®] immunoassay tests; a duplicate calibration is performed for each set of samples tested (see the instructions in the User's Guide)

D. Method blank - field analyze the contents of an unused extraction jar

- E. Site-specific matrix background field analysis collect and field analyze uncontaminated sample from site matrix to document matrix effect
- F. Duplicate sample field analysis field analyze duplicate sample to document method repeatability; at least one of every 20 samples should be analyzed in duplicate
- G. Confirmation of field analysis provide confirmation of the quantitation of the analyte via an EPA-approved method different from the field method on at least 10% of the samples; choose at least two representative samples testing below the action level or lowest test level and at least two representative samples testing above the action level; provide chain of custody and documentation such as gas chromatograms, mass spectra, etc.
- H. Performance evaluation sample field analysis (optional, but strongly recommended) field analyze performance evaluation sample daily to document method/operator performance
- I. Matrix spike field analysis (optional) field analyze matrix spike to document matrix effect on analyte measurement

Further Questions?

EnSys technical support personnel are always prepared to discuss your quality needs to help you meet your data quality objectives.

	Data	a She	et for F	PCB F	RISC® S	Soil Test
Operator:		- <u></u>	Date:	Location:		
Sample ID	∆OD standards	OD sample	Interpretation	OD sample	Interpretation	Comments
<u> </u>						
		······································				
				,	•	
						,
					·	

.

ı

APPENDIX D

CHAIN OF CUSTODY FORM



CHAIN OF CUSTODY RECORD

SURVEY:				SAMP	LED BY:				
LOCATION:				ORGA	NIZATIC	N:			
STATION NUMBER	SAMPLE LOCATION	DATE Collected	TIME Collected	sample Matrix	COMP. OR GRAB	NO. Of Containers	ANAL REQU	ysis Ired	
									_
	· · · · · · · · · · · · · · · · · · ·								
lelinquished By:		DATE	TIME	Received	By:			DATE	TI
elinquished By:		DATE	TIME	Received	By:			DATE	TI
elinquished By:		DATE	TIME	Received	by Laborato	ry:		DATE	TI

COMMENTS:

METHOD OF SHIPMENT:

O'BRIEN & GERE LABORATORIES, INC. an O'Brien & Gere Company 5000 Brittonfield Parkway / P.O. Box 4942 / Syracuse, NY 13221 / (315) 437-0200 FAX (315) 463-7554

EXHIBIT A

HEALTH AND SAFETY PLAN

· · ·

.

HEALTH AND SAFETY PLAN

PCB/PAH/VOC Soils Excavation

Prepared for: Former Accurate Die Casting Site Fayetteville, NY

February 1995



O'Brien & Gere Technical Services, Inc. 5000 Brittonfield Pkwy. E. Syracuse, NY 13057 (315) 437-6400

. .



This Health and Safety Plan has been prepared to meet the requirements of 29 CFR 1926.65, Hazardous Waste Operatons and Emergency Response.

Signed,

OBG TECHNICAL SERVICES, INC.

ora

Jeffrey R. Parsons, CIH Corporate Safety Coordinator

TABLE OF CONTENTS

SECTION 1	Introduction		1	
	1.1	Corporate Protocols		
		1.1.1 Employee Tra	ining Assignments	
		1.1.2 Personal Prote	ective Equipment	
		1.1.3 Medical Surve	eillance	
		1.1.4 Air Monitorin	ıg	
		1.1.5 Site Control		
		1.1.6 Confined Space	ce Entry Procedures	
	1.2	Site-Specific Protocols		
SECTION 2	Background and Scope of Work			3
	2.1	Background		
	2.2	Scope of Work		
SECTION 3	Project Personnel		5	
SECTION 4	Hazard Evaluation and Hazard Control		6	
	4.1 Potential Health Hazards			
	4.2	Operations and Tasks to be Performed		
	4.3	Site-Specific Personal Protection		
SECTION 5	Air Monitoring and Action Levels		10	
	5.1	Wind Indicator		
	5.2	Air Monitoring Equipment to b	e used	
	5.3	Work Zone Air Monitoring Procedures and Action Levels		
	5.4	Community Air Monitoring Pla	n	
SECTION 6	Site Control		13	
	6.1	Potential Confined Space Areas	4	
	6.2	Work Areas		
SECTION 7	Emergency Response and Contingency Plan			14
	7.1	7.1 Directions to Hospital		
	7.2	Location of Nearest Available Telephone		
	7.3	Emergency Telephone Numbers		
	7.4	Safe Refuge		
	7.5	Spill Control		

.
TABLE OF CONTENTS cont.

EXHIBITS

Table 1	Summary of Potential Health Effects		
Table 2	Hazard Evaluation and Control		
Table 3	Work Zone Action Levels		
Table 4	Community Action Levels		
Figure 1	Site Map		
Figure 2	Site Plan		
Figure 3	Soil Sampling Locations/Analytical Results		
Figure 4	Hospital Route Map		

ATTACHMENTS

Attachment 1

HSP Compliance Agreement

APPENDICES

Appendix A	Health and Safety Training
Appendix B	EPA Levels of Personal Protective Equipment
Appendix C	Medical Surveillance
Appendix D	Airborne Materials Exposure
Appendix E	Hazardous Waste Operations
Appendix F	Respiratory Protection
Appendix G	Excavations
Appendix H	Hazardous Communication
Appendix I	Confined Spaces
	•

SECTION 1 - INTRODUCTION

The following Health and Safety Plan (HSP) has been developed in accordance with the requirements set forth in 29 CFR Part 1926.65, *Hazardous Waste Operations and Emergency Response*, and in conjunction with the O'Brien & Gere Technical Services (OBG Tech) Hazardous Waste Operations Standard Operating Procedures for OBG Technical Services' on-site activities. OBG Tech's Hazardous Waste Operations Standard Operating Procedures are in Section 4.05 of OBG Technical Services' Corporate Health and Safety Manual (see Appendix E). Section 4.05 as well as other pertinent Sections are appended to this HSP (Appendices A through I).

The purpose of this HSP is to assign site-specific responsibilities, site-specific training requirements, establish sitespecific personnel protective requirements, and to provide guidance for site-specific contingencies that may arise for OBG Technical Services' and subcontracted employees. In addition to this site-specific HSP, a copy of the OBG Technical Services Corporate Health and Safety Manual will be kept on site for reference purposes. All OBG Tech personnel and subcontractor personnel must be familiar with this site specific HSP and the OBG Tech Corporate Health and Safety Manual prior to participation in site activities.

The front portion of this HSP is designed to include site-specific information, so that it can be separated from the Corporate protocols in the Appendices and distributed and posted as necessary.

This HSP may be provided to interested parties for informational purposes, however, the HSP is specifically intended for the conduct of activities within the scope of work for OBG Technical Services.

1.1 Corporate Protocols

The following are the sections required by 29 CFR 1926.65 that can be found in the Corporate Health & Safety Manual and are appended to this HSP.

1.1.1 Employee Training Assignments

The employee training assignments are found in Section 2.02 of the Corporate Health and Safety Manual and are included in Appendix A. The description for the hazardous waste pre-entry briefing can be found in the hazardous waste protocols in Section 4.05 of the Corporate Health and Safety Manual. The hazardous waste protocols are included in Appendix E. There are no additions to these protocols.

1.1.2 Personal Protective Equipment (PPE)

The EPA definitions of personal protective equipment (PPE) and donning and decontamination procedures are found in Section 5.01 of the Corporate Health and Safety Manual, which is included in Appendix B. The site-specific personal protective equipment can be found in Section 4.

1.1.3 Medical Surveillance

The Corporate Medical Surveillance program can be found in Section 2.04 and is included in Appendix C. There are no site-specific additions to these protocols.

1.1.4 Air Monitoring

The corporate air monitoring program and calibration and maintenance procedures for air

monitoring equipment is found in Section 4.03 of the Corporate Health and Safety Manual and is attached in Appendix D. The site-specific air monitoring program is found in Section 5.

1.1.5 Site Control

The Corporate Site Control Program is found in the Hazardous Waste Protocols, attached in Appendix E. The site-specific site control program can be found in Section 6.

1.1.6 Confined Space Entry Procedures

The Corporate Confined Space Entry Procedures are found in Section 3.05 of the Corporate Health and Safety Manual and are attached in Appendix I. Site-specific confined space entry information is included in Section 6.1.

1.2 Site-Specific Protocols

The following are the sections required by 29 CFR 1910.120 that can be found in this HSP. Section 2 is the project background and Scope of Work. Personnel involved in the project can be found in Section 3. Section 4 includes the hazard evaluation and hazard control. Air monitoring procedures are included in Section 5. Section 6 outlines the site-specific site control measures. Section 7 includes the emergency response and spill control procedures.

SECTION 2 - BACKGROUND AND SCOPE OF WORK

2.1 Background

The former Accurate Die Casting Facility located in Fayetteville, New York is situated on a 32-acre parcel at 547 East Genesee Street (Figure 1). Refer to Figure 2 for an overview of site features. The facility was constructed in the 1950's and had been used as a die casting operation until 1988, when it was abandoned. ITT Commercial Finance Company ("ITT"), which possessed a mortgage on the real property, foreclosed on the premises and was the sole and successful bidder at the auction. Shortly thereafter, an environmental assessment for property transfer was conducted by Stearns & Wheler for ITT, the asset lender. As a result of the site assessment, it was concluded that potential for environmental contamination existed at the site and that further investigation would be necessary. Stearns & Wheler was subsequently retained to perform a Phase II assessment of the site, completing it in September 1990.

Upon review of the work completed to date at the former Accurate Die Casting facility, the NYSDEC has required additional investigation activities which have been completed in accordance with the Remedial Investigation/Feasibility Study Work Plan (Stearns & Wheler, May 1992). The Draft Report - Remedial Investigation (Stearns & Wheler, January 1993) concluded that:

- 1. Ground water quality, both overburden and bedrock, has been impacted by trichloroethene (TCE). The overburden TCE plume extends from near MW-3 north towards Bishop Brook.
- 2. TCE in soils is limited to the vicinity of MW-3, with concentrations ranging from 0.39 mg/kg at MW-10 to 7,500 mg/kg at MW-3. The highest TCE concentrations were observed at about 20 to 25 feet below grade, at the interface between the sand/gravel and till layers.
- 3. The PCB/PAH/VOC soils area contains polychlorinated biphenyls (PCBs), polynuclear aromatic hydrocarbons (PAHs), and volatile organic hydrocarbons (VOCs) above Technical Administrative Guidance Memorandum (TAGM) soil cleanup objectives for New York State.
- 4. The septic tank contains materials that if released could result in VOC and metal concentrations above TAGM soil cleanup levels.

The area containing TCE in subsurface soil outside the Northeast corner of the building was remediated in June 1994 as part of an Interim Remedial Measure (IRM). The objective of that IRM was to remediate the subsurface soils in the vicinity of MW-3 to reduce the potential for further migration of TCE from the source to the ground water.

2.02 Scope of Work

This site-specific Health and Safety Plan (HSP) addresses activities that are to be conducted in association with the characterization and removal of soils located in the PCB/PAH/VOC soils area and sludges contained in a septic tank. The activities include excavation of PCB/PAH/VOC contaminated soils, removal and disposal of septic tank sludge, and restoration of the two areas in an environmentally sound manner.

- 1. Five soil borings will be completed in the PCB/PAH/VOC soils areas located in the northern portion of the site immediately south of MW-4, as shown in Figure 3. Soil borings will be advanced to a depth of approximately six feet, and up to two samples from each boring will be taken for PCBs, PAHs, and VOCs.
- 2. Excavation of contaminated soil from the PCB/PAH/VOC soils area.
- 3. Removal of sludges from the septic tank
- 4. Excavation and disposal of the septic tank
- 5. Site restoration (backfilling, grading, etc.)
- 6. Equipment decontamination

SECTION 3 - PROJECT PERSONNEL

Section 4.05 of OBG Technical Services' Corporate Health and Safety Manual (see Appendix E) contains the specific responsibilities for the below personnel.

Project Officer:

.

Sr. Project Supervisor:

Corporate Safety Coordinator:

Health and Safety Specialist:

. . . 1 TT . 14

Field Supervisor/Site Safety and Health Coordinator:

To Be Determined

Terry L. Brown, PE

Jeffrey R. Parsons, CIH

Anthony J. Geiss

Kelly L. Fisher

SECTION 4 - HAZARD EVALUATION AND HAZARD CONTROL

4.1 Potential Health and Safety Hazards

Previous soil analysis in the PCB/PAH/VOC soils area has shown the following:

- 1. PCBs present at concentrations up to 2.6 milligrams of PCBs per kilogram of soil (mg/kg)
- 2. PAHs present at concentrations up to 113 mg/kg.
- 3. Dichloroethylene (VOCs) at concentrations up to 190 parts per million (ppm)

Septic tank sludge analysis detected the presence of several metals, but only zinc was elevated when compared to other soil samples taken from the site. No VOCs were detected.

Health and safety information for PCBs, PAHs, and VOCs (trichloroethylene and dichloroethylene) are included in Table 1. Because metals concentrations are low, they are not considered a hazard for this project.

The exposure pathways of concern are inhalation of materials volatilized from the soil and groundwater; inhalation of dusts containing the contaminants; accidental ingestion of soil and groundwater containing contaminants; and dermal exposure to soil and groundwater containing contaminants. Based upon anticipated site activities and prudent safety practices during site work, ingestion of site contaminants is unlikely. Due to the use of personal protective equipment, direct contact with site contaminants poses a low health hazard. Direct-reading air monitoring equipment will be used to monitor the airborne volatile organic compounds and dusts, thus minimizing the inhalation of VOCs and dust from the soil and sludge.

4.2 Operations and Tasks to be performed

OBG Technical Services employees are required to use personal protective equipment appropriate to their work task and potential exposures, as detailed in Section 5 of the OBG Technical Services' Corporate Health and Safety Manual. Specific sections related to this site include "EPA Levels of Protection for Hazardous Waste Sites" and "Respiratory Protection."

In addition, the soil amendment material(s) used during the soils aeration activities have the potential to cause health effects. All hazard information regarding the soil amendment material(s) should be reviewed by site personnel prior to initiating site treatment activities. The SSHC is responsible for administering the Hazard Communication program (Appendix H), including informing all employees about the amendment material(s) and the location of the material safety data sheet (MSDS) for the amendment material(s), as well as all other MSDS's for materials brought on-site.

The levels of protection assigned to each activity below are based on available information and represent an estimate of exposure potential and appropriate protective equipment. The SSHC may revise these levels in accordance with Section 4.05 of OBG Technical Services' Corporate Health & Safety Manual.

When dry and/or dusty conditions are observed by the SSHC or dust levels are above the action level specified in Section 5.3, a water spray will be applied to the contaminated soils. In addition, excavated materials will be covered.

Chemical	Location	PEL	IDLH	Characteristics	Routes of Exposure	Symptoms of Exposure
Trichloroethylene	Soil Groundwater	100 ppm TWA 200 ppm Ceiling (1989 - 50 ppm TWA 200 ppm STEL)	1000 ppm	Colorless liquid with a chloroform-like odor. Lowest reported odor threshold is 0.2 ppm.	Inhalation Ingestion Contact	Headache, vertigo; visual disturbances, tremors, somnolence, nausea, vomiting; irritated eyes; dermatitis; cardiac arrhythmia, paresthesia; potential carcinogen.
Dichloroethylene	Soil	200 ppm TWA	4000 ppm	Colorless liquid with a slightly acrid, chloroform-like odor. Lowest reported odor threshold is 0.1 ppm	Inhalation Ingestion Contact	Irritation of the eyes and respiratory tract. Central nervous system depression (i.e. alcohol-like effects)
Polychlorinated Biphenyls (PCBs)	Soil	1 mg/m³	10 mg/m³	Coloriess to light colored, viscous liquid with a mild hydrocarbon odor.	Inhalation Absorption Ingestion Contact	Eye irritation, Chloracne, liver damage, carcinogen
Polynuclear Aromatic Hydrocarbons (PAHs) ²	Soil	0.2 mg/m ³	700 mg/m³	Black or dark brown amorphous residue.	Inhalation Contact	Skin dermatitis, bronchitis, kidney damage, carcinogen

TABLE 1 - SUMMARY OF POTENTIAL HEALTH EFFECTS

An appeals court decision forced OSHA to revert the PELs published in 1989 to the PELs published in 1971. The PELs listed here reflect the 1971 PELs with 1989 PELs listed in parentheses.

1

2

Exposure limits for PAHs are referenced by OSHA as coal tar pitch volatiles, benzene soluble fraction. This includes the following compounds: anthracene, benzo-a-pyrene, phenanthrene, acridine, chrysene, and pyrene.

7

Operation/Task	Hazards	Hazard Control			
MOBILIZATION General Site Workers performing non- intrusive activities including mobilization, equipment set-up	 Safety/Physical Hazards Noise Use of Heavy Equipment Overhead Utilities Heat or cold stress Exposure to toxic plants, insects, and reptiles 	 Level D PPE Hearing protection when necessary Eye protection when necessary Review hazard recognition and prevention at toolbox safety meeting Stay 20 feet from overhead utility lines 			
EXCAVATION AND HA	NDLING OF CONTAMINATED SOILS				
Operator	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Noise Use of heavy equipment Overhead utilities Underground utilities Exposure to excavations Heat or cold stress Confined space entry for excavations greater than 4 feet deep. 	 Modified Level D PPE (includes skin protection - coveralls, gloves, boots, eye protection) Hearing Protection is required while operating equipment. Use trained operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety rules and inspection forms found in Appendix G Call utilities before breaking ground Review hand signals with laborer Follow confined space entry procedure. Air monitor as detailed in Section 5 			
Laborer	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Safety/Physical Hazards Slip/trip/fall hazards Exposure to heavy equipment Noise Exposure to toxic plants, insects, and reptiles Exposure to excavations Heat or cold stress Confined space entry for excavations greater than 4 feet. 	 Modified Level D PPE Hearing Protection Review hand signals with operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety procedures (see Appendix G) when entering an excavation Follow confined space entry procedure Air monitor as detailed in Section 5 			
REMOVAL OF SOIL AND SLUDGES FROM SEPTIC TANK					
Laborer	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Noise Slip/trip/fall hazards Exposure to heavy equipment Exposure to toxic plants, flying insects, and reptiles Exposure to excavations Heat or cold stress Confined space entry 	 Modified Level D PPE Hearing protection Review hand signals with operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety procedures (see Appendix G) Follow confined space entry procedures found in Appendix I Air monitor as detailed in Section 5 			

TABLE 2 - HAZARD EVALUATION AND CONTROL

i.

Operation/Task	Hazards	Hazard Control
SOIL BORING & SAM	PLING CONTAMINATED SOIL	
Sampler	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Exposure to heavy equipment Noise Pinch points Heat or cold stress 	 Modified Level D PPE Hearing protection Review hazard recognition and prevention at toolbox safety meeting Collect sample using excavator bucket Air monitor as detailed in Section 5
DECONTAMINATION	ACTIVITIES	· · · · · · · · · · · · · · · · · · ·
Decontamination of Equipment	 Contact with contaminated tools and equipment Safety/Physical Hazards Noise Exposure to Heavy Equipment Heat or cold stress Burns from steam decon equip. 	 Level C PPE Hearing protection Review SOPs/operating instructions for decontamination equipment

4.3 Site-specific Personal Protection

The site-specific personal protective equipment is to be used in accordance with the protocols in Section 5.01 of the Corporate Health and Safety Manual. These protocols include the definitions of the EPA levels of protection for Level B, Level C and Level D to be used on-site, as well as donning and decontamination procedures. The site-specific personal protective equipment is as follows:

Coveralls:

Poly-coated Tyvek[®] when exposed to dirt/dusts Saranex[®]-coated Tyvek[®] when exposed to site waters/liquids

Outer Gloves:

Viton Gloves (Note: Work gloves dedicated to the exclusion zone may be used over the viton gloves)

Respirator:

Level C: Full-face air purifying respirator with organic vapor cartridges and high efficiency particulate (HEPA) filters

Level B: Positive pressure, full-facepiece Self-Contained Breathing Apparatus (SCBA) or Positive pressure, supplied-air respirator with escape SCBA

Hard hats, steel-toed boots, and safety glasses are mandatory on-site.

SECTION 5 - AIR MONITORING AND ACTION LEVELS

Air monitoring is to be performed in accordance with Appendix D (Section 4.03 of the Corporate Health and Safety Manual - Airborne Materials Exposure) and Appendix E (Section 4.05 of the Corporate Health and Safety Manual - Hazardous Waste Operations). Presented below is the site-specific information.

5.1 Wind Indicator

Wind direction will be monitored each day of on-site activities by using a portable wind indicator.

5.2 Air Monitoring Equipment to be used

Photoionization Detector (PID) with 10.2 eV lamp

Combustible gas/oxygen meter

Direct-Reading Dust Monitor

5.3 Work Zone Air Monitoring Procedures and Action Levels

The following describes the methods and parameters to be used on-site for PPE upgrades and work cessation. The action levels are based upon total organic vapors detected by a photoionization detector (PID) calibrated to isobutylene. PID readings will be adjusted in accordance with manufacturer's guidelines to read in ppm of dichloroethylene.

Туре	Frequency	Action Level	Action
Volatile Organic Compounds (VOCs) (Using PID)1. Initially when a new task begins and hourly thereafter 2. When obvious contamination is encountered 3. Prior to and continuously during entry into excavations over 4 feet	1. Initially when a new task begins and hourly thereafter	25 ppm above background in breathing zone for 5 minutes	Increase to Level C PPE
	200 ppm above background in breathing zone for 5 minutes	 Notify SSHC Increase to Level B PPE Consider engineering controls. 	
	500 ppm above background in breathing zone for 5 minute	 Stop work - consult SSHC; use engineering controls (e.g. ventilation) to decrease concentrations 	
Combustible Gases (Lower Explosive Limit - LEL)	Prior to and continuously during entry into an excavation over 4 feet	10 %	Stop work - increase ventilation to bring below 10 % LEL

TABLE 3 - WORK ZONE ACTION LEVELS

Туре	Frequency	Action Level	Action
Oxygen	Prior to and continuously during entry into an excavation over 4 feet	19.5 %	Stop work - increase ventilation to bring above 19.5 % oxygen

5.4 Community Air Monitoring Plan

Real-time monitoring for volatile compounds and particulate levels at the perimeter of the work area will be performed.

Туре	Frequency	Action Level	Action
VOCs (using PID)	Initially when a new task begins, every fifteen minutes for the first hour and every two hours thereafter	5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area	 Stop work until emission controls (covering excavated areas) are in place and the concentrations detected by the PID are reduced to below 5 ppm at the downwind perimeter of the work area. Air monitor as directed below. When work continues monitoring at the downwind perimeter of the work area will be performed at least every one-half hour.
	Monitoring will be performed within 20 feet of the perimeter of the nearest residential or commercial structure when 5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area	1 ppm above background sustained for one minute	 Stop work and continue to monitor. Contact Corporate Safety Coordinator for appropriate action. Work may continue when concentrations detected by the PID are reduced to below 1 ppm. Monitor with pumps and charcoal tubes following NIOSH 1022 (see below)
VOCs (pumps and charcoal tubes following NIOSH 1022)	 Daily for the first week of excavation actives in contaminated area - one upwind and two downwind When 5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area is detected using PID 	1/2 of PEL after subtracting upwind from downwind	The SSHC and Project Supervisor will re-evaluate vapor suppression techniques
Particulates	Continuously downwind during excavation of contaminated materials integrated over 15 minute periods	100 ug/m³ 150 ug/m³	 Measure the upwind background level. Implement dust suppression techniques (e.g. using water sprays, covering excavated materials and areas) Notify the SSHC STOP WORK Measure upwind background level immediately Implement additional dust suppression techniques

TABLE 4 - COMMUNITY ACTION LEVELS

12

SECTION 6 - SITE CONTROL

The elements of site control include site work zones (exclusion zone, contamination reduction zone, and support zone), the "buddy" system, site communication, and site security. Protocols for site control can be found in Section 4.04 of the Corporate Health and Safety Manual (see Appendix E).

Only approved, trained personnel will be allowed to enter the exclusion zone and contamination reduction zone. Personnel must understand this HSP and the potential on-site hazards before being allowed on site.

6.1 Potential Confined Space Areas

The excavations (greater than 4 feet) do pose a potential confined space entry area. When entry occurs then the confined space procedures in the Corporate Health & Safety Manual will be followed (section 3.05 - see Appendix I). Specifically, the following precautions will apply:

- 1. Confined Space Entry permit will be filled out by the SSHC.
- 2. Air monitoring will be performed continuously for oxygen deficiency and LEL.
- 3. Personal protective equipment as outlined previously will be used.
- 4. Barriers and signs will be used to protect opening.
- 5. Provisions will be made for confined space rescue.

Specific procedures, such as coordination with Owner, training, entrant responsibilities. attendant responsibilities, supervisor/foreman responsibilities, and atmospheric testing can be found in Section 3.05 of the Corporate Health & Safety Manual (see Appendix I).

In addition, the safety requirements for excavations will be followed, as outlined in Section 10.06 of the Corporate Health and Safety Manual (See Appendix G).

6.2 Work Area

A map of the work areas depicting the exclusion and contamination reduction zones will be prepared and submitted with the Excavation plan and performance specifications.

SECTION 7 - EMERGENCY RESPONSE AND CONTINGENCY PLAN

Sections 3.01, Medical Services and First Aid, 3.03, Emergency Action, and 4.05, Hazardous Waste Operations, of the Corporate Health and Safety Manual provide guidance on emergency response. This section provides the site-specific emergency response information.

7.1 Directions to Hospital - To be posted in all site trailers (See Figure 4).

Genesee Street West to I-481 North. I-481 to I-690 West I-690 West to Townsend Street Left onto Townsend Street Turn left onto Genesee Street Turn right onto Irving Ave The hospital on right.

7.2 Location of Nearest Available Telephone

The telephone will be located in the site field office.

7.3 Emergency Telephone numbers - To be posted in site office

Police	911
Crouse Irving Memorial Hospital	470- 7111
Fire Department	911 .
OBG Technical Services, Inc.	437-6400

7.4 Safe Refuge

The field office will serve as point of safe refuge. An air horn will be kept in the field office. If the air horn is sounded, all on-site personnel must return to the office.

7.5 Spill Control

A major spill is not anticipated at the site. Should a spill of any type occur, the personnel present at the spill should report it immediately to the SSHC and/or the OBG Tech emergency contacts, who will make arrangements for the proper clean-up of the spill and contact the appropriate local emergency groups and regulatory agencies. On-site personnel should immediately secure the area to prevent unauthorized entry into the spill area. On-site personnel must evaluate the extent of the hazard(s) and if available, utilize engineering controls and proper safety equipment to contain the spill until response personnel are on site. The emergency response personnel will be contacted immediately by SSHC in the event that the spill can not be immediately contained by on-site materials.







لـ

~~**]** .



FIGURE 3





HOSPITAL ROUTE MAP

FIGURE 4

Health and Safety Plan Compliance

-			
	·		
=			_
	_		
TEC	HNICA	I SERV	JICES

Project Name: _

This is to certify that I have read, fully understand, and agree to comply fully with the attached Health and Safety Plan furnished to me by OBG TECHNICAL SERVICES, INC.

NAME	SIGNATURE	COMPANY	DATE
·		· · · · · · · · · · · · · · · · · · ·	
······································			
	······	·	•
	· · · · · · · · · · · · · · · · · · ·		
·			
			· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·		
		······································	
<u> </u>			

Appendix A

SECTION: 2.02 SUPERSEDES: 5/28/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 1 of 2

2.02 Health and Safety Training

A. Introduction

All employees must recognize and understand the potential hazards to health and safety associated with the tasks that they perform. The objectives of safety training programs are:

- 1. To make employees aware of the potential hazards they may encounter;
- 2. To provide employees with the knowledge and skill necessary to perform the work with minimai risk to employee health and sarety;
- 3. To make employees aware of the purpose and limitations of equipment; and
- 4. To provide employees with information to avoid or procedures to follow in the event of an emergency.

B. <u>New Employee</u>

All new employees to OBG Technical Services, Inc. will receive training that will provide an introduction to the Safery Program, employees rights and responsibilities on safety, electrical safety, minimizing back strain on the job, Hazard Communication, and substance abuse-free workplace policy.

C. Job specific

Field Supervisors/Foreman will be responsible for training new employees on site-specific information, such as safe work practices and procedures to follow in the event of an emergency.

D. <u>Haz Com</u>

All OBG Tech employees will receive training on the OSHA Hazardous Communication Standard ("Right-to-Know"). Each employee receives a copy of the written program with their job offer letter. However, formal training is conducted to verify that all employees have been made aware of the plan, its requirements, and be informed of their right to receive information about hazardous materials.

Each supervisor is responsible for training employees on new materials introduced on specific jobs.

E. Other Safety Training

1. "Toolbox" Safety Meetings

There are a number of OSHA regulations which require training. Many of these are specific to jobs and should be documented by "tool box" meetings. These include: medical services and emergency action, fire protection, accident prevention, and poweroperated and woodworking hand tools, stairways and ladders, guardrails and handrails, welding and cutting, scaffolds, concrete and masonry construction, and demolition.

Supervisors/Foreman review the applicable topics in weekly "tool box" safety meetings

SECTION: 2.02 SUPERSEDES: 5/23/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 2 of 2

on every project. Names of employees attending the meeting and topics covered are forwarded to the Safety Coordinator for recordkeeping purposes.

2. Quarterly Staff and Safety Meetings

Starf and safety meetings are held quarterly for all employees. These meetings cover starf and personnel issues, as well as selected topics related to safety.

3. Haz Waste Training

Employees who are assigned to perform duties on hazardous waste sites will receive the-OSHA initial 40-hour health and safety training prior to on-site activities at a hazardous waste site. Applicable employees will receive yearly 8-hour refresher courses. In addition, Foreman, Supervisors, and Project Managers will receive the OSHA 8-hour health and safety supervisors training.

4. Other Safety Training

Periodically, employees will be sent to safety training courses relevant to their duties, such as confined space entry.

F. <u>Records</u>

.

The Safety Coordinator will maintain safety training records on each individual.



.

SECTION 5.01 SUPERSEDES: None DATE: May 14, 1992 REVIEW: February 12,1993 PAGE: 1 of 4

Section 5.01 EPA Levels of Personal Protective Equipment

A. Introduction

Due to the nature of the business of OBG Tech, employees may be required to work on hazardous waste sites. Use of personal protective equipment (PPE) is required by OSFLA. EPA has defined four Levels of Protection: Levels A. B. C. and D. These levels are defined below and may be used as a starting point for PPE on sites, but must be tailored to the specific situation.

B. Levels of Personal Protection

Level A

Level A protection provides the highest available level of respiratory, skin and eye protection. The material in the suit, gloves, and boots must be compatible with the substances involved.

- 1. Positive pressure, full-facepiece SCBA or positive pressure supplied-air respirator with escape SCBA, approved by the National Institute for Occupational Safety and Health (NIOSH).
- 2. Fully-encapsulating, chemical-resistant suit.
- 3. Outer chemical-resistant gloves.
- Inner chemical-resistant gloves.
- 5. Chemical-resistant safety boots.
- 6. Two-way communications.

Options:

- 1. Cooling unit.
- 2. Coveralls.
- 3. Long-underwear.
- 4. Other PPE as required in Section 5.02 through 5.09 of this Health and Safety Manual.

Level B

Level B protection provides the highest level of respiratory protection but less skin protection than Level A. Use only when airborne chemicals are not hazardous to the skin or not capable of being absorbed through the intact skin.

1. Positive-pressure, full-facepiece SCBA or positive-pressure supplied-air respirator

SECTION 5.01 SUPERSEDES: None DATE: May 14, 1992 REVIEW: February 12,1993 PAGE: 2 of 4

with escape SCBA (NIOSH approved).

- 2. Chemical-resistant suit.
- 3. Inner and outer chemical-resistant gloves.
- 4. Chemical-resistant safety boots.
- 5. Two-way communications.

Options:

- 1. Cooling unit.
- 2. Coverails.
- 3. Long-underwear.
- 4. Other PPE as required in Section 5.02 through 5.09 of this Health and Safety Manual.

Level C

Level C protection provides less skin protection as Level A and a lower level of respiratory protection than Levels A and B.

- 1. Full-facepiece, air-purifying, canister-equipped respirator (NIOSH approved).
- 2. Chemical-resistant suit.
- 3. Inner and outer chemical-resistant gloves.
- 4. Chemical-resistant safety boots or chemical-resistant boot covers.
- 5. Two-way communications.

Options:

- L Cooling unit.
- 2. Coveralls.
- 3. Long underwear.
- 4. Other PPE as required in Sections 5.02 through 5.09 of this Health and Safety Manual.

SECTION 5.01 SUPERSEDES: None DATE: May 14, 1992 REVIEW: February 12.1993 PAGE: 3 of 4

Levei D

Level D protection provides minimal skin protection and no respiratory protection.

- 1. Coveralls or long pants and long-sleeved shirt.
- 2. Safety boots.

Options:

1. Other PPE as required in Sections 5.02 through 5.09 of this Health and Safety Manual.

C. <u>PPE donning procedures</u>

- Inspect the PPE before domaing with the procedures outlined in Sections 5.02 through 5.09.
 Make adjustments to hard hat to fit user's head, if necessary.
- 3. Standing or sitting, step into legs of the suit; evaluate proper placement of feet within the suit; then gather suit and pull sleeves over arms and secure suit front.
- 4. Put on chemical-resistant safety boots over the feet of the suit. Tape the leg cuff over the tops of the boots.
- 5. Put on air tanks and harness assembly of the SCBA (if applicable). Don the facepiece or respirator and adjust it to be secure, but comfortable. Do not connect the breathing hose of the SCBA. Open valve on the air tank (if applicable).
- 6. Perform negative and positive respirator facepiece seal test procedures.
- 7. Put on inner gloves.
- 8. Put on other PPE (e.g hard hat, hearing protectors).
- 9. Raise hood over head carefully so that the face seal of the respirator is not disrupted.
- 10. Connect the breathing hose while opening the main valve (if applicable).
- 11. Have assistant observe the wearer for a period of time to evaluate whether the wearer is comfortable, stable, and that the PPE is functioning properly.

D. <u>PPE Decontamination Procedures</u>

Station 1: Equipment Drop

Deposit equipment used on-site (tools, sampling devices, clipboards, etc) onto plastic drop cloths. During hot weather a cool down station may be set up within this area.

SECTION 5.01 SUPERSEDES: None DATE: June 3, 1992 REVIEW: February 12,1993 PAGE: 4 of 4

Station 2: Outer Garment, Boots, and Gloves Wash and Rinse

Scrub outer boots, outer gloves and chemical-resistant splash suit with decontamination solution or detergent water. Rinse off using large amounts of water.

Station 3: Outer Boot and Glove Removal

Remove outer boots and gloves. Deposit in appropriate area.

Station 4: Tank, Canister, or Mask Change

If worker leaves exclusion zone to change air tank, canister, filters, or mask, this is the last step in the decontamination procedure. Worker's air tank, canister, filters, or mask is exchanged, new outer gloves and boots are donned, joints taped, and worker returns to dury.

Station 5: Boot, Gioves, and Outer Garment Removal

Boots, chemical-resistant splash suit, inner gloves removed and deposited in separate containers lined with plastic.

Station 6: SCBA or Face-piece Removal

SCBA backpack and/or facepiece is removed. Avoid touching face with finger. SCBA deposited on plastic sheets.

Station 7: Field Wash

Hands and face are thoroughly washed. Shower if appropriate.

E. Upgrading/Downgrading of PPE Levels

The PPE used and the overall level of protection should be reevaluated periodically as the amount of information on the site increases, and as workers as required to perform different tasks.

Reasons to upgrade the level of PPE may include:

- Known or suspected presence of skin contact hazards.
- Occurance or likely occurance of gas or vapor emission.
- Change in work task that will increase contact or potential contact with hazardous materials.

Appendix C

.

SECTION: 2.04 SUPERSEDES: 5/18/92 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 1 of 2

2.04 Medical Surveillance

A. Introduction

OBG Technical Services provides medical and environmental surveillance of the health of those individuals exposed to materials or work conditions which are associated with adverse health conditions or who participate in the Health & Exercise Program.

B. <u>Medical Examinations</u>

1. Warehouse Employees

The examination for warehouse employees, for the purposes of medical examinations, includes any employee who works on regular basis at the warehouse, and is not covered under another examination (e.g. hazardous waste examination). This is the examination for DOT driver certification. This examination is required every two years. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes the Medical Questionnaire, a Medical Examination. Sudiology Test, DOT Driver Certification, and testing for complete bilod count and chemistry profile.

2. Field Employees

The examination for field employees, for the purposes of medical examinations, is any employee who visits or works at a job site on a regular basis. Field employees, for the purpose of medical examination, includes some Supervisors, Foreman, and Laborers. This examinations is required by OSHA in 29 CFR 1910.120, the Hazardous Waste Operations regulations. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes the OSHA required Medical Questionnaire, Respirator Suitability Form, a Medical Examination, Audiology Test, Pulmonary Function Test, DOT Driver Certification, and testing for complete blood count and chemistry profile, and phenols (quantitative).

3. Supervisor/Managerment Employees

A Supervisor/Management employee is any employee who periodically visits a job site (less than 30 days a year). Supervisors/Management employees, for the purpose of medical examinations, includes Officers, Project Managers, and all levels of Supervisor not covered elsewhere (i.e. hazardous waste examination). This examination is required at a frequency based on the individuals age. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes the Medical Questionaire, a Medical Examination, Audiology Test, and testing for complete blood count, and chemistry profile.

4. Office Personnel

SECTION: 2.04 SUPERSEDES: 5/18/92 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 2 of 2

Office personnel, for the purposes of medical examinations, include administrative and office personnel. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes a Medical Examination, and testing for complete blood count and chemistry profile.

5. Drug Screen

The drug screen is mandatory for all newly hire, "regular status" employees. Temporary status employees may also require a drug screen, based upon a variety of factors. Please refer to OBG Technical Services' "Substance Abuse-Free Workplace" policy for more information.

C. <u>Frequency</u>

1. Baseline examinations

Individuals who are assigned temporarily or permanently to field work will receive a baseline examination prior to job assignment.

2. Periodic Examinations

Individuals who are assigned temporarily or permanently to field work will receive periodic examinations yearly.

3. Termination Examinations

Field employees permanently leaving the company will receive an exit examination.

4. Possible Exposure Examinations

As soon as possible upon notification by an employee that the employee has developed signs or symptoms indicating possible overexposure to hazardous substances or health hazards, or that employee has been injured or exposed above the permissible exposure limits in an emergency situation, that employee will be required to receive medical attention.

D. Examination Results

OBG Tech must receive a letter from the attending physician stating the parameters of the examination and whether or not the individual is able to work with or without restriction. This letter will be filed by the Office Administrator and a copy is distributed to the employee.

.

Appendix D

.

SECTION: 4.03 SUPERSEDES:5/14/92 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 1 of 2

Section 4.03 Airborne Materials Exposure

A. Introduction

OSHA. in 29 CFR 1910.1000, specifies that an employee's exposure to substances listed in Table Z-1-A. Z-2, or Z-3 shall be limited in accordance with the requirements of the section. OSHA, in 29 CFR 1926.55, specifies that an employee's exposure to those specified in the American Conference of Governmental Industrial Hygienists (ACGIH) Threshold Limit Values (TLVs) for Airborne Contaminants for 1970 shall be avoided. Where the requirements of the two standards overlap, the most stringent of the two will be enforced.

B. <u>Administrative and Engineering Controls</u>

To achieve compliance with airborne materials exposure, administrative or engineering controls must first be implemented to achieve compliance.

C. <u>Respirator Use</u>

When effective engineering or administrative controls are not feasible to control airborne exposures, or while they are being instituted, appropriate respirators will be used. Training, selection, issuance, and standard operating procedures are outlined in Section 5.06 of this Health and Safety Manual.

D. <u>Action Specific Hazards</u>

As specified in 29 CFR 1910 Subpart Z, action specific hazards, i.e. lead, asbestos, and formaldehyde, have certain requirements. Such material will be monitored as directed under specific regulatory requirements.

E. Monitoring

1. Monitoring for IDLH and other dangerous conditions

During confined space entry and other situations where the quality of the air is unknown, air monitoring will be conducted for combustible or oxygen deficient atmospheres, as well as for volatile organics. Calibration and maintenance procedures for direct-reading equipment are included as Appendix A.

2. General On-site Monitoring

Site conditions may change during site activities. Air monitoring will be conducted when:

- a. Work begins on a different portion of the site.
- b. A different type of operation is initiated.
- c. Employees are working in obvious contamination.
- 3. Personal Monitoring

SECTION: 4.03 SUPERSEDES: 5/14/92 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 2 of 2

Personal monitoring will be performed on high-risk workers who are closest to the source of contaminant generation. This approach is based upon the rationale that the probability of significant exposure varies directly with distance from the source. If workers who are closest to the source are not significantly exposed, then all other workers are, presumably, also not significantly exposed and probably do not need to be monitored.

However, any employee may request personal monitoring be performed if a potential risk exists.

Personal air monitoring methodology will be determined by the Safety Coordinator using the appropriate National Institute for Occupational Safety and Health (NIOSH) or other appropriate methods.

F. Notification

Employees will be notified of the results of the monitoring performed in accordance with the procedures specified in Section 2.05.

Appendix A

Calibration and Maintenance Procedures

A. Loan of Equipment

1. Sign-out Procedures

The health and safety monitoring equipment can be obtained from the Warehouse Supervisor or the Safety Coordinator. The information that will be required prior to a piece of equipment leaving the warhouse equipment room will be:

- Serial Number of equipment
- Name of Person responsible for equipment
- Job Number and Location
- Expected Return Date

2. Equipment Return

Never use carrying cases as shipping cases.

Should a piece of equipment be damaged or in need of maintenance, it should be tagged with appropriate instructions.

Upon return, the piece of equipment will be checked to see if it is in the same condition as it left in. If it is broken, an assessment will be made to determine whether it was broken by mishandling or by normal wear and tear. It will also be determined whether the repair can be performed in-house or must be sent to the manufacturer. If the equipment was determined to be broken due to mishandling and the equipment must be sent to the manufacturer, the Project Manager in charge will be contacted to determine if the project can absorb the cost of the repair.

B. Photoionization Detectors (PIDs)

1. Calibration Procedures

This is to be performed, at a minimum, on a daily basis or whenever the detector is used.

- a. Turn on photoionization detector (PID) and allow to warm un.
- b. Zero instrument in "clean" air. (Note: "Clean" air refers to upwind of a waste site).
- c. Connect Span Gas cylinder to PID with a piece of clean tubing.
- d. Open the valve on the cylinder until a steady reading is obtained.
- e. Adjust the SPAN control, if necessary, until reading is the same as the Span Gas concentration.
- f. Close the valve on the Span Gas cylinder. Disconnect the cylinder from the PID.
- g. Sample again in "clean" air. Adjust the zero, if necessary.
- 2. Maintenance Procedures

Keeping the PID in top operating shape means charging the battery, cleaning the lamp window, and replacing filters. The exterior of the PID can be wiped clean with a damp cloth and mild detergent, if necessary.

B. Combustible Gas and Oxygen Meters (CGI/O. Meters)

1. Calibration Procedures

Appendix A

Calibration and Maintenance Procedures

- a. Compustible Gas Meter
 - i. Turn on the instrument and allow to warm up.
 - ii. Zero % LEL meter in "clean" air.
 - iii. Connect calibration gas with clean piece of tubing to meter.
 - iv. Cpen the valve on the cylinder until a stendy reading is obtained.
 - v. Adjust the combustible gas calibration control, if zecessary, until reading is the same as the calibration gas concentration.
 - vi. Close the valve on the calibration gas cylinder. Disconnect the cylinder from the meter.
 - vii. Sample again in "clem" air. Adjust the zero, if necessary.
- b. Oxygen Meter
 - i. Turn on the instrument in "clean" air and allow to warm up.
 - ii. If the % oxygen stabilizes at a value other than 20.3%, adjust the oxygen calibration control until the reading is 20.3%.
- 2. Maintenance Procedures

Keeping the meter in top operating shape means charging the battery and replacing worm parts. The exterior of the meter can be wiped clean with a damp cloth and mild detergent, if necessary.

- C. Flame Ionization Detector (FID)
 - 1. Calibration Procedures

This is to be performed on a daily basis or, at a minimum, whenever the instrument is used.

- Place instrument in normal operation with CALIBRATE Switch set to X10). Allow 20
 minutes for warm up and stabilization.
- b. Introduce a methane sample of known concentration (between 90 and 100 ppm) and adjust the GAS SELECT dial so the meter reading corresponds to the known sample.

Periodically (once a month), the FID should be calibrated over the full range using the trimpots. Please refer to the Operations and Maintenance Manual for further instructions.

- 2. Maintenance Procedures
 - a. The battery pack can be recharged overnight; however care should be taken to completely discharge the battery prior to recharging.
 - b. A supply of analytical grade hydrogen is needed to recharge the 75 cc hydrogen tank in the OVA.
- 3. Shipping

Compressed hydrogen cannot be shipped by air. A supply of compressed hydrogen must be available locally.

D. <u>Air Sampling Pumps</u>

1. Calibration Procedures

Calibrate personal air sampling pumps before and after each day of use using a primary calibrator. If it is not feasible to use a primary calibrator, a precision rotameter may be used.

Allow the pump to run 5 minutes prior to calibration.
Appendix A

Calibration and Maintenance Procedures

- b. Check the battery charge.
- c. Connect the collection device in the same manner in which the sample will be collected. Please note that the sampling media used for calibration is not to be used for sample collection.
- d. Wet the inside of the glass tube with soap solution.
- e. Turn on the pump and adjust the pump rotameter to the appropriate flow rate setting.
- f. Draw two or three bubbles up the glass tube.
- g. If the flowrate is not within the range of accuracy, adjust the flowrate and draw more bubbles up the glass tube.
- h. Record the average of three runs on the sample data sheet.
- i. Repeat the procedures for all pumps to be used. The same sampling media may be used for all calibrations.
- 2. Maintenance
 - a. Battery

The NiCad battery pack should be completely discharged from time to time to minimize the potential for "memory effect" which occurs frequently with rechargeable batteries. "Memory effect" would prevent a pump from running a full eight-hour period in some cases. The "Full Cycle" mode on the batery charger will fully discharge the battery pack, then recharge it and switch to a trickle automatically. A status lamp on the battery charger shows the status of the attached battery pack.

b. Pump Inlet Filter

A filter/trap inside the clear plastic intake port housing prevents particulates and liquids from being drawn into the pump mechanism. Replace the filter as needed.

E. Dust Monitor

- 1. Zero
 - a. Inlet should be in clear position.
 - b. Selector switch should be in 0-2 position, and the letter "m" should appear to the right or the display reading, indicating that the instrument is set to read concentration measurements.
 - e. Place time constant switch in 2-second position.
 - d. Allow 1-minute warm-up period.
 - e. If required, slowly adjust the ZERO control (while lifting its protective spring loaded cap) until an average reading of 0.000 m is obtained.
- 2. Calibration
 - Keep inlet in clear position.
 - b. Unlock hinged flow chamber and place it in the horizontal position.
 - e. Push reference scatterer knob (REF SCAT) inwards until a positive stop is reached. It is important to ensure the plunger has been fully inserted to the limit of its travel. Insertion of the reference scatter will automatically shut off the pump.
 - d. The letter "K" should be flashing at the upper right side of the digital display indicating that the reference scatterer has been inserted in the sensing beam.

Appendix A

Calibration and Maintenance Procedures

- Place the time constant in the 2-second position. £
 - Move range selector switch to its 0-20 position and allow about 30 seconds for the reading to fully stabilize. The display should read approximately the calibration number indicated on the RAM-1 calibration serial number label within the flow chamber.
- If the indicated readings differ substantially (i.e. by more than 5%) from the factoryg. labelled value, unlock the CAL control and adjust until the desired reading obtained. Relock CAL control.
- Puil reference scatterer knob completely out of sensing region until the "X" indicator on h. the display ceases to flash.
- i. Close flow champer cover by tightening two thumb-screws at its upper corner.
- F. Drager Putto

đ

- 1. Pump Leak Test
 - Insert unopened tube into socket. a.,
 - Squeeze pump completely and release. 'n.,
 - Pump is adequately leak-proof if the end of stroke indicator has not appeared after 15 c. minutes.
 - đ. Remove tube from socket.
 - Press button to reset counter to zero; button is recessed to prevent accidental resetting. e.
- 2. To clean or replace the exhaust valve
 - Lift lower pump plate up with a coin or screwdriver. **a**..
 - Pull vaive disc from its seating. Ъ.
 - Wipe valve seating with a damp cloth. с.
 - đ. Moisten the stem on a new valve disc and push ti firmly into the center hole of the valve seat.
 - Check correct seating by pulling on valve disc lightly. e.
 - f. Put back lower pump plate and press it into place.
 - Perform leak test discribed above. g.
- 3. Replacing the socket and the membrane
 - Press spring hook down with a suitable tool (e.g. screwdriver). 2.
 - Ъ. Take out cover place.
 - Press spring flanges together and remove socket. c.
 - đ. Insert new socket.
 - Using a new End-of-Stroke Indicator Membrane, place parts into the housing.
 - e. f. lasert spring hook of cover plate and push on firmly.
 - Perform leak test described above. g.

.

.

· · ·

.

Appendix E

.

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL	SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95
	PAGE: 1 of 11

Section 4.05 Hazardous Waste Operations

A. Introduction

Due to the nature of the business conducted by OBG Tech, employees may be working on hazardous waste sites. OSHA has requirements for such things as air monitoring, health and safety plans, site control, and emergency response. This section is to define OBG Tech's Standard Operating Procedures (SOPs) for hazardous waste sites and to outline the elements for inclusion in the Site-specific Health and Safety Plan.

B. Other Pertinent Sections of OBG Tech's Health and Safery Manual

Section	Topic
1.01	Safery Policy
2.01	Health and Safery Records
2.02	Health and Safery Training
2.03	Injury/Illness and Accident Report Process
2.04	Medical Surveillance
2.05	Exposure Monitoring
2.09	General Safery Rules
3.01	Medical Services and First Aid
3.02 -	Sanitation
3.03	Emergency Action
3.04	Confined Space Entry
4.01	Occupational Noise Exposure
4.02	Radiation Exposure
4.03	Airborne Materials Exposure
5	Personal Protective and Life Saving Equipment
6	Fire Protection and Prevention
8	Materials, Handling, Storage, Use, and Disposal

C. <u>Hazardous Waste Standard Operating Procedures</u>

1. Contractors and Sub-contractors

All contractors and sub-contractors retained by OBG Tech for work in hazardous waste will be informed of emergency response procedures and any potential fire, explosion, health, safety, or other hazards of the hazardous waste operation that have been identified by OBG Tech.

2. Program Availability

The written safety and health program will be made available to: any contractor or subcontractor; OBG Tech employees; OSHA personnel; and to personnel of other Federal, State, or local agencies.

3. Project Personnel

Certain individuals have specifically designated responsibilities on hazardons waste sites.

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 2 of 11

<u>Project Manager</u> - The Project Manager is responsible for the over-all management of the project. The Project Manager manages administrative requirements.

<u>Project Supervisor</u> - The Project Supervisor is responsible for coordinating between office and field personnel. The Project Supervisor is responsible for the day-to-day activities of the project. The Project Supervisor will oversee field and related activities.

Site Safery and Health Coordinator - The Site Safery and Health Coordinator (SSHC) will establish operating standards and coordinate overall project safery and health activities for the site. The SSHC will review project plans and revisions to plans to determine that safety and health procedures are maintained throughout the project. The specific responsibilities of the SSHC are outlined in Part 16, below.

4. Pre-entry Briefing

Pre-entry briefings will be held prior to initiating any site activity and at other times as accessary to inform employees of the site-specific health and safety plan. In situations covered by OBG Tech's Haz Com Program or the 40-hour Hazardous Waste Operations Health and Safety Training, training required by that program need not be duplicated.

5. Effectiveness of site-specific health and safery plan

Inspections will be made by the Site Health and Safery Coordinator or the Project Supervisor to determine the effectiveness of the site safery and health plan.

6. Personal Protective Equipment

Personal protective equipment (PPE) will be provided in accordance with Section 5 of this Health and Safety Manual. PPE will be selected and used which will provide protection against known or suspected hazardous substances and health hazards.

7. Initial Site Entry Monitoring

.

.

When information on the site shows that the potential for ionizing radiation, for IDLH conditions, or when site information is not sufficient to reasonably eliminate the following conditions:

- a. Monitoring with direct-reading instruments for hazardous levels of ionizing radiation (See Section 4.02).
- b. Monitoring the air with direct-reading instruments for IDLH conditions (See Section 4.03).
- c. Visually observing for signs of actual of potential IDLH or other dangerous conditions.
- 8. On-going Air Monitoring Program

An on-going monitoring program will be established for every site where there is a potential for employee exposure to hazardous concentrations of hazardous substances. This on-going monitoring is to evaluate proper selection engineering controls, work practices, and PPE, so that employees are not exposed above OSHA PELs and published

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 3 of 11

exposure levels. Specifically the on-going monitoring will be conducted:

- when work begins on a different part of the site;
- when contaminants other than those previously identified are being handled;
- when a different type of operation is initiated;
 - when employees are handling leaking drums or containers or working in areas with obvious liquid contamination (e.g. spill or lagoon)

Personal monitoring will be performed on those employees likely to have the highest exposures to airborne materials in accordance with Section 4.03 of this Health and Safety Manual.

9. Site Control

The elements of the site control program will include a site map; site work zones; site communications, including alerting means for emergencies; and identification of nearest medical assistance. OSHA also requires the use of the "buddy" system, which is defined below and standard operating procedures, which are encompassed into this Health and Safety Manual. These need not be repeated in the site-specific health and safety plan when this Health and Safety Manual is on-site.

a. Site Work Zones

- Exclusion Zone: The exclusion zone or the Hot Zone is the area where contamination does or could occur. The exclusion zone boundary should be clearly marked by lines, placards, hazard tape and/or signs or enclosed by physical barriers, such as chains, tences, or ropes. Access control points should be established at the periphery of the exclusion zone to regulate the flow of personal and equipment into an out of the zone and to help verify that proper procedures for entering and exiting are followed.

The required level of PPE in the exclusion zone can vary according to job assignment. This will allow a flexible, effective, and less costly operation, while still maintaining a high degree of safety.

- Contamination Reduction Zone:

The contamination reduction zone or the decontamination zone is the transition area between the contaminated area and the clean area. At least two lines of decontamination stations should be set. up within the contamination reduction zone: one for personnel and one for heavy equipment. Personnel entering the contamination reduction zone should be required to wear PPE prescribed for working in the contamination reduction zone. To reenter the support zone, workers must remove any PPE. Personnel stationed in the contamination reduction zone includes the Site Health and Safety Coordinator, the Project Supervisor, personnel assisting in decontamination procedures, and

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 4 of 11

emergency response personnel.

The contamination reduction zone should be designed to facilitate: decontamination, emergency response, equipment resupply, sample packaging, and temporary work rest area.

- Support Zone:

The support zone is the location of the administrative and other support functions needed to keep the operations in the exclusion and contamination reduction zone running smoothly. Any function that need not or cannot be performed in a hazardous atmosphere is performed here. Personnel may wear normal work clothes within this zone. Any potentially contaminated clothing, equipment and samples must remain in the contamination reduction zone until decontaminated. All emergency telephone numbers, change for the telephone (if necessary), evacuation route maps, and vehicle keys should be kept in the support zone.

b. "Buddy" System

Most activities in a contaminated or otherwise hazardous areas should be conducted with a buddy who is able to:

- Provide his or her partner with assistance.
- Observe his or her partner for signs of chemical or heat exposure.
- Periodically check the integrity of his or her partner's protective clothing.
- Notify the Project Supervisor or the Site Health and Safety Coordinator if emergency help is needed.

c. Site Communication

Internal communication among personnel on site and external communication between on site and off-site personnel should be established.

Verbal communication at a site can be impeded by on-site background noise and the use of personal protective equipment. In the absence of site-specific communication signals, the following will be used for emergencies:

Hand clutching throat: Out of air/can't breathe

Thumbs up: OK/Tm alright/I understand

Thumbs down: No, negative

Grip partner's wrist or both hands around partner's waist: Leave area immediately

Hands on top of head: Need assistance

10. Site Security

Site security is necessary to prevent the exposure of unauthorized, unprotected people to the site, to avoid the increased hazards of vandals, to prevent their, and to avoid

OBG TECHNICAL SERVICES. INC.	SECTION: 4.05
HEALTH AND SAFETY MANUAL	SUPERSEDES: 6/3/93
	DATE: 12/31/93
	REVIEW: 12/31/95
	PAGE: 5 of 11

interference with safe working procedures.

A physical barrier will be constructed to help prevent exposure of unauthorized and unprotected people to site hazards.

The Site-specific health and safery plan will include any other provisions for maintaining security at the Site.

11. Engineering Controls, work practices, and personal protective equipment for employee protection

Engineering controls and work practices must be first instituted to reduce and maintain employee exposure to or below the OSHA PELs, except to the extent that such controls and work practices are not feasible.

Engineering controls which may be feasible include the use of remotely operated material handling equipment. Work practices which may be feasible include wetting down dusty operations and locating employees upwind of possible hazards.

Whenever engineering controls and work practices are not feasible, PPE will be used in accordance with Section 5 of this Health and Safety Manual to reduce employee exposure to or below the OSHA PELs.

Engineering controls, work practices, and personal protective equipment will be used to reduce and maintain employee exposure to or below published exposure levels not regulated by OSHA. PPE will be used in accordance with Section 5 of this Health and Safety Manual to reduce employee exposure to or below the published exposure levels.

Guidance for PPE selection can be obtained from 29 CFR 1910.120 Appendix B or the NIOSH/OSHA/USCG/EPA document "Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities." The guidance document is usually handed out in the OSHA 40-hour course. However, should an employee desire a copy of 29 CFR 1910.120 Appendix B or the guidance document, the safety coordinator can provide one.

12. Levels of Protection

OBG Tech orten employs the use of PPE based upon the EPA Levels of Protection. They are listed in Section 5 as a starting point, however site-specific PPE may be required. Site-specific PPE will be addressed in the site-specific health and safety plan. The type of equipment used and the overall level of protection should be re-evaluated periodically as the amount of information about the site increases and as workers as required to perform different tasks.

13. PPE Decontamination Procedures

General PPE decontamination procedures are outlined in Section 5.01. Site-specific decontamination procedures, if any, will be included in the Site-specific Health and Safety Plan.

14. Emergency Response Procedures

SECTION: 4.05
SUPERSEDES: 6/3/93
DATE: 12/31/93
REVIEW: 12/31/95
PAGE: 6 of 11

The nature of hazardous waste work makes emergencies a continuous potential, no matter how infrequently they occur. The communications network, as outlined in Section 4.05, part 9.c, must be conveyed to all workers so that emergency information may quickly distributed. Equipment will be necessary for emergency situations. At a minimum each site will be equipped with a fire extinguisher, eye wash bottle, and first aid kit. Other equipment, which should be considered on a project by project basis, are emergency showers, safety harnesses, and spiil-containment equipment.

Follow-up procedures, as follows, must be implemented before activities resume onsite:

- Appropriate governmental agencies must be notified as required.
- Restock, replace, and/or clean all equipment and supplies.
- The Site Health and Safety Coordinator, the Project Supervisor, and the employee affected should fill out an accident report, as outlined in Section 2.03.
- 15. Hazards of hazardous waste sites

Chemical

Most sites contain a variety of chemicals that may be in a gaseous, liquid, or solid form. These substances can be hazardous through inhalation, skin absorption, ingestion, or through injection (a puncture wound).

Acute chemical exposures usually occur during or shortly after exposure to a high concentration of a chemical. Caronic usually refers to exposures to low concentrations of a chemical over a long period of time. Both of these are dependant on the chemical and may be temporary or reversible or may be permanent. Some chemicals may exhibit warning signs, while other chemicals are odorless, colorless, and tasteless.

Inhalation is the primary route of exposure at a hazardous waste site. Therefore the selection and use of the proper respiratory protection is extremely important where there is a potential for inhalation of hazardous materials. Direct skin and eyes contact is also a potential route of exposure, therefore proper selection and use of PPE is extremely important. Ingestion and injection are the least significant routes of exposure at a hazardous waste site. However, personal habits such as eating, drinking, or smoking, and safety hazards such as puncture wounds can be potential routes, therefore following SOPs and use of PPE can reduce the hazards from both of these routes of exposure.

Further information is presented in OBG Tech's Haz Com and OSHA 40-hour hazardous waste operations health and safety training.

Explosion and Fire

Due to the nature of activities at hazardous waste sites, there is a potential for explosions and fires. To minimize the hazards from fire and explosion: monitor for explosive and flammable atmospheres following the procedures in Section 4.03; keep all potential ignition sources away from an explosive or flammable environment; use non-sparking, explosion-proof equipment; and follow safe work practices.

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL	SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 7 of 11

Oxygen Deficiency

Oxygen deficiency may result from the displacement of oxygen by another gas, or the consumption of oxygen by another chemical reaction. Confined spaces and low-lying areas are particularly vulnerable and should always be monitored prior to entry following the procedures in Section 3.04.

Ionizing Radiation

Monitoring for ionization radiation is required to be performed when there is not sufficient information to eliminate the possibility of it being present on-site (e.g. uncontrolled hazardous waste sites, where little or no information is available on past history). The procedure in Section 4.02 will be followed for monitoring.

Biological Hazards

Wastes from hospitals and research facilities may contain biological materials that may cause infections to site personnel. Other biological hazards that may be present at hazardous waste sites include poisonous plants, insects, and animals. PPE can reduce the potential for exposure. The Safety Coordinator can assist in determining the correct PPE for the hazard present.

Safety Hazards

Hazardous waste sites contain numerous potential safety hazards such as: holes, ditches, drums, boards, nails, broken glass, slippery surfaces, steep grades, and uneven terrains. The work itself may be a potential safety hazard. Site personnel should constantly look out for potential safety hazards and should immediately inform the Project Supervisor or the Site Health and Safety Coordinator of any new hazards.

Electrical Hazards

As in all construction work, overhead power lines, electrical wires and cables, site electrical equipment, and lightning also pose a potential hazard to site workers. Section 10.02 provides guidance on safe electrical practices.

Heat Stress

Heat stress is potentially a major hazard for workers wearing protective clothing. Due to the impervious nature of the PPE to keep chemicals away from the skin, body heat and moisture are trapped within the PPE. Careful training and frequent monitoring of personal who wear protective clothing, scheduling of work and rest periods, and the frequent replacement of fluids can protect against this hazard.

Heat stress can be minimized by taking the following steps:

- Adjusting work schedules
- Provided air conditioned or shaded rest areas
- A total of 1 to 1.6 gallons of fluid intake recommended, but more may be necessary to maintain body weight
- Acclimatize workers to site work conditions
- Provide cooling devices to aid natural body heat exchange during prolonged work or

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 8 of 11

severe heat exposure

Review recognition and treatment of heat stress with workers

For workers wearing semi-permeable or impermeable PPE and when the temperature in the work area is above 70°F, measure:

- Heart Rate: count the radial pulse during a 30-second period as early as possible in the rest period.

If heart rate exceeds 110 beats per minute at the beginning of the rest period, shorten the work cycle by one-third and keep the rest period the same.

If the heart rate exceeds 110 beats per minute at the next rest period, shorten the following work cycle by one-third.

Oral Temperature: use a clinical thermometer or similar device to measure the oral temperature at the end of the work period (before intake of fluids).

If the oral temperature exceeds 99.6'F, shorten the next work cycle by one-third without changing the rest period.

- If the oral temperature still exceeds 99.6°F at the beginning of the next rest cycle, shorten the following work cycle by one-third.
- Body Water Loss, if possible: measure weight on a scale accurate to \pm -0.25 lb at the beginning and end of each work day to see if enough fluids are being taken in to prevent dehydration. Weights should taken while the employee is wearing similar clothing each day. The body water loss should not exceed 1.5 percent total body weight loss in a work day.

The frequency of physiological monitoring depends on the air temperature adjusted for solar radiation and the level of physical work. The suggested frequency of physiological monitoring for fit and acclimatized workers.

Adjusted Temp.	Normal Work Ensemble	Impermeable Ensemble
90°F and above	45 minutes	15 minutes
87.5F to 90°F	60 minute	30 minutes
82.5F to 87.5F	90 minutes	60 minutes
77 .5F to 82.5F	120 minutes	90 minutes
72.5F to 7 7.5°F	150 minutes	120 minutes

Cold Exposure

Cold stress and impaired ability to work are dangers at low temperatures and when the wind-

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 9 of 11

chill factor is low. To guard against cold stress: wear appropriate clothing; have warm clothing and warm shelter readily available; and carefully monitor workers' physical conditions.

Noise

Work around large equipment often creates excess noise. The procedures set forth in Section 4.01 for occupational noise exposure and Section 5.04 for hearing protection should be followed.

16. Responsibilities of Site Health and Safery Coordinator

The Site Health and Safery Coordinator advises the Project Manager and the Project Supervisor on the matters of health and safery on the site. Specifically the responsibilities of the Site Health and Safery Coordinator include:

- a. Aiding the selection of protective clothing and equipment.
- b. Periodically inspecting protective clothing and equipment.
- c. Maintaining proper storage or protective clothing and equipment.
- d. Monitors the workers for signs of heat stress, cold stress, and fatigue.
- e. Monitors on-site hazards and conditions.
- f. Conducts periodic surveillance to evaluate effectiveness of Site-specific Health and Safety Plan.
- g. Has knowledge of emergency procedures, evacuation routes, and the telephone numbers of the ambulance, local hospital, poison control center, fire department, and police department.
- h. Posts the directions to the hospital and the telephone numbers of the ambulance, local hospital, poison control center, fire department, and police department.
- i. Notifies, when necessary, local public emergency officials.
- j. Coordinates emergency medical care.
- 17. Safe Work Practices
 - a. No eating, smoking, eating, drinking, or application of cosmetics in the Contamination Reduction Zone or the Exclusion Zone.
 - b. No matches or lighters in the Contamination Reduction Zone or the Exclusion Zone.
 - c. Enter and exit following procedures in the Site-specific Health and Safety Plan.
 d. Wear the PPE specified in the site-specific Health and Safety Plan in the
 - Exclusion Zone.
 - e. Use the "buddy" system.
 - f. Report any unusual conditions to the Project Supervisor or the Site Health and Safety Coordinator immediately.

D. Requirements for Site-Specific Health and Safery Plans

As a company policy, site-specific health and safety plans are required on projects. The following site-specific information must be included in the health and safety plan, as well as this manual attached as an appendix. The Safety Coordinator will prepare and/or review each site-specific health and safety plan.

1. Organizational structure of site program

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 10 of 11

The following information, at a minimum, will be included in the site-specific health and safety plan:

- Project Manager's name and telephone number
 - Project Supervisor's name and telephone number
- Site Safery and Health Coordinator's name and telephone number
- Cther personnel needed for emergency response

This organizational structure will be reviewed and updated as necessary to reflect the current status of the hazardous waste site operations.

2. Workplan Summary

A summary of the workplan, including location and approximate size of the site, will be included in the site-specific health and safety plan addressing the anticipated activities.

3. Safery and Health Hazard Analysis

A safety and health hazard analysis will be prepared for the site-specific health and safety plan, including pathways for hazardous substance dispersion. The safety and health hazard analysis will include hazardous substances and health hazards involved or expected at the site, and their chemical and physical properties.

Information to consider to include in the site-specific health and safety plan:

- Exposures exceeding the permissible exposure limits and published exposure levels
- IDLH situations
- Potential skin absorption and irritation sources
- Potential eye irritation sources
- Explosion sensitivity and flammability ranges
- Oxygen dericiency
- 4. Site-specific Training Assignments

÷

ŝ

Initial 40-hour worker, Supervisor, and Refresher training are addressed in Section 2.02, Health and Safery Training. Hazard Communication training is also addressed in this section. Training specific to the site will be addressed in the site-specific health and safety plan.

5. Personal Protective Equipment

Personal protective equipment to be used by employees for each of the site tasks and operations being conducted will be addressed in the site-specific health and safety plan. The following information will be included in the site-specific plan: PPE selection based upon site hazards; work mission duration; site-specific information on PPE decontamination and disposal;

OBG TECHNICAL SERVICES, INC.	SECTION: + 05
HEALTH AND SAFETY MANUAL	SUPERSEDES: 6/3/93
	DATE: 12/31/93
	REVIEW: 12/31/95
	PAGE: 11 of 11

6. Frequency and types of air monitoring

Frequency and types of air monitoring, personnel monitoring, and environmental sampling techniques and instrumentation to be used will be addressed in the site-specific health and safety plan. Methods of maintenance and calibration of monitoring equipment can be found in Section 4.03.

7. Site Control Program

The site-specific site control program will include a site map and the identification of the nearest medical assistance.

8. Site Security

Site-specific information for maintaining security will be included in the site-specific health and safety plan.

9. Emergency Response Plan

Site-specific information such as safe distances and places of refuge, evacuation routes and procedures, and procedures for reporting incidents to local, state, and federal agencies not covered in OBG Tech's SOPs will be covered in the site-specific health and safety plan. The nearest telephone for emergency communication will be identified in the site-specific emergency response plan.

10. Spill Containment Program

A spill containment program unique to the site will be developed for the site-specific health and safety plan, if applicable.

- 11. Site-specific information other than in SOPs
 - a. Medical surveillance requirements unique to the site, other than what is covered in OBG Tech's SOP in Section 2.04.
 - b. Employee training assignments unique to the site other than what is covered in OBG Tech's SOP in Section 2.02.
 - c. Decontamination procedures which are unique to the site other than what is covered in OBG Tech's SOP in Section 4.04 (E).
 - d. Safe work practices unique to the site, other than what is covered in OBG Tech's SOP in this section.

SECTION: 4.05 SUPERSEDES: None DATE: May 23, 1992 REVIEW: March 24, 1994 PACE: 1 of 1

4.06 Process Safety

: :

A. Introduction

Releases of toxic, reactive, or flammable liquids and gases in processes involving highly hazardous chemicals have been reported for many years. In an effort to control these releases, CSHA has published a regulation, 29 CFR 1910.119, Process Safety Management of Highly Hazardous Chemicals. In that regulation is a section on contract employer (such as OBG Tech) responsibilities. This section has been developed in compliance with the regulation.

B. <u>Requirements</u>

1. Each employee is provided health and safety training, as outlined in Section 2.02 of this Health and Safety Manual. Specific training required at a jobsite regarding the known potential fire, explosion, or toxic release hazards related to his/her job and the process, and the applicable provisions of the emergency action plan will be the responsibility of the Project Manager.

2. Each employee will sign a document stating that they have received and understood the training that is outlined above.

3. Each employee is expected to follow the safety rules of the facility.

4.

In addition to the training required above, employees will be informed of unique hazards presented by the project or any of the hazards found by the work.

Appendix F

OBG TECHNICAL SERVICES. NO.	SECTIÓN: 5.06
HEALTH AND SAFETY MANUAL	SUPERSEDES: None
	DATE: May 15, 1990
	REVIEW: January 2, 1993
	PAGE: 1 OF 5
ويبيب ومستعلق تبصيني مكالية المنتكلية المناكرين والمتحد ومستدي مستنا والمستعدي والم	

5.06 Respiratory Protection

A. Introduction

OSHA, in 29 CFR 1910.134, specifies that "when effective engineering controls are not feasible, or while they are being instituted, appropriate respirators will be used." OSHA references exposure to air contaminated with harmful dusts, fogs, fumes, mists, gases, smokes, sprays, or vapors.

Due to the nature of the business conducted by the company, respiratory protection may be necessary for specific activities performed by our employees. This section will serve as the respiratory protection program and is adopted by the company in an effort to assist ensuring a safer working environment for our employees during work activities requiring respiratory protection as dictated in the site-specific Health and Safety Plan. In order to comply with regulatory requirements, this program is developed pursuant to 29 CFR 1910.134 and 29 CFR 1926.103. This program and use of respirators are instituted according to the site-specific Health and Safety Plan and only after exhausting all feasible engineering controls.

B. <u>Employer and Employee Responsibility</u>

1. Employer Responsibilities

The Safety Coordinator will see that approved respirators, cartridges, and spare parts will be provided by the company. The Safety Coordinator will also be responsible for the establishment and maintenance of this respiratory protection program and the upkeep of records for fit testing, medical surveillance, and training.

2. Employee Responsibilities

It is the responsibility of the employee to use the respiratory protection in accordance with instructions and training received. The employee will maintain the respirator to insure that cartridges and parts are replaced when necessary. The employee will report any problems with his respirator to his supervisor or the Safery Coordinator.

C. <u>Training of Employees</u>

All employees required to wear respirators on the job will be trained prior to the use of respirators. That training will cover the topics required by 29 CFR 1910.120 (Hazardous waste operations and emergency response) and 29 CFR 1910.134 (Respiratory Protection), and include the following:

- basics of respiration
- basics of respiratory hazards
- capabilities and limitations of respirators
- inspection of respirators
- how a respirator should be worn
- cleaning and disinfecting respirators
- storage of respirators
- respirator-specific training
- fit-checking procedures

SECTION: 5.06 SUPERSEDES: None DATE: May 15, 1992 REVIEW: January 2, 1993 PAGE: 2 OF 5

All employees will be given the opportunity to wear their respirator in an uncontaminated atmosphere and a test atmosphere for a period of time to become familiar with the use of respirators.

All training is documented and is filed in the Syracuse office.

D. <u>Physician's Approval</u>

All employees required to wear a respirator will have a physician's written approval to wear a respirator prior to being required to wear one. The respirator user's medical status will be reviewed annually.

The local physician will determine which health and physical conditions are pertinent. A description of the respiratory hazards, specifics of employee's job functions while wearing a respiratory and copies of all applicable regulations will be provided to the physicians to further aid the decision process.

The physical will be provided at no cost to the employee. A copy of the written report will be made to the employee upon request.

All medical examinations are recorded and are filed in the Syracuse office.

E Selection of Respirators

Respirators will be selected on the basis of the following:

- chemical and physical hazards
- characteristics of the hazardous operation of process
- face piece to face fit
- comioπ
- utilization of NIOSH recommendations
- utilization of manufacturer's recommendations
- the guidance of American National Standard Practices for Protection Z38.2-1969.

All respirators selected will be NIOSH/MSHA approved for the hazards encountered.

The Safety Coordinator will be adequately instructed to insure that the correct respirator is selected and that the appropriate personal modifications are made such as corrective lens for full-face masks.

F. Issuance of Respirators

The Warehouse Supervisor will be responsible for the issuance of a properly selected respirator to each employee. Each employee will be given his own respirator and will be responsible for bringing it to the jobsite. Employees should mark his respirator so that it will not be confused with others.

OBG TECHNICAL SERVICES, INC.	SECTION: 5.06
HEALTH AND SAFETY MANUAL	SUPERSEDES: None
	DATE: May 15, 1992
	REVIEW: January 2, 1993
	PAGE: 3 OF 5

G. Fit-Testing of Respirators

To insure a proper fit of negative pressure respirators, respirator fit-testing will be performed. Where fit-testing is not required by specific hazard regulation, as it is with lead or asbestos, the qualitative irritant vapor or smoke protocol of the asbestos standard will be adopted (29 CFR 1910.1001. Appendix C). Fit-testing will be performed to select respirators and be performed at the discretion of the Safety Coordinator thereafter unless required by law to be performed more often or unless there is sufficient need to do so (i.e., denture replacement, scarting of face, weight change).

Fit-test (allure will result in selection of a different size respirator. Continued test failure will result in selection of a different manufacturer's respirator.

All fit-testing information such as the employee, the date, and the type of respirator is recorded and filed in the Syracuse office.

H. Inspection of Respirators

Respirators will be inspected for damage before and after each use. Each employee, after training, will responsible for inspection. The following areas will be inspected:

- tightness of connections
- face piece
- headbands
- inhalation valve
- exhalation valve
- cartridge or filter fittings
- pliability of rubber or elastomer parts
- signs of deterioration

Any malformation, distortion, missing parts, cracks, etc. will be sufficient to issue replacement parts or if necessary, a new respirator.

L Standard Operating Procedures

Before entering any potentially contaminated environment, each employee will:

- 1. Carefully inspect the respirator following the procedures specified in Section 8.
- 2. Duct tape should be removed from cartridges (if applicable).
- 3. The respirator should be donned and checked for a proper fit using the following tests:
 - a. <u>Positive Pressure Test</u> close off the exhalation valve with your hand. Breathe into the mask. The face-to-facepiece seal is satisfactory if some pressure can be built up inside the mask and sustained.

SECTION: 5.06 SUPERSEDES: None DATE: May 15, 1991 REVIEW: January 1, 1993 PAGE: 4 OF 5

- b. <u>Negative Pressure Test</u> close off the inlet openings of the cartridge with the paim of your hand. Inhale gently so that a vacuum occurs inside the mask. Hold your breath for 10 seconds. If the vacuum is sustained, and no inward leakage is detected, the respirator fits properly.
- 4. Inside the contaminated environment, respirators will not be removed except in a medical emergency such as a suspected heart attack.
- Respirators will be worn with straps inside the disposable garment allowing a worker to maintain respiratory protection while removing contaminated garments.

J. <u>Cleaning and Disinfecting of Respirators</u>

Respirators will be cleaned after each use. Manufacturers may have specific recommendations for cleaning and those should be followed. In absence of manufacturers recommendations, the following procedures should be used:

- L. Remove the cartridges and headbands
- 2. Disassemble all respirator parts
- 3. Wash all respirator parts (except cartridges and headbands) in a cleaner disinfectant solution or use soap and hot water
- 4. Rinse completely in clean, warm water
- 5. Air dry in a clean area
- 6. Re-assemble the respirator

No alcohol will be used to clean the respirator. If a disinfecting solution is not used, a disinfecting spray will be used at least weekly, but preferably after each use.

Respirator wipes will be provided to employees in order to clean respirators during work shifts between uses. The employee will be allowed to leave work area and remove respirator to wash face in order to prevent rashes and discomfort. The respirators will be wiped out at each of these times.

K. Storage of Respirators

Respirators will be stored in clean plastic bags and protected against dust, sunlight, heat, extreme cold, excessive moisture, or damaging chemicals. Respirator carridges will have the inhalation holes covered with duct tape (or acceptable substitute tape) immediately after leaving a contaminated area. The tape will be left on until the respirator is donned for the nest entry into a contaminated area. This tape will prevent any contaminants from being dislodged from the cartridge.

Respirators should be packed or stored so that the facepiece and exhalation valve will rest in a normal position and function will not be impaired by the elastomer setting in an abnormal position.

OBG TECHNICAL SERVICES. INC.	SECTION: 5.06
HEALTH AND SAFETY MANUAL	SUPERSEDES: None
	DATE: May 15, 1992
	REVIEW: January 2, 1993
	PAGE: 5 OF 5

L Periodic Surveillance

Work areas will be monitored as required by specific hazard regulations or on a periodic timetable as set by the site-specific Health and Safety Officer and the site-specific Health and Safety Plan. This surveillance is required to ensure that the proper level of protection is provided to employees. Whenever new hazards are encountered or a substantial change in magnitude of the existing hazard occurs, then additional monitoring will take place.

M. Evaluation of Respiratory Protection Program

In order to maintain an effective program, the respiratory protection program will be re-evaluated on at least an annual basis. This evaluation will address:

- employee acceptance of program and respirators
- methods of surveillance of hazards and results
- regulatory compliance
- changing job functions
- changes in hazards

Employees are encouraged to express any concerns about respirator protection, such input is critical for evaluating the program.

Each employee will be made aware of this written program and any annual changes.

Frequent random inspections will be conducted by the Safety Coordinator to assure that respirators are properly selected, used, cleaned, and maintained.

N. Hazard Specific Respiratory Protection

As specified in 29 CFR Parts 1910, action specific hazards, i.e. lead, asbestos, and formaldehyde, require specific respiratory protection. Any such material will be monitored as directed under specific regulatory requirements, and respiratory protection will issued pursuant to specific regulatory requirement.

Appendix G

SECTION: 10.06 SUPERCEDES: 5/3/92 DATE: 12/31.93 REVIEW: 12/31.95 MARCH: 1 of 3

10.06 Excavations

A. <u>Introduction</u>

This section applies to all open excavations made in the earth's surface. Excavations are defined to include trenches.

B. <u>General Requirements</u>

- 1. All surface encumbrances that are located so as to create a hazard to employees will be removed or supported, as necessary, to safeguard employees.
- 2. Underground installations.
 - a. The estimated location of utility installations, such as sewer, telephone, fuel, electric, water lines, or any other underground installations that reasonably may be expected to be encountered during excavation work, will be determined prior to opening an excavation.
 - b. Utility companies or owners will be contacted within established or sustomary local response times, advised of the proposed work, and asked to establish the location of the utility underground installations prior to the start of actual excavation. When utility companies or owners cannot respond to a request to locate underground utility installations within 24 hours (unless a longer period is required by state or local law), or cannot establish the exact location of these installations, the work may proceed, provided caution is taken, and provided detection equipment or other acceptable means to locate utility installations are used.
 - c. When excavation operations approach the estimated location of underground installations, the exact location of the installations will be determined by safe and acceptable means.
 - d. While the excavation is open, underground installations will be protected, supported or removed as necessary to safeguard employees.

C. Access and egress

- 1. Structural ramps.
 - a. Structural ramps that are used solely by employees as a means of access or egress from excavations will be designed by a competent person. Structural ramps used for access or egress of equipment will be designed by a competent person qualified in structural design, and will be constructed in accordance with the design.
 - b. Ramps and runways constructed of two or more structural members will have the structural members connected together to prevent displacement.
 - c. Structural members used for ramps and runways will be of uniform thickness.
 - d. Clears or other appropriate means used to connect runway structural members will be attached to the bottom of the runway or will be attached in a manner to prevent tripping.
 - e. Structural ramps used in lieu of steps will be provided with clears or other surface treatments on the top surface to prevent slipping.

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL	SECTION: 10.06 SUPERCEDES: 6/3/91	
	DATE: 12/31/93	
	REVIEW: 12/31/95	
	MARCH: 2 of 3	

2. Means of egress from trench excavations. A stairway, ladder, ramp or other safe means of egress will be located in trench excavations that are 4 feet (1.22 m) or more in depth so as to require no more than 25 feet (7.62 m) of lateral travel for employees.

D. Exposure to vehicular traffic

Employees exposed to public vehicular raffic will be provided with, and will wear, warning vests or other suitable garments marked with or made of reflectorized or high-visibility material.

E. Exposure to failing loads

No employee will be permitted underneath loads handled by lifting or digging equipment. Employees will be required to stand away from any vehicle being loaded or unloaded to avoid being struck by any spillage or failing materials. Operators may remain in the cabs of vehicles being loaded or unloaded when the vehicles are equipped to provide adequate protection for the operator during loading and unloading operations.

F. Warning system for mobile equipment

When mobile equipment is operated adjacent to an excavation, or when such equipment is required to approach the edge of an excavation, and the operator does not have a clear and direct view of the edge of the excavation, a warning system will be utilized such as barricades, hand or mechanical signals, or stop logs. If possible, the grade should be away from the excavation.

G. <u>Hazardous atmospheres</u>

1. Testing and controls

In addition to the procedures set forth in Sections 3 and 4 of this Health and Safery Manual to prevent exposure to harmful levels of atmospheric contaminants and to assure acceptable atmospheric conditions, the following requirements will apply:

- a. Where oxygen deficiency (atmospheres containing less than 19.5 percent oxygen) or a hazarcous atmosphere exists or could reasonably be expected to exist, such as in excavations in areas where hazardous substances are stored nearby, the atmospheres in the excavation will be tested before employees enter excavations greater than 4 feet (1.22 m) in depth.
- b. Adequate preclutions will be taken to prevent employee exposure to atmospheres containing less than 19.5 percent oxygen and other hazardous atmospheres. These preclutions include providing proper respiratory protection or ventilation.
- c. Adequate precaution will be taken such as providing ventilation, to prevent employee exposure to an atmosphere containing a concentration of a flammable gas in excess of 20 percent of the lower flammable limit of the gas.
- d. When controls are used that are intended to reduce the level of atmospheric contaminants to acceptable levels, the testing will be conducted as often as necessary to ensure that the atmosphere remains safe.
- 2. Emergency rescue equipment
 - a. Emergency rescue equipment, such as breathing apparatus, a safety harness and line, or a basket stretcher, will be readily available where hazardous atmospheric

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL	SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: (L/31/93 REVIEW: 12/31.95
	MARCH: 3 of 3

conditions exist or may reasonably be expected to develop during work in an excavation. This equipment will be attended when in use.

b. Employees entering bell-bottom pier holes, or other similar deep and confined tooting excavations, will wear a harness with a life-line securely attached to it. The lifeline will be separate from any line used to handle materials, and will be individually attended at all times while the employee wearing the lifeline is in the excavation.

H. Protection from hazards associated with water accumulation

- 1. Employees will not work in excavations in which there is accumulated water, or in excavations in which water is accumulating, unless adequate precautions have been taken to protect employees against the hazards posed by water accumulation. The precautions necessary to protect employees adequately vary with each situation, but could include special support or shield systems to protect from cave-ins, water removal to control the level of accumulating water, or use of a safety harness and lifeline.
- 2. If water is controlled or prevented from accumulating by the use of water removal equipment, the water removal equipment and operations will be monitored by a competent person to ensure proper operation.
- 3. If excavation work interrupts the natural drainage of surface water (such as streams), diversion ditches, dikes, or other suitable means will be used to prevent surface water from entering the excavation and to provide adequate drainage of the area adjacent to the excavation. Excavations subject to runoff from heavy rains will require an inspection by a competent person.

L Stability of adjacent structures

- 1. Where the stability of adjoining buildings, walls, or other structures is endangered by excavation operation, support systems such as shoring, bracing, or underpinning will be provided to ensure the stability of such structures for the protection of employees.
- 2. Excavation below the level of the base or footing of any foundation or retaining wall that could be reasonably expected to pose a hazard to employees will not be permitted except when:
 - a. A support system, such as underpinning, is provided to ensure the safety of employees and the stability of the structure; or
 - b. The excavation is in stable rock; or
 - c. A registered professional engineer has approved the determination that such excavation work will not pose a hazard to employees.
 - d. Sidewalks, pavements, and appurtenant structure will not be undermined unless a support system or another method of protection is provided to protect employees from the possible collapse of such structures.

J. Protection of employees from loose rock or soil

- 1. Adequate protection will be provided to protect employees from loose rock or soil that could pose a hazard by failing or rolling from an excavation face. Such protection will consist of scaling to remove loose material; installation of protective barricades at intervals as necessary on the face to stop and contain failing material; or other means that provide equivalent protection.
- 2. Employees will be protected from excavated or other materials of equipment that could pose a hazard by falling or rolling into excavations. Protection will be provided by

SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: 12/31.93 REVIEW: 12/31/95 MARCH: 4 of 3

placing and keeping such materials or equipment at least 2 feet (.61 m) from the edge of excavations, or by the use of retaining devices that are sufficient to prevent materials or equipment from falling or rolling into excavations, or by a combination of both if necessary.

K. <u>Inspections</u>

1.

- Daily inspections of excavations, the adjacent areas, and protective systems will be made by a competent person for evidence of a situation that could result in possible cave-ins, indications of failure of protective systems, hazardous atmospheres, or other hazardous conditions. An inspection will be conducted by conducted by the competent person prior to the start of work and as needed throughout the shift. Inspections will also be made after every rainstorm or other hazard increasing occurrence. These inspections are only required when employee exposure can be reasonably anticipated.
- 2. Where the competent person finds evidence of a situation that could result in a possible cave-in, indications of failure of protective systems, hazardous atmospheres, or other hazardous conditions, exposed employees will be removed from the hazardous area until the necessary precautions have been taken to ensure their safety.

L. Fail protection

1

.

- 1. Where employees or equipment are required or permitted to cross over excavations, walkways or bridges with standard guardrails will be provided.
- 2. Adequate barrier physical protection will be provided at all remotely located excavations. All wells, pits, sharts, etc., will be barricaded or covered. Upon completion of exploration and similar operations, temporary wells, pits, sharts, etc., will be backrilled.

M. Requirements for protective systems

- 1. Protection of employees in excavations.
 - a. Each employee in an excavation will be protected from cave-ins by an adequate protective system designed in accordance with paragraph 2 or 3 of this section except when:
 - 1. Excavations are made entirely in stable tock: or
 - 2. Excavations are less than 5 feet (1.52 m) in depth and examination of the ground by a competent person provides no indication of a potential cave-in.
 - b. Protective systems will have the capacity to resist without failure all loads that are intended or could reasonably be expected to be applied or transmitted to the system.

SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 5 of 8

2. Design of sloping and benching systems. The slopes and configurations of sloping and benching systems will be in accordance with the requirements of option 1; or, in the alternative, option 2; or. in the alternative, option 3, or, in the alternative, option 4, as follows:

Option (1) - Allowable configurations and slones.

- Excavations will be sloped at an angle not steeper than one and one-half horizontal to one vertical (34 degrees measured from the horizontal), unless one of the other options listed below is used.
- 2. Slopes specified in paragraph 1 of this section, will be excavated to form configurations that are in accordance with the slopes shown for Type C soil in Appendix 3 to this subpart.

Option (2) - Determination of slopes and configurations using 19 CFR 1925 Subpart P Appendices A and B

Maximum allowable slopes, and allowable configurations for sloping and benching systems, will be determined in accordance with the conditions and requirements set forth in appendices A and B to this subpart.

Cotion (3) - Designs using other tabulated data

- 1. Designs of sloping or benching systems will be selected from and be in accordance with tabulated data, such as tables and charts.
- 2. The tabulated data will be in written form and will include all of the following:
 - A. Identification of the parameters that affect the selection of a sloping or benching system drawn from such data;
 - B. Identification of the limits of use of the data, to include the magnitude and configuration of slopes determined to be safe;
 - C. Explanatory information as may be necessary to aid the user in making a correct selection of a protective system from the data.
- 3. At least one copy of the tabulated data which identifies the registered professional engineer who approved the data, will be maintained at the jobsite during construction of the protective system. After that time the data may be stored off the jobsite, but a copy of the data will be made available to the Secretary upon request.

Option (4) - Design by a registered processional engineer

- 1. Sloping and benching systems not utilizing Option (1) or Option (2) or Option (3) under paragraph 2 of this section will be approved by a registered professional engineer.
- 2. Designs will be in written form and will include at least the following:
 - A. The magnitude of the slopes that were determined to be safe for the particular project;
 - B. The configurations that were determined to be safe for the particular project; and
 - C. The identity of the registered professional engineer approving the design.

3. At least one copy of the design will be maintained at the jobsite while the slope is being constructed. After that time the design need not be at the jobsite, but a copy will be made available to OSHA upon request.

3. Design of support systems, shield systems, and other protective systems. Designs of support systems shield systems, and other protective systems will be in accordance with the requirements of option 1; or, in the alternative, option 2; or, in the alternative, option 3; or, in the alternative, option 4 as follows:

OBG TECHNICAL SERVICES, INC.	
HEALTH AND SAFETY MANUAL	

а.

2

Ъ.

:: :

÷

SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 6 of 3

<u>Option (1) - Designs using appendices A. C. and D</u> Designs for imber shoring in trenches will be determined in accordance with the conditions and requirements set forth in appendices A and C to this subpart. Designs for aluminum hydraulic shoring will be in accordance with paragraph (c)(2) of this section, but if manufacturer's tabulated data cannot be utilized, designs will be in accordance with Appendix D.

Option (2) - Designs Using Manufacturer's Tabulated Data

- a. Design of support systems, shield systems, or other protective systems that are drawn from manufacturer's tabulated data will be in accordance with all specifications, recommendations, and limitations issued or made by the manufacturer.
- b. Deviation from the specifications, recommendations, and limitations issued or made by the manufacturer will only be allowed after the manufacturer issues specific written approval.
- c. Manufacturer's specifications, recommendations, and limitations, and manufacturer's approval to deviate from the specifications, recommendations, and limitations will be in written form at the jobsite during construction of the protective system. After that time this data may be stored off the jobsite, but a copy will be made available to OSEA upon request.

Option (3) - Designs using other tabulated data

- Designs of support systems, shield systems, or other protective systems will be selected from and be in accordance with tabulated data, such as tables and charts.
- b. The tabulated data will be in written form and include all of the following:
 - 1. Identification of the parameters that affect the selection of a protective system drawn from such data;
 - Leastification of the limits of use of the data;
 - 3. Explanatory information as may be necessary to aid the user in making a correct selection of a protective system from the data.
- c. At least one copy of the tabulated data, which identifies the registered professional engineer who approved the data, will be maintained at the jobsite during construction of the protective system. After that time the data may be stored off the jobsite, but a copy of the data will be made available upon request.

Option (4) - Design by a registered professional engineer

- Support systems, shield systems, and other protective systems not utilizing Cytion 1, Option 2, or Cytion 3, above, will be approved by a registered professional engineer.
 - Designs will be in written form and will include the following:
 - 1. A plan indicating the sizes, types, and configurations of the materials to be used in the protective system; and
 - 2. The identity of the registered professional engineer approving the design.

SECTION: 10.06 SUPERCEDES: 5/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 7 of 8

- c. At least one copy of the design will be maintained at the jobsite during construction of the protective system. After that time, the design may be stored off the jobsite, but a copy of the design will be made available to OSHA upon request.
- 4. Materials and equipment.
 - a. Materials and equipment used for protective systems will be free from damage or defects that might impair their proper function.
 - b. Manufactured materials and equipment used and maintained in a manner that is consistent with the recommendations of the manufacturer, and in a manner that will prevent employee exposure to hazards.
 - c. When material or equipment that is used for protective systems is damaged, a competent person will examine the material or equipment and evaluate its suitability for continued use. If the competent person cannot assure the material or equipment is able to support the intended loads or is otherwise suitable for safe use, then such material or equipment will be removed from service, and will be evaluated and approved by a registered professional engineer before being returned to service.
- 5. Installation and removal of support
 - a. General.

b.

1.

- 1. Members of support systems will be securely connected together to prevent sliding, falling, kickouts, or other predictable failure.
- Support systems will be installed and removed in a manner that protects employees from cave-ins, structural collapses, or from being struck by members of the support system.
- 3. Individual members of support systems will not be subjected to loads exceeding those which those members were designed to withstand.
- 4. Before temporary removal of individual members begins, additional precautions will be taken to ensure the safety of employees, such as installing other structural members to carry the loads imposed on the support system.
- 5. Removal will begin at, and progress from, the bottom of the excavation. Members will be released slowly so as to note any indication of possible failure of the remaining members of the structure or possible cave-in of the sides of the excavation.
- 6. Backfilling will progress together with the removal of support systems from excavations.
- Additional requirements for support systems for trench excavations.
 - Excavation of material to a level no greater than 2 feet (.61 m) below the bottom of the members of a support system will be permitted, but only if the system is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open of a possible loss of soil from behind or below the bottom of the support system.

SECTION: 10.06 SUPERCEDES: 5/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: \$ of \$

2. Installation of a support system will be closely coordinated with the excavation of renches.

N. Sloping and benching systems

Employees will not be permitted to work on the faces of sloped or benched excavations at levels above other employees except when employees at the lower levels are adequately protected from the hazard of failing, rolling, or sliding material or equipment.

- O. <u>Shield systems</u>
 - 1. Generai
 - a. Shield systems will not be subjected to loads exceeding those which the system was designed to withstand.
 - b. Shield will be installed in a manner to restrict lateral or other hazardous movement of the shield in the event of the application of sudden lateral loads.
 - c. Employees will be protected from the hazard of cave-ins when entering or exiting the areas protected by shields.
 - d. Employees will not be allowed in shield when shields are being installed, removed, or moved vertically.
 - 2. Additional requirement for shield systems used in trench excavations. Excavations of earth material to a level not greater than 2 feet (.61 m) below the bottom of a shield will be permitted, but only if the shield is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open of a possible loss of soil from behind or below the bottom of the shield.

CONSTRUCTION STANDARDS

(1) Option (1)—Designs using sppenlices A. C. and D. Designs for timber shorting in trenches shall be determined in accordance with the conditions and recuirements set forth in appendices A and C to this subpart. Designs for aluminum hydraulic shoring shall be in accordance with paragraph (c)(2) of this section, but if manufacturer's tabulated data cannot be utilized, designs shall be in accordance with appendix D.

(2) Option (2)—Designs Using Manufacturer's Tabulated Data. (i) Design of support systems, shield systems, or other protective systems that are drawn from manufacturer's tabulated data shall be in accordance with all specifications, recommendations, and limitations issued or made by the manufacturer.

(ii) Deviation from the specifications, recommendations, and limitations issued or made by the manufacturer shall only be allowed after the manufacturer issues specific written approval.

(ii) Manufacturer's specifications, recommendations, and ilimitations, and manufacturer's approval to deviate from the specifications, recommendations, and limitations shall be in written form at the jobsite during construction of the protective system. After that time this data may be stored off the jobsite, but a copy shall be made available to the Secretary upon request.

(3) Option (3)—Designs using other tabulated data. (i) Designs of support systems, shield systems, or other protective systems shall be selected from and be in accordance with tabulated data, such as tables and charts.

(ii) The tabulated data shall be in written form and include all of the following:

(A) Identification of the parameters that affect the selection of a protective system drawn from such data;

(B) Identification of the limits of use of the data;

(C) Explanatory information as may be necessary to aid the user in making a correct selection of a protective system from the data.

(iii) At least one copy of the tabulated data, which identifies the registered professional engineer who approved the data, shall be maintained at the jobsite during construction of the protective system. After that time the data may be stored off the jobsite, but a copy of the data shall be made available to the Secretary upon request.

(4) Option (4)—Design by a registered professional engineer. (i) Support systems, shield systems, and other protective systems not utilizing Option 1. Option 2 or Option 3, above, shall be approved by a registered professional engineer.

(ii) Designs shall be in written form and shall include the following:

(A) A plan indicating the sizes, types, and configurations of the materials to be used in the protective system; and (B) The identity of the registered professional engineer approving the design.

(iii) At least one copy of the design shall be maintained at the jobsite during construction of the protective system. After that time, the design may be stored off the jobsite, but a copy of the design shall be made available to the Secretary upon request.

(d) Materials and equipment. (1) Materials and equipment used for protective systems shall be free from damage or defects that might impair their proper function.

(2) Manufactured materials and equipment used for protective systems shall be used and maintained in a manner that is consistent with the recommendations of the manufacturer, and in a manner that will prevent employee exposure to hazards.

(3) When material or equipment that is used for protective systems is damaged, a competent person shall examine the material or equipment and evaluate its suitability for continued use. If the commetent person cannot assure the material or equipment is able to support the intended loads or is otherwise suitable for safe use, then such material or equipment shall be removed from service, and shall be evaluated and approved by a registered professional engineer before being returned to service.

(e) Installation and removal of support—(1) General. (i) Members of support systems shall be securely connected together to prevent slicing, failing, kickouts, or other predictable failure.

(ii) Support systems shall be installed and removed in a manner that protects employees from cave-ins, structural collapses, or from being struck by members of the support system.

(iii) Individual members of support systems shall not be subjected to loads excoording those which those members were designed to withstand.

(iv) Before temporary removal of individual members begins, additional precautions shall be taken to ensure the safety of employees, such as installing other structural members to carry the loads imposed on the support system.

(v) Removal shall begin at, and progress from, the bottom of the excavation. Members shall be released slowly so as to note any indication of possible failure of the remaining members of the structure or possible cave-in of the sides of the excavation.

(vi) Backfilling shall progress together with the removal of support systems from excavations.

(2) Additional requirements for support systems for trench excavations. (i) Excavation of material to a level no greater than 2 feet (.61 m) below the bottom of the members of a support system shall be permitted, but only if the system is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open of a possible loss of soil from behind or below the bottom of the support system.

(ii) installation of a support system, shall be slosely coordinated with the encavation of tranches.

(f) Stoping and benching systems. Employees shall not be permitted to work on the faces of sloped or benched excavations at levels above other employees except when employees at the lower levels are adequately protected from the hazard of failing, rolling, or sliding material or equipment.

(g) Shield systems—(1) General. (i) Shield systems shall not be subjected to loads exceeding those which the system was designed to withstand.

(ii) Sateids shall be installed in a manner to restrict lateral or other hazardous movement of the shield in the event of the application of sudden lateral loads.

(iii) Employees shall be protected from the bazard of mye-ins when entering or exiting the areas protected by shields.

(iv) Employees shall not be allowed in shields when shields are being installed, removed, or moved vertically.

(2) Additional requirement for shield systems used in trench excavations. Excavations of earth material to a level not greater than 2 feet (.51 m) below the bottom of a shield shall be permitted, but only if the shield is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open of a possible loss of soil from behind or below the bottom of the shield.

Appendix A to Subpart P

Soil Classification

(a) Scope and application-(1) Scope. This appendix describes a method of classifying soil and rock deposits based on site and environmental conditions, and on the structure and composition of the earth deposits. The appendix contains definitions, sets forth requirements, and describes acceptable visual and manual tests for use in classifying soils.

(2) Application. This appendix applies when a sloping or benching system is designed in accordance with the requirements set forth in §1925.552(b)(2) as a method of protection for employees from cave-ins. This appendix also applies when umber shoring for excavations is designed as a method of protection from cave-ins in accordance with appendix C to subpart P of part 1926, and when aluminum hydraulic shoring is designed in accordance with appendix D. This Appendix also applies if other protective systems are designed and selected for use from data prepared in accordance with the requirements set forth in §1926.652(c), and the use of the data is predicated on the use of the soil classification system set forth in this appendix.

(b) Definitions. The definitions and examples given below are based on, in whole or in part, the following: American Society for Testing Materiais (ASTM) Standards D653 -85 and D2488: The Unified Soils Classification System, The U.S. Department of Agriculture (USDA) Textural Classification 01:3074

Schemet and The National Burrau of Standams Report BSS-121.

Comented soil means a soil in which the particles are held together by a chemical agent, such as calcium carbonate; such that a hand-sized sample cannot be crushed into powder or individual soil particles by inger pressure.

Cohesive soil means clay (fine grained soil), or soil with a high clay content, which has cohesive strength. Cohesive soil does not crumble, can be excavated with vertical sidesiopes, and is plastic when moist. Cohesive soil is hard to break up when dry, and exhibits significant cohesion when submerged. Cohesive soils include clayey sill, sandy clay, silty clay, clay and organic clay.

Dry soil means soil that does not exhibit visible signs of moisture content.

Fistured means a soil materia: that has a tendency to break along definite planes of fracture with little resistance, or a material that exhibits open cracks, such as tension cracks, in an exposed surface.

Granular soil means gravel, sand, or silt, (coarse grained soil) with little or no clay content. Granular soil has no cohesive strength. Some moist granular soils exhibit apparent cohesion. Granular soil cannot be molded when moist and crumbles easily when dry.

Layered system means two or more distincity different soil or rock types arranged in layers. Micheous seams or weakened planes in rock or shale are considered layered.

Moist soil means a condition in which a soil looks and feels damp. Moist conesive soil can easily be shaped into a ball and rolled into small diameter threads before crumbling. Moist granular soil that contains some cohesive material will exhibit signs of cohesion between particles.

Plestic means a property of a soil which allows the soil to be deformed or molded without cracking, or appreciable volume change.

Saturated soil means a soil in which the voids are filled with water. Saturation does not require flow. Saturation, or near saturation, is necessary for the proper use of instruments such as a pocket penetrometer or sheer vane.

Soil classification system means, for the purpose of this subpart, a method of categorizing soil and rock deposits in a hierarchy of Stable Rock. Type A. Type B. and Type C, in decreasing order of stability. The categories are determined based on an analysis of the properties and performance characteristics of the deposits and the environmental conditions of exposure.

Stable rock means natural solid mineral matter that can be excavated with vertical sides and remain intact while exposed.

Submerged soil means soil which is underwater or is free seeping.

Type A means cohesive soils with an uncommed compressive strength of 1.5 ton per square foot (sf) (144 kPa) or greater. Examples of cohesive soils are: ciay, silty day, sandy ciay, ciay loam and, in some cases silty clay loam and sandy clay loam. Cemented soils such as caliche and hardpan are also considered Type A. However, no soil is Type A if:

(i) The soil is fissured: or

-

..

(ii) The soil is subject to vibration from heavy traffic, pile triving, or similar effects: or

(iii) The soil has been providually disturbed: or

(iv) The soil is part of a sloped, layered system where the layers dip into the excavation on a slope of four horizontal to one vertical (4H: 1V) or greater; or

(v) The material is subject to other factors that would require it to be classified as a less stable material.

Type 3 means:

(i) Cohesive soil with an unconfined compressive strongth greater than 0.5 uf (48 kPa) but less than 1.5 uf (144 kPa); or

(ii) Granular schesionless soils including: angular gravei (similar to crushed rock), silt silt loam, sandy loam and, in some cases silty day loam and sandy day loam.

(iii) Previously disturbed soils except those which would otherwise be classed is Type C soil.

(iv) Soil that meets the unconfined compressive strength or comentation requirements for Type A, but is fissured or subject to vibration; or

(v) Dry rock that is not stable; or

(vi) Material that is part of a sloped, layered system where the layers dip into the excavation on a slope less steep than four horizontal to one vertical (4H:1 V), but only if the material would otherwise be classified as Type 3.

Type C means:

(i) Cohesive soil with an unconfined compressive strength of 0.5 tsf (48 kP2) or less; or

(ii) Granular soils including gravel, sand, and loamy sand, or

(iii) Submerged soil or soil from which water is freely seecing; or

(iv) Submerged rock that is not stable, or

(v) Material in a sloped, layered system where the layers dip into the excavation or a slope of four horizontal to one vertical (4H:1V) or steeper.

Unconfined compressive strength means the load per unit area at which a soil will fail in compression. It can be determined by laboratory testing, or estimated in the field using a pocket penetrometer, by thumb penetration tests, and other methods,

Wet soil means soil that contains significantly more moisture than moist soil, but in such a range of values that cohesive material will slump or begin to dow when vibrated. Granular material that would exhibit cohesive properties when moist will lose those cohesive properties when wet.

(c) Requirements—(1) Classification of soil and rock deposits. Each soil and rock deposit shall be classified by a competent person as Stable Rock. Type A, Type B, or Type C in accordance with the definitions set forth in paragraph (b) of this appendix.

(2) Basis of classification. The classification of the deposits shall be made based on the results of at lenst one visual and at lenst one manual analysis. Such analyses shall be conducted by a competent person using tests described in paragraph (d) below, or in other recognized methods of soil classification and testing such as those adopted by the America Society for Testing Materials, or the U.S. Department of Agriculture textural classification system.

(3) Visual and manual analyses. The visual and manual analyses, such as those noted is being isoeptable in paragraph (d) of this isopendix, shall be lettered and conducted to provide sufficient quantitative and qualitative information is may be necessary to identify property the properties. factors, and conditions infecting the classification of the deposits.

(4) Layered systems. In a layered system, the system shall be classified in accordance with its weakest layer. However, each layer may be classified individually where a more stable layer lies under a less stable layer.

(5) Reclassification. If after classifying a deposit, the properties, factors, or conditions infecting its classification change in any way, the changes shall be evaluated by a competent person. The deposit shall be reclassified as necessary to reflect the changed direumstances.

(d) Acceptable visual and manual lests-(1) Visual tests. Visual analysis is conducted to determine qualitative information regarding the excavation site in general, the soil adjacent to the excavation, the soil forming the sides of the open excavation, and the soil taken as samples from excavatied matemal.

(i) Observe samples if soil that are excavated and soil in the sides of the excavation. Estimate the range of particle sides and the relative amounts of the particle sides. Soil that is primarily composed of the-grained material is tablesive material. Soil composed primarily of coarse-grained sand of gravel is granular material.

(ii) Observe soil as it is excavated. Soil that remains in clumps when excavated is observe. Soil that breaks up easily and does not stay in clumps is granular.

(iii) Observe the side of the open excavation and the surface area adjacent to the excavation. Grack-like openings such as tension cracks could indicate fissured material. If chunks of soil spail of a vertical side, the soil could be fissured. Small spails are evidence of moving ground and are indications of potentially hazardous situations.

(iv) Observe the area adjacent to the excavation and the excavation itself for evidence of existing utility and other underground structures, and to identify previously disturbed soil.

(v) Observe the opened side of the excavation to identify layered systems. Examine layered systems to identify if the layers slope toward the excavation. Estimate the degree of slope of the layers.

(vi) Observe the area adjacent to the excavation and the sides of the opened excavation for evidence of surface water, water seeping from the sides of the excavation, or the location of the level of the water table.

(vii) Observe the area adjacent to the excavation and the area within the excavation for sources of vibration that may affect the stability of the excavation face.

(2) Manual tests. Manual analysis of soil samples is conducted to determine quantitative is well as qualitative properties of soil and to provide more information in order to classify soil property.

(i) Plasticity. Moid a moist or wet sample of soil into a ball and attempt to roll it into threads as thin as /b-lach in diameter. Cohesive material can be successfully rolled into threads without crumbling. For example, if at least a two inch (50 mm) length of //zinch thread can be held on one and without tearing, the soil is cohesive.

CONSTRUCTION STANCARDS

tuantitative as well as qualitative properties of soil and to provide more information in order to classify soil property.

(i) Plasmany. More a morst or wet sample of soil into a bad and attempt to roll it into threads as thin as "would in diameter. Conserve material can be successfully rolled into threads without grumbling. For example, if at least a two inch (50 mm) length of Winch thread can be beid on one end without tearing, the soil is concerve.

(ii) Dry strength. If the soil is dry and crumbles on its own or with moderate pressure into individual grains or fine powder. It is granular (any combination of gravel, sand, or sill). If the soil is dry and falls into diamos which break up into smaller diamps, but the smaller diamos can only be broken up with difficulty. It may be day in any combination with gravel, sand or sill. If the dry soil breaks unto diamps which do not break up into small diamos which do not break up into small diamos which do not break up into small diamos which do not break up into small diamos which do not break up into small difficulty, and there is no way be considered unifisative.

(iii) Thumb penetration. The inumb penetration test and be used to estimate the uncomfined compressive strength of conesive soils. This lest is based on the thumb penetradoo test described in American Somery for Testing and Matemala (ASTM) Standard designation D2435-"Standard Recommended Practice for Description of Sous (Visual-Manual Procedures,") Type A soils with an uncomfined compressive strength of 1.5 taf can be readily indented by the thumb: however, they can be penetrated by the thumb only with very great affort. Type C soils with an unconfined compressive strength of 0.5 isf can be easily penetrated several inches by the thumb, and can be moided by light dinger pressure. This test should be conducted on an undisturbed soil sample, such as a large clump of spoil, as soon as practicable after excavation to keep to a minimum the effects of exposure to drying influences. If the excavation is later exposed to wetting influences (rain, flooding), the classification of the soil must be changed accordingiy.

(iv) Other strength tests. Estimates of unconfined compressive strength of soils can also be obtained by use of a pocket penemometer or by using a hand-operated snearwade.

(v) Drying less. The basic purpose of the drying test is to differentiate between conssive material with fissures, unfissured conssive material, and granular material. The procedure for the drying test involves drying a sample of soil that is approximately one inch thick (2.54 cm) and six inches (15.24 cm) in diameter until it is thoroughly dryi

(A) If the sample develops cracks as it dries, significant fissures are indicated.

(3) Samples that dry without cracking are to be broken by hand. If considerable force is necessary to break a sample, the soil has significant conesive maternal content. The soil can be classified as a unitsoured conesive maternal and the unconfined compressive strength should be determined.

(C) if a sample breaks easily by hand, it is either a fissured cohosive material or a granular material. To distinguish between the two, pulverne the direct ourings of the sample by band or by stepping on them. If the clumps do not pulverne easily, the material is cohosive with fissures. If they pulverize easily into very small fragments, the material is granular.

Appendix 3 to Subpart P

Sloping and Benching

(a) Scope and application. This appendix contains specifications for sloping and benching when used as methods of protecting employees working in excavations from asveins. The requirements of this appendix apply when the design of sloping and benching protective systems is to be performed in accordance with the requirements set forth in § 1922.652(b)(2).

(b) Definitions.

Actual slope means the slope to which an excavation face is excavated.

Distress means that the soil is in a condition where a cave-in is imminent or is likely to occur. Distress is evidenced by such phenomena as the development of fissures in the face of or aciacont to an open axcavanon: the subsidence of the edge of an excavation; the simpling of material from the face or the builting or beaving of material from the bottom of an excavation; the spalling of material from the face of an excavation; and raveiling, i.e., small amounts of material such as peboles or litue clumos of material sucherity separating from the face of an excavation and mobiling or rolling down into the excavation.

Maximum allowable slope means the steepest incline of an excavation face that is acceptable for the most favorable site conditions as protection against cave-ins, and is expressed as the ratio of horizontal distance to vernical rise (HiW).

Short term excessive means a period of time less than or equal to 24 hours that an excession is open.

(c) Requirementa-(1) Soli stassification. Soli and rock deposits shall be tlassified in accordance with appendix A to subpart P of part 1922.

(2) Maximum silowobie sicce. The maximum allowable slope for a soil or rock deposit shall be determined from Table 3-1 of this appendix.

(3) Actual slope, (3) The actual slope shall not be steeper than the maximum allowable slope.

(ii) The actual slope shall be lass steep than the maximum allowable slope, when there are signs of distress. If that situation occurs, the slope shall be cut back to an actual slope which is at least to horizontal to one vertical ("which") less steep than the maximum allowable slope.

(iii) When surnarge loads from stored material or equipment, operating equipment, or traffic are present, a competent person shall determine the degree to which the actual slope must be reduced below the maximum allowable slope, and shall assure that such reduction is achieved. Surcharge loads from adjacent structures shall be evaluated in accordance with § 1925.851(i).

(4) Configurations. Configurations of sloping and benching systems shall be in accordance with Figure 3-1.

8-242 31:3182.3

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, C.C. 20027

[Appendix 3]

31:3182.4

REFERENCE FILE

TABLE B-1 MAXIMUM ÀLLOWABLE SLOPES_

SOIL OR ROCK TYPE	MAXIMUM ALLOWABLE SLOPES (H:V) [1] FOR EXCAVATIONS LESS THAN 20 FEET DEEP [3]
STABLE ROCK	VERTICAL (90 ⁻)
TYPE A [2]	3/4:1 (53 ^c)
TYPE B	1:1 (45 ⁻)
TYPE C	1 ¹ / ₂ :1 (34)

NOTES:

- 1. Numbers shown in parentheses next to maximum allowable slopes are angles expressed in decrees from the horizontal. Angles have been rounded off.
- A short-term maximum allowable slope of 1/2H:1V (63°) is allowed in excavations in Type A soil that are 12 feet (3.67 m) or less in depth. Short-term maximum allowable slopes for excavations greater than 12 feet (3.67 m) in depth shall be 3/4H:1V (53°).
- 3. Sloping or benching for excavations greater than 20 feet deep shall be designed by a registered professional engineer.

Figure B-1

Slope Configurations

(All slopes stated below are in the horizontal to vertical ratio)

B-1.1 Excavations made in Type A soil.

1. All simple slope excavation 20 feet or less in depth shall have a maximum allowable slope of %:1.



Simple Slope-General

Exception: Simple slope excavations which are open 24 hours or less (short term) and which are 12 feet or less in depth shall have a maximum allowable slope of 1/2:1.

[Sec. 1925.652, Table 8-1]

Occupational Safety & Hearth Reporter

70

CONSTRUCTION STANDARDS

2



Simple Slope-Short Term

2. All benched excavations 22 feet or less in depth shall have a maximum allowable slope of % to 1 and maximum bench dimensions as follows:







Multiple Bench

•

S. All excevations 8 feet or less in depth which have unsupported vertically sided lower portions shall have a maximum vertical side of 3% feet.

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

71

[Sec. 1926.652, Table B-1]



Unsupported Vertically Sided Lower Portion-Maximum 8 Feet in Depth

All excavations more than 3 feet but not more than 12 feet in depth which unsupported vertically sided lower portions shall have a maximum allowable slope of 1:1 and a maximum vertical side of 3% feet.



Unsupported Vertically Sided Lower Portion-Maximum 12 Feet in Depth

All excevations 20 feet or less in depth which have vertically sided lower portions that are supported or shielded shall have a maximum allowable slope of %:1. The support or shield system must extend at least 18 inches above the top of the vertical side.



Suported or Shielded Vertically Sided Lower Portion

4. All other simple slope, compound slope, and vertically sided lower portion excavations shall be in accordance with the other options permitted under § 1926.652(b).

B-1.2 Excevations Made in Type B Soil

1. All simple slope excavations 20 feet or less in depth shall have a maximum allowable slope of 1:1.

[Sec. 1926.652, Table B-1.2]

Occupational Safety & Health Reporter
CONSTRUCTION STANDARDS

STANDARDS 31:3132.

Simple Slope

2. All benched excavations 20 feet or less in depth shall have a maximum allowable slope of 1:1 and maximum bench dimensions as follows:



Single Bench



Multiple Bench

3. All excavations 20 feet or less in depth which have vertically sided lower portions shall be shielded or supported to a beight at least 13 inches above the top of the vertical side. All such excavations shall have a maximum allowable slope of 1:1.

12-6-89

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

73

[Sec. 1926.652, Figure 8-1.3]



Vertically Sided Lower Portion

4. All other sloped excavations shall be in accordance with the other options permitted in § 1928.852(b). B-1.3 Excavations Made in Type C Soil

1. All simple slope excavations 20 feet or less in depth shall have a maximum allowable slope of 1%:1.



Simple Slope

2. All excevations 20 feet or less in depth which have vertically sided lower portions shall be shielded or supported to a height at least 18 inches above the top of the vertical side. All such excevations shall have a maximum allowable slope of 11/21.



Vertical Sided Lower Portion

3. All other sloped excavations shall be in accordance with the other options permitted in § 1926.652(b).

B-1.4 Excavations Made in Layered Soils

1. All excavations 20 feet or less in depth made in layered soils shall have a maximum allowable slope for each layer as set forth below. [Sec. 1926.652, Figure B-1.4]

Occupational Safety & Health Reporter

7⊥



.

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

.



2. All other sloped excavations shall be in accordance with the other options permitted in § 1923.652(b).

Appendix C to Subpart P

÷

Timber Shoring for Trenches

(a) Scope. This appendix contains information that can be used timber shoring is provided as a method of protection from cave-ins in trenches that do not exceed 20 feet (6.1 m) in depth. This appendix must be used when design of timber shoring protective systems is to be performed in accordance with § 1923.652(c)(1). Other timber shoring configurations: other systems of support such as hydraulic and pneumatic systems; and other protective systems such as aloping, benching, shielding, and freezing

Occupational Safety & Health Reporter

systems must be designed in accordance with the requirements set forth in § 1925.652(b) and § 1925.652(c).

(b) Soil Classification. In order to use the data presented in this appendix, the soil type or types in which the excavation is made must first be determined using the soil

[Appendix C]

7\$

CONSTRUCTION STANDARDS

dessification method set form in appendix A of suppart P of this part.

(c) Presentation of Information.

Information is presented in several forms as follows:

(1) information is presented in tabular form in Tables C-1.1, C-1.2 and C-1.3, and Tables C-2.1, C-2.2 and C-2.3 following paragraph (2) of the appendix. Each table presents the minimum sizes of timber members to use in a shoring system, and each table contains data only for the particular soil type in which the excavation or portion of the excavation is made. The data are arranged to allow the user the flexibility to select from among several acceptable configurations of members based on varying the horizontal spacing of the crossbraces. Stable rock is exempt from shoring requirements and therefore, no data are presented for this condition.

(1) Information concerning the basis of the tabular data and the limitations of the data is presented in paragraph (d) of this appendix. and on the tables themselves.

(3) Information explaining the use of the tabular data is presented in paragraph (e) of this appendix.

(4) information illustrating the use of the tabular data is presented in paragraph (f) of this appendix.

(3) Miscellaneous notations regarding Tables C-1.1 through C-1.3 and Tables C-2.1 through C-2.2 are presented in paragraph (g) of this Appendix.

(d) Basis and limitations of the data.-(1) Dimensions of timber members. (i) The sizes of the timber members listed in Tables C-1.1 through C-1.3 are taken from the National Bureau of Standards (NBS) report. "Recommended Technical Provisions for Construction Practice in Shoring and Sloping of Trenches and Excavations." In addition, where NBS did not recommend specific sizes of members, member sizes are based on an analysis of the sizes required for use by existing codes and on empirical practice.

(ii) The required dimensions of the members listed in Tables C-1.1 through C-1.3 refer to actual dimensions and not nominal dimensions of the timber. Employers wanting to use nominal size shoring are directed to Tables C-2.1 through C-2.3. or have this choice under § 1928.352(c)(3), and are referred to The Corps of Engineers. The Bureau of Reclamation or data from other acceptable sources. (2) Limitation of application. (i) It is not intended that the timber shoring specification apply to every situation that may be experienced in the field. These data were developed to apply to the situations that are most commonly experienced in current benching practice. Shoring systems for use in situations that are not covered by the data in this appendix must be designed as specified in § 1923.652(c).

(ii) When any of the following conditions are present, the members specified in the tables are not considered adequate. Either an alternate timber shoring system must be designed or another type of protective system designed in accordance with § 1928.852.

(A) When loads imposed by structures or by stored material adjacent to the trench weigh in excess of the load imposed by a two-foot soil surcharge. The term "adjacent" as used here means the area within a horizontal distance from the edge of the trench equal to the depth of the trench.

(B) When vertical loads imposed on cross braces exceed a 240-pound gravity load distributed on a one-foot section of the center of the crossbrace.

(C) When surcharge loads are present from equipment weighing in excess of 20.000 pounds.

(D) When only the lower portion of a trench is shored and the remaining portion of the trench is sloped or benched unless: The sloped portion is sloped at an angle less steep than three horizontal to one vertical; or the members are selected from the tables for use at a depth which is determined from the top of the overall trench, and not from the top of the sloped portion.

(e) Use of Tables. The members of the shoring system that are to be selected using this information are the cross braces, the uprights, and the wales, where wales are required. Minimum sizes of members are specified for use in different types of soil There are six tables of information, two for each soil type. The soil type must first be determined in accordance with the soil classification system described in appendix A to subpart P of part 1928. Using the appropriate table, the selection of the size and spacing of the members is then made. The selection is based on the depth and width of the trench where the members are to be installed and, in most instances, the selection is also based on the horizontal

spacing of the crossbraces. Instances where a choice of horizontal spacing of crossbracing is available, the horizontal spacing of the crossbraces must be chosen by the user before the size of any member can be determined. When the soil type, the width and depth of the rench, and the horizontal spacing of the crossbraces are known, the size and vertical spacing of the crossbraces, the size and vertical spacing of the wales, and the size and horizontal spacing of the uprights can be read from the appropriate table.

(1) Examples to Illustrate the Use of Tables C-1.1 through C-1.1.

(1) Example 1.

A rrench dug in Type A soil is 13 feet deep and five feet wide.

From Table C-1.1, for acceptable arrangements of timber can be used.

Arrangement =1

Space 4×4 crossbraces at six feet horizontally and four feet vertically. Wales are not required.

Space 3×3 uprights at six feet horizontally This arrangement is commonly called "skip shorms."

Arrangement =2

Soace 4×8 crossbraces at eight feet horizontally and four feet vertically.

Space 3×3 wales at four feet vertically. Space 2×8 uprights at four feet horizontally.

Arrangement =3

Space 6×8 crossbraces at 10 feet horizontally and four feet vertically. Space 3×10 wales at four feet vertically.

Space 2×6 uprights at five feet horizontally.

Arrangement 🛋

Space #X8 crossbraces at 12 feet

horizontally and four feet vertically. Space 10×10 wales at four feet vertically. Spaces 3×8 uprights at six feet

horizontally.

(2) Example 2

A trench dug in Type B soil in 13 feet deep and five feet wide. From Table C-1.2 three acceptable arrangements of members are listed.

Arrangement #1

Space 8×8 crossoraces at six feet borizontally and five feet vertically. Space 8×3 wales at five feet vertically. Space 2×6 uprights at two feet borizontally.

Arrangement #2

Space 5×3 crossbraces at eight feet horizontally and five feet vertically. Space 10×10 wales at five feet vertically. Space 2×6 uprights at two feet horizontally.

Arrangement =3

Space 8×8 crossbraces at 10 feet horizontally and five feet vertically. Space 10×12 wales at five feet vertically.

Space 2×8 uprights at two feet vertically. (3) Example 3. A trench dug in Type C soil is 13 feet deep

and five feet wide.

From Table C-1.3 two acceptable arrangements of members can be used.

Arrangement =1

Space 8×5 crossoraces at six feet

homionically and five feet vertically. Space 10×12 wales at five feet vertically. Position 2×6 uprights as closely together as possible.

if water must be retained use special

tongue and groove uprights to form light sheeting.

Arrangement =2

Space 8 × 10 crossbraces at eight feet horizontally and five feet vertically.

Space 12×12 wales at five feet vertically. Position 2×5 uprights in a close sheeting configuration unless water pressure must be resisted. Tight sheeting must be used where water must be retained.

(4) Example 4.

A trench dug in Type C soil is 20 feet deep and 11 feet wide. The size and spacing of members for the section of trench that is over 15 feet in depth is determined using Table C-1.3. Only one arrangement of members is provided. Space 3 × 10 crossbraces at six feet homoontaily and five feet vertically. Space 12× 12 wales at five feet vertically. Use 3 × 6 upht sheeting. Use of Tables C-21 through C-23 would [ollow the same procedures. (g) Notes for all Tables.

1. Member sizes at spacings other than indicated are to be determined as specified in § 1925.652(c), "Design of Protective Systems."

2. When conditions are saturated or submerged use Tight Sheeting. Tight Sheeting refers to the use of specially-edged timber planks (e.g., tongue and groove) at least three inches thick, steel sheet piling, or similar construction that when driven or placed in position provide a tight wall to resist the lateral pressure of water and to prevent the loss of backfill material. Close Sheeting refers to the placement of planks side-by-side allowing as little space as possible between them.

3. All spacing indicated is measured center to center.

4. Wales to be installed with greater dimension horizontal.

L if the vertical distance from the center of the lowest crossbrace to the bottom of the trench exceeds two and one-half feet uprights shall be firmly embedded or a mudsill shall be used. Where uprights are embedded, the vertical distance from the center of the lowest crossbrace to the bottom of the trench shall not exceed 30 inches. When mudsills are used, the vertical distance shall not exceed 42 inches. Mudsills are wales that are installed at the toe of the trench side.

5. Trench jacks may be used in lieu of or in combination with timber crossbraces.

7. Placement of crossbraces. When the vertical spacing of crossbraces is four feet, place the top crossbrace no more than two feet below the top of the trench. When the vertical spacing of crossbraces is five feet, place the top crossbrace no more than 25 feet below the top of the trench.

Ţ	AB	IL E	C -	ł	1
-		_	_	-	 <u> </u>

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

CONSTRUCTION STANDARDS

31:3182.13

P_a = 25 X H + 72 psf (2 ft Surcharge) SOIL TYPE A

	[\$17	E LACTU	IAL) AND	SPACING	OF MEMBE	RS **				
DEPTH			CROS	S DRACE	5				[5		Ų	PRIGHTS		
	HORIZ	WI	DTIL OF	TRENCH	(FEET)		VERT.		VERT.	MAXIMU	ALLONA	BILE HORI	ZONTAL S	J'AC ING
(FEET)	SPACING	UP TO	UP TO	UP TO	UP TO	UP TO	SPACING	SIZE	SPACING			(FEET)		• • • •
(1121)	(FEET)	4	6	9	12	15	(FEET)	(IN)	(FEET)	CLOSE	4	5	6	. 8
······	UP TO		· ·					flot						
5	6	<u>4X4</u>	<u>4X4</u>	<u>4x6</u>	<u>6X6</u>	<u>6X6</u>	- 4	<u>Rea'd</u>		. <u></u> .			<u>2X6</u>	
	UP TO							llot						
10	8.	4X4	<u>4X1</u>	4 X.5	<u>6X5</u>	6X6	4	<u>Req'd</u>						2 X B
	UP TO								I ·			•		
10	10	<u>4×6</u>	<u> 4X6 </u>	4X6	<u>6X6</u>	<u>_6X6</u> _	4	<u> 8×8 </u>	4			_286_		
	UP TO		476	(16	6 M 6	640		0.40					0.44	
	12	486	470	0X0	010	010	4	878	4			·	2×6	
	UP TO		•					Not				l.		
10	6	<u>4X4</u>	<u>4X4</u>	<u>4x6</u>	<u>6X6</u>	<u>6X6</u>	4	<u>Reo'd</u>					<u>3X8</u>	
	UP TO													1
1 10	<u> </u>	<u>4x6</u>	<u>4x6</u>	676	<u>6X0</u>	<u> </u>	4	<u>-8x8</u>	4		<u>2X6</u>			
	UP TO	ine	C.V.P.		C 10	C V O		0410				276		
15		<u> 6X0</u>	<u> </u>	<u> </u>	<u>-010</u>	010	<u> </u>	-4216	1			<u> </u>		[
1	UP 10	646	646	646	670	670		10710					370	
	<u> </u>	0.00		<u> </u>	0.0		[-10010						
		616	616	686	6¥8	688	4	638	۵	386		· .		
15														
то	8	6X6	6X6	6X6	6X8	6X8	4	8X8	4	3X6	1. A.			ł
	UP TO	1]
20	10	8X8	<u>8X8</u>	<u>8X8</u>	8X8	<u>8X10</u>	4	8X10		3X6				
	UP TO	1					·							
	12	<u>8X8</u>	8X8	<u>8X8</u>	<u>8X8</u>	<u>8X10</u>	4	10X10	4	<u>3X6</u>				I
OVER]								•					
20	I SEE NOT	E 1												

* Mixed oak or equivalent with a bending strength not less than 850 psi. ** Manufactured members of equivalent strength may by substituted for wood.

12-6-89

ч The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 200 J7

[Sec. 1926.652, Table C-1.1]

3

.

TABLE C-1.2

····

.

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS .

SOIL TYPE B P = 45 X H + 72 paf (2 ft. Surcharge)

DEPTH				-	<u></u>	_{ACTU	AL)_AND_S	SPACING	QF_MEMBE	ISAA				
0.00			CROS	5 BRACE	S			<u> </u>	ES		<u></u> U1	RIGHTS		
TRENCH	HORIZ.	<u> </u>	TH OF	TRENCH	(FEET)		VERT.		WERT	HAXTHUR	ALLOVAL	BLE HORTZ	OUTAL S	PACHIG
(FEET)	SPACING	መግለ	UP 10	OP TO	UP TO	UP TO	SPACING	SIZE	SPACING			(FEET)		
	(FEĘT)	· 4	6	9	12	15	(FEET)	(11)	(FEET)	CLOSE	2	3		
5	UP TO 6	4×6	4x6	6X6	686	6X6	5	6X8	5			286		
то	UP TO B	6 <u>x6</u>	6X6	6X6	<u>6X8</u>	<u>6X8</u>	5	<u>8x10</u>	5			286		
10	UР ТО 10	6X6	6X6	6X6	6X8	6X8	5	10X10	5			2X6		
	See Note 1											[
10	UP TO 6	6X6	<u>6X6</u>	6X6	6X8	6X8	5	8X8	<u> </u>		286			
то	UP TO 8	6X8	<u>6X8</u>	<u>6X8</u>	<u>8x8</u>	8X8	5	10x10	5		286			
15	UP TO 10	<u>8x8</u>	<u>8X8</u>	<u>8x8</u>	_8X8_	_8X10	5	10X12	5		2X6			
	Sce Note l								· · · · · · · · · · · · · · · · · · ·					· ·
15	UP TO 6	6X8	6XB	6X8	8X8	8X8	5	8X10	5	3X6	. <u> </u>			
то	UP TO 8	8XA	8x8	8X8	BX8	8X10	5	10X12	5	386			·	
20	UP TO	<u>8x10</u>	<u>8x10</u>	<u>8x10</u>	<u>8x10</u>	<u>10X10</u>	5	12X12	5	<u></u>				
	See Note 1												· ·	
OVER 20	SEE NOT	1 31												

• Hixed oak or equivalent with a bending strength not less than 850 pst.

** Manufactured members of equivalent strength may by substituted for wood.

Occupational Safety & Health Reporter

N 84 - 8

....

80

REFERENCE FILE

· A.

TABLE C-1.3

.

TIMBER TRENCH SHORING -- MINIMULTIMBER REQUIREMENTS *

SOIL TYPE C P a = 80 X H + 72 psf (2 ft. Surcharge)

OF TRENCH (FEET) CHOSS BRACES UP TO UP TO SPACING UIDTU OF TRENCH (FET) GPACING UP TO UP TO G UP TO UP TO UP TO BX10 <	
TRENCI (PEET) UIDTL OF TRENCI (FEET) VERT. SPACING STZE VERT. SPACING UP TO SPACING IAXIIIUH ALLOHABLE NORIZOFTAL (FEET) 5 4 6 9 12 15 STZE <t< td=""><td></td></t<>	
(PEET) SPACING (FEET) up to 4 up to 6 up to 9 up to 12 up to 15 up to (FEET) SIZE (III) STACING SACING (FEET) UR to STACING (FEET) (FEET) (III) (III) (III) (IIII) (IIIII) (IIIII) (IIIIII) (IIIIIII) (IIIIIIIIIIIIII) (IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	SPACING
(FEET) 4 6 9 12 15 (FEET) (III) (FEET) CLOSE	o(e 2)
5 UP TO 6 6x8 6x8 6x8 8x8 ></td>	
5 6 6×8 6×8 6×8 8×10 5 10×12 5 2×6 10 10 6 8×8 8×8 8×8 8×10 5 10×12 5 2×6 10 10 6 8×8 8×8 8×10 10×10 5 12×12 5 2×6	
TO B BXI0 Size	
TO B 8X8 8X8 8X8 8X8 8X10 5 10x12 5 2x6 10 10 8x10 8x10 8x10 8x10 8x10 10x10 5 12x12 5 2x6 10 See 1 1 10x10 5 12x12 5 2x6 1 10 6 8x8 8x8 8x8 8x10 5 10x12 5 2x6 10 6 8x8 8x8 8x8 8x10 5 10x12 5 2x6 10 6 8x8 8x8 8x8 8x10 5 10x12 5 2x6 10 6 8x8 8x8 8x8 8x10 5 10x12 5 2x6 10 70 8 8x10 8x10 8x10 10x10 5 12x12 5 2x6 15 Note 1 15 0 8x10 8x10 8x10 10x10 5 12x12 5 3x6 15 0 6 8x10 8x10 8x10 10x10 5	
UP TO 8x10 8x10 8x10 8x10 10x10 5 12x12 5 2x6	
10 6x10 8x10 8x10 8x10 10x10 5 12x12 5 2x6	1
See I	
UP TO 8x8 8x8 8x8 8x8 8x10 5 10x12 5 2x6	
10 6 8x8 8x8 8x8 8x10 5 10x12 5 2x6	
10 UP TO 8 8x10 8x10 8x10 8x10 8x10 10x10 5 12x12 5 2x6 15 See Note 1	
TO 8 8x10 8x10 8x10 8x10 8x10 10x10 5 12x12 5 2x6	
See Note 1	
15 Note 1	
See Note 1 See	
Note 1 <	
15 UP TO 6 8x10 8x10 8x10 8x10 8x10 10x10 5 12x12 5 3x6 TO See Note 1	
15 6 8x10 8x10 8x10 8x10 8x10 10x10 5 12x12 5 3x6 See <td></td>	
See See Note 1	
TO Note 1 20 Note 1 See	
20 Sea Note 1 Note 1	
20 Note 1	1
See Note 1	}
	l
OVER 20 SEE NOTE 1	

A Mixed Oak or equivalent with a bending strength not less than 850 psi. A# Manufactured members of equivalent strength may be substituted for wood.

CONSTRUCTION STANDARDS

5-242 31:3182.15

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

12-6-89 •

TABLE C-2.1

···· · · · · · ·

.

pational Safety & Health Reports

8

TIMBER TRENC	<u>H SH</u>	DRING		<u>MI</u>	<u>11W</u>	<u>IM T</u>	IHBE	<u>R</u> R	EQUIREMENTS	<u>;</u> •
SOIL TYPE A	Pa	- 25	X	H ±	72	paf	(2	fc.	Surcharge))

DEPTH					512	E (\$45	AND_SP/	CING_OF	MEMBERS	**				
OF			CRO	SS BRAC	ES			1/Δ	LES		0	PRICHTS		
TRENCH	HORIZ.	<u> </u>	DTH OF	TRENCH	(FEET)	·	VERT.		VERT.	HAXTHU	ALLOVA	BLE HORA	ZONTAL S	PACING
(FEET)	SPACING	UP TO	UP TO	UP TO	UP TO	UP TO	SPACING	SIZE	SPACING			(FEET)	·····	
	(FEET)		<u> </u>	2		-15	(FEET)	(11)	(FEET)_	CLOSE		5	<u>_6</u>	<u> </u>
5	UР ТО 6	<u>4x4</u>	<u>4x4</u>	<u>4x4</u>	<u>4x4</u>	<u>4x6</u>	4	Not Req'd	Rég ¹ d				4x6	·
то	UP TO 8	4X4	4x4	4x4	4X6	4X6	4	Noy Req ⁴ d	Reg ¹ d					4X8
	UP TO	4X6	4X6	486	6X6	6X6	4	8X8	4			4X6		
	UP TO	4x6	· 4x6	4X6	6X6	6X6	. 4	8x8	4				4X6	
	UP TÓ 6	4X4	4X4	4X4	6X6	6X6	4	Nor Req d	Not Reg d				4X10	
то	UP TO 8	4×6	486	486	6X6	6X6	4	6X8	4		486			
	UP TO EO	6X6	6X6	6X6	6X6	6X6	4	8X8	4			4X8		
13	UP TO 12	6X6	6X6	6X6	6X6	6X6	4	8x10	4		4X6		<u>4X10</u>	
15	UP TO 6	6X6	6X6	6X6	686	6X6	4	6X8	4	_ 3 X6				
то	UP TO 8	6X6	.686	6X6	686	6X6	4	axa	_4	386	4x12		·	
20	UP TO	6X6	6X6	6X6	6X6	6X8	4	8X10	4	386				
	UP TO 12	6X6	686	6X6	6X8	6X8	4	8X12	4	3X6	4X12			
OVER 20	SEE NOTI	E 1											ı	

A Douglas fir or equivalent with a bending strength not less than 1500 psi. AA Hanufactured members of equivalent strength may be substituted for wood.

TABLE C-2.2

_

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

SOIL TYPE B P = 45 X II + 72 psf (2 ft. Surcharge)

DEPTH		······			SIZĘ -	(545) /	ND SPACE	NG OF M	EHBERS A	· · · · · · · · · · · · · · · · · · ·				
OF			CROS	SS_BRACI	:s			WA	ES		<u> </u>	PRIGHTS		
TRENCI	HORTZ.	¥I	DTIL OF	TRENCH	(FEET)		VERT.		VERT.	HAXTHUR	1 ALLOVA	BLE HORI	ZONTAL S	SPACIEG
(FEET)	SPACING	UP TO	OT AU	UP TO	UP TO	OT 96	SPACING	ST2.E	SPACING			(FEET)		
	(FEET)	4	6	9	12	15	(FEET)	(111)	(FEET)	CLOSE	2	3	4	6
5	ир то 6	4X6	4X6	4x6	6X6	6X6	5 -	6X8	5			3X12 4X8		4X12
то	UP TO 8	4X6	4X6	6X6	6X6	6X6	5	8X8	5		<u>3x8</u>		4X8	
10	UP TO 10	4X6	4X6	6X6	6X6	6X8	5	8X10	5			4X8		
	See Note I	:												-
10	UP TO 6	<u>6X6</u>	6X6	6X6	6X8	6X8	5	8X8	5	3X6	4X10			
то	UP TO 8	6x8	6X8	6X8	8X8	8X8	5	10210	5	3X6	4X10			
15	UP TO 10	6X8	6X8	8X8	8x8	8x8	5	10X12	5	3X6	4810		·	
	See Note I	[. <u>.</u>	
15	UP TO 6	6X8	6X8	6X8	6X8	8x8	. 5	8X10	5	4X6				
то	UP TO 8	6XB	6X8	6X8	8X8	8x8	5	10x12	5	4x6				
20	ир то 10	8X8	8X8	8X8	8X8	8X8	5	12X12	<u> </u>	486			·	
	See Note I	<u> </u>					<u> </u>						•	
OVER 20	SEE NOTE	2 1								,				

* Douglas fir or equivalent with a bending strength not less than 1500 ps1.

** Manufactured members of equivalent strength may be substituted for wood.

Pusiisned by The SUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 2002

12-6-89

:

[Sec. 1925.552, Table C-2.2]

CONSTRUCTION STANDARDS

TABLE C-2.3

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

SOIL TYPE C P = 80 X II + 72 psf (2 ft. Surcharge)

DEPTH					SIZE	(\$4\$)	AND SPAC	LUIG_OF_1	MEMBERS_	Å Å				
OF			CRO	S BRAC	ES			VAL	<u>[S</u>		<u></u>	RICHTS		
	HORIZ.	WI	OTH OF	TRENCH	(FEET)		VERT.		VERT.	NAX INUH	ALLOVAB	LE HORTZ	ONTAL SI	ACING
TRENCH	SPACING	UP TO	UP TO	UP TO	UP TO	UP TO	SPACING	SIZE	SPACING			(FEET)		
(1221)	(FEET)	4	6	_2	12	15	(FEET)	<u>_(117)</u>	(FEET)	CLOSE				
	UP TO													
5	6	6X6	6X6	6X6	6X6	8X8	5	8X8	5	3x6				
	UP TO													
TO .	8	6X0	0X0	6X6	988	8X8	<u>``</u>	10X10	<u> </u>	3X6				
	UP TO												1	
10	10	6X6	6X6	8X8	8X8	8X8	5	10X12	5	3X6			ļ	ł
	See													
	Note L								ļ				ļ	l
	UP TO													
10	6	6X8	6X8	6X8	8X8	8X8	5	<u>10X10</u>	5	<u>4x6</u>				
	UP TO													
то	8	888	8X8	8X8	<u>8x8</u>	8X8	5	12X12	5	4X6		[
	See								· · ·					
15	Note L								<u> </u>					·
	See			·										
	Notel ·]												
	UP TO	0.00				0			·					
15	6	828	888	BXB	8X10	8X10	<u> </u>	10212	<u> </u>	4X6				
	See													
то	Note 1]					
	See								Į					
20	NOLE !								·				······	
	See							·						
J	Note							I	I	L				l <u></u>
OVER	SEE NOTE	E 1	•										r	
20														

* Douglas fir or equivalent with a bending strength not less than 1500 psi.

** Hanufactured members of equivalent strength may be substituted for wood.

BILLING CODE 4510-25-C

. . . .

Appendix D to Subpart P

Aluminum Hydraulia Shoring for Trenches

(al Score. This appendix contains information that can be used when aluminum hydraulic snoring is provided as a method of protection against cave-ins in trenches that do not exceed 20 feet (6.1m) in depth. This appendix must be used when design of the aluminum hydraulic protective system cannot be performed in accordance with

} 1929.652(c)(2).

(b) Soil Classification. In order to use data presented in this appendix, the soil type or types in which the excavation is made must first be determined using the soil classification method set forth in appendix A of subpart P of part 1925.

 (c) Presentation of Information.
 Information is presented in several forms as follows:

(1) Information is presented in tabular form in Tables D-1.1, D-1.2, D-1.3 and E-1.4. Each toble presents the maximum vertical and horizontal spacings that may be used with various aluminum member sizes and various hydraulic cylinder sizes. Each table contains data only for the particular soil type in which the excavation or portion of the excavation is made. Tables D-1.1 and D-1.2 are for vertical shores in Types A and B soil. Tables D-1.3 and D1.4 are for horizontal waler systems in Types B and C soil.

(2) Information concerning the basis of the tabular data and the limitations of the data is presented in paragraph (d) of this appendix.

(3) information explaining the use of the tabular data is presented in paragraph (e) of this appendix.

(4) Information illustrating the use of the tabular data is presented in paragraph [f] of this appendix.

(5) Miscellaneous notations (footnotes) regarding Table D-1.1 through D-1.4 are presented in parsgraph (g) of this appendix.

(6) Figures, illustrating typical installations of hydraulic shoring, are included just prior to the Tables. The illustrations page is entitled "Aluminum Hydraulic Shoring: Typical Installations."

(d) Basis and limitations of the data.
(1) Vertical shore rails and horizontal wales are those that meet the Section Modulus requirements in the D-1 Tables. Aluminum material is 6061-T6 or material of equivalent strength and properties.

(2) Hydraulic cylinders specifications. (1) 2inch cylinders shall be a minimum 2-inch inside diameter with a minimum safe working capacity of no less than 18.000 pounds axial compressive load at maximum extension. Maximum extension is to include full range of dylinder extensions as recommended by product manufacture.

(ii) 3-inch cylinders shell be a minimum 3inch inside diameter with a safe working cubacity of not less than 30,000 pounds axial compressive load at extensions as recommended by product manufacturer. -

(3) Limitation of application.

(i) It is not intended that the aluminum hydraulic specification apply to every situation that may be experienced in the field. These data were developed to apply to the situations that are most commonly expenenced in current trenching practice. Shoring systems for use in situations that are not covered by the data in this appendix must be otherwise designed as specified in § 1929.552(c).

(ii) When any of the following conditions are present, the members specified in the Tables are not considered adequate. In this case, an alternative aluminum hydraulic shoring system or other type of protective system must be designed in accordance with § 1928.852.

(A) When vertical loads imposed on cross braces exceed a 100 Pound gravity load distributed on a one loot section of the center of the hydraulic cylinder.

(B) When surcharge loads are present from equipment weighing in excess of 20,000 pounds.

(C) When only the lower portion or a trench is shored and the remaining portion of the trench is sloped or benched unless: The sloped portion is sloped at an angle less steep than three horizontal to one vertical: or the members are selected from the tables for use at a dopth which is determined from the top of the overall trench, and not from the top of the sloped portion.

(e) Use of Tables D-1.1, D-1.2, D-1.3 and D-1.4. The members of the shoring system that are to be selected using this information are the hydraulic cylinders, and either the vertical shores or the horizontal wales. When a waler system is used the vertical timber sheeting to be used is also selected from these tables. The Tables D-1.1 and D-1.2 for vertical shores are used in Type A and B soils that do not require sheeting. Type B soils that may require sheeting, and Type C soils that always require sheeting are found in the horizontal wale Tables D-1.3 and D-1.4. The soil type must first be determined in accordance with the soil classification system described in appendix A to suppart P of part 1928. Using the appropriate table, the selection of the size and spacing of the

1

[Appendix 0]

members is made. The selection is based on the depun and width of the trench where the members are to be installed. In these tables the vertical spacing is held constant at four feet on center. The tables show the maximum horizontal spacing of cylinders allowed for each size of wale in the water system tables, and in the vertical shore tables, the hydraulic cylinder horizontal spacing is the same as the vertical shore spacing.

(f) Example to illustrate the Use of the Taples:

(1) Example 1:

A trench dug in Type A soil is 6 feet deep and 3 feet wide. From Table D-1.1: Find vertical shores and 2 inch diameter cylinders spaced 3 feet on center (0.0.) horizontally and 4 feet on center (0.0.) vertically. (See Figures 1 & 3 for typical installations.)

(2) Example 2:

A trench is dug in Type 3 soil that does not require sheeting, 13 feet deep and 5 feet wide. From Table D-1.2: Find vertical shores and 2 such diameter cylinders spaced 3.5 feet o.c. horizontally and 4 feet o.c. vertically. (See Figures 1 & 3 for typical installations.)

(3) A trench is dug in Type B soil that does not require sheeting, but does expenence some minor raveling of the trench face. The trench is 16 feet deep and 9 feet wide. From Table D-1.2: Find vertical shores and 2 inch diameter cylinder (with special oversleeves as designated by foothote =2) spaced 5.5 feet o.c. horizontally and 4 feet o.c. vertically, piywood (per foothote (g)(7) to the D-1 Table) should be used behind the shores. (See Figures 2 & 3 for typical installations.)

(4) Example 4: A trench is dug in previously disturbed Type 3 soil, with characteristics of a Type C soil, and will require sheeting. The trench is 18 feet deep and 12 feet wide, 8 foot horizontal spacing between cylinders is desired for working space. From Table D-1.3: Find horizontal wale with a section modulus of 14.0 spaced at 4 feet o.c. vertically and 3 inch diameter cylinder spaced at 9 feet maximum o.c. horizontally. 3 x 12 timber sheeting is required at close spacing vertically. (See Figure 4 for typical installation.)

(5) Example 5: A trench is dug in Type C soil, 9 feet deep and 4 feet wide. Horizontal cylinder spacing in excess of 8 feet is desired for working spece. From Table D-1.4: Find horizontal wale with a section modulus of 7.0 and 2 inch diameter cylinders spaced at 5.5 feet o.c. homeontaily. Or, find homeontai wale with a 14.3 section modulus and 2 inch diameter cylinder spaced at 10 feet o.c. homeontaily. Both wales are spaced 4 feet o.c. vertically. 3 × 12 timber sheeting is required at close spacing vertically. (See Figure 4 for typical installation.)

(g) Footnotes, and general notes, for Tables -.D-1.1, D-1.2, D-1.3, and D-1.4.

(1) For applications other than those listed in the tables, refer to § 1928.652(c)(2) for use of manufacturer's tabulated data. For trench depths in excess of 20 feet, refer to § 1925.652(c)(2) and § 1928.652(c)(3).

(2) 2 inch diameter cylinders, at this width, shall have structural steel tube $(3.5 \times 3.5 \times 0.1375)$ oversleeves, or structural oversleeves of manufacturer's specification, extanding the full, collapsed length.

(3) Hydraulic cylinders capacities. (i) 2 inch cylinders shall be a minimum 2-inch inside diameter with a safe working capacity of not less than 13.000 pounds axial compressive load at maximum extension. Maximum extension is to include full range of cylinder extensions as recommended by product manufacturer.

(ii) 34nch cylinders shall be a minimum 3inch inside diameter with a safe work capacity of not less than 30,000 pounds axial compressive load at maximum extension. Maximum extension is to include full range of cylinder extensions as recommended by product manufacturer.

(4) All spacing indicated is measured center to center.

(5) Vertical shoring rails shall have s
 minimum section modulus of 0.40 inch.
 (6) When vertical shores are used, there

must be a minimum of three shores spaced equally, horizontally, in a group.

(7) Plywood shall be 1.125 in. thick softwood or 0.75 inch. thick, 14 ply, arctic white birch (Finland form). Please note that plywood is not intended as a structural member. but only for prevention of local raveling (sloughing of the trench face) between shores.

(8) See appendix C for timber specifications.

(9) Wales are calculated for simple span conditions.

(10) See appendix D. item (d), for basis and limitations of the data.

[Appendix D]

36

Cocupational Safety & Health Reporter

CONSTRUCTION STANDARDS

5-240 31:3182.21



Published by The SUREAU OF NATIONAL AFFAIRS. INC., Washington, D.C. 20037

TABLE D - 1.1 ALUMINUM HYDRAULIC SHORING VERTICAL SHORES FOR SOIL TYPE A

		HYDRAULIC	CYLINDERS	· · · · · · · · · · · · · · · · · · ·	
DEPTH	ΜΑΧΙΜΙΙΜ	ΜΑΧΙΜΙΙΜ	WI	DTH OF TRENCH (FI	EET)
OF TRENCH	HORIZONTAL SPACING	VERTICAL SPACING	UP TO 8	OVER 8 UP TO 12	OVER 12 UP TO 15
(FEET)	(FEET)	(FEET)			
OVER 5 UP TO 10	8				
OVER 10 UP TO 15	8	4	2 INCH DIAMETER	2 INCH DIAMETER NOTE (2)	3 INCH DIAMETER
OVER 15 UP TO 20	7				
OVER 20		NOTE (1)			

Footnotes to tables, and general notes on hydraulic shoring, are found in Appendix D, Item (g)

Note (1): See Appendix D, Item (g) (1)

. ...

Note (2): See Appendix D, Item (g) (2)

....

.. .

88

31:3182.22

÷

ABFERENCE FILE

		IIYDRAULIC	CYLINDERS		
		MINUX	IIM	DTH OF TRENCH (FE	ET)
OF	HIORIZONTAL	VERTICAL			
TRENCH	SPACING	SPACING	UP TO 8	OVER 8 UP TO 12	OVER 12 UP TO 15
(FEET)	(FEET)	(FEET)			
OVER 5 UP TO 10	8				
OVER 10 UP TO 15	6.5	4	2 INCH DIAMETER	2 INCII DIAMI:TER NOTE (2)	3 INCH DIAMETIER
OVER 15 UP TO 20	3.5				
OVER 20		(I) illon			-
Footnotes to tables, a Note (1): See Apper Note (2): See Apper	nd general notes on h dix D, ltem (g) (l) dix D, ltem (g) (2)	ydraulic shoring, arc f	ound in Appendix D, I	lem (y)	

S-240 31:3182.00

12-6-89

ALUMINUM HYDRAULIC SHORING

TABLED - 1.2

1

VERTICAL SHORES FOR SOIL TYPE B

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

•

39

[Sec. 1926.652, Table D-1.2]

TABLE D - 1.3 ALUMINUM HYDRAULIC SHORING WALER SYSTEMS FOR SOIL TYPE B

ſ		WA	LES		IIY	(DRAULIC	CYLINDI	ERS		TIMBI	R UPRI	GHTTS
	DEDTU				WI	ЭТН ОГ ТІ	ENCH (FI	ED)		ILXAM 10)	ORIZ.SP	ACING ER)
	OF TRENCH	VERTICAL SPACING	• SECTION MODULUS	UP '	TO 8	OVER 8	UP TO 12	OVER 12	UP TO15	SOLID	2 FT.	3 FT.
	(FEET)	(FEET)	(IN ³)	HORTZ. SPACING	CYLINDER DIAMETER	HORIZ. Spacing	CYLINDER DIAMETER	HORIZ. SPACING	CYLINDER DIAMETER	SFREET		
	OVER		3.5	8.0	2 IN	8.0	2 IN NOTE(2)	8.0	3 IN			
	5 UP TO	4	7.0	9.0	2 IN	9.0	2 IN NOTE(2)	9.0	3 IN			3x12
	10		14.0	12.0	3 IN	12.0	3 IN	12.0	3 IN			
ſ	OVER		3.5	6.0	2 IN	6.0	2 IN NOTE(2)	6.0	3 IN			
	10 LIP TO	4	7.0	8.0	3 IN	8.0	3 IN	8.0	3 IN		3x12	
	15		14.0	10.0	3 IN	10.0	3 IN	10.0	3 IN			
·	OVER		3.5	5.5	2 IN	5.5	2 IN NOTE(2)	5.5	3 IN			
	15 UP TO	4	7.0	6.0	3 IN	6.0	3 IN	6.0	3 IN	3x12		
	20		14.0	9.0	3 IN	9.0	3 IN	9.0	3 IN			
ſ	OVER 20			NOTE (I))							

Footnotes to tables, and general notes on hydraulic shoring, are found in Appendix D, Hem (g)

Notes (1): See Appendix D, item (g) (1)

.

Notes (2): See Appendix D, liem (g) (2)

* Consult product manufacturer and/or qualified engineer for Section Modulus of available wales.

. . .

1

8

31:3182.24

· · · ·

REFERENCE FILE

CONSTRUCTION STANDARDS

TABLED - 1.4 **ALUMINUM HYDRAULIC SHORING WALER SYSTEMS** FOR SOIL TYPE C

	WAI	ES		НУ	DRAULIC	CYLIND	IRS -		тімы	ER UPRI	GHTS
DEPTH			·	WIE	DTH OF TR	KENCH (FF	EET)		H XAM (O)	ORIZ SP 4 CENTI	ACING R)
OF	VERTICAL SPACING	SECTION MODULUS	UP	TO 8	OVER 8	UP TO 12	OVER 12	UP TO 15	SOLID	2 FT.	3 IFE.
(FEET)	(FEET)	(iN³)	HORIZ, SPACING	CYLINDER DIAMETER	HORIZ. Spacing	CYLINDER DIAMETER	HORIZ. Spacing	CYLINDER DIAMETER	SHFIA		
OVER		3.5	6.0	2 IN	6.0	2 IN Note(2)	6.0	3 IN			
- 5 UP TO	4	7.0	6.5	2 IN	6.5	2 IN NOTE(2)	6.5	3 IN	3x12		
10		14.0	10.0	3 IN	10.0	3 IN	10.0	3 IN			
OVER		3.5	4.0	2 IN	4.0	2 IN NOTE(2)	4.0	3 IN			
	4	7.0	5.5	3 IN	5.5	3 IN	5.5	3 IN	3x12		
15		14.0	8.0	3 IN	8.0	3 IN	8.0	3 IN			
OVER		3.5	3.5	2 IN	3.5	2 IN NOTE(2)	3.5	3 IN			
15 UP TO	. 4	7.0	5.0	3 IN	5.0	<u>3 IN</u>	5.0	<u>3 IN</u>	3x12		
20		14.0	6.0	3 IN	6.0	3 IN	6.0	3 IN			
OVER 20			NOTE (1))							

Footnotes to tables, and general notes on hydraulic shoring, are found in Appendix D, Item (g)

Notes (1): See Appendix D, item (g) (1)

Notes (2): See Appendix D, Item (g) (2)

* Consult product manufacturer and/or qualified engineer for Section Modulus of available wales.

9

[Sec. 1925.652. Table D-1.4]

12-5-89

Published

g 쿭

BUREAU OF

NATIONAL AFFAIRS, INC., Washington, D.C. 20037

5-2-2 31:3182.25

Appendix E to Subpart P-Alternatives to Timber Shoring





Figure 2. Pneumatic/hydraulic Shoring



Figure 3. Trench Jacks (Screw Jacks)







[Appendix E]

12-6-89

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

Appendix F to Subpart P-Selection of Protective Systems

The following figures are a graphic summary of the requirements contained in subpart ? for excavations 20 feet or less in depth. Protective systems for use in excavations more than 20 feet in depth must be designed by a registered professional engineer in accordance with § 1926.652 (b) and (c).



FIGURE 1 - PRELIMINARY DECISIONS

Occupational Safety & Health Reporter

CONSTRUCTION STANDARDS

8-242 31:3182.29



12-6-89

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

35

[Appendix F]

.

REFERENCE FILE

Shoring or shielding selected as the method of protection.

Soil classification is required when shoring or shielding is used. The excavation must comply with one of the following four options:

Option 1 \$1925.652 (c)(1) which requires Appendices A and C to be followed (e.g. timber shoring).

Option 2 §1926.652 (c)(2) which requires manufacturers data to be followed (e.g. hydraulic shoring,trench jacks, air shores, shields).

Option 3 \$1926.652 (c)(3) which requires tabulated data (see definition) to be followed (e.g. any system as per the tabulated data).

Option 4 \$1926.652 (c)(4) which requires the excavation to be designed by a registered professional engineer (e.g. any designed system).

FIGURE 3 - SHORING AND SHIELDING OPTIONS

[Appendix F]

Appendix H

.

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 1 of 5

4.04 Hazard Communication

A. Introduction

OBG Technical Services. Inc. is committed to providing a safe and healthy work environment for all employees through compliance with the Federal Hazard Communication Standards. It is recognized that certain work activities and procedures require the use of chemicals which have potentially hazardous properties. When using such chemicals, it is important that workers are aware of the identity and potentially hazardous properties of the chemicals. Thus, this section represents the Written Program and has been established to summarize the components of the Hazard Communication Program instituted by OBG Tech.

This Written Hazard Communication Program is available to all employees, their designated representatives, and representatives of the Federal Occupational Safety and Health Administration. This Program has been developed in compliance with the federal regulations contained in 29 CFR 1910.1200 and 29 CFR 1926.59. Further information on the program may be obtained from the Safety Coordinator in the Syracuse office.

B. <u>Materials Exempted</u>

The following materials are exempt from ALL aspects of the Federal Hazard Communication regulations:

- 1. Tobacco or tobacco products;
- Wood or wood products;
- 3. Articles (defined as manufactured items, intended for use in specific shape or design, which do not release, or otherwise result in exposure to a hazardous chemical);
- 4. Food, drugs, cosmetics, or alcoholic beverages in a retail establishment packaged for sale to consumers;
- 5. Any consumer product or hazardous substance (as governed by the Consumer Product Safety Commission) which is used in the workplace in the same manner as normal consumer use, which results in a frequency and duration of exposure which is not greater than exposures experienced by consumers;
- 6. Drugs in solid, final form for direct administration to the patient (i.e., tablets or pills); and
- 7. Hazardous wastes, as defined under the Solid Waste Disposal Act and the Resource Conservation and Recovery Act.

C Labeling prógram

1. No hazardous chemicals or chemical products will be used or accepted for use by OBG Tech, unless labeled with at least the following information:

i.	Identity	of the	e hazardous	chemical(s):

- ii. Appropriate hazard warnings;
- iii. Name and address (and phone number, if available) of the chemical manufacturer, importer, distributer, supplier, or other responsible party.

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 2 of 5

All OBG Tech employees are responsible for compliance with these requirements.

- 2. The identity of the chemical that appears on the manufacturer's label will be the same name to identify the chemical on the List of Hazardous Chemicais, and the MSDS for that substance.
- 3. All labels will be legible, in English, and prominently displayed on the container.
- 4. In the situation of a stationary container, the label may be replaced by a sign, placard, process sheet, batch ticket, or other means to convey the identity of the hazardous chemical and the appropriate hazard warnings. If these other forms of warning are used, they will be readily accessible to employees in their work area throughout each work shift.
- 5. The Standard does not require a label to be placed on portable containers into which hazardous chemicals are transferred from labeled containers, and which are intended only for the immediate use of the employee who performs the transfer. However, by labeling the portable container appropriately, accidental misuse of the chemical by other employees may be prevented.
- 6. No label is to be defaced or removed unless the container is immediately marked with the required information. Any container without an appropriate label should be reported immediately to the job-site Supervisor.
- 7. Exemptions from these labeling requirements are as follows:
 - Pesticides subject to the Federal Insecticide, Fungicide and Rodenticide Act:
 Food, food additives, color additives, drugs, cosmetics, or medical or veterinary devices, as governed by the Food and Drug Administration;
 - iii. Alcoholic beverages, wine or malted beverages intended for non-industrial use;
 - iv. Any consumer product or hazardous substance as defined by the Consumer Product Safety Commission;

D. Comprehensive List of Hazardous Chemicals

Every chemical or product mixture of chemicals in any of the OBG Tech jobsites which is included in any of the four following lists of hazardous chemicals is listed in the List of Hazardous Chemicals for each jobsite. This List will be kept and maintained by the job-site Supervisor in a central location at each jobsite, such as the field office trailer. The four defining lists of hazardous chemicals are as follows:

- 1. Those compounds with designated OSHA Permissible Exposure Limits, as listed in 29 CFR 1910, Subpart Z ("Toxic and Hazardous Substances");
- 2. Those compounds listed in "Threshold Limit Values for Chemical Substances in the Work Environment" as published in the latest edition by the American Conference of Governmental Industrial Hygienists;
- 3. The latest edition of the "Annual Report on Carcinogens", published by the National Toxicology Program;
- 4. The Monographs published by the International Agency for Research on Cancer.

If a product contains a mixture of chemicals, it is considered to be hazardous if it contains 1% or more of any of the chemicals contained in any of the four lists referenced above. In the case of

OBG TECHNICAL SERVICES, INC.	SECTION: 4.04
HEALTH AND SAFETY MANUAL	SUPERSEDES: None
	DATE: May 14, 1992
	REVIEW: January 3, 1993
	PAGE: 3 of 5
میں میں بین میں میں میں میں ایک اور میں میں میں میں میں میں میں <u>میں میں میں میں میں میں میں میں میں میں </u>	

carcinogens, a mixture is considered to be hazardous if it contains 0.1% or more of any of the carcinogens within the lists referenced above.

The identity of each chemical, as noted in the List of Hazardous Chemicals, will be the same as the identity that appears on the container label and on the MSDS. A copy of the List of Hazardous Chemicals will be located at each Field Office along with a corresponding MSDS file for each compound on the list. Updating of the Comprehensive List of Hazardous Chemicals will be done regularly, upon acquisition of new materials and/or elimination of "old" materials. Each job-site Supervisor or his designate within OBG Tech will be responsible for updating each jobsite's list and MSDS file.

E. Material Safety Data Sheets

Copies of Material Safety Data Sheets for all hazardous chemicals to which employees may be exposed will be kept at a central location at each jobsite, such as the field office trailer and are readily accessible to employees in the work area during each work shift. The job-site Supervisor or his designate is responsible for maintaining and updating the file of Material Safety Data Sheets.

Each Material Safety Data Sheet shall be in English and shall contain the following information:

- 1. The identity used on the label.
- 2. Physical and chemical characteristics of the hazardous chemical.
- 3. Physical and health hazards of the hazardous chemical.
- 4. The primary routes of entry.
- 5. The OSHA permissible exposure limit, ACGIH threshold limit value, and any other exposure limit used or recommended by the chemical manufacturer, importer, or employer preparing the material safety data sheet, where available.
- 6. Whether the chemical is found to be a potential carcinogen.
- 7. Precautions for safe handling.
- 8. Control measures, such engineering controls, work practice, or appropriate personal protective equipment.
- Emergency and first aid procedures.
- 10. The date of the preparation of the material safety data sheet or the last change to it; and
- 11. The name, address, and telephone number of the responsible party for preparation or distribution of the material safety data sheet.

F. Employee Information and Training

Employees working with or potentially exposed to hazardous chemicals appropriately informed and trained concerning the potential hazards of the materials to which they may be exposed at the time of their initial work assignment. Training will be appropriately supplemented whenever a new hazard is introduced into the work area.

Employee information includes:

1. 2

- The requirements of the Hazard Communication Standard;
- Any job-related activities involving hazardous chemicals;

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 4 of 5

3. The locations and availability of the Written Hazard Communication Program, the List(s) of Hazardous Chemicals, and the Material Safety Data Sheet File.

Employee training includes:

- Methods and observations that may be used to detect the presence or release of a hazardous 1. chemical in the work area. Such procedures include monitoring, visual appearance and/or odor recognition, symptoms, etc.;
- 2 The physical and health hazards of the chemicals in the work area.
- Measures employees can take to protect themselves from such hazards. This includes 3. specific procedures instituted by OBG Tech to protect employees from exposure to hazardous chemicals such as appropriate work practices, emergency procedures, and personal protective equipment suitable for use.
- The details of the Hazard Communication Program developed by OBG Tech, including an 4. explanation of the labelling system and the Material Safety Data Sheets, and how employees can obtain and use the appropriate hazard information.

The Safety Coordinator is responsible for overseeing the employee information and training program. All training is documented and is filed in the Syracuse office.

G. Provisions for Non-routine Tasks and Sub-Contractors

Before any non-routine task is performed that could involve exposure to hazardous chemicals, the supervisor for the area involved will evaluate the task for appropriate work practice procedures, potential chemical hazards of the task, safety and protective measures are reviewed and understood by the employee.

Prior to beginning work on OBG Tech jobsite, all sub-contractors will be given a copy of the Written Hazard Communication Program. The particular hazards associated with the work area(s) will be identified, along with all appropriate MSDSs for the materials to be used and/or encountered. Sub-Contractors are responsible for training their own employees. The job-site Supervisor or Foreman is responsible for verifying that training has been performed by the sub-contractors.

H. Multi-Employer Workplaces

OBG Technical Services, Inc. employees may use or store hazardous chemicals at a workplace in such a way that the employees of another employer or employers may be exposed. If this occurs, then the following steps must be taken.

- Prior to the beginning of work on a multi-employer site, the Supervisor will notify all of 1 the other contractors and sub-contractors where the MSDSs for all the hazardous chemicals, brought on-site by OBG Tech, are available.
- The Supervisor will inform the other employers of precautionary measures which need to 2 be taken to protect employees from hazardous chemicals used by OBG Tech employees during normal operating conditions and in foreseeable emergencies. This communication will also include a brief description of the container labeling system used by OBG Tech. 3.
 - OBG Technical Services, Inc. will request similar information from all the other contractors

4.

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 5 of 5

working on this site.

OBG Tech employees will receive training in the hazards of new chemicals that they will use, and the chemicals to be used by other contractors at a tool box safety meeting to be held prior to the beginning of work.

Appendix I

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 1 of 6

3.05 Confined Spaces

A. Introduction

OSHA has a standard, 29 CFR 1910.146, for permit-required confined spaces. The standard includes provisions for testing and entering confined spaces. The employee who enters a confined space may be subject to multiple hazards. The purpose of this section is to outline procedures to reduce these hazards.

B. <u>Definitions</u>

"Permit-required confined space" means a confined space (see definition below) that has one or more of the following characteristics:

- 1. Contains or has a potential to contain a hazardous atmosphere (see definition below).
- 2. Contains a material that has the potential for engulfing an entrant.
- 3. Has an internal configuration such that an entrant could be trapped or asphyxiated by inwardly converging walls, or by a floor which slopes downward and tapers to a smaller cross-section, or
- 4. Contains any other recognized serious safety or health hazard.

"Confined Space" means a space that:

- 1. Is large enough and so configured that an employee can bodily enter and perform assigned work; and
- 2. Has limited or restricted means for entry or exit (for example, tanks, vessels, silos, storage bins, hoppers, vaults, and pits are spaces that may have limited means of entry): and
- 3. Is not designed for continuous employees occupancy.

"Hazardous Atmosphere" means an atmosphere that may expose employees to the risk of death, incapacitation, impairment of ability to self-rescue, injury, or acute illness from one or more of the following causes:

- 1. Flammable gas, vapor, or mist in excess of 10 percent of its lower flammable limit (LFL);
- 2. Airborne combustible dust at a concentration that meets or exceeds its LFL (Note: this concentration may be approximated as a condition in which dust obscures vision at a distance of 5 feet (1.52 meters) or less);
- 3. Atmospheric oxygen concentration below 19.5% or above 23.5%;
- 4. Atmospheric concentration of any substance for which a dose or a permissible exposure limit is published in 29 CFR 1910 Subparts G or Z and which could result in employee exposure in excess of its dose or permissible exposure limit (Note: An atmospheric concentration of any substance that is not capable of causing death, incapacitation, impairment or ability to self-rescue, injury, or acute illness due to its health effects is not covered).

5.

Any other atmospheric condition that is immediately dangerous to life or health.

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 2 of 6

Note: For air contaminants for which OSHA has not determined a dose or permissible exposure limit, other sources of information, such as Material Safety Data Sheets that comply with the Hazard Communication Program, published information, and internal documents can provide guidance in establishing acceptable atmospheric conditions.

"Non-permit confined space" means a confined space that does not contain or, with respect to atmospheric hazards, have the potential to contain any hazard capable of causing death or serious physical harm.

C. <u>Confined Space Entry Program</u>

1. Coordination with Host Employer (Owner)

The Supervisor/Foreman in charge of entry will:

- a. Chain any available information regarding permit space hazards and entry operations from the host employer (Cwaer);
- b. Coordinate entry operations with the host employer (Owner), when both host employer personnel and OBG Tech personnel will be working near permit spaces;
- c. Inform the host employer (Owner) of the permit space program that OBG Tech personnel will follow and of any hazards confronted or created in permit spaces, either through a debriefing or during the entry operation.
- d. Follow the Host Employers (Owners) confined space entry program whenever possible.
- 2. Determine classification of Confined Space

Should OBG Tech employees need to enter a confined space that has not been classified, then the following procedures will be used:

- a. Determine whether the space meets the definition of a confined space as defined in Part B of this Section.
- b. If the space meets the definition of a confined space, then determine whether it meets the definition of a permit-required confined space. If the confined space does not meet the definition of a permit-required confined space, employees may enter the space after the appropriate safety pre-cautions have been taken.
- c. If the confined space requires a permit, refer to Part 3 of this section for the procedures to be followed under these conditions.
- 3. Preparations for Permit-Required Confined Spaces

The Supervisor/Foreman in charge of entry is responsible for:

- a. Implementing the measures necessary to prevent unauthorized entry.
- b. Identifying and evaluating the hazards before employees enter the permit space.
 - c. Specifying acceptable entry conditions on the "Safety, Work, Flame, Tank Entry, & Gas Test Permit" (An example is attached. Four-copy pressure sensitive forms are available from the Syracuse office).
 - d. Isolating the permit space and following lock-out/tag-out procedures as necessary.
 - e. Purging, inerting, flushing or ventilating the permit space as necessary to eliminate or control atmospheric hazards.
- f. Providing signs and pedestrian, vehicle, or other barriers as necessary to protect entrants from external hazards.
- g. Verifying that conditions in the permit space are acceptable for entry throughout the duration of an authorized entry.
- h. Confirming that the necessary equipment is available and that employees are trained to use it as outlined on the permit.
- i. Performing testing as described in Section 6.
- j. Verifying that at least one attendant is stationed outside the permit space into which entry

k.

I.

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 3 of 6

is authorized for the duration of the entry.

- Verifying that the emergency rescue team has been notified (see Section 10).
- Coordinating entry operations when employees of more than one employer are simultaneously as authorized attendants in a permit space, so that the employees of one employer do not endanger the employees of any other employers.
- m. Prior to entry, signing the permit to authorize entry.
- n. Posting the completed permit at the entry portal or by any other equally effective means, so that the entrants can confirm that pre-entry preparations have been completed.
- o. When a ladder is required to enter the confined space, the ladder must be used securely and must not be removed while anyone is in the confined space.
- p. Adequate lighting of an approved safety type must be provided.
- q. All necessary safety equipment to be used by the person entering the confined space as well as for the safety watch will be checked.
- r. Emergency procedures will be reviewed.
- s. Smoking is prohibited inside of and within twenty feet of the confined space.
- t. Spark proof hand tools and explosion proof equipment will be used.
- u. If weiding is to be performed in the confined space that previously or now contain combustibles, all residues, including dry scale or sediment must be removed. If it is not possible to remove all combustible materials, they must be covered with a noncombustible blanker.
- v. At least one 20 lb. ABC multi-purpose fire extinguisher must be available for instant use in a confined space containing flammable gases or vapors.
- w. Each person involved shall be trained on the hazards, as outlined in Section 4, how to recognize the hazards, and how to protect themselves from the hazards. Additionally, the attendants will be training on the use of rescue equipment and other duties outlined Section 3.

4. Training

Each employee who is required to enter a permit-required confined space will be trained in the safe performance of duties.

a.

ь.

Training will be provided to each affected employee:

(1) before an employee is assigned duties which require entry into permit-required confined spaces,

(2) whenever there is a change in permit space operations that presents a hazard about which an employee has not previously been trained, or

(3) whenever there is reason to believe either that there are deviations from the permitrequired confined space entry procedures or that there are inadequacies in the employee's knowledge or use of these procedures.

- The training will be documented by the employees name, the signatures or initials of the trainers, and the dates of training. This documentation will be filed in the employees safety file and is available for inspection.
- 5. Procedures during entry

a. The duration of the permit may not exceed the time required to complete the assigned task or job identified on the permit.

b. The Foreman/Supervisor in charge of entry will terminate and cancel the entry permit when:

(1) Entry operations covered by the entry permit have been completed; or

(2) A condition that is not allowed under the entry permit arises in or near the permit space.

6.

Procedures for Atmospheric Testing

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 4 of 6

- Before an employee enters that space, the internal atmosphere will be tested, with a calibrated direct-reading instrument to determine if acceptable entry conditions exist, for the following conditions in the order given:
 - (1) Oxygen content
 - (2) Fianmable gases and vapors
 - (3) Potential toxic air contaminants
- b. If isolation of the space is infeasible because the space is large or is part of a continuous system, such as a sewer, pre-entry testing will be performed to the extent feasible before entry is authorized and, if entry is authorized, entry conditions will be continuously monitored in the areas where authorized entrants are working.
- c. Test or monitor the permit space as necessary to determine if acceptable entry conditions are being maintained during the course of the entry operations.
- d. Evaluation and interpretation of data should be done or reviewed by the Safety Coordinator.
- e. When monitoring for entries involving a descent into atmospheres that may be stratified, the atmospheric envelope should be tested a distance of 4 feet in the direction of travel and to each side. If a sampling probe is use, the entrant's rate of progress should be slowed to accommodate the sampling speed and detector response.
- 7. Authorized Entrants

The Foreman/Supervisor in charge of entry is responsible for confirming that all authorized entrants:

- a. Know the hazards they face during entry, including information on the mode, signs or symptoms, and consequences of exposure;
- b. Property use equipment as required by this section;
- c. Communicate with the attendant as necessary to enable the attendant to monitor entrants stanus and to enable the attendant to alert entrants of the need to evacuate the space;
- d. Alert the attendant whenever:

(1) The entrant recognizes any warning sign or symptom of exposure to a dangerous situation;

(2) The entrant detects a prohibited condition;

- Exit from the permit space as quickly as possible whenever:
 - (1) An order to evacuate is given by the attendant or the entry supervisor,

(2) The entrant recognizes any warning sign or symptom of exposure to a dangerous situation

- (3) The entrant recognizes a prohibited condition, or
- (4) An evacuation alarm is activated.

8. Attendants

e.

The Foreman/Supervisor is charge of entry is responsible for confirming that attendants:

- a. Know the hazards that may be faced during entry, including information on the mode, signs or symptoms, and consequences of exposure;
- b. Is aware of possible behavioral effects of hazard exposure in authorized entrants;
- c. Initial entry will be accomplished with an attendant stationed outside for the purpose of immediate assistance.
- d. The designated attendant will never enter into the permit space for the purpose of an attempt to rescue the entrants.

ь.

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 5 of 6

- e. Attendants will use any rescue equipment provided for their use and perform other rescue and emergency duties, without entering the permit space.
- f. The attendant will be responsible for continuously maintaining an accurate count of all authorized entrants in the space.
- g. Communication between attendants and entrants will be maintained continuously during entry.
 - Attendants will order authorized entrants to evacuate a space immediately when:
 - (1) The attendant observes a prohibited condition.
 - (2) The attendant detects behavioral effects harnrd exposure.
 - (3) The attendant detects a situation outside the space which could endanger entrants.
 - (4) The attendant detects an uncontrolled situation within the permit space.

(5) The attendant is monitoring entry in more than one permit space and must focus attention on the rescue of entrants from one of those spaces,

(6) The attendant must leave the work station.

(7) If the attendant cannot effectively and safely perform all the duties required under this section.

- i. Attendants will summon rescue and other emergency services as soon as the attendant determines that the authorized entrants may need assistance to escape from the confined space hazards.
- j. Attendants will wara unauthorized persons away from the space, request that unauthorized persons exit immediately if they have entered the space, and inform authorized entrants if unauthorized persons have entered the space.
 - Attendants will perform no duties that might interfere with the attendant's primary duty to monitor and protect the authorized entrants.
- 9. Supervisor/Foreman

k.

Employees of OBG Tech authorizing or in charge of entry will:

- a. Know the hazards that may be faced during entry, including information on the mode, signs or symptoms, and consequences of exposure;
- b. Determine that the entry permit is properly filled out.
- b. Determine that the entrants and attendants are properly trained.
- c. Determine that necessary procedures, practices and equipment for safe entry are in place, including verifying rescue services are available and that the means of summoning them are operable.
- d Remove unauthorized individuals who enter or attempt to enter the permit space during entry operations.
- e. Periodically monitor to determine that confined space operations remain consistent with the terms of the entry permit and that acceptable entry conditions are present.
- f. Cancel authorization and terminate entry whenever entry conditions are not present.
- g. Take the necessary measures for concluding an entry operation, such as closing off a
- permit space and cancelling the permit, once the authorized work has been completed.
- h. The supervisor/foreman in charge of authorizing the confined space entry may also be an entrant or an attendant.
- 10. Rescue and Emergency Services
 - a. Arrangements must be made prior to entry under which a rescue team will respond to a request for rescue activities. The outside rescue team will be informed of the bazards they may confront when called to the rescue.
 - b. To facilitate non-entry rescue, retrieval systems or methods will be used whenever an authorized entrant enters a permit-required confined space, unless the retrieval equipment would increase the overall risk of entry or would not contribute to the rescue of the
| OBG TECHNICAL SERVICES. INC. | SECTION: 3.05 |
|---------------------------------------|------------------------|
| HEALTH AND SAFETY MANUAL | SUPERSEDES: 5/14/92 |
| | DATE: April 12, 1993 |
| | REVIEW: March 31, 1994 |
| · · · · · · · · · · · · · · · · · · · | PAGE: 6 of 6 |
| | |

entrant. Retrieval systems will meet the following requirements:

(1) Each authorized entrant will use a chest or full body harness, with a retrieval line attached at the center of the entrants back near shoulder level, or above the entrant's head. Wristlets may be used in lieu of the chest or full body harness if it can be demonstrated that the use of a chest or full body harness is infeasible or creates a greater hazard and that the use of wristlets is the safest and most effective alternative.

(2) The other end of the retrieval line will be attached to a mechanical device or fixed point outside the permit space in such a manner that rescue can begin as soon as the rescuer becomes aware that rescue is necessary. A mechanical device will be available to retrieve personnel from vertical type permit-required confined spaces more than 5 feet deep.

c. If an injured entrant is exposed to a substance for which a Material Safety Data Sheet (MSDS) or other similar written information is required to be kept at the worksite, that MSDS or written information will be made available to the medical facility treating the exposed entrant.

D. <u>Records</u>

۰.

ł,

.

Permits will be filed with other pertinent project health and safety information, as well as with the Safety Coordinator.

F. Evaluation of Program

The Permit-Required Confined Space Entry Program will be reviewed once per year, using the cancelled permits and revise the program as necessary to confirm that employees participating in entry operations are protected from permit space hazards.

G. Exception to Permit Requirements

If the only hazard posed by the confined space is an actual or potential hazardous atmosphere and it can be demonstrated that continuous forced air ventilation alone is sufficient to maintain that confined space for safe entry, then a permit is not required under the regulations. However, it is OBG Tech's policy to require a permit for entries that meet the definition of a "permit-required confined space" (see Section B). If a project requires regular entry into a confined space that does not require a permit under the regulations, the Safety Coordinator must be contacted and the permit waiver will be considered on a project by project basis. Special procedures to be followed under these conditions will be developed by the Safety Coordinator.

OEG TECHNICAL SERVICES INC. SAFETY, WORK, FLAME, TANK ENTRY & GAS TEST PERMIT

PROJECT: LOCATION OF WORK: PLANT IPROJ.NUMBER

FLCOR

DESCRIPTION OF WORK TO BE PERFORMED:

SPECIAL SAFETY PRECAUTIONS TO BE OBSERVED:

HAZARD COMMUNICATION Work crew supervisor and all work crew personel have been notified of chemicals and physical hazards in the work area, MSDS location, sarety equipment required and emergency instructions. Work personnel have been instructed to report any unsafe or unusual conditions.

Gcqg:es :	Ventiation Equip.	Safety Be	eit	1
Face Shield	Water Hose	i Plastic A	pron	
Nitrile or Vinvi Gloves .	Vinvi Suit	Uncoated	Tyvek Suit	ļ
Ruccer Boots	Acid Suit	Coated T	yvek Suit	1
Hall-lace Resourator	Air-ine Resourator	Ladder/S	carfoid	1
Full-lace Resolvator	Cannoge Type	Non-spa	rking Tools	,
GFI I	Ere Extinguisher	Vaporpro	of Ext. Lign	t
Locxeut I	Grounding Equipment	Other		
Communication Equip: Barriers & Signs		Cther		
			1.10	
UREUN EACH QUESTION: Y	ES, NO CHINCI APPUCABLE	<u> </u>		I NA
* Have valves been properly			_	
* has electric been locked or	en? Test iccal starter button.	1	i	1
* has night heen disconder	Ted or Disover off?	1	1	i

Every hrs.

Confined Soa	de entry
Potential Haz	ards: -
Satar Superai	
Citity Subervi	
Attendant(s):	
	······································
Entrants:	
	·
	· · · · · · · · · · · · · · · · · · ·
Rescue Proce	oures (Equipment to use and
Numpers to C	ail):
· -	
Monitoring Pr	ocedures:
Communicati	an Brandurom

OCKOUT # Locks	# Tags	Air 1	ests P	EL.	Required?	PreEntry	Initial	Chour	2 hours	4 hours	6 hours	8 hours
	1.	Öxy	gen 11	9.5 - 23%	4	1	1				1	1
	1	LEL	10	9.46			1		1	1		1
			3	5 000	1	i	Ī		1	1	T	1
	1	i H2S	11) ppm	1	ī —	1			1	1	1
	ł	Vola	tiles 5	opmSTE!	j		1		_	1	7	ļ
	i	Cthe	ers		1	-	1			1		1

1

Authorizations	
Owner:	
Agent:	
Subcontractor:	

Has vessel/cloing been drained.cleaned or purgeo?

Is a stand-by fire watch required? If so, present? Tests have been conducted for Flammable Vapors?

Can sparks ignite material in vicinity sewers, lower floors?

Tests for Toxic Gases have been conducted and approved?

Have Safety Belt and Harness been provided for work?

Has OSHA approved shoring been provided for work?

Is Gas or Air test necessary? Cont?

Tests have been conduted for Cxygen?

Has a pre-entry briefing been conducted?

Is adjacent equipment sale?

PERMIT VALIDATION	(Valid for one shift only)	:
FROM: DATE	TIME	
TO: DATE	TIME	

Work Completed?	Yes	No	NA	Comments:		
Lockout/Tagout Protect	ction has	been re	moved an	id equipment can	be placed into service	Yes
Protection Removed B	ly:			Time:	Date:	

APPENDIX C

.

SPDES PERMIT

91-20-2a (1/89)		•		SPDES No .:	<u> #734052</u>
				Part 1, Page	<u>1</u> of
ACCURATE DIE CASTING SIT	E, ONONDAGA COUN	TY			
EFFLUENT LIMITATIONS A	AND MONITORING REC	UIREMENTS			
During the period beginning	December 1,	1993			
and lasting until	December 1,	1998		·	
the discharges from the permitte	d facility shall be limited	i and monitored	by the perr	nittee as specified	below:
• .				:	
				Monitoring	a Requiremen
Outfall Number &	Discharoe	Limitations		Measurement	Sar
Effluent Parameter	Daily Avg.	Daily Max.	Units	Frequency	T
Remedial Discharge to Bishop Bi	rook (0-66-11-P26-37-6-2	2-4), Class <u>C (TS</u>)	È		
Flow	Monitor	150.000	GPD	Continuous	Meter
σH	(6.5 - 8.5)		SU	2/week	Grab
Solids, Total Suspended	Monitor	20	mg/l	Weekly	3-hr. comp
Solids, Total Dissolved	Monitor	500	mg/l	Weekly	3-hr. comp
CBOD ₅	Monitor	Monitor	mg/l	Monthly	3-hr. comp
TKN	Monitor	Monitor	mg/i	Monthly	3-hr. comp
	Monitor	15 7	mg/l	Monthly	Calculated
Dissolved Oxygen	Monitor	/ mun.	mg/i	MOILLINY	Giab
Aluminum, Dissolved	Monitor	200	µg∕l	2/month	3-hr. comp
Antimony, Total	Monitor	100	µg/l	2/month	3-hr. comp
Chromium, Total	Monitor	500	i∖gų	2/month	3-hr. comp
Cobait. Total	Monitor	10	µg/l	2/month	3-hr. comp.
Copper, Total	Montor	100	μg/I	2/month	3-nr. comp.
ron, lotal	Monitor	300 -	μg/I	2/monut	3-hr. comp.
Jeau, Total	Monitor	20 10 B	י∕פ י ו עם/ו	2/month	3-hr. comp.
Nickel. Total	Monitor	200	<u>на/i</u>	2/month	3-hr. comp.
Silver, Total	Monitor	100	μg/i	2/month	3-hr. comp.
anadium, Total	Monitor	30 .	μg/l	2/month	3-hr. comp.
Eine, Total	Monitor	300	µg∕ì	2/month	3-hr. comp.
Cis-1,2-Dichloroethylene	Monitor	10	`µg∕l	2/month	Grab
rans-1,2-Dichloroethylene	Monitor	10	μg/i	2/month	Grab
Aethylene Chloride	Monitor	50	μg/l	2/month	Grab
,1,2,2-Tetrachloroethane	Monitor	30	μg/l	2/month	Grad
	Monitor	20	μg/!		Grab
	Monitor	2U ·	49/l	2/month	Grab
cetone	Monitor	1000	۲ <u>۳۷</u> ۲	2/month	Grab
	1735/A 6165/JI		P9/1		
-Hexanone	Monitor	1000	ua/l	2/month	Grab

SPDES No.: #734052____

Part 1, Page _2 of _2

Special Conditions:

Authorization is valid only for the period noted above but may be renewed if appropriate. A request for renewal must t received 6 months prior to the expiration date to allow for a review of .nonitoring data and reassessment of monitoring requirements.

Only site generated wastewater is authorized for treatment and discharge.

Discharge is not authorized until such time as an engineering report, plans and specifications are submitted detailing the proposed method of treatment and approval is granted by the Department.

*TOD = $1.5 \times \text{CBOD}_5 + 4.5 \times \text{TKN}$.

SPDES PERMIT FACT SHEET

Prepared by:	Shavne Mitchell Date: <u>12/02/93</u>
Company: <u>Accurate Die Casting</u>	Permit No.: _734052
Location: <u>Onondaga County, Fayetteville</u>	Industrial Code No.:
Industrial Segment: <u>NA</u>	Part No.:
Type of Processing & Production Rate:	
Site Remediation.	
Basis for Technology Effluent Limitations:	
Best Professional Judgement (BPJ).	
PARAMETER	BASIS FOR PERMIT CONDITION
Dutfall No.: 001 : <u>Remedial</u>	Discharge: Nominal Flow: 0.15 MGD
Flow SH TSS	BPJ WQ BPJ
TDS BOD KN	WQ Monitor Monitor
DD JD Juminum	WQ WQ
ntimony hromium obalt	BPJ BPJ WQ
opper ron ead	BPJ WQ WQ
ercury ickel ilver	WQ/Detection Limit BPJ BPJ
anadium iloride Zinc is-1.2-DCE	WQ WQ BPJ
rans-1.2-DCE ≥thylene Chloride .1.2.2-PCA	BPJ BPJ BPJ

SPDES PERMIT FACT SHEET (contd) Page 2

PARAMETER

Outfall No.: 001 ; Remedial (contd)

PCE Toluene TCE Acetone 2-Hexanone 4-Methy1-2-Pentanone

BASIS FOR PERMIT CONDITION

WQ BPJ BPJ BPJ BPJ BPJ Discharge: Nominal Flow: 0.15 MGD_

APPENDIX D

.

SOIL BORING PROCEDURES

ŝ,

SOIL BORING PROCEDURES

Soil borings shall be completed using hollow stem augers and/or other applicable drilling methods (i.e. air rotary, spun casing) to the approximate depth specified in the work plan.

Samples of the encountered subsurface materials will be collected continuously to provide physical descriptions of the material at each location. The sampling method employed will be the ASTM Method D- 1586-84 using either a standard 2-foot long, 2-inch outside diameter split-barrel sampler with a 140 lb. hammer or a 3-inch outside diameter sampler with a 300 lb. hammer. Upon retrieval of the sampling barrel, the collected sample shall be described, labelled, and placed in a glass jar.

A hydrogeologist will be on-site during the drilling and sampling operations to fully describe each soil sample including but not limited to: 1) soil type, 2) color, 3) percent recovery, 4) moisture content, 5) odor and other observations, such as organic content and cohesiveness. The Wentworth Soil Classification System will be used to describe the soil samples. The hydrogeologist will be responsible for retaining a representative portion of each sample in a glass jar labeled at a minimum with: 1) site name; 2) boring number; 3) sample interval; 4) date; and, 5) time of sample collection. Split-spoon soil samples will be field screened for volatile organic compounds (VOCs).

The VOC screening procedure will be performed by placing a portion of each soil sample into a plastic self-sealing baggie or a glass jar sealed with aluminum foil. The sample will be retained until it equilibrates with ambient temperature. The sample will then be gently agitated to volatilize potential VOCs and the headspace will be monitored with a PID.

The drilling contractor will be responsible for obtaining accurate and representative samples, informing the supervising hydrogeologist of changes in drilling pressure, and keeping a separate general log of soils encountered. Included in this log will be a record of blow counts (i.e. the number of blows from a 140 or 300 pound soil sampling drive weight required to drive the split barrel sampler 6 inches).

Soil sampling equipment, including split spoon samplers, will be decontaminated prior to the initial boring and following advancement of each successive boring in accordance with the decontamination protocol provided below:

Equipment Decontamination

A decontamination pad will be constructed adjacent to the PCB/PAH/VOC soils area for cleaning the drill rig and apparatus. The pad will be lined with polyethylene sheeting and sloped and curbed. Wash and rinse water will be pumped from the decontamination pad and collected in 55-gallon drums. Once the collected liquids are characterized, the drums will be transported to an appropriate off-site treatment or disposal facility.

Decontamination procedures will be followed for drilling and sampling activities. The split barrel sampling equipment will be decontaminated after each use. Drilling equipment mobilized to the site will receive an initial decontamination. Decontamination will consist of steam cleaning of the entire rig and associated equipment to the satisfaction of the hydrogeologist. The rear portion of the drill rig will be decontaminated as necessary by steam cleaning between soil boring installations. In addition, equipment used for sample collection will be decontaminated using a high pressure steam cleaner to remove soil and volatilize organics. Drilling equipment will be decontaminated prior to removing the equipment from the site.

The field sampling equipment cleaning and decontamination procedures will be as follows:

- 1. Non-phosphate detergent wash.
- 2. Tap water rinse.
- 3. Distilled water rinse

Information concerning decontamination methodology, date, time, and personnel will be recorded in the field log book.

APPENDIX E

GROUND WATER MONITORING PLAN OUTLINE

DRAFT TABLE OF CONTENTS

Ground Water Monitoring Plan

1. Introduction

- 1.1 General
- 1.2 Purpose of Remediation System
 - 1.2.1 Basis of Operation
 - 1.2.2 Remedial Action Objectives
- 1.3 Background Ground Water Quality

2. Ground Water Monitoring Requirements

- 2.1 General
- 2.2 Quarterly Sampling and Analyses
- 2.3 Annual Sampling and Analyses
- 2.4 Laboratory QC Samples
- 2.5 Ground Water Level Measurements
- 2.6 Submittals to the NYSDEC
- 3. Operations Evaluation
 - 3.1 Criteria for Evaluating System Effectiveness
 - 3.2 Procedure to Petition Continued Operations

Tables

- 1. Monitoring Well Construction Details
- 2. Ground Water Monitoring Schedule

Figures

- 1. Site Map
- 2. Site Plan

Appendices

A. Ground Water Sampling Procedures

Separately Bound Documents

• Health & Safety Plan

APPENDIX F

.

PROGRESS REPORT FORMS

•

FORMER ACCURATE DIE CASTING SITE FAYETTEVILLE, NEW YORK

Monthly Progress Report for: _____

(a) Activities Performed

This section will provide a brief description of the activities completed and documents submitted to or received from the NYSDEC during the month for which this report was prepared. For activities not complete, an estimated percentage of completion will be provided.

(b) Sampling and Test Results

This section will present the data obtained during the month as a result of sampling and test efforts. However, an interpretation of the data, and conclusions made based on the data, may not be presented in this Report. Interpretations and conclusions will be presented within the documents proposed within the RD/RA Plan.

(c) **Projected Activities**

This section will present a brief description of the activities proposed to be undertaken during the next month.

(d) **Project Schedule**

This section will identify if there have been any conditions encountered that may delay completion of the project. Included will be a brief description of the condition, measures taken to remedy it, and anticipated impact to the completion of the remedial efforts.

(e) Work Plan Modifications

This section will identify NYSDEC approved modifications made during the month, and identify proposed modifications that may be necessary in the future.

(f) Activities in support of Community Relations Plan

This section will identify activities performed and proposed in support of the Community Relations Plan.

APPENDIX G

•

OPERATION AND MAINTENANCE PLAN OUTLINE

DRAFT TABLE OF CONTENTS

Operation and Maintenance Manual

- i. Emergency Telephone Numbers/Project Contacts
- 1. Introduction
 - 1.1 General
 - 1.2 Purpose of Remediation System
 - 1.3 Basis of Operation
 - 1.4 Purpose of O&M Manual
 - 1.5 Operation Precautions
- 2. Operator Requirements
 - 2.1 General
 - 2.2 Operator Qualifications
 - 2.3 Operator Responsibilities
 - 2.3.1 System Performance Monitoring and Recordkeeping
 - 2.3.2 Preventative Maintenance and Repair
 - 2.3.3 Environmental Monitoring Sample Collection
 - 2.3.4 SPDES Discharge Monitoring Reports
- 3. System Components
 - 3.1 General
 - 3.2 Control Logic and Alarms
 - 3.3 Recovery Wells
 - 3.4 Influent Stabilization Tank
 - 3.5 Bag Filters
 - 3.5 Granular Activated Carbon (GAC) Vessels
 - 3.6 Treated Water Discharge
- 4. Disposal of Waste GAC
- 5. Emergency Contingency Plan
 - 5.1 Emergency Spill Response
 - 5.2 Personal Injury
 - 5.3 Toxic Exposures
 - 5.4 Public Notification

Tables

- 1. Normal System Gauge Readings
- 2. Preventative Maintenance and Inspection Schedule

DRAFT TABLE OF CONTENTS

Operation and Maintenance Manual (Continued)

Figures

- 1. Site Map
- 2. Site Plan
- 3. Process Schematic

Appendices

R

- A. Maintenance and Inspection Checklist
- B. Daily Monitoring Report Form
 - C. SPDES Permit

Separately Bound Documents

- Record Drawings
- Manufacturer Provided Installation, Operation and Maintenance Manuals
- Health and Safety Plan
- Ground Water Monitoring Plan

Exhibits



ļ

EXHIBIT A

. .

HEALTH AND SAFETY PLAN

0

HEALTH AND SAFETY PLAN

Remedial Design/Remedial Actions

Prepared for: Former Accurate Die Casting Site Fayetteville, NY

February 1995 Revised March 7, 1995



O'Brien & Gere Technical Services, Inc. 5000 Brittonfield Pkwy. E. Syracuse, NY 13057 (315) 437-6400



This Health and Safety Plan has been prepared to meet the requirements of 29 CFR 1926.65, Hazardous Waste Operatons and Emergency Response.

Signed,

OBG TECHNICAL SERVICES, INC.

cria

Jeffrey R. Parsons, CIH Corporate Safety Coordinator

Revision Summary

.

.

Date	DEC Comments to HSP	Revision Incorporated
3/7/95	Particulate monitoring should be included for the work zone.	Section 5.3 page 12
3/7/95	In Table 2, hazard control such as the respiratory protection was not provided for the inhalation of contaminated materials.	Respiratory protection is listed as a hazard control for protection from inhaling contaminated materials. Personal protective equipment (PPE) levels are listed in the hazard control column along with a description in Section 4.3, Site- specific personal protection. Further upgrade in PPE level to include respiratory protection is also outlined in the Air Monitoring Section, Table 3 & 4 of this HSP.
3/7/95	In Table 3, action level 200 ppm for upgrading to Level B seems to be high.	Action levels have been changed to accommodate levels of TCE are encountered. Action levels are now 100 ppm and 300 ppm. See page 11.
3/7/95	Section 7 should include the DEC's toll free number 1-800-342-9296 and the regional office telephone number (315)426-7551 with Charlie Branagh	Section 7.3 on page 15.
3/7/95	As proposed for the previous IRM activities, documentation air monitoring should be carried out at the site during the implementation of the excavation work.	Section 1.1.7 and section 5's introduction paragraph. Also see attachments 5-7.

TABLE OF CONTENTS

SECTION 1	Introd	uction	1
	1.1	Corporate Protocols	
		1.1.1 Employee Training Assig	gnments
		1.1.2 Personal Protective Equi	pment
		1.1.3 Medical Surveillance	-
		1.1.4 Air Monitoring	
		1.1.5 Site Control	
		1.1.6 Confined Space Entry Pr	rocedures
		1.1.7 Health and Safety Forms	ł
	1.2	Site-Specific Protocols	
SECTION 2	Backg	ound and Scope of Work	3
	2.1	General	
	2.2	Background	
	2.3	NYSDEC Selected Remedial Alternative	
	2.4	Scope of Work	
SECTION 3	Projec	Personnel	. 6
SECTION 4	Hazar	Evaluation and Hazard Control	7
	4.1	Potential Health Hazards	
	4.2	Operations and Tasks to be Performed	
	4.3	Site-Specific Personal Protection	
SECTION 5	Air M	onitoring and Action Levels	11
	5.1	Wind Indicator	
	5.2	Air Monitoring Equipment to be used	
	5.3	Work Zone Air Monitoring Procedures an	d Action Levels
	5.4	Community Air Monitoring Plan	
SECTION 6	Site C	ontrol	14
	6.1	Potential Confined Space Areas	
	6.2	Work Areas	
SECTION 7	Emerg	ency Response and Contingency Plan	15
	7.1	Directions to Hospital	
	7.2	Location of Nearest Available Telephone	· · ·
	7.3	Emergency Telephone Numbers	
	7.4	Safe Refuge	
	7.5	Spill Control	

.

.

TABLE OF CONTENTS cont.

EXHIBITS

•

•

Table 1	Summary of Potential Health Effects	8
Table 2	Hazard Evaluation and Control	9
Table 3	Work Zone Action Levels	11
Table 4	Community Action Levels	13
	· ·	

Figure 1	Site Map	
Figure 2	Site Plan	
Figure 3	Hospital Route Map	

ATTACHMENTS

HSP Compliance Agreement
Accident Investigation Form
Toolbox Meeting Record
Entry/Exit Log
Air Monitoring Instrument Calibration Record
Air Monitoring Log
Air Sampling Form
Quantitative Fit Test Form

APPENDICES

Appendix A	Health and Safety Training
Appendix B	EPA Levels of Personal Protective Equipment
Appendix C	Medical Surveillance
Appendix D	Airborne Materials Exposure
Appendix E	Hazardous Waste Operations
Appendix F	Respiratory Protection
Appendix G	Excavations
Appendix H	Hazardous Communication
Appendix I	Confined Spaces

SECTION 1 - INTRODUCTION

The following Health and Safety Plan (HSP) has been developed in accordance with the requirements set forth in 29 CFR Part 1926.65, *Hazardous Waste Operations and Emergency Response*, and in conjunction with the O'Brien & Gere Technical Services (OBG Tech) Hazardous Waste Operations Standard Operating Procedures for OBG Technical Services' on-site activities. OBG Tech's Hazardous Waste Operations Standard Operating Procedures are in Section 4.05 of OBG Technical Services' Corporate Health and Safety Manual (see Appendix E). Section 4.05 as well as other pertinent Sections are appended to this HSP (Appendices A through I).

The purpose of this HSP is to assign site-specific responsibilities, site-specific training requirements, establish sitespecific personnel protective requirements, and to provide guidance for site-specific contingencies that may arise for OBG Technical Services' and subcontracted employees. In addition to this site-specific HSP, a copy of the OBG Technical Services Corporate Health and Safety Manual will be kept on site for reference purposes. All OBG Tech personnel and subcontractor personnel must be familiar with this site specific HSP and the OBG Tech Corporate Health and Safety Manual prior to participation in site activities.

The front portion of this HSP is designed to include site-specific information, so that it can be separated from the Corporate protocols in the Appendices and distributed and posted as necessary.

This HSP may be provided to interested parties for informational purposes, however, the HSP is specifically intended for the conduct of activities within the scope of work for OBG Technical Services.

1.1 Corporate Protocols

The following are the sections required by 29 CFR 1926.65 that can be found in the Corporate Health & Safety Manual and are appended to this HSP.

1.1.1 Employee Training Assignments

The employee training assignments are found in Section 2.02 of the Corporate Health and Safety Manual and are included in Appendix A. The description for the hazardous waste pre-entry briefing can be found in the hazardous waste protocols in Section 4.05 of the Corporate Health and Safety Manual. The hazardous waste protocols are included in Appendix E. There are no additions to these protocols.

1.1.2 Personal Protective Equipment (PPE)

The EPA definitions of personal protective equipment (PPE) and donning and decontamination procedures are found in Section 5.01 of the Corporate Health and Safety Manual, which is included in Appendix B. The site-specific personal protective equipment can be found in Section 4.

1.1.3 Medical Surveillance

The Corporate Medical Surveillance program can be found in Section 2.04 and is included in Appendix C. There are no site-specific additions to these protocols.

1.1.4 Air Monitoring

The corporate air monitoring program and calibration and maintenance procedures for air

monitoring equipment is found in Section 4.03 of the Corporate Health and Safety Manual and is attached in Appendix D. The site-specific air monitoring program is found in Section 5.

1.1.5 Site Control

The Corporate Site Control Program is found in the Hazardous Waste Protocols, attached in Appendix E. The site-specific site control program can be found in Section 6.

1.1.6 Confined Space Entry Procedures

The Corporate Confined Space Entry Procedures are found in Section 3.05 of the Corporate Health and Safety Manual and are attached in Appendix I. Site-specific confined space entry information is included in Section 6.1.

1.1.7 Health and Safety Forms

The following OBG Tech health and safety forms will be available for use on-site:

- Accident Investigation Form
- Health and Safety Plan Compliance Form
- Toolbox Meeting Record
- Entry/Exit Log
- Air Monitoring Log
- Air Monitoring Instrument Calibration Record
- Air Sampling Form
- Quantitative Fit Test Form

1.2 Site-Specific Protocols

The following are the sections required by 29 CFR 1910.120 that can be found in this HSP. Section 2 is the project background and Scope of Work. Personnel involved in the project can be found in Section 3. Section 4 includes the hazard evaluation and hazard control. Air monitoring procedures are included in Section 5. Section 6 outlines the site-specific site control measures. Section 7 includes the emergency response and spill control procedures.

SECTION 2 - BACKGROUND AND SCOPE OF WORK

2.1 General

This document is the Health and Safety Plan prepared in support of the Remedial Design/Remedial Action (RD/RA) activities to be conducted at the former Accurate Die Casting Site located at 547 East Genesee Street in the Village of Fayetteville, New York (Figure 1). Based on the results of the Remedial Investigation and Feasibility Study completed in August 1994, the New York State Department of Environmental Conservation (NYSDEC) has published a Record of Decision (ROD) directing that remedial design/remedial action be performed.

This section presents background information pertaining to the site, a description of the selected remedy for the site, and the RD/RA activities to be conducted on-site.

2.2 Background

The facility was constructed in 1950 as a die casting industry. Die Casting activities were conducted at the site until mid-1988 when the then owners, the former Accurate Die Casting Corporation and George and Theresa Slyman, abandoned the facility and filed for bankruptcy. In December 1988, ITT Commercial Finance Corporation (ITT) assumed control of the property as mortgagee-in-possession as a result of the foreclosure process.

Upon acquiring the property, ITT conducted a Phase I (June 1989) and Phase II (September 1990) environmental assessment. During the Phase II environmental assessment:

- Approximately 70 drums of waste found at the site after foreclosure and located inside the building were characterized and disposed;
- Sludge from an abandoned trichloroethylene (TCE) degreasing system was removed and the system was decontaminated; and
- A TCE free product pool, which was discovered adjacent to and outside the northeast corner of the building proximal to monitoring well MW-3 (Figure 2), was pumped and disposed until no TCE free product was detected in samples.

Upon reviewing the Phase II assessment of the Former Accurate Die Casting site, the NYSDEC required that a Remedial Investigation be conducted. The investigation was completed in accordance with the Remedial Investigation (RI)/Feasibility Study (FS) Work Plan (Stearns & Wheler, May 1992). In the RI Report (Stearns & Wheler, December 1993), it was concluded among other things that:

- Ground water quality had been impacted by TCE. An overburden TCE plume extends from outside the northeast corner of the facility north towards Bishop Brook;
- The highest TCE concentrations in soils were observed at about 20 to 25 feet below grade outside the northeast corner of the facility, at the interface between the sand/gravel and clayey till layers;
- Surface and subsurface soils contain residues of polychlorinated biphenyls (PCBs), polyaromatic hydrocarbons (PAH), and volatile organic compounds (VOCs) in the location referred to as the PCB/PAH/VOC Soils Area shown on Figure 2; and
- The material inside the abandoned septic tank contains levels of zinc that could present a risk to the environment.

In May 1994, an Interim Remedial Measure Work Plan was prepared to separately address the soils exhibiting TCE outside the northeast corner of the facility and the TCE impacted overburden aquifer. The IRM Work Plan was subsequently approved by the NYSDEC in a letter dated May 23, 1994. In accordance with the approved Work Plan, the soil remediation activities were initiated on May 24, 1994 and completed on June 22, 1994, as documented in the Summary Report dated October 1994, and activities to address the overburden aquifer were initiated in September 1994.

As part of the soil remediation activities conducted outside the northeast corner of the facility, approximately 3,000 cubic yards (CY) of soil were excavated. Of that amount, approximately 1,250 CY of soil exhibited TCE levels above the established Remedial Action Objective (RAO) of 0.7 mg/kg, but below 15 mg/kg. The soil exhibiting TCE above 0.7 mg/kg was processed on site through a mechanical volatilization system. Once processed, the average residual TCE concentration was 0.13 mg/kg. The treated soils, as well as the excavated soils that did not contain TCE above the RAO, were subsequently used to backfill the excavation.

In association with the soil remediation activities conducted outside the northeast corner of the facility, a ground water collection sump was constructed on top of the clay layer at the base of the excavation, prior to backfilling. The sump is located approximately 20 feet from the north wall of the east building addition, and approximately 20 feet east of the main building, as shown in Figure 2. The purpose of the ground water collection sump is to monitor the quality and facilitate removal of overburden ground water in the previously excavated area.

In accordance with approved IRM Work Plan (O'Brien & Gere, May 1994), an overburden aquifer test recovery well, RW-1, was also installed on site (Figure 2) for the purpose of conducting a 24-hour pump test and hydrogeologic evaluation. The results of the 24-hour pump test and hydrogeologic evaluation are presented in the Basis of Design Report (O'Brien & Gere, December 1994) which also presents a description of the proposed overburden ground water recovery and treatment system.

Upon approval of the Basis of Design Report dated December 1994 by the NYSDEC, the overburden ground water treatment system will be constructed in accordance with IRM Work Plan.

2.3 NYSDEC Selected Remedial Alternative

Based upon the results of the RI/FS at the former Accurate Die Casting site and the criteria identified for the evaluation of alternatives, the NYSDEC has further directed that the contaminated soil and sludge be excavated and disposed off-site and the TCE impacted bedrock ground water be extracted and treated onsite. The components of the selected remedy identified in the ROD are as follows:

- The PCB/PAH/VOC soils from the former oil spill area, located on the northwest portion of the site, will be excavated and disposed of in a permitted landfill. The excavated area will be backfilled with clean soil;
- The sludge from the septic tank, located on the north-east portion of the site, will be excavated and disposed of in a permitted landfill;
- The TCE impacted bedrock ground water will be extracted and treated on-site. The treated ground water will be discharged to Bishop Brook;
- Soil samples will be obtained from the area northeast of the facility, where soils containing TCE were excavated as part of an Interim Remedial Measure, to verify that a potentially significant source of TCE is no longer present in the vadose zone soil; and
- A long-term ground water monitoring program will be implemented to monitor the effectiveness of the ground water (overburden and bedrock) and soil remediation program.

2.4 Scope of Work

The RD/RA activities to be performed to implement the NYSDEC selected remedial alternative are described in the RD/RA Work Plan dated February 1995. The specific field activities for which this HSP has been prepared include:

- The completion of soil borings, in the area outside the northeast corner of the facility, to obtain soil samples where TCE contaminated soils were previously excavated;
 - The completion of soil borings, in the location of previous oil spill, to obtain soil

samples. The soils in this area contain residues of polychlorinated biphenyls (PCBs), polynuclear aromatic hydrocarbons (PAHs), and volatile organic compounds (VOCs);

- The excavation of soils in the PCB/PAH/VOC area, and the removal of the septic tank and its contents. The septic tank contains a sludge exhibiting levels of zinc potentially harmful to the environment.
- The completion of a bedrock aquifer ground water recovery well outside the northeast corner of the facility. This area exhibited the highest levels of TCE in the ground water on site.
- The installation of the ground water recovery and treatment systems. Construction activities will include excavation of a pipe trench between the existing recovery well RW-1 and the facility, installation of tanks and other apparatus inside the facility, and the excavation of a pipe trench between the proposed bedrock ground water recovery well RW-2 and the facility; and

• The collection of periodic ground water level measurements and samples.

SECTION 3 - PROJECT PERSONNEL

Section 4.05 of OBG Technical Services' Corporate Health and Safety Manual (see Appendix E) contains the specific responsibilities for the below personnel.

Project Officer:

Sr. Project Supervisor:

Corporate Safety Coordinator:

Health and Safety Specialist:

Kelly L. Fisher

Field Supervisor/Site Safety and Health Coordinator:

To Be Determined

Terry L. Brown, PE

Jeffrey R. Parsons, CIH

Anthony J. Geiss

SECTION 4 - HAZARD EVALUATION AND HAZARD CONTROL

4.1 Potential Health and Safety Hazards

Previous soil analysis in the PCB/PAH/VOC soils area has shown the following:

- 1. PCBs present at concentrations up to 2.6 milligrams of PCBs per kilogram of soil (mg/kg)
- 2. PAHs present at concentrations up to 113 mg/kg.
- 3. Dichloroethylene (VOCs) at concentrations up to 190 parts per million (ppm)

Septic tank sludge analysis detected the presence of several metals, but only zinc was elevated when compared to other soil samples taken from the site. No VOCs were detected.

The highest TCE concentrations were observed in the previously excavated soil area at about 20 to 25 feet below grade. In addition, ground water TCE concentrations were noted in bedrock MW-10 and bedrock MW-11 located in the previously excavated soil area.

Health and safety information for PCBs, PAHs, and VOCs (trichloroethylene and dichloroethylene) are included in Table 1. Because metals concentrations are low, they are not considered a hazard for this project.

The exposure pathways of concern are inhalation of materials volatilized from the soil and groundwater; inhalation of dusts containing the contaminants; accidental ingestion of soil and groundwater containing contaminants. Based upon anticipated site activities and prudent safety practices during site work, ingestion of site contaminants is unlikely. Due to the use of personal protective equipment, direct contact with site contaminants poses a low health hazard. Direct-reading air monitoring equipment will be used to monitor the airborne volatile organic compounds and dusts, thus minimizing the inhalation of VOCs and dust from the soil and sludge.

4.2 Operations and Tasks to be performed

OBG Technical Services employees are required to use personal protective equipment appropriate to their work task and potential exposures, as detailed in Section 5 of the OBG Technical Services' Corporate Health and Safety Manual. Specific sections related to this site include "EPA Levels of Protection for Hazardous Waste Sites" and "Respiratory Protection."

In addition, the soil amendment material(s) used during the soils aeration activities have the potential to cause health effects. All hazard information regarding the soil amendment material(s) should be reviewed by site personnel prior to initiating site treatment activities. The SSHC is responsible for administering the Hazard Communication program (Appendix H), including informing all employees about the amendment material(s) and the location of the material safety data sheet (MSDS) for the amendment material(s), as well as all other MSDS's for materials brought on-site.

The levels of protection assigned to each activity below are based on available information and represent an estimate of exposure potential and appropriate protective equipment. The SSHC may revise these levels in accordance with Section 4.05 of OBG Technical Services' Corporate Health & Safety Manual.

When dry and/or dusty conditions are observed by the SSHC or dust levels are above the action level specified in Section 5.3, a water spray will be applied to the contaminated soils. In addition, excavated materials will be covered.

7

Chemical	Location	PEL ¹	IDLH	Characteristics	Routes of Exposure	Symptoms of Exposure
Trichloroethylene	Soil Groundwater	100 ppm TWA 200 ppm Ceiling (1989 - 50 ppm TWA 200 ppm STEL)	1000 ppm	Colorless liquid with a chloroform-like odor. Lowest reported odor threshold is 0.2 ppm.	Inhalation Ingestion Contact	Headache, vertigo; visual disturbances, tremors, somnolence, nausea, vomiting; irritated eyes; dermatitis; cardiac arrhythmia, paresthesia; potential carcinogen.
Dichloroethylene	Soil	200 ppm TWA	4000 ppm	Colorless liquid with a slightly acrid, chloroform-like odor. Lowest reported odor threshold is 0.1 ppm	Inhalation Ingestion Contact	Irritation of the eyes and respiratory tract. Central nervous system depression (i.e. alcohol-like effects)
Polychlorinated Biphenyls (PCBs)	Soil	1 mg/m³	10 mg/m ³	Colorless to light colored, viscous liquid with a mild hydrocarbon odor.	Inhalation Absorption Ingestion Contact	Eye irritation, Chloracne, liver damage, carcinogen
Polynuclear Aromatic Hydrocarbons (PAHs) ²	Soil	0.2 mg/m ³	700 mg/m ³	Black or dark brown amorphous residue.	Inhalation Contact	Skin dermatitis, bronchitis, kidney damage, carcinogen

 TABLE 1 - SUMMARY OF POTENTIAL HEALTH EFFECTS

3-6-001

¹ An appeals court decision forced OSHA to revert the PELs published in 1989 to the PELs published in 1971. The PELs listed here reflect the 1971 PELs with 1989 PELs listed in parentheses.

² Exposure limits for PAHs are referenced by OSHA as coal tar pitch volatiles, benzene soluble fraction. This includes the following compounds: anthracene, benzo-a-pyrene, phenanthrene, acridine, chrysene, and pyrene.

word at 7 Water - 22:4 let - Imple 24.45 m& w/ 4.24.45 100 5 100 8

TABLE 2 - HAZARD EVALUATION AND CONTROL

Operation/Task	Hazards	Hazard Control		
MOBILIZATION				
General Site Workers performing non- intrusive activities including mobilization, equipment set-up	 Safety/Physical Hazards Noise Use of Heavy Equipment Overhead Utilities Heat or cold stress Exposure to toxic plants, insects, and reptiles 	 Level D PPE Hearing protection when necessary Eye protection when necessary Review hazard recognition and prevention at toolbox safety meeting Stay 20 feet from overhead utility lines 		
EXCAVATION AND HA	ANDLING OF CONTAMINATED SOILS			
Operator	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Noise Use of heavy equipment Overhead utilities Underground utilities Exposure to excavations Heat or cold stress Confined space entry for excavations greater than 4 feet deep. 	 Modified Level D PPE (includes skin protection - coveralls, gloves, boots, eye protection) Hearing Protection is required while operating equipment. Use trained operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety rules and inspection forms found in Appendix G Call utilities before breaking ground Review hand signals with laborer Follow confined space entry procedure. Air monitor as detailed in Section 5 		
Laborer	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Slip/trip/fall hazards Exposure to heavy equipment Noise Exposure to toxic plants, insects, and reptiles Exposure to excavations Heat or cold stress Confined space entry for excavations greater than 4 feet. 	 Modified Level D PPE Hearing Protection Review hand signals with operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety procedures (see Appendix G) when entering an excavation Follow confined space entry procedure Air monitor as detailed in Section 5 		
REMOVAL OF SOIL AND SLUDGES FROM SEPTIC TANK				
Laborer	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Noise Slip/trip/fall hazards Exposure to heavy equipment Exposure to toxic plants, flying insects, and reptiles Exposure to excavations Heat or cold stress Confined space entry 	 Modified Level D PPE Hearing protection Review hand signals with operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety procedures (see Appendix G) Follow confined space entry procedures found in Appendix I Air monitor as detailed in Section 5 		

9

Operation/Task	Hazards	Hazard Control			
SOIL BORING & SAMP	SOIL BORING & SAMPLING CONTAMINATED SOIL				
Sampler	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Exposure to heavy equipment Noise Pinch points Heat or cold stress 	 Modified Level D PPE Hearing protection Review hazard recognition and prevention at toolbox safety meeting Collect sample using excavator bucket Air monitor as detailed in Section 5 			
GROUNDWATER RECO	OVERY AND TREATMENT SYSTEMS				
Laborer	 Contact with contaminated materials Inhalation of contaminated materials Safety/Physical Hazards Slip/trip/fall hazards Exposure to heavy equipment Noise Exposure to toxic plants, insects, and reptiles Exposure to excavations Handling pipe Heat or cold stress Confined space entry 	 Modified Level D PPE Hearing protection Review hand signals with operators Review hazard recognition and prevention at toolbox safety meeting Follow excavation safety procedures (see Appendix G) when entering an excavation Follow confined space entry procedures (see Appendix I) Air monitor as detailed in Section 5 			
DECONTAMINATION ACTIVITIES					
Decontamination of Equipment	1. Contact with contaminated tools and equipment 2. Safety/Physical Hazards -Noise -Exposure to Heavy Equipment -Heat or cold stress -Burns from steam decon equip.	 Level C PPE Hearing protection Review SOPs/operating instructions for decontamination equipment 			

4.3 Site-specific Personal Protection

.

The site-specific personal protective equipment is to be used in accordance with the protocols in Section 5.01 of the Corporate Health and Safety Manual. These protocols include the definitions of the EPA levels of protection for Level B, Level C and Level D to be used on-site, as well as donning and decontamination procedures. The site-specific personal protective equipment is as follows:

Coveralls:	Poly-coated Tyvek [®] when exposed to dirt/dusts Saranex [®] -coated Tyvek [®] when exposed to site waters/liquids
Outer Gloves:	Viton Gloves (Note: Work gloves dedicated to the exclusion zone may be used over the viton gloves)
Respirator:	Level C: Full-face air purifying respirator with organic vapor cartridges and high efficiency particulate (HEPA) filters
	Level B: Positive pressure, full-facepiece Self-Contained Breathing Apparatus (SCBA) or Positive pressure, supplied-air respirator with escape SCBA

Hard hats, steel-toed boots, and safety glasses are mandatory on-site.

SECTION 5 - AIR MONITORING AND ACTION LEVELS

Air monitoring is to be performed in accordance with Appendix D (Section 4.03 of the Corporate Health and Safety Manual - Airborne Materials Exposure) and Appendix E (Section 4.05 of the Corporate Health and Safety Manual - Hazardous Waste Operations). Air monitoring results will be documented using appropriate forms (see Attachments 5, 6, & 7). Presented below is the site-specific information.

5.1 Wind Indicator

Wind direction will be monitored each day of on-site activities by using a portable wind indicator.

5.2 Air Monitoring Equipment to be used

Photoionization Detector (PID) with 10.2 eV lamp

Combustible gas/oxygen meter

Direct-Reading Dust Monitor

5.3 Work Zone Air Monitoring Procedures and Action Levels

The following describes the methods and parameters to be used on-site for PPE upgrades and work cessation. The action levels are based upon total organic vapors detected by a photoionization detector (PID) calibrated to isobutylene. PID readings will be adjusted in accordance with manufacturer's guidelines to read in ppm of dichloroethylene.

Туре	Frequency	Action Level	Action
Volatile Organic Compounds (VOCs) (Using PID) 1. Initially when a new task begins and hourly thereafter 2. When obvious contamination is encountered 3. Prior to and continuously during entry into excavations over 4 feet	 Initially when a new task begins and hourly thereafter When obvious contamination is encountered Prior to and continuously during entry into excavations 	25 ppm above background in breathing zone for 5 minutes	Increase to Level C PPE
		100 ppm above background in breathing zone for 5 minutes	 Notify SSHC Increase to Level B PPE Consider engineering controls.
	over 4 feet	300 ppm above background in breathing zone for 5 minute	 Stop work - consult SSHC; use engineering controls (e.g. ventilation) to decrease concentrations
Combustible Gases (Lower Explosive Limit - LEL)	Prior to and continuously during entry into an excavation over 4 feet	10 %	Stop work - increase ventilation to bring below 10 % LEL

TABLE 3 - WORK ZONE ACTION LEVELS

Туре	Frequency	Action Level	Action
Oxygen	Prior to and continuously during entry into an excavation over 4 feet	19.5 %	Stop work - increase ventilation to bring above 19.5 % oxygen
Particulate	Continuously downwind during excavation of contaminated materials integrated over 15 minute periods	150 ug/m ³	Monitor the upwin background level immediately. If the working site dust measurement is greater then 100 ug/m ³ , then dust suppression techniques will be implemented (e.g. using water sprays, covering excavated materials and areas)
5.4 Community Air Monitoring Plan

Real-time monitoring for volatile compounds and particulate levels at the perimeter of the work area will be performed.

Frequency	Action Level	Action
Initially when a new task begins, every fifteen minutes for the first hour and every two hours thereafter	5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area	 Stop work until emission controls (covering excavated areas) are in place and the concentrations detected by the PID are reduced to below 5 ppm at the downwind perimeter of the work area. Air monitor as directed below. When work continues monitoring at the downwind perimeter of the work area will be performed at least every one-half hour.
Monitoring will be performed within 20 feet of the perimeter of the nearest residential or commercial structure when 5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area	1 ppm above background sustained for one minute	 Stop work and continue to monitor. Contact Corporate Safety Coordinator for appropriate action. Work may continue when concentrations detected by the PID are reduced to below 1 ppm. Monitor with pumps and charcoal tubes following NIOSH 1022 (are below)
 Daily for the first week of excavation actives in contaminated area - one upwind and two downwind When 5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area is detected using PID 	1/2 of PEL after subtracting upwind from downwind	The SSHC and Project Supervisor will re-evaluate vapor suppression techniques
Continuously downwind during excavation of contaminated materials integrated over 15 minute periods	100 ug/m ^a 150 ug/m ^a	 Measure the upwind background level. Implement dust suppression techniques (e.g. using water sprays, covering excavated materials and areas) Notify the SSHC STOP WORK Measure upwind background level immediately
	Frequency Initially when a new task begins, every fifteen minutes for the first hour and every two hours thereafter Monitoring will be performed within 20 feet of the perimeter of the nearest residential or commercial structure when 5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area 1. Daily for the first week of excavation actives in contaminated area - one upwind and two downwind 2. When 5 ppm above background sustained for one minute in breathing zone at downwind sustained for one minute in breathing zone at downwind perimeter of work area is detected using PID Continuously downwind during excavation of contaminated materials integrated over 15 minute periods	FrequencyAction LevelInitially when a new task begins, every fifteen minutes for the first hour and every two hours thereafter5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work areaMonitoring will be performed within 20 feet of the perimeter of the nearest residential or commercial structure when 5 ppm above background sustained for one minute in breathing zone at downwind perimeter of work area1 ppm above background sustained for one minute1. Daily for the first week of excavation actives in contaminated area - one upwind and two downwind perimeter of work area at downwind perimeter of work area is detected using PID1/2 of PEL after subtracting upwind from downwindContinuously downwind during excavation of contaminated materials integrated over 15 minute periods100 ug/m²

TABLE 4 - COMMUNITY ACTION LEVELS

13

SECTION 6 - SITE CONTROL

The elements of site control include site work zones (exclusion zone, contamination reduction zone, and support zone), the "buddy" system, site communication, and site security. Protocols for site control can be found in Section 4.04 of the Corporate Health and Safety Manual (see Appendix E).

Only approved, trained personnel will be allowed to enter the exclusion zone and contamination reduction zone. Personnel must understand this HSP and the potential on-site hazards before being allowed on site.

6.1 Potential Confined Space Areas

The excavations (greater than 4 feet) do pose a potential confined space entry area. When entry occurs then the confined space procedures in the Corporate Health & Safety Manual will be followed (section 3.05 - see Appendix I). Specifically, the following precautions will apply:

- 1. Confined Space Entry permit will be filled out by the SSHC.
- 2. Air monitoring will be performed continuously for oxygen deficiency and LEL.
- 3. Personal protective equipment as outlined previously will be used.
- 4. Barriers and signs will be used to protect opening.
- 5. Provisions will be made for confined space rescue.

Specific procedures, such as coordination with Owner, training, entrant responsibilities. attendant responsibilities, supervisor/foreman responsibilities, and atmospheric testing can be found in Section 3.05 of the Corporate Health & Safety Manual (see Appendix I).

In addition, the safety requirements for excavations will be followed, as outlined in Section 10.06 of the Corporate Health and Safety Manual (See Appendix G).

6.2 Work Area

A map of the work areas depicting the exclusion and contamination reduction zones will be prepared and submitted with the Construction plans and performance specifications.

SECTION 7 - EMERGENCY RESPONSE AND CONTINGENCY PLAN

Sections 3.01, Medical Services and First Aid, 3.03, Emergency Action, and 4.05, Hazardous Waste Operations, of the Corporate Health and Safety Manual provide guidance on emergency response. This section provides the site-specific emergency response information.

7.1 Directions to Hospital - To be posted in all site trailers (See Figure 3).

Genesee Street West to I-481 North. I-481 to I-690 West I-690 West to Townsend Street Left onto Townsend Street Turn left onto Genesee Street Turn right onto Irving Ave The hospital on right.

7.2 Location of Nearest Available Telephone

The telephone will be located in the site field office.

7.3 Emergency Telephone numbers - To be posted in site office

Police	911
Crouse Irving Memorial Hospital	470-7111
Fire Department	911
DEC	(800)342-9296
DEC Regional - Charlie Branagh	426-7551
OBG Technical Services, Inc.	437-6400

7.4 Safe Refuge

The field office will serve as point of safe refuge. An air horn will be kept in the field office. If the air horn is sounded, all on-site personnel must return to the office.

7.5 Spill Control

١

A major spill is not anticipated at the site. Should a spill of any type occur, the personnel present at the spill should report it immediately to the SSHC and/or the OBG Tech emergency contacts, who will make arrangements for the proper clean-up of the spill and contact the appropriate local emergency groups and regulatory agencies. On-site personnel should immediately secure the area to prevent unauthorized entry into the spill area. On-site personnel must evaluate the extent of the hazard(s) and if available, utilize engineering controls and proper safety equipment to contain the spill until response personnel are on site. The emergency response personnel will be contacted immediately by SSHC in the event that the spill can not be immediately contained by on-site materials.

Health and Safety Plan Compliance



Project Name: Project Number:

This is to certify that I have read, fully understand, and agree to comply fully with the attached Health and Safety Plan furnished to me by O'BRIEN & GERE TECHNICAL SERVICES, INC.

NAME	SIGNATURE	COMPANY	DATE
		-	
		· · · · · · · · · · · · · · · · · · ·	
		•	
· ·			
			· · · ·
		<u> </u>	

Revision: 12/29/94 (File Name: HSPCOMP.FRM)

ACCIDENT INVESTIGATION FORM

FORWARD TO SAFETY COORDINATOR WITHIN 24 HOURS!



Instructions: This form must be filled out be the employees' Manager, Supervisor, or Foreman. Mark N.A. (not applicable) for those items which do not apply to case. Write "none", "unknown", or "don't know" for applicable items for which there is no answer. NEVER LEAVE A BLANK. Write a description of the incident on the back, including any information on conditions that may have contributed to the incident.

MPLO	YEE DATA										<u></u>									
Name	1								Subco	stractor?		Yes -	Nume of	Company	n.			\$\$#	:	
Job T	īde:								Time is Job:				Age	;	Sex: M					
CCIDE	INT DETAILS				_															
Date:		Days					Т	me:		Treates	ent al	u .								
Loca	ion of accident;			Job b	ing perf	formed:		-		Witnes	KG6:									
How	did the socident h	appen?			•		1			<u> </u>										
What	is the extent of th	e injury?					•													
сстре	ENT TYPE																-			
	Contact w/Ten	m. Extreme		Strain/Ov	erexertic	n .		Caught Betw	vocto		Ca	ught On		Struck B	ly .		c	onisci w/Ha	ız Subet.	
	Contact w/Elec	nt: Current		Different	Level F	sil		Suza Lovel	Fall		Cat	ught in		Struck A	gainet		0	uher:		
EASO	NS FOR WHAT	WAS DONE	OR WAS FA	ILED T	O BE D	ONE														_
	Unaware of jo	b hazard			1	Tried to	gain/sav	ve time			Ì	Acted to a	avoid disc	- notation				failuence o	f illness	
	Unaware of sa	fe methods				Defective	e hearin	g/vision				Influence	of intexic	ونمح	. <u></u>			influence o	f emotions	
	influence of fa	ligue	•		1	Tried w	avoid e	xura effort				Low level	l of job si	<u>1811</u>				Instantive	to hazarda	1
NSAFI	E CONDITIONS					r						,								
	Hazardous De	lects of equipe	neni			Inst	ednare :	sisle, spsce, e	tc,		Uaguarded		improperty placed (includes househeeping)							
	instequately p	laced against u	ndesired me	tion		lrad	oquato	ventilation			inadequately guarded			Un- or in- adequately labeled						
	Hazardous me	thods and/or p	rocedures	·		lmp	roper ill	hmination	•		Ungrounded			Improperty piled						
	Lack of or ina	dequate warnij	ng systems			Fire	or expl	osion hazard			Uninsulated			inadequate clearance for moving parts						
NSAF	E ACTS									<u></u>										
,	Cicaning, oilin	, adjusting of	moving equi	ip.	_		use of	hands or body	у рыла		Making safety device inoperative				improper use of equipment					
	Working in an	electrically ch	arged cuviro	•	!	Insticutio	n io lo	otings or surr	oundings	<u> </u>	Taking an unsafe position or posture				Using unsafe equipment					
	Failure to wear	safe personal	attire			Operatio	g or we	whing at unsul	is speed	[Fail to sccu	ne or war	8			Fai	lure to wear	PPE	
	Homeplay, dist	recting, teasin	g, etc.					· · · · ·												
CT10	NS TO PREVEN	T ACCIDENT	REOCCU	ANCE				·								- ; · · ·	,			
	Reinstruction of	f person(s) inv	volved			<u> </u>	Ac	tion to improv	e inspecti	oa		installat	tion of gu	ard or sal	'ety device		_	Equipmen	t repair o	replacemen
	Convenistion r	ocord with per	son(s) involv	ed		<u></u>	_ Ac	tion to improv	e class-u	<u>-</u>		Correct	ion of un	nearana ny	congestion	·	_	Inform all	forcmen/	Repervisions
	Discipling of p	orson(s) izvolv	red				<u>_</u>	eck with mam	ufacturer			Perman	cut reasi	gnunent o	f person		-	Order use	of safer i	macrials
Reinstruction of others doing job Action to improve dosign Order regular pre-job instructions A					Action to	шрюче	costruction													
	Improvement o	f personal pro	tective protec	tive equi	panear		<u> </u>									<u> </u>				
CASE TYPE - TO BE FILLED IN BY SAFETY COORDINATOR																				
	First Aid Injury - lost time Occupational illness Injury - No Lost Time Restricted work Fetality																			
N		-									Ne-					13 <i>01</i> 2				Dete
javes	tigned Bv:	- xo- 000009999			as 4830	2.26793 	an an an an an an an an an an an an an a		-49.94	:-*** X	Proi	act Manager		<u>eren 384</u>	2 (2011 M.L	99.47 XXXX			- 499.888 (3	
Supe	rvisor/Foremen:	-								Safety Constitutor:										
												,					•			

Revision: 12/27/94 (File Name: ACCIDENT.FRM)

SAFETY/TOOLBOX MEETING



.

DATE: _____ PROJECT NUMBER: _____

PROJECT LOCATION:

CONDUCTED BY:_____

EMPLOYEE NAME	COMPANY	Social Security #
	· · · · · · · · · · · · · · · · · · ·	•
· -		

Safety Meeting Topic(s):

Comments:

Revision: 01/03/95 (File Name: TOOLBOX.FRM)

ENTRY/EXIT LOG

O'BRIEN 5 GERE TECHNICAL SERVICES

Site:

Date:

Name	Company	Time In	Time Out	Reason
				· · · · · · · · · · · · · · · · · · ·
·	<u>, , , , , , , , , , , , , , , , , , , </u>			
	·		=	
				· · · · · · · · · · · · · · · · · · ·
			· · · · ·	
· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·			

Revision: 01/03/95 (File Name: ENTRYLOG.FRM)

Attachment 5 DIRECT READING INSTRUMENT CALIBRATION RECORD

Instrument:

Serial No.:

Calibration concentration

Date	Initial Reading	Adjusted Reading	Performed By
	· ·		· ·
	·		
· · ·			
			·
	·		
		· · · · · · · · · · · · · · · · · · ·	



TECHNICAL SERVICES

Air Monitoring Log-Organic Vapors, Oxygen, LEL

Log- Attachment 6



Date:	Project Name:	Project Location:
Job Number:	SSHC:	
Background:	Wind Direction:	Temperature:
Activities:		

Time	Location	HNU (ppm)	LEL	O ₂
			· · · · · · · · ·	
	- · · · · · · · · · · · · · · · · · · ·	·		
	· · · · · · · · · · · · · · · · · · ·			
	· · · · · · · · · · · · · · · · · · ·			
	,			
·				
· · · · · · · · · · · · · · · · · · ·				· · · · · · · · · · · · · · · · · · ·

Revision: 12/30/94 (File Name: AIRVAPOR.FRM)

.

AIR	MONIT	ORING	DATA	SHEET
-----	-------	-------	------	-------

. . .

LOCATION:

PROJECT NUMBER:____

______ SURVEY DATE:

Ритр Туре:			 Manufacturer:	Model:
Soriel Number	Pre-Cel	Post-Cel	 COMMENTS - Include Resson for Sampling, Operations, and Processes	
•				

	Y
Collection Media:	Analysis Required:

Sample: No.	Sample Media, & Pump ID No.	Person (neme & BSN) OR Semple Location:	Start Time	End Time	Duretion	Flow Rate	Shift Duration	Analysis Required
					`			
		· · · · · · · · · · · · · · · · · · ·						
		······································						
				·				
		· · · · · · · · · · · · · · · · · · ·					<u> </u>	<u> </u>
-	· · · · · · · · · · · · · · · · · · ·							

•

SAMPLER(S):

CLIENT:

OBRIEN & GERE

OUALITATIVE FIT TEST



Name:
SS#:
Company:
Respirator Brand:
Model: Size:
Procedure: Irritant Smoke Banana Oil
NOTE: Failure in any one of the following exercises indicates unacceptable respirator fit.
1 Positive and negative fit test
2 Breath normally
3 Turn head side to side
4 Nod head
5 Jog in place
6 Breath normally and read the following:
When the sunlight strikes raindrops in the air, they act like a prism and form a rainbow. The rainbow is a division of white light into many beautiful colors. These take the shape of a long round arch, with its path high above, and its two ends apparently beyond the horizon. There is, according to legend, a boiling pot of gold at one end. People look, but no one ever finds it. When a man looks for something beyond his reach, his friends say he is looking for the pot of gold at the end of the rainbow.
Test performed by: Date:

Revision: 01/03/95 (File Name: FITTEST.FRM)









HOSPITAL ROUTE MAP

FIGURE 3

يتريده والرور والمنتقية بمسابق الالتحقية المعترة والافاق سابقا فالمار براعم سابقا فال

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL

SECTION: 2.02 SUPERSEDES: 5/28/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 1 of 2

2.02 Health and Safety Training

A. Introduction

All employees must recognize and understand the potential hazards to health and safety associated with the tasks that they perform. The objectives of safety training programs are:

- 1. To make employees aware of the potential hazards they may encounter;
- 2. To provide employees with the knowledge and skill necessary to perform the work with minimal risk to employee health and safety;
- 3. To make employees aware of the purpose and limitations of equipment; and
- 4. To provide employees with information to avoid or procedures to follow in the event of an emergency.

B. <u>New Employee</u>

All new employees to OBG Technical Services, Inc. will receive training that will provide an introduction to the Safety Program, employees rights and responsibilities on safety, electrical safety, minimizing back strain on the job, Hazard Communication, and substance abuse-free workplace policy.

C. Job specific

Field Supervisors/Foreman will be responsible for training new employees on site-specific information, such as safe work practices and procedures to follow in the event of an emergency.

D. <u>Haz Com</u>

All OBG Tech employees will receive training on the OSHA Hazardous Communication Standard ("Right-to-Know"). Each employee receives a copy of the written program with their job offer letter. However, formal training is conducted to verify that all employees have been made aware of the plan, its requirements, and be informed of their right to receive information about hazardous materials.

Each supervisor is responsible for training employees on new materials introduced on specific jobs.

E. <u>Other Safety Training</u>

٠.

1. "Toolbox" Safety Meetings

There are a number of OSHA regulations which require training. Many of these are specific to jobs and should be documented by "tool box" meetings. These include: medical services and emergency action, fire protection, accident prevention, and poweroperated and woodworking hand tools, stairways and ladders, guardrails and handrails, welding and cutting, scaffolds, concrete and masonry construction, and demolition.

Supervisors/Foreman review the applicable topics in weekly "tool box" safety meetings

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL

SECTION: 2.02 SUPERSEDES: 5/18/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 2 of 2

on every project. Names of employees attending the meeting and topics covered are forwarded to the Safety Coordinator for recordkeeping purposes.

2. Quarterly Staff and Safety Meetings

Staff and safety meetings are held quarterly for all employees. These meetings cover staff and personnel issues, as well as selected topics related to safety.

3. Haz Waste Training

Employees who are assigned to perform duties on hazardous waste sites will receive the OSHA initial 40-hour health and safety training prior to on-site activities at a hazardous waste site. Applicable employees will receive yearly 8-hour refresher courses. In addition, Foreman, Supervisors, and Project Managers will receive the OSHA 8-hour health and safety supervisors training.

4. Other Safety Training

Periodically, employees will be sent to safety training courses relevant to their duties, such as confined space entry.

F. Records

The Safery Coordinator will maintain safety training records on each individual.

Appendix B

ţ

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL

SECTION 5.01 SUPERSEDES: None DATE: May 14, 1992 REVIEW: February 12,1993 PAGE: 1 of 4

Section 5.01 EPA Levels of Personal Protective Equipment

A. Introduction

Due to the nature of the business of OBG Tech, employees may be required to work on hazardous waste sites. Use of personal protective equipment (PPE) is required by OSHA. EPA has defined four Levels of Protection: Levels A. B. C. and D. These levels are defined below and may be used as a starting point for PPE on sites, but must be tailored to the specific situation.

B. Levels of Personal Protection

Level A

Level A protection provides the highest available level of respiratory, skin and eye protection. The material in the suit, gloves, and boots must be compatible with the substances involved.

- 1. Positive pressure, full-facepiece SCBA or positive pressure supplied-air respirator with escape SCBA, approved by the National Institute for Occupational Safety and Health (NIOSH).
- 2. Fully-encapsulating, chemical-resistant suit.
- 3. Outer chemical-resistant gloves.
- 4. Inner chemical-resistant gloves,
- 5. Chemical-resistant safety boots.
- 6. Two-way communications.

Options:

- 1. Cooling unit.
- 2. Coveralls.
- 3. Long-underwear.
- 4. Other PPE as required in Section 5.02 through 5.09 of this Health and Safety Manual.

Level B

Level B protection provides the highest level of respiratory protection but less skin protection than Level A. Use only when airborne chemicals are not bazardous to the skin or not capable of being absorbed through the intact skin.

1. Positive-pressure, full-facepiece SCBA or positive-pressure supplied-air respirator

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL

SECTION 5.01 SUPERSEDES: None DATE: May 14, 1992 REVIEW: February 12,1993 PAGE: 2 of 4

with escape SCBA (NIOSH approved).

- 2. Chemical-resistant suit.
- 3. Inner and outer chemical-resistant gloves.
- 4. Chemical-resistant safety boots.
- 5. Two-way communications.
 - Options:
 - 1. Cooling unit.
 - 2. Coveralls.
 - 3. Long-underwear.
 - 4. Other PPE as required in Section 5.02 through 5.09 of this Health and Safety Manual.

Level C

Level C protection provides less skin protection as Level A and a lower level of respiratory protection than Levels A and B.

- 1. Full-facepiece, air-purifying, canister-equipped respirator (NIOSH approved).
- 2. Chemical-resistant suit.
- Inner and outer chemical-resistant gloves.
- 4. Chemical-resistant safety boots or chemical-resistant boot covers.
- 5. Two-way communications.

Options:

- 1. Cooling unit.
- 2. Coveralls.
- 3. Long underwear.
- 4. Other PPE as required in Sections 5.02 through 5.09 of this Health and Safety Manual.

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL

SECTION 5.01 SUPERSEDES: None DATE: May 14, 1992 REVIEW: February 12,1993 PAGE: 3 of 4

Level D

Level D protection provides minimal skin protection and no respiratory protection.

- 1. Coveralls or long pants and long-sleeved shirt.
- 2. Safety boots.

Options:

1. Other PPE as required in Sections 5.02 through 5.09 of this Health and Safety Manual.

C. <u>PPE donning procedures</u>

- 1. Inspect the PPE before donning with the procedures outlined in Sections 5.62 through 5.09.
- 2. Make adjustments to hard hat to fit user's head, if necessary.
- 3. Standing or sitting, step into legs of the suit; evaluate proper placement of feet within the suit; then gather suit and pull sleeves over arms and secure suit front.
- 4. Put on chemical-resistant safety boots over the feet of the suit. Tape the leg cuff over the tops of the boots.
- 5. Put on air tanks and harness assembly of the SCBA (if applicable). Don the facepiece or respirator and adjust it to be secure, but comfortable. Do *not* connect the breathing hose of the SCBA. Open valve on the air tank (if applicable).
- 6. Perform negative and positive respirator facepiece seal test procedures.
- 7. Put on inner gloves.
- 8. Put on other PPE (e.g hard hat, hearing protectors).
- 9. Raise hood over head carefully so that the face seal of the respirator is not disrupted.
- 10. Connect the breathing hose while opening the main valve (if applicable).
- 11. Have assistant observe the wearer for a period of time to evaluate whether the wearer is comfortable, stable, and that the PPE is functioning properly.

D. <u>PPE Decontamination Procedures</u>

Station 1: Equipment Drop

Deposit equipment used on-site (tools, sampling devices, clipboards, etc) onto plastic drop cloths. During hot weather a cool down station may be set up within this area.

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL

SECTION 5.01 SUPERSEDES: None DATE: June 3, 1992 REVIEW: February 12,1993 PAGE: 4 of 4

Station 2: Outer Garment, Boots, and Gloves Wash and Rinse

> Scrub outer boots, outer gloves and chemical-resistant splash suit with decontamination solution or detergent water. Rinse off using large amounts of water.

Station 3:

Outer Boot and Glove Removal

Remove outer boots and gloves. Deposit in appropriate area.

Station 4:

Tank, Canister, or Mask Change

If worker leaves exclusion zone to change air tank canister, filters, or mask, this is the last step in the decontamination procedure. Worker's air tank, canister, filters, or mask is exchanged, new outer gloves and boots are donned, joints taped. and worker returns to dury.

Station 5: Boot, Gioves, and Outer Garment Removal

> Boots, chemical-resistant splash suit, inner gloves removed and deposited in separate containers lined with plastic.

Station 6: SCBA or Face-piece Removal

> SCBA backpack and/or facepiece is removed. Avoid touching face with finger. SCBA deposited on plastic sheets.

Station 7: Field Wash

Hands and face are thoroughly washed. Shower if appropriate.

E. Upgrading/Downgrading of PPE Levels

The PPE used and the overall level of protection should be reevaluated periodically as the amount of information on the site increases, and as workers as required to perform different tasks.

Reasons to upgrade the level of PPE may include:

- Known or suspected presence of skin contact hazards.
- Occurance or likely occurance of gas or vapor emission.
- Change in work task that will increase contact or potential contact with hazardous materials.

Appendix C •:.

.

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL

SECTION: 2.04 SUPERSEDES: 5/18/92 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 1 of 2

2.04 Medical Surveillance

A. <u>Introduction</u>

OBG Technical Services provides medical and environmental surveillance of the health of those individuals exposed to materials or work conditions which are associated with adverse health conditions or who participate in the Health & Exercise Program.

B. <u>Medical Examinations</u>

1. Warehouse Employees

The examination for warehouse employees, for the purposes of medical examinations, includes any employee who works on regular basis at the warehouse, and is not covered under another examination (e.g. hazardous waste examination). This is the examination for DOT driver certification. This examination is required every two years. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes the Medical Questionnaire, a Medical Examination. Sudiology Test. DOT Driver Certification, and testing for complete biled count and chemistry profile.

2. Field Employees

The examination for field employees, for the purposes of medical examinations, is any employee who visits or works at a job site on a regular basis. Field employees, for the purpose of medical examination, includes some Supervisors, Foreman, and Laborers. This examinations is required by OSHA in 29 CFR 1910.120, the Hazardous Waste Operations regulations. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes the OSHA required Medical Questionnaire. Respirator Suitability Form, a Medical Examination, Audiology Test, Pulmonary Function Test, DOT Driver Certification, and testing for complete blood count and chemistry profile, and phenols (quantitative).

3. Supervisor/Managerment Employees

A Supervisor/Management employee is any employee who periodically visits a job site (less than 30 days a year). Supervisors/Management employees, for the purpose of medical examinations, includes Officers, Project Managers, and all levels of Supervisor not covered elsewhere (i.e. hazardous waste examination). This examination is required at a frequency based on the individuals age. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes the Medical Questionaire, a Medical Examination, Audiology Test, and testing for complete blood count, and chemistry profile.

Office Personnel

4.

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL	SECTION: 2.04 SUPERSEDES: 5/18/97
	DATE: 12/31/93
· .	REVIEW: 12/31/95
	PAGE: 2 of 2

Office personnel, for the purposes of medical examinations, include administrative and office personnel. This examination fulfills the requirement for employee participation in the Firm's Health & Exercise Program.

The examination includes a Medical Examination, and testing for complete blood count and chemistry profile.

5. Drug Screen

The drug screen is mandatory for all newly hire, "regular status" employees. Temporary status employees may also require a drug screen, based upon a variety of factors. Please refer to OBG Technical Services' "Substance Abuse-Free Workplace" policy for more information.

C. <u>Frequency</u>

1. Baseline examinations

Individuals who are assigned temporarily or permanently to field work will receive a baseline examination prior to job assignment.

2. Periodic Examinations

Individuals who are assigned temporarily or permanently to field work will receive periodic examinations yearly.

3. Termination Examinations

Field employees permanently leaving the company will receive an exit examination.

4. Possible Exposure Examinations

As soon as possible upon notification by an employee that the employee has developed signs or symptoms indicating possible overexposure to hazardous substances or health hazards, or that employee has been injured or exposed above the permissible exposure limits in an emergency situation, that employee will be required to receive medical attention.

D. Examination Results

OBG Tech must receive a letter from the attending physician stating the parameters of the examination and whether or not the individual is able to work with or without restriction. This letter will be filed by the Office Administrator and a copy is distributed to the employee.

Appendix D

. .

·

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL

SECTION: 4.03 SUPERSEDES: 5/14/92 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 1 of 2

Section 4.03 Airborne Materials Exposure

A. <u>Introduction</u>

OSHA, in 29 CFR 1910.1000, specifies that an employee's exposure to substances listed in Table Z-1-A, Z-2, or Z-3 shall be limited in accordance with the requirements of the section. OSHA, in 29 CFR 1926.55, specifies that an employee's exposure to those specified in the American Conference of Governmental Industrial Hygienists (ACGIH) Threshold Limit Values (TLVs) for Airborne Contaminants for 1970 shall be avoided. Where the requirements of the two standards overlap, the most stringent of the two will be emorced.

B. <u>Administrative and Engineering Controls</u>

To achieve compliance with airborne materials exposure, administrative or engineering controls must first be implemented to achieve compliance.

C. <u>Respirator Use</u>

When effective engineering or administrative controls are not feasible to control airborne exposures, or while they are being instituted, appropriate respirators will be used. Training, selection, issuance, and standard operating procedures are outlined in Section 5.06 of this Health and Safety Manual.

D. <u>Action Specific Hazards</u>

As specified in 29 CFR 1910 Subpart Z, action specific hazards, i.e. lead, asbestos, and formaldehyde, have certain requirements. Such material will be monitored as directed under specific regulatory requirements.

E. <u>Monitoring</u>

1. Monitoring for IDLH and other dangerous conditions

During confined space entry and other situations where the quality of the air is unknown, air monitoring will be conducted for combustible or oxygen deficient atmospheres, as well as for volatile organics. Calibration and maintenance procedures for direct-reading equipment are included as Appendix A.

2. General On-site Monitoring

Site conditions may change during site activities. Air monitoring will be conducted when:

- a. Work begins on a different portion of the site.
- b. A different type of operation is initiated.
- c. Employees are working in obvious contamination.
- 3. Personal Monitoring

OBG TECHNICAL SERVICES, INC.	SECTION: 4.03
HEALTH AND SAFETY MANUAL	SUPERSEDES: 5/14/92
·	DATE: 12/31/93
	REVIEW: 12/31/95
	PAGE: 2 of 2

Personal monitoring will be performed on high-risk workers who are closest to the source of contaminant generation. This approach is based upon the rationale that the probability of significant exposure varies directly with distance from the source. If workers who are closest to the source are not significantly exposed, then all other workers are, presumably, also not significantly exposed and probably do not need to be monitored.

However, any employee may request personal monitoring be performed if a potential risk exists.

Personal air monitoring methodology will be determined by the Safety Coordinator using the appropriate National Institute for Occupational Safety and Health (NIOSH) or other appropriate methods.

F. Notification

Employees will be notified of the results of the monitoring performed in accordance with the procedures specified in Section 2.05.

• •

Calibration and Maintenance Procedures

A. Loan of Equipment

1. Sign-out Procedures

The health and safety monitoring equipment can be obtained from the Warehouse Supervisor or the Safety Coordinator. The information that will be required prior to a piece of equipment leaving the warhouse equipment room will be:

- Serial Number of equipment
- Name of Person responsible for equipment
 Job Number and Location
- Expected Return Date

2. Equipment Return

Never use carrying cases as shipping cases.

Should a piece of equipment be damaged or in need of maintenance, it should be tagged with appropriate instructions.

Upon return, the piece of equipment will be checked to see if it is in the same condition is it left in. If it is broken. an assessment will be made to determine whether it was broken by mishandling or by normal wear and tear. It will also be determined whether the repair can be performed in-house or must be sent to the manufacturer. If the equipment was determined to be broken due to mishandling and the equipment must be sent to the manufacturer, the Project Manager in charge will be contacted to determine if the project can absorb the cost of the repair.

Β. Photoionization Detectors (PIDs)

1. Calibration Procedures

This is to be performed, at a minimum, on a daily basis or whenever the detector is used.

- Turn on photoionization detector (PID) and allow to warm up. a.
- Zero instrument in "clean" air. (Note: "Clean" air refers to upwind of a waste site). Ъ.
- Connect Span Gas cylinder to PID with a piece of clean tubing. c.
- Open the valve on the cylinder until a steady reading is obtained. d.
- Adjust the SPAN control, if necessary, until reading is the same as the Span Gas e. concentration.
- Close the valve on the Span Gas cylinder. Disconnect the cylinder from the PID. f.
- Sample again in "clean" air. Adjust the zero, if necessary. g٠
- 2. Maintenance Procedures

Keeping the PID in top operating shape means charging the battery, cleaning the lamp window, and replacing filters. The exterior of the PID can be wiped clean with a damp cloth and mild detergent, if necessary.

.

- Combustible Gas and Oxygen Meters (CGI/O, Meters)
 - 1. Calibration Procedures

B.

Calibration and Maintenance Procedures

- Combustible Gas Meter
 - i. Turn on the instrument and allow to warm up.
 - ii. Zero % LEL meter in "clean" air.
 - iii. Connect calibration gas with clean piece of tubing to meter.
 - iv. Cpen the valve on the cylinder until a steady reading is obtained.
 - v. Adjust the combustible gas calibration control, if necessary, until reading is the same as the calibration gas concentration.
 - vi. Close the valve on the calibration gas cylinder. Disconnect the cylinder from the meter.
 - vii. Sample again in 'clean' air. Adjust the zero, if necessary.
- b. Oxygen Meter
 - i. Turn on the instrument in "clean" air and allow to warm up.
 - ii. If the % oxygen stabilizes at a value other than 20.3%, adjust the oxygen calibration control until the reading is 20.3%.
- 2. Maintenance Procedures

Keeping the meter in top operating shape means charging the battery and replacing worn parts. The exterior of the meter can be wiped clean with a damp cloth and mild detergent, if necessary.

- C. Flame Ionimion Detector (ED)
 - 1. Calibration Procedures

This is to be performed on a daily basis or, at a minimum, whenever the instrument is used.

- 2. Place instrument in normal operation with CALIBRATE Switch set to X10). Allow 20 minutes for warm up and subilization.
- b. Introduce a methane sample of known concetration (between 90 and 100 ppm) and adjust the GAS SELECT dial so the meter reading corresponds to the known sample.

Periodically (once a month), the FID should be calibrated over the full range using the trimpots. Please refer to the Operations and Maintenance Manual for further instructions.

- 2. Maintenance Procedures
 - a. The battery pack can be recharged overnight; however care should be taken to completely discharge the battery prior to recharging.
 - b. A supply of analytical grade hydrogen is needed to recharge the 75 cc hydrogen tank in the OVA.
- 3. Saipping

Compressed hydrogen cannot be shipped by air. A supply of compressed hydrogen must be available locally.

D. Air Sampling Pumps

1. Calibration Procedures

Calibrate personal air sampling pumps before and after each day of use using a primary calibrator. If it is not feasible to use a primary calibrator, a precision rotameter may be used.

Allow the pump to run 5 minutes prior to calibration.

Calibration and Maintenance Procedures

- b. Czeck the battery charge.
- c. Connect the collection device in the same manner in which the sample will be collected. Please note that the sampling media used for calibration is not to be used for sample collection.
- d. Wet the inside of the glass tube with soap solution.
- e. Turn on the pump and adjust the pump rotameter to the appropriate flow rate setting.
- f. Draw two or three bubbles up the glass tube.
- g. If the flowrate is not within the range of accuracy, adjust the flowrate and draw more bubbles up the glass tube.
- h. Record the average of three runs on the sample data sheet.
- i. Repeat the procedures for all pumps to be used. The same sampling media may be used for all calibrations.
- 2. Maintenance
 - a. Battery

The NiCad battery pack should be completely discharged from time to time to minimize the potential for "memory effect" which occurs frequently with rechargeable batteries. "Memory effect" would prevent a pump from running a full eight-hour period in some cases. The "Full Cycle" mode on the battery charger will fully discharge the battery pack, then recharge it and switch to a trickle automatically. A status lamp on the battery charger shows the status of the attached battery pack.

b. Pump Inlet Filter

A filter/trap inside the clear plastic intake port housing prevents particulates and liquids from being drawn into the pump mechanism. Replace the filter as needed.

E. <u>Dust Monitor</u>

:

. .

- 1. Zero
 - a. Inlet should be in clear position.
 - b. Selector switch should be in 0-2 position, and the letter "m" should appear to the right of the display reading, indicating that the instrument is set to read concentration measurements.
 - Place time constant switch in 2-second position.
 - d. Allow 1-minute warm-up period.
 - e. If required, slowly adjust the ZERO control (while lifting its protective spring loaded cap) until an average reading of 0.000 m is obtained.

2. Calibration

c.

- a. Keep inlet in clear position.
- b. Unlock hinged flow chamber and place it in the horizontal position.
- c. Push reference scatterer knob (REF SCAT) inwards until a positive stop is reached. It is important to ensure the plunger has been fully inserted to the limit of its travel. Insertion of the reference scatter will automatically shut off the pump.
- d. The letter "K" should be flashing at the upper right side of the digital display indicating that the reference scatterer has been inserted in the sensing beam.

.

Calibration and Maintenance Procedures

- Place the time constant in the 2-second position. e.
- f. Move range selector switch to its 0-20 position and allow about 30 seconds for the reading to fully stabilize. The display should read approximately the calibration number indicated on the RAM-1 calibration serial number label within the flow chamber.
- If the indicated readings differ substantially (i.e. by more than 5%) from the factoryg. labelled value, unlock the CAL control and adjust until the desired reading obtained, Relock CAL control.
- Ъ. Puil reference scatterer knob completely out of sensing region until the "X" indicator on the display ceases to flash.
- i. Close flow chamber cover by fightening two thumb-screws at its upper corner.

F. Drager Pump

- Pump Leak Test 1.
 - insert unopened tube into socket. 2.
 - ы Squeeze pump completely and release.
 - Pump is adequately leak-proof if the end of stroke indicator has not appeared after 15 c. minutes.
 - đ. Remove tube from socket.
 - Press oution to reset counter to zero; button is recessed to prevent accidental resetting. e.
- 2. To clean or replace the exhaust valve
 - **a**. Lift lower pump plate up with a coin or screwdriver.
 - Ъ. Pull vaive disc from its seating.
 - Wine valve seating with a damp cloth. C.
 - Moisten the stem on a new valve disc and push ti firmly into the center hole of the valve d. seat.
 - Check correct seating by pulling on valve disc lightly. e.
 - f. Put back lower pump place and press it into place.
 - Perform leak test discribed above. g.
- 3. Replacing the socket and the membrane
 - Press spring hook down with a suitable tool (e.g. screwdriver). 2
 - Ъ. Take out cover plate.
 - Press spring flanges together and remove socket.
 - с. d. Insert zew socket.
 - Using a new End-of-Stroke Indicator Membrane, place parts into the housing.

.

- e. f. Insert spring book of cover plate and push on firmly.
- Perform leak test described above. g.

· · · · ·

Appendix E

н — -

and a second second second second second second second second second second second second second second second

OBG TECHNICAL SERVICES. INC.	SECTION: 4.05
HEALTH AND SAFETY MANUAL	SUPERSEDES: 6/3/93
· .	DATE: 12/31/93
	REVIEW: 12/31/95
	PAGE: 1 of 11

Section 4.05 Hazardous Waste Operations

A. <u>Introduction</u>

Due to the nature of the business conducted by OBG Tech, employees may be working on hazardous waste sites. OSHA has requirements for such things as air monitoring, health and safety plans, site control, and emergency response. This section is to define OBG Tech's Standard Operating Procedures (SOPs) for hazardous waste sites and to outline the elements for inclusion in the Site-specific Health and Safety Plan.

B. Other Pertinent Sections of OBG Tech's Health and Safery Manual

<u>Section</u>	Topic
1.01 2.01 2.02 2.03 2.04 2.05 2.09 3.01 3.02 - 3.03 3.04 4.01 4.02 4.03	Safety Policy Health and Safety Records Health and Safety Training Injury/Illness and Accident Report Process Medical Surveillance Exposure Monitoring General Safety Rules Medical Services and First Aid Sanitation Emergency Action Confined Space Entry Occupational Noise Exposure Radiation Exposure Airborne Materials Exposure
5	Personal Protective and Life Saving Equipment
8	Materials, Handling, Storage, Use, and Disposal

C. Hazardous Waste Standard Operating Procedures

1. Contractors and Sub-contractors

All contractors and sub-contractors retained by OBG Tech for work in hazardous waste will be informed of emergency response procedures and any potential fire, explosion, health, safety, or other hazards of the hazardous waste operation that have been identified by OBG Tech.

2. Program Availability

The written safety and health program will be made available to: any contractor or subcontractor; OBG Tech employees; OSHA personnel; and to personnel of other Federal, State, or local agencies.

3. Project Personnel

Certain individuals have specifically designated responsibilities on hazardous waste sites.

OBG TECHNICAL SERVICES, INC.	SECTION: 4.05
HEALTH AND SAFETY MANUAL	SUPERSEDES: 6/3/93
· .	DATE: 12/31/93
	REVIEW: 12/31/95
·	PAGE: 2 of 11

<u>Project Manager</u> - The Project Manager is responsible for the over-all management of the project. The Project Manager manages administrative requirements.

<u>Project Supervisor</u> - The Project Supervisor is responsible for coordinating between office and field personnel. The Project Supervisor is responsible for the day-to-day activities of the project. The Project Supervisor will oversee field and related activities.

Site Safety and Health Coordinator - The Site Safety and Health Coordinator (SSHC) will establish operating standards and coordinate overall project safety and health activities for the site. The SSHC will review project plans and revisions to plans to determine that safety and health procedures are maintained throughout the project. The specific responsibilities of the SSHC are outlined in Part 16, below.

4. Pre-entry Briefing

Pre-entry briefings will be held prior to initiating any site activity and at other times as necessary to inform employees of the site-specific health and safery plan. In situations covered by OBG Tech's Haz Com Program or the 40-hour Hazardous Waste Operations Health and Safety Training, training required by that program need not be duplicated.

5. Effectiveness of site-specific health and safety plan

Inspections will be made by the Site Health and Safety Coordinator or the Project Supervisor to determine the effectiveness of the site safety and health plan.

6. Personal Protective Equipment

Personal protective equipment (PPE) will be provided in accordance with Section 5 of this Health and Safety Manual. PPE will be selected and used which will provide protection against known or suspected hazardous substances and health hazards.

7. Initial Site Entry Monitoring

When information on the site shows that the potential for ionizing radiation, for IDLH conditions, or when site information is not sufficient to reasonably eliminate the following conditions:

- a. Monitoring with direct-reading instruments for hazardous levels of ionizing radiation (See Section 4.02).
- b. Monitoring the air with direct-reading instruments for IDLH conditions (See Section 4.03).
- c. Visually observing for signs of actual of potential IDLH or other dangerous conditions.

8. On-going Air Monitoring Program

An on-going monitoring program will be established for every site where there is a potential for employee exposure to hazardous concentrations of hazardous substances. This on-going monitoring is to evaluate proper selection engineering controls, work practices, and PPE, so that employees are not exposed above OSHA PELs and published

OBG TECHNICAL SERVICES. INC. HEALTH AND SAFETY MANUAL

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 3 of 11

exposure levels. Specifically the on-going monitoring will be conducted:

- when work begins on a different part of the site;
 - when contaminants other than those previously identified are being handled;
- when a different type of operation is initiated;
 - when employees are handling leaking drums or containers or working in areas with obvious liquid contamination (e.g. spill or lageon)

Personal monitoring will be performed on those employees likely to have the highest exposures to airborne materials in accordance with Section 4.03 of this Health and Safety Manual.

9. Site Control

The elements of the site control program will include a site map; site work zones; site communications, including alerting means for emergencies; and identification of nearest medical assistance. OSHA also requires the use of the "buddy" system, which is defined below and standard operating procedures, which are encompassed into this Health and Safery Manual. These need not be repeated in the site-specific health and safery plan when this Health and Safery Manual is on-site.

a. Site Work Zones

- Exclusion Zone:

The exclusion zone or the Hot Zone is the area where contamination does or could occur. The exclusion zone boundary should be clearly marked by lines, placards, hazard tape and/or signs or enclosed by physical barriers, such as chains, fences, or ropes. Access control points should be established at the periphery of the exclusion zone to regulate the flow of personal and equipment into an out of the zone and to help verify that proper procedures for entering and exiting are followed.

The required level of PPE in the exclusion zone can vary according to job assignment. This will allow a flexible, effective, and less costly operation, while still maintaining a high degree of safety.

- Contamination Reduction Zone:

The contamination reduction zone or the decontamination zone is the transition area between the contaminated area and the clean area. At least two lines of decontamination stations should be set. up within the contamination reduction zone: one for personnel and one for heavy equipment. Personnel entering the contamination reduction zone should be required to wear PPE prescribed for working in the contamination reduction zone. To reenter the support zone, workers must remove any PPE. Personnel stationed in the contamination reduction zone should be for working in the support zone, the Site Health and Safety Coordinator, the Project Supervisor, personnel assisting in decontamination procedures, and
SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 4 of 11

emergency response personnel.

The contamination reduction zone should be designed to facilitate: decontamination, emergency response, equipment resupply, sample packaging, and temporary work rest area.

- Succort Zone:

The support zone is the location of the administrative and other support functions needed to keep the operations in the exclusion and contamination reduction zone running smoothly. Any function that need not or cannot be performed in a hazardous atmosphere is performed here. Personnel may wear normal work clothes within this zone. Any potentially contaminated clothing, equipment and samples must remain in the contamination reduction zone until decontaminated. All emergency telephone numbers, change for the telephone (if necessary), evacuation route maps, and vehicle keys should be kept in the support zone.

b. "Buddy" System

Most activities in a contaminated or otherwise hazardous areas should be conducted with a buddy who is able to:

- Provide his or her partner with assistance.
- Observe his or her parmer for signs of chemical or heat exposure.
- Periodically check the integrity of his or her partner's protective clothing.
- Notify the Project Supervisor or the Site Health and Safety Coordinator if emergency help is needed.

c. Site Communication

Internal communication among personnel on site and external communication between on site and off-site personnel should be established.

Verbal communication at a site can be impeded by on-site background noise and the use of personal protective equipment. In the absence of site-specific communication signals, the following will be used for emergencies:

Hand clutching throat: Out of air/can't breathe

Thumbs up: OK/I'm alright/I understand

Thumbs down: No, negative

Grip partner's wrist or both hands around partner's waist: Leave area immediately

Hands on top of head: Need assistance

10. Site Security

Site security is necessary to prevent the exposure of unauthorized, unprotected people to the site, to avoid the increased hazards of vandals, to prevent theft, and to avoid

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 5 of 11

interference with safe working procedures.

A physical barrier will be constructed to help prevent exposure of unauthorized and unprotected people to site hazards.

The Site-specific health and safery plan will include any other provisions for maintaining security at the Site.

11. Engineering Controls, work practices, and personal protective equipment for employee protection

Engineering controls and work practices must be first instituted to reduce and maintain employee exposure to or below the OSHA PELs, except to the extent that such controls and work practices are not feasible.

Engineering controls which may be feasible include the use of remotely operated material handling equipment. Work practices which may be feasible include wetting down dusty operations and locating employees upwind of possible hazards.

Whenever engineering controls and work practices are not feasible, PPE will be used in accordance with Section 5 of this Health and Safery Manual to reduce employee exposure to or below the OSHA PELs.

Engineering controls, work practices, and personal protective equipment will be used to reduce and maintain employee exposure to or below published exposure levels not regulated by OSHA. PPE will be used in accordance with Section 5 of this Health and Safety Manual to reduce employee exposure to or below the published exposure levels.

Guidance for PPE selection can be obtained from 29 CFR 1910.120 Appendix B or the NIOSH/OSHA/USCG/EPA document "Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities." The guidance document is usually handed out in the OSHA 40-hour course. However, should an employee desire a copy of 29 CFR 1910.120 Appendix B or the guidance document, the safety coordinator can provide one.

12. Levels of Protection

OBG Tech often employs the use of PPE based upon the EPA Levels of Protection. They are listed in Section 5 as a starting point, however site-specific PPE may be required. Site-specific PPE will be addressed in the site-specific health and safety plan. The type of equipment used and the overall level of protection should be re-evaluated periodically as the amount of information about the site increases and as workers as required to perform different tasks.

13. PPE Decontamination Procedures

General PPE decontamination procedures are outlined in Section 5.01. Site-specific decontamination procedures, if any, will be included in the Site-specific Health and Safety Plan.

1

14. Emergency Response Procedures

OBG TECHNICAL SERVICES. INC.	SECTION: 4.05
HEALTH AND SAFETY MANUAL	SUPERSEDES: 6/3/93
	DATE: 12/31/93
	REVIEW: 12/31/95
	PAGE: 6 of 11

The nature of hazardous waste work makes emergencies a continuous potential, no matter how infrequently they occur. The communications network, as outlined in Section 4.05, part 9.c. must be conveyed to all workers so that emergency information may quickly distributed. Equipment will be necessary for emergency similations. At a minimum each site will be equipped with a fire extinguisher, eye wash bottle, and first aid kit. Other equipment, which should be considered on a project by project basis, are emergency showers, safety harnesses, and spiil-containment equipment.

Follow-up procedures, as follows, must be implemented before activities resume onsite:

Appropriate governmental agencies must be notified as required.

Restock, replace, and/or clean all equipment and supplies.

The Site Health and Safery Coordinator, the Project Supervisor, and the employee affected should fill out an accident report, as outlined in Section 2.03.

15. Hazards of hazardous waste sites

Chemical

Most sites contain a variety of chemicals that may be in a gaseous, liquid, or solid form. These substances can be hazardous through inhalation, skin absorption, ingestion, or through injection (a puncture wound).

Acute chemical exposures usually occur during or shortly after exposure to a high concentration of a chemical. *Caronic* usually refers to exposures to low concentrations of a chemical over a long period of time. Both of these are dependent on the chemical and may be temporary or reversible or may be permanent. Some chemicals may exhibit warning signs, while other chemicals are odorless, colorless, and tasteless.

Inhalation is the primary route of exposure at a hazardous waste site. Therefore the selection and use of the proper respiratory protection is extremely important where there is a potential for inhalation of hazardous materials. Direct skin and eyes contact is also a potential route of exposure, therefore proper selection and use of PPE is extremely important. Ingestion and injection are the least significant routes of exposure at a hazardous waste site. However, personal habits such as eating, drinking, or smoking, and safety hazards such as puncture wounds can be potential routes, therefore foilowing SOPs and use of PPE can reduce the hazards from both of these routes of exposure.

Further information is presented in OBG Tech's Haz Com and OSHA 40-hour hazardous waste operations health and safety training.

Explosion and Fire

Due to the nature of activities at hazardous waste sites, there is a potential for explosions and fires. To minimize the hazards from fire and explosion: monitor for explosive and flammable atmospheres following the procedures in Section 4.03; keep all potential ignition sources away from an explosive or tlammable environment; use non-sparking, explosion-proof equipment; and follow safe work practices.

OBG TECHNICAL SERVICES, INC. HEALTH AND SAFETY MANUAL	SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 7 of 11

Oxygen Deficiency

Oxygen deficiency may result from the displacement of oxygen by another gas, or the consumption of oxygen by another chemical reaction. Confined spaces and low-lying areas are particularly vulnerable and should always be monitored prior to entry following the procedures in Section 3.04.

Ionizing Radiation

Monitoring for ionization radiation is required to be performed when there is not sufficient information to eliminate the possibility of it being present on-site (e.g. uncontrolled hazardous waste sites, where little or no information is available on past history). The procedure in Section 4.02 will be followed for monitoring.

Biological Hazards

Wastes from hospitals and research facilities may contain biological materials that may cause infections to site personnel. Other biological hazards that may be present at hazardous waste sites include poisonous plants, insects, and animals. PPE can reduce the potential for exposure. The Safety Coordinator can assist in determining the correct PPE for the hazard present.

Safety Hazards

Hazardous waste sites contain numerous potential safety hazards such as: holes, ditches, drums, boards, nails, broken glass, slippery surfaces, steep grades, and uneven terrains. The work itself may be a potential safety hazard. Site personnel should constantly look out for potential safety hazards and should immediately inform the Project Supervisor or the Site Health and Safety Coordinator of any new hazards.

Electrical Hazards

As in all construction work, overhead power lines, electrical wires and cables, site electrical equipment, and lightning also pose a potential hazard to site workers. Section 10.02 provides guidance on safe electrical practices.

Heat Stress

Heat stress is potentially a major bazard for workers wearing protective clothing. Due to the impervious nature of the PPE to keep chemicals away from the skin, body heat and moisture are trapped within the PPE. Careful training and frequent monitoring of personal who wear protective clothing, scheduling of work and rest periods, and the frequent replacement of fluids can protect against this hazard.

Heat stress can be minimized by taking the following steps:

- Adjusting work schedules
- Provided air conditioned or shaded rest areas
- A total of 1 to 1.6 gallons of fluid intake recommended, but more may be necessary to maintain body weight
- Acclimatize workers to site work conditions
- Provide cooling devices to aid natural body heat exchange during prolonged work or

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 8 of 11

severe heat exposure

Review recognition and treatment of heat stress with workers

For workers wearing semi-permeable or impermeable PPE and when the temperature in the work area is above 70[°]F, measure:

- Heart Rate: count the radial pulse during a 30-second period as early as possible in the rest period.

If heart rate exceeds 110 beats per minute at the beginning of the rest period, shorten the work cycle by one-third and keep the rest period the same.

If the heart rate exceeds 110 beats per minute at the next rest period, shorten the following work cycle by one-third.

Oral Temperature: use a clinical thermometer or similar device to measure the oral temperature at the end of the work period (before intake of fluids).

If the oral temperature exceeds 99.6°F, shorten the next work cycle by one-third without changing the rest period.

If the oral temperature still exceeds 99.6°F at the beginning of the next rest cycle, shorten the following work cycle by one-third.

Body Water Loss, if possible: measure weight on a scale accurate to $\pm 1/-0.25$ lb at the beginning and end of each work day to see if enough fluids are being taken in to prevent dehydration. Weights should taken while the employee is wearing similar clothing each day. The body water loss should not exceed 1.5 percent total body weight loss in a work day.

The frequency of physiological monitoring depends on the air temperature adjusted for solar radiation and the level of physical work. The suggested frequency of physiological monitoring for fit and acclimatized workers.

Adjusted Temp.	Normal Work Ensemble	Impermeable Ensemble
90°F and above	45 minutes	15 minutes
87.5°F to 90°F	60 minute	30 minutes
82.5°F to 87.5°F	90 minutes	60 minutes
77.5°F to 82.5°F	120 minutes	90 minutes
72.5°F to 77.5°F	150 minutes	120 minutes

Cold Exposure

Cold stress and impaired ability to work are dangers at low temperatures and when the wind-

OBG TECH	NICAL SER	VICES. INC.
HEALTH AN	ND SAFETT	MANUAL

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 9 of 11

chill factor is low. To guard against cold stress: wear appropriate clothing; have warm clothing and warm shelter readily available; and carefully monitor workers' physical conditions.

Noise

Work around large equipment often creates excess noise. The procedures set forth in Section 4.01 for occupational noise exposure and Section 5.04 for hearing protection should be followed.

16. Responsibilities of Site Health and Safety Coordinator

The Site Health and Safety Coordinator advises the Project Manager and the Project Supervisor on the matters of health and safety on the site. Specifically the responsibilities of the Site Health and Safety Coordinator include:

- a. Aiding the selection of protective clothing and equipment.
- b. Periodically inspecting protective clothing and equipment.
- c. Maintaining proper storage of protective clothing and equipment.
- d. Monitors the workers for signs of heat stress, cold stress, and fatigue.
- e. Monitors on-site hazards and conditions.
- f. Conducts periodic surveillance to evaluate effectiveness of Site-specific Health and Safety Plan.
- g. Has knowledge of emergency procedures, evacuation routes, and the telephone numbers of the ambulance, local hospital, poison control center, fire department, and police department.

h. Posts the directions to the hospital and the telephone numbers of the ambulance, local hospital, poison control center, fire department, and police department.

- i. Notifies, when necessary, local public emergency officials.
- j. Coordinates emergency medical care.
- 17. Safe Work Practices
 - a. No eating, smoking, eating, drinking, or application of cosmetics in the Contamination Reduction Zone or the Exclusion Zone.
 - b. No matches or lighters in the Contamination Reduction Zone or the Exclusion Zone.
 - c. Enter and exit following procedures in the Site-specific Health and Safety Plan.
 - d. Wear the PPE specified in the site-specific Health and Safery Plan in the Exclusion Zone.
 - e. Use the "buddy" system.
 - f. Report any unusual conditions to the Project Supervisor or the Site Health and Safety Coordinator immediately.

D. Requirements for Site-Specific Health and Safety Plans

As a company policy, site-specific health and safety plans are required on projects. The following site-specific information must be included in the health and safety plan, as well as this manual attached as an appendix. The Safety Coordinator will prepare and/or review each site-specific health and safety plan.

1. Organizational structure of site program

OBG TECHNICAL SERVICES. INC.	SECTION: 4.05
HEALTH AND SAFELY MANUAL	SUPERSEDES: 0/3/93
	DATE: 12/31/93
	REVIEW: 12/31/95
	PAGE: 10 of 11

The following information, at a minimum, will be included in the site-specific health and safety plan:

- Project Manager's name and telephone number
 - Project Supervisor's name and telephone number
- Site Safery and Health Coordinator's name and telephone number
- Cther personnel aeeded for emergency response

This organizational structure will be reviewed and updated as necessary to reflect the current status of the bazardous waste site operations.

2. Workelan Summary

A summary of the workplan, including location and approximate size of the site, will be included in the site-specific health and safety plan addressing the anticipated activities.

3. Safety and Health Hazard Analysis

A safety and health hazard analysis will be prepared for the site-specific health and safety plan, including pathways for hazardous substance dispersion. The safety and health hazard analysis will include hazardous substances and health hazards involved or expected at the site, and their chemical and physical properties.

Information to consider to include in the site-specific health and safety plan:

- Exposures exceeding the permissible exposure limits and published exposure levels
- IDLH situations

- Potential skin absorption and irritation sources
- Potential eye irritation sources
- Explosion sensitivity and flammability ranges
- Oxygen deficiency
- 4. Site-specific Training Assignments

Initial 40-hour worker, Supervisor, and Refresher training are addressed in Section 2.02, Health and Safery Training. Hazard Communication training is also addressed in this section. Training specific to the site will be addressed in the site-specific health and safety plan.

5. Personal Protective Equipment

Personal protective equipment to be used by employees for each of the site tasks and operations being conducted will be addressed in the site-specific health and safety plan. The following information will be included in the site-specific plan: PPE selection based upon site hazards; work mission duration; site-specific information on PPE decontamination and disposal;

SECTION: 4.05 SUPERSEDES: 6/3/93 DATE: 12/31/93 REVIEW: 12/31/95 PAGE: 11 of 11

6. Frequency and types of air monitoring

Frequency and types of air monitoring, personnel monitoring, and environmental sampling techniques and instrumentation to be used will be addressed in the site-specific health and safety plan. Methods of maintenance and calibration of monitoring equipment can be found in Section 4.03.

7. Site Control Program

The site-specific site control program will include a site map and the identification of the nearest medical assistance.

8. Site Security

Site-specific information for maintaining security will be included in the site-specific health and safety plan.

9. Emergency Response Plan

Site-specific information such as safe distances and places of refuge, evacuation routes and procedures, and procedures for reporting incidents to local, state, and federal agencies not covered in OBG Tech's SOPs will be covered in the site-specific health and safety plan. The nearest telephone for emergency communication will be identified in the site-specific emergency response plan.

10. Spill Containment Program

A spill containment program unique to the site will be developed for the site-specific health and safety plan, if applicable.

- 11. Site-specific information other than in SOPs
 - a. Medical surveillance requirements unique to the site, other than what is covered in OBG Tech's SOP in Section 2.04.
 - b. Employee training assignments unique to the site other than what is covered in OBG Tech's SOP in Section 2.02.
 - c. Decontamination procedures which are unique to the site other than what is covered in OBG Tech's SOP in Section 4.04 (E).
 - d. Sare work practices unique to the site, other than what is covered in OBG Tech's SOP in this section.

SECTION: 4.05 SUPERSEDES: None DATE: May 23, 1992 REVIEW: March 24, 1994 PAGE: 1 of 1

4.06 Process Safety

A. Introduction

Releases of toric, reactive, or flammable liquids and gases in processes involving highly harardous chemicals have been reported for many years. In an effort to control there releases, CSHA has published a regulation, 29 CFR 1910.119, Process Safety Management of Highly Harardous Chemicals. In that regulation is a section on contract employer (such as OBC Tech) responsibilities. This section has been developed in compliance with the regulation.

B. <u>Requirements</u>

1.

2

Each employee is provided health and safety training, as outlined in Section 2.02 of this Health and Safety Manual. Specific training required at a jobsite regarding the known potential fire, explosion, or toxic release hazards related to his/her job and the process, and the applicable provisions of the emergency action plan will be the responsibility of the Project Manager.

Each employee will sign a document stating that they have received and understood the training that is outlined above.

3. Each employee is expected to follow the safety rules of the facility.

4. In addition to the training required above, employees will be informed of unique hazards presented by the project or any of the hazards found by the work.

Appendix F

•

OBG TECHNICAL SERVICES. NO. HEALTH AND SAFETY MANUAL	SECTION: 5.06 SUPERSEDES: None
	DATE: May 15, 1992
	REVIEW: January 1, 1993
	PAGE: 1 CF 5

5.06 Respiratory Protection

A. Introduction

OSHA, in 29 CFR 1910.134, specifies that "when effective engineering controls are not feasible, or while they are being instituted, appropriate respirators will be used." OSHA references exposure to air contaminated with harmful dusts, fogs, fumes, mists, gases, smokes, sprays, or vapors.

Due to the nature of the business conducted by the company, respiratory protection may be necessary for specific activities performed by our employees. This section will serve as the respiratory protection program and is adopted by the company in an effort to assist ensuring a safer working environment for our employees during work activities requiring respiratory protection as dictated in the site-specific Health and Safery Plan. In order to comply with regulatory requirements, this program is developed pursuant to 29 CFR 1910.134 and 29 CFR 1926.103. This program and use of respirators are instituted according to the site-specific Health and Safery Plan and only after exhausting all feasible engineering controls.

B. <u>Employer and Employee Responsibility</u>

1. Employer Responsibilities

The Safety Coordinator will see that approved respirators, cartridges, and spare parts will be provided by the company. The Safety Coordinator will also be responsible for the establishment and maintenance of this respiratory protection program and the upkeep of records for fit testing, medical surveillance, and training.

Employee Responsibilities

It is the responsibility of the employee to use the respiratory protection in accordance with instructions and training received. The employee will maintain the respirator to insure that cartridges and parts are replaced when necessary. The employee will report any problems with his respirator to his supervisor or the Safety Coordinator.

C. Training of Employees

All employees required to wear respirators on the job will be trained prior to the use of respirators. That training will cover the topics required by 29 CFR 1910.120 (Hazardous waste operations and emergency response) and 29 CFR 1910.134 (Respiratory Protection), and include the following:

- basics of respiration
- basics of respiratory hazards
- capabilities and limitations of respirators
- inspection of respirators
- how a respirator should be worn
- cleaning and disinfecting respirators
- storage of respirators
- respirator-specific training
- fit-checking procedures

OBG	TECHN	CAL SERV	ACES, INC.
HEAI	LTH AND	SAFETY	MANUAL

SECTION: 5.06 SUPERSEDES: None DATE: May 15, 1992 REVIEW: January 2, 1993 PAGE: 2 OF 5

All employees will be given the opportunity to wear their respirator in an uncontaminated atmosphere and a test atmosphere for a period of time to become familiar with the use of respirators.

All training is documented and is filed in the Syracuse office.

D. <u>Physician's Approval</u>

All employees required to wear a respirator will have a physician's written approval to wear a respirator prior to being required to wear one. The respirator user's medical status will be reviewed annually.

The local physician will determine which health and physical conditions are pertinent. A description of the respiratory hazards, specifics of employee's job functions while wearing a respiratory and copies of all applicable regulations will be provided to the physicians to further aid the decision process.

The physical will be provided at no cost to the employee. A copy of the written report will be made to the employee upon request.

All medical examinations are recorded and are filed in the Syracuse office.

E <u>Selection of Respirators</u>

Respirators will be selected on the basis of the following:

- chemical and physical hazards
- characteristics of the hazardous operation of process
- face piece to face fit
- comfort
- utilization of NIOSH recommendations
- utilization of manufacturer's recommendations
- the guidance of American National Standard Practices for Protection Z38.2-1969.

All respirators selected will be NIOSHMSHA approved for the hazards encountered.

The Safety Coordinator will be adequately instructed to insure that the correct respirator is selected and that the appropriate personal modifications are made such as corrective lens for full-face masks.

F. <u>Issuance of Respirators</u>

The Warehouse Supervisor will be responsible for the issuance of a property selected respirator to each employee. Each employee will be given his own respirator and will be responsible for bringing it to the jobsite. Employees should mark his respirator so that it will not be confused with others.

OBG TECHNICAL SERVICES, INC.	SECTION: 5.26
HEALTH AND SAFETY MANUAL	SUPERSEDES: None
	DATE: May 15, 1992
	REVIEW: January 2, 1993
	PAGE: 3 CF 5

G. Fit-Testing of Resolutions

To insure a proper it of negative pressure respirators, respirator fit-testing will be performed. Where fit-testing is not required by specific hazard regulation, as it is with lead or asbestos, the qualitative irritant vapor or smoke protocol of the asbestos standard will be adopted (29 CFR 1910.1001. Appendix C). Fit-testing will be performed to select respirators and be performed at the discretion of the Safety Coordinator thereafter unless required by law to be performed more often or unless there is sufficient need to do so (i.e., denture replacement, somring of face, weight change).

Fit-test failure will result in selection of a different size respirator. Continued test failure will result in selection of a different manufacturer's respirator.

All fit-testing information such as the employee, the date, and the type of respirator is recorded and filed in the Syracuse office.

H. Inspection of Respirators

Respirators will be inspected for damage before and after each use. Each employee, after training, will responsible for inspection. The following areas will be inspected:

- tightness of connections
- face piece
- headbands
- inhalation valve
- exhalation valve
- cararidge or filter fittings
- pliability of rubber or elastomer parts
- signs of deterioration

Any malformation, distortion, missing parts, cracks, etc. will be sufficient to issue replacement parts or if necessary, a new respirator.

L Standard Operating Procedures

Before entering any potentially contaminated environment, each employee will:

- 1. Carefully inspect the respirator following the procedures specified in Section 8.
- 2. Duct tape should be removed from cartridges (if applicable).
- 3. The respirator should be donned and checked for a proper fit using the following tests:
 - a. <u>Positive Pressure Test</u> close off the exhalation valve with your hand. Breathe into the mask. The face-to-facepiece seal is satisfactory if some pressure can be built up inside the mask and sustained.

SECTION: 5.06 SUPERSEDES: None DATE: May 15, 1992 REVIEW: January 2, 1993 PAGE: 4 OF 5

- b. <u>Negative Pressure Test</u> close off the inlet openings of the cartridge with the paim of your hand. Inhale gently so that a vacuum occurs inside the mask. Hold your breath for 10 seconds. If the vacuum is sustained, and no inward leakage is detected, the respirator fits properly.
- 4. Inside the contaminated environment, respirators will not be removed except in a medical entergency such as a suspected heart attack.
- Respirators will be worn with straps inside the disposable garment allowing a worker to maintain respiratory protection while removing contaminated garments.

I. <u>Cleaning and Disinfecting of Respirators</u>

Respirators will be cleaned after each use. Manufacturers may have specific recommendations for cleaning and those should be followed. In absence of manufacturers recommendations, the following procedures should be used:

- Remove the cartridges and headbands
- 2. Disassemble all respirator parts
- 3. Wash all respirator parts (except cartridges and headbands) in a cleaner disinfectant solution or use soap and hot water
- Rinse completely in clean, warm water
- 5. Air dry in a clean area
- 6. Re-assemble the respirator

No alcohol will be used to clean the respirator. If a disinfecting solution is not used, a disinfecting spray will be used at least weekly, but preferably after each use.

Respirator wipes will be provided to employees in order to clean respirators during work shifts between uses. The employee will be allowed to leave work area and remove respirator to wash face in order to prevent rashes and discomfort. The respirators will be wiped out at each of these times.

K. Storage of Respirators

Respirators will be stored in clean plastic bags and protected against dust, sunlight, heat, extreme cold, excessive moisture, or damaging chemicals. Respirator cartridges will have the inhalation holes covered with duct tape (or acceptable substitute tape) immediately after leaving a contaminated area. The tape will be left on until the respirator is donned for the nest entry into a contaminated area. This tape will prevent any contaminants from being dislodged from the cartridge.

Respirators should be packed or stored so that the facepiece and exhalation valve will rest in a normal position and function will not be impaired by the elastomer setting in an abnormal position.

SECTION: 5.06 SUPERSEDES: None DATE: May 15, 1992 REVIEW: January 1, 1993 PAGE: 5 OF 5

L. <u>Periodic Surveillance</u>

Work areas will be monitored as required by specific hazard regulations or on a periodic timetable as set by the site-specific Health and Safery Officer and the site-specific Health and Safery Plan. This surveillance is required to ensure that the proper level of protection is provided to employees. Whenever new hazards are encountered or a substantial change in magnitude of the existing hazard occurs, then additional monitoring will take place.

M. Evaluation of Respiratory Protection Program

In order to maintain an effective program, the respiratory protection program will be re-evaluated on at least an annual basis. This evaluation will address:

- employee acceptance of program and respirators
- methods of surveillance of hazards and results
- regulatory compliance
- changing job functions
- changes in hazards

Employees are encouraged to express any concerns about respirator protection, such input is critical for evaluating the program.

Each employee will be made aware of this written program and any annual changes.

Frequent random inspections will be conducted by the Safety Coordinator to assure that respirators are properly selected, used, cleaned, and maintained.

N. Hazard Specific Respiratory Protection

As specified in 29 CFR Parts 1910, action specific hazards, i.e. lead, asbestos, and formaldehyde, require specific respiratory protection. Any such material will be monitored as directed under specific regulatory requirements, and respiratory protection will issued pursuant to specific regulatory requirement.



								_				
Project:			Pro	Project Number:								
Date:	Time:		Cor	Competent Person:								
Where was the sample	nple taken from:											
Excavation Length:			Excavation	n Depth:			Excavation Width:					
		, ,		VISUAL	TEST							
Particle type: Fine grained (cohesive) Granular (sand/silt or gravel)				gravel)	<u> </u>		<u> </u>					
Water conditions:	Wet Surface	water present			Dry Se Submerged	eping Water						
· ·						<u> </u>	·					
Previously disturbed so	oils:				Lavered soils dipping	nto excavatio	n:					
Underground utilities: If yes, what type?					Excavation exposed to If yes, from what?	o vibrations:	•					
Lavered soils? Note: The	less stable layer d	contrais soil type.			Creck like openings of	r spallings obs	erved:					
				······································					·			
			R	MANUAL	TEST				MANUAL TEST			
Plasticity: Cohesive Non-cohesive												
Plasticity: Col	nesive	- Non-cor	nesive	_				•				
Plasticity: Col Dry strength: Gra	nesive inular (crumble	s easily)	Cohesive	(broken v	with difficulty)			·				
Plasticity: Col Dry strength: Gra Wet shake: Wa OTE: The following unc	nesive anular (crumble ter comes to s anfined compr	Non-cor s easily) urface (granular ma ressive strength test	Cohesive Cohesive Iterial) ts should be	(broken v	with difficulty) Surface remains dry (led on undisturbed soil	clay material) s.						
Plasticity: Col Dry strength: Gra Wet shake: Wa OTE: The following unc <u>numb Test</u> used to estin Test performed:	nesive anular (crumble ter comes to s onfined compr mate unconfine Yes	Non-cor is easily) urface (granular ma ressive strength test ad compressive stre	Cohesive Cohesive (teriel) ts should be ngth of coh	(broken v	with difficulty) Surface remains dry (red on undisturbed soil I:	clay material) s.						
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>numb Test</u> used to estir Test performed: Ty	nesive nular (crumble ter comes to s onfined compr mate unconfine Yes pe A - soil inde	Non-cor is easily) urface (granular ma ressive strength test ed compressive stre anted by thumb wit	Cohesive Cohesive Iterial) ts should be ngth of coh No h very great	(broken v b perform esive soil	with difficulty) Surface remains dry (ed on undisturbed soil I:	clay material) s.						
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>numb Test</u> used to estir Test performed: Ty Ty	nesive anular (crumble ter comes to s onfined compr mate unconfine Yes pe A - soil inde pe B - soil inde	Non-cor is easily) urface (granular ma ressive strength test ed compressive stre anted by thumb with ented by thumb with	Cohesive Cohesive (terial) ts should be ngth of coh No h very great	(broken v o perform essive soil t effort	with difficulty) Surface remains dry (ed on undisturbed soil l:	clay material) s.						
Plasticity: Col Dry strength: Gra Wet shake: Wa OTE: The following unc numb Test used to estin Test performed: Ty Ty Ty Ty	nesive anular (crumble ter comes to s onfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge	Non-cor is easily) urface (granular ma- ressive strength test ed compressive stre ented by thumb with ented by thumb with enter by thumb with	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to	(broken v e perform essive soil t effort ort y thumb v o surface	with difficulty) Surface remains dry (ed on undisturbed soil l: 	clay material) s.						
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>numb Test</u> used to estin Test performed: Ty Ty Ty Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge ne used to esti	Non-cor is easily) urface (granular ma- ressive strength test ed compressive stre ented by thumb with ented by thumb with ily penetrated sever id, seeping water, a mate unconfined co	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to pompressive	(broken) b perform essive soil t effort ort y thumb) b surface strength	with difficulty) Surface remains dry (ed on undisturbed soil I: with little or no effort. water, runoff, exposed of cohesive soils:	clay material) s.						
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>uumb Test</u> used to estin Test performed: Ty Ty Ty Ty Ts instrometer or Shearva	nesive anular (crumble ter comes to s onfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge ne used to esti	Non-cor is easily) urface (granular ma- ressive strength test anted by thumb with ented by thumb with illy penetrated sever ad, seeping water, so mate unconfined cor No	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to compressive Device us	(broken v e perform esive soil t effort ort y thumb v surface strength	with difficulty) Surface remains dry (ed on undisturbed soil: 1: 	clay material) s. d to wetting.						
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>numb Test</u> used to estir Test performed: Ty Ty Ty Ty Ty stretrometer or Shearva Test performed: Yes	nesive nular (crumble ter comes to s onfined compr mate unconfine Yes pe A - soil inde pe C - soil easi oil is submerge ne used to esti s pe A - soil with	Non-cor is easily) urface (granular ma ressive strength test ed compressive stre ented by thumb with ented by thumb with illy penetrated sever ed, seeping water, a mate unconfined compre- h unconfined compre-	Cohesive Cohesive terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to perfecte us Device us	(broken v o perform esive soil t effort ort y thumb v o surface strength eed: ngth of 1.	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils:	clay material) s.						
Plasticity: Col Dry strength: Grz Wet shake: Wa DTE: The following unc <u>numb Test</u> used to estir Test performed: Ty Ty Ty Ty Ty Ty Ty Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe C - soil easi oil is submerge ne used to esti g pe A - soil with pe B - soil with	Non-cor is easily) urface (granular ma- ressive strength test ed compressive stre anted by thumb with tented by thumb with ily penetrated sever ed, seeping water, so mate unconfined compre- th unconfined compre- th unconfined compre- th unconfined compre-	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h very great h some effo ral inches by subjected to compressive Device us ressive streat	(broken v e perform esive soil t effort ort y thumb v o surface strength led: ngth of 1.	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and let	d to wetting.						
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>sumb Test used to estin</u> Test performed: Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe C - soil easi oil is submerge ne used to est a pe A - soil with pe B - soil with pe C - soil with	Non-cor is easily)	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to ompressive Device us ressive streat subjected to	(broken v e perform essive soil t effort y thumb v o surface strength ned: ngth of 1. o surface	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and let .5 tsf or less water, runoff, exposed	d to wetting.						
Plasticity: Col Dry strength: Grz Wet shake: Wa DTE: The following unc <u>numb Test</u> used to estir Test performed: Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge ne used to esti pe A - soil with pe B - soil with pe B - soil with pe C - soil with pe C - soil with	Non-cor is easily) urface (granular ma- ressive strength test ed compressive stre anted by thumb with tented by thumb with ity penetrated sever anted by thumb with ity penetrated sever anted by thumb with ity penetrated sever anted by thumb with ity penetrated sever anted by thumb with ity penetrated sever anted unconfined compr h unconfined compr h unconfined compr h unconfined compr h unconfined compr h unconfined compr h unconfined compr h unconfined compr	Cohesive Cohesive terial) ts should be ngth of coh No h very great h very great h some effo ral inches by subjected to Device us ressive strer ressive strer subjected to to vibration	(broken v e perform esive soil t effort ort y thumb v o surface strength ed: ngth of 1. ngth great ngth of 0. o surface	with difficulty) Surface remains dry (ed on undisturbed soil i: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and let .5 tsf or less water, runoff, exposed siy disturbed, layered	d to wetting.	f Kcavation on a slope	of 4H: 1				
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following und iumb Test used to estin Test performed: Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge ne used to esti pe A - soil with pe B - soil with pe B - soil with pe C - soil with pe C - soil with	Non-cor is easily)	Cohesive Cohesive (terial) ts should be ngth of coh No	(broken v perform esive soil t effort y thumb v surface strength ed: ngth of 1 ngth great ngth of 0. surface surface	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and let .5 tsf or less water, runoff, exposed isiy disturbed, layered of FICATION	d to wetting.	f Kcavation on a slope	of 4H: 1				
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>umb Test</u> used to estin Test performed: Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi a pe A - soil with pe B - soil with pe B - soil with pe C - soil with oil is submerge a type A if soil	Non-cor is easily)	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to compressive Device us ressive streat essive streat subjected to to vibration SOIL Type B	(broken v e perform esive soil t effort ort y thumb v o surface strength ed: ngth of 1.	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and les swater, runoff, exposed sily disturbed, layered a FICATION Type C	d to wetting.	f kcavation on a slope	of 4H: 1				
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>tumb Test</u> used to estir Test performed: Ty	nesive anular (crumble ter comes to s confined compr mate unconfine Yes pe A - soil inde pe C - soil easi oil is submerge ne used to esti s pe A - soil with pe B - soil with pe B - soil with oil is submerge s type A if soil	Non-cor is easily) urface (granular ma ressive strength test anted by thumb with ented by thumb with ity penetrated sever ad, seeping water, s mate unconfined compre- h unconfined c	Cohesive Cohesive terial)	(broken v o perform esive soil t effort ort y thumb v o surface strength red: ngth of 1 ngth of 1 o surface o, previou	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and let .5 tsf or less water, runoff, exposed water, runoff, exposed sity disturbed, layered of FICATION Type C	d to wetting.	f Kcavation on a slope	of 4H: 1				
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc numb Test used to estin Test performed: Ty	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge ne used to esti a pe A - soil with pe B - soil with pe B - soil with oil is submerge s type A if soil	Non-cor is easily)	Cohesive Cohesive terial) ts should be ngth of coh No h very great h some effo ral inches by subjected to performerssive Device us ressive stren essive stren subjected to to vibration SOIL Type B TON OF PR	(broken v e) perform essive soil t effort ort y thumb v o surface strength ed: ngth of 1. ngth great ngth of 0. o surface b, previou	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf end les swater, runoff, exposed sily disturbed, layered i FICATION Type C E SYSTEM (Appendix	d to wetting.	f xcavation on a slope	of 4H: 1	· · · · · · · · · · · · · · · · · · ·			
Plasticity: Col Dry strength: Gra Wet shake: Wa DTE: The following unc <u>numb Test used to estin</u> Test performed: Ty J	nesive anular (crumble ter comes to s anfined compr mate unconfine Yes pe A - soil inde pe B - soil inde pe C - soil easi oil is submerge ne used to esti s pe A - soil with pe B - soil with pe C - soil with oil is submerge s type A if soil	Non-cor is easily)	Cohesive Cohesive (terial) ts should be ngth of coh No h very great h some effo al inches by subjected to ompressive Device us ressive streat essive streat subjected to to vibration SOIL Type B	(broken v e perform esive soil t effort ort y thumb v e surface strength ed: ngth of 1 o surface surface strength ed: o surface o surface	with difficulty) Surface remains dry (ed on undisturbed soil l: with little or no effort. water, runoff, exposed of cohesive soils: .5 tsf or greater ter than 0.5 tsf and les .5 tsf or less water, runoff, exposed sity disturbed, layered FICATION Type C E SYSTEM (Appendix	clay material) s. d to wetting. d to wetting dipping into ex f) Aluminum hys	f Kcavation on a slope	of 4H: 1V				

Ű

		1				<u> </u>				
Date:	Time: C		Competent Person:							
Soil Type:								1		
Soil Classification:	· · · · · · · · · · · · · · · · · · ·		Excavation	Depth: Excavation Width:				n Width:		
Type of Protective Sys	stem Used:		<u> </u>							
Excevation Entrents:	Time In Tim	e Out Pa	Iramoter	Pro-E	ntry.	Initial	hours	hours	hours	- s-,
		LE	L (<10%)							
		٥,	, (19.5%-22%)							
		Va	platiles (<5 ppm)							
		Ot	ther							
				1						
Yes No N/A	Description			Yes	No	N/A	Description			
GENERAL INSPECTION	OF JOBSITE			υτιμι	IES					
	Employees protected rock or soil that coul excavation	from cav e- ir d fall or roll i	ns and loose into the				Utility comp located and	oanies conta I marked	cted and/or	utiliti
	Spoils, materials, and least 2 feet from the	i squipment : edge of the	set back at excavation				Overhead to precautions does not co	ransmission taken to en ome in conta	lines are not sure that eq ct with then	eciar uipm
	Walkways and bridges over excavations are equipped with standard guardrails and toeboards						Utilities cro from above exist	ssing the ex , and protec	cavation sup ted from fall	porte ing m
	Adequate signs have barricades provided	been posted	and the t				Undergroun supported, open	nd installation or removed	ns protected when excave	, ation
WET CONDITIONS	· · · <u>-</u> -			HAZA	RDOUS	ATMOSP	HERE			
	Precautions taken to protect employees from the accumulation of water						Air in the end deficiency, contaminan	xcavation te combustible its	sted for oxy s, or other	gen
	Surface water or runoff diverted or controlled to prevent accumulation in the excavation					_	Ventilation containing l other hazar	or supplied a less than 19 dous substa	tir used in at .5 % oxyger nce	tmos n and
	Inspection mede after every rainstorm or other hazard increasing occurrence		torm or other				Ventilation	provided to	keep LEL be	low 1
MEANS OF ACCESS A	AND EGRESS						Emergency hazardous	equipment a atmospheres	vailable who could or do	ere exist
	Ladder or ramp egress no greater than 25 feet						Safety harn individually bottom or o	ess and lifeli attended wi other deep co	ne used and nen entering onfined exca	bell Ivatio
	Ladders used secure above the edge of th	d and extend e trench	ed 3 feet	OTHER		MENTS:				
	Ramps constructed o thickness, cleated to equipped with no-slip	of materials o gether on the surface	of uniform e bottom,							
	Employees protected entering or exiting th	from cave-ir e excavation	ns when					-		
							<u></u>			

•

a and the second second second second second second second second second second second second second second se

رسي کافت دينې د د

• • • • •

SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 1 of 3

10.06 Excavations

A. <u>Introduction</u>

This section applies to all open excavations made in the earth's surface. Excavations are defined to include trenches.

B. General Requirements

- 1. All surface encumbrances that are located so as to create a harard to employees will be removed or supported, as necessary, to safeguard employees.
- 2. Undergrouze installations.
 - a. The estimated location of utility installations, such as sewer, telephone, fuel, electric, water lines, or any other underground installations that reasonably may be expected to be encountered during excavation work, will be determined prior to opening an excavation.
 - b. Utility companies or owners will be connected within established or sustomary local response times, advised of the proposed work, and taked to establish the location of the utility underground installations prior to the start of actual excavation. When utility companies or owners cannot respond to a request to locate underground utility installations within 14 hours (unless a longer period is required by state or local law), or cannot establish the exact location of these installations, the work may proceed, provided caution is taken, and provided detection equipment or other acceptable means to locate utility installations are used.
 - c. When excavation operations approach the estimated location of underground installations, the exact location of the installations will be determined by safe and acceptable means.
 - d. While the excavation is open, underground installations will be protected, supported or removed as necessary to safeguard employees.
- C. Access and egress
 - 1. Structural camps.
 - a. Structural ramps that are used solely by employees as a means of access or egress from excavations will be designed by a competent person. Structural ramps used for access or egress of equipment will be designed by a competent person qualified in structural design, and will be constructed in accordance with the design.
 - b. Ramps and runways constructed of two or more structural members will have the structural members connected together to prevent displacement.
 - c. Scructural members used for ramps and runways will be of uniform thickness.
 - d. Clears or other appropriate means used to connect runway structural members will be attached to the bottom of the runway or will be attached in a manner to prevent tripping.
 - e. Structural tamps used in lieu of steps will be provided with clears or other surface treatments on the top surface to prevent slipping.

SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 2 of 3

2. Means of egress from trench excavations. A stnirway, ladder, ramp or other safe means of egress will be located in trench excavations that are 4 feet (1.22 m) or more in depth so as to require no more than 25 feet (7.62 m) of lateral travel for employees.

D. <u>Exposure to vehicular traffic</u>

Employees exposed to public vehicular mattic will be provided with, and will wear, warning vests or other suitable garments marked with or made or reflectorized or high-visibility material.

E. Exposure to failing loads

No employee will be permitted underneath loads handled by lifting or digging equipment. Employees will be required to stand away from any vehicle being loaded or unloaded to avoid being struck by any spillage or failing materials. Operators may remain in the cabs of vehicles being loaded or unloaded when the vehicles are equipped to provide adequate protection for the operator during loading and unloading operations.

F. <u>Warning system for mobile equipment</u>

When mobile equipment is operated adjacent to an excavation, or when such equipment is required to approach the edge of an excavation, and the operator does not have a clear and direct view of the edge of the excavation, a warning system will be utilized such as barricades, hand or mechanical signals, or stop logs. If possible, the grade should be away from the excavation.

G. <u>Hazardous armospheres</u>

1. Testing and controls

In addition to the procedures set forth in Sections 3 and 4 of this Health and Safety Manual to prevent exposure to harmful levels of atmospheric contaminants and to assure acceptable atmospheric conditions, the following requirements will apply:

- a. Where oxygen deficiency (atmospheres containing less than 19.5 percent oxygen) or a hazardous atmosphere exists or could reasonably be expected to exist, such as in excavations in areas where hazardous substances are stored nearby, the atmospheres in the excavation will be tested before employees enter excavations greater than 4 feet (1.22 m) in depth.
- b. Acequate precautions will be taken to prevent employee exposure to atmospheres containing less than 19.5 percent oxygen and other hazardous atmospheres. These precautions include providing proper respiratory protection or ventilation.
- c. Adequate precaution will be taken such as providing ventilation, to prevent employee exposure to an atmosphere containing a concentration of a flammable gas in excess of 20 percent of the lower flammable limit of the gas.
- gas in excess of 20 percent of the lower flammable limit of the gas.
 d. When controls are used that are intended to reduce the level of atmospheric contaminants to acceptable levels, the testing will be conducted as often as necessary to ensure that the atmosphere remains safe.
- 2. Emergency rescue equipment
 - a. Emergency rescue equipment, such as breathing apparatus, a safety harness and line, or a basket stretcher, will be readily available where hazardous atmospheric

SECTION: 10.05 SUPERCEDES: 6/3/92 DATE: 12/31/93 REVIEW: 12/31.95 MARCH: 3 of 3

conditions exist or may reasonably be expected to develop during work in an excavation. This equipment will be attended when in use.

b. Employees entering bell-bottom pier holes, or other similar deep and confined footing excavations, will wear a harness with a life-line securely annohed to it. The lifeline will be separate from any line used to handle materials, and will be individually attended at all times while the employee wearing the lifeline is in the excavation.

H. Protection from hazards associated with water accumulation

- 1. Employees will not work in excavations in which there is accumulated water, or in excavations in which water is accumulating, unless adequate precautions have been taken to protect employees against the hazards posed by water accumulation. The precautions necessary to protect employees adequately vary with each situation, but could include special support or shield systems to protect from cave-ins, water removal to control the level of accumulating water, or use of a safety harness and lifeline.
- 2. If water is controlled or prevented from accumulating by the use of water removal equipment, the water removal equipment and operations will be monitored by a competent person to ensure proper operation.

3. If excavation work interrupts the natural drainage of surface water (such as streams), diversion ditches, dikes, or other suitable means will be used to prevent surface water from entering the excavation and to provide adequate drainage of the area adjacent to the excavation. Excavations subject to runoff from heavy rains will require an inspection by a competent person.

L <u>Stability of adjacent structures</u>

- 1. Where the stability of adjoining buildings, walls, or other structures is endangered by excavation operation, support systems such as shoring, bracing, or underpinning will be provided to ensure the stability of such structures for the protection of employees.
- 2. Excavation below the level of the base or footing of any foundation or retaining wall that could be reasonably expected to pose a bazard to employees will not be permitted except when:
 - a. A support system, such as underpinning, is provided to ensure the safety of employees and the stability of the structure; or
 - b. The excavation is in stable rock; or
 - c. A registered professional engineer has approved the determination that such excavation work will not pose a hazard to employees.
 - d. Sidewalks, pavements, and appurtenant structure will not be undermined unless a support system or another method of protection is provided to protect employees from the possible collapse of such structures.

J. <u>Protection of employees from loose rock or soil</u>

- 1. Adequate protection will be provided to protect employees from loose rock or soil that could pose a hazard by failing or rolling from an excavation face. Such protection will consist of scaling to remove loose material; installation of protective barricades at intervals as necessary on the face to stop and contain falling material; or other means that provide equivalent protection.
- 2. Employees will be protected from excavated or other materials of equipment that could pose a hazard by failing or rolling into excavations. Protection will be provided by

SECTION: 10.06 SUPERCEDES: 6/3/92 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 4 of 3

placing and keeping such materials or equipment at least 2 feet (.61 m) from the edge of excavations, or by the use of remining devices that are sufficient to prevent materials or equipment from falling or colling into excavations, or by a combination of both if necessary.

K. Inspections

- 1. Daily inspections of excavations, the adjacent areas, and protective systems will be made by a competent person for evidence of a situation that could result in possible cave-ins, indications of failure of protective systems, azzardous atmospheres, or other hazardous conditions. An inspection will be conducted by conducted by the competent person prior to the start of work and as needed throughout the shift. Inspections will also be made after every rainstorm or other hazard increasing occurrence. These inspections are only required when employee exposure can be reasonably anticipated.
- 2. Where the competent person finds evidence of a situation that could result in a possible cave-in, indications of failure of protective systems, hazardous annospheres, or other hazardous conditions, exposed employees will be removed from the hazardous area until the necessary precautions have been taken to ensure their safery.

L. Fall protection

- 1. Where employees or equipment are required or permitted to cross over excavations, walkways or bridges with standard guardrails will be provided.
- Adequate barrier physical protection will be provided at all remotely located excavations. All wells, pits, sharts, etc., will be barricaded or covered. Upon completion of exploration and similar operations, temporary wells, pits, sharts, etc., will be backrilled.

M. <u>Requirements for protective systems</u>

b.

- 1. Protection of employees in excavations.
 - Each employee in an excavation will be protected from cave-ins by an adequate protective system designed in accordance with paragraph 2 or 3 of this section except when:
 - 1. Excavations are made entirely in stable rock; or
 - 2. Excavations are less than 5 feet (1.52 m) in depth and examination of the ground by a competent person provides no indication of a potential cave-in.
 - Protective systems will have the capacity to resist without failure all loads that are intended or could reasonably be expected to be applied or transmitted to the

system.

SECTION: 10.06 SUPERCEDES: 6/3/91 DATE: 12/31/93 REVIEW: 12/31/95 MARCH: 5 of 3

2. Design of sloping and benching systems. The slopes and configurations of sloping and benching systems will be in accordance with the requirements of option 1; or, in the alternative, option 2; or, in the alternative, option 3, or, in the alternative, option 4, as follows:

Option (1) - Allowable configurations and slopes

- Excavations will be sloped at an angle not steeper than one and one-half horizontal to one vertical (34 degrees measured from the horizontal), unless one of the other options listed below is used.
- Slopes specified in paragraph 1 of this section, will be excavated to form configurations that are in accordance with the slopes shown for Type C soil in Appendix B to this subpart.

Option (2) - Determination of slopes and configurations using 19 CFR 1926 Subpart P Appendices A and B

Maximum allowable slopes, and allowable configurations for sloping and benching systems, will be determined in accordance with the conditions and requirements set forth in appendices A and B to this subpart.

Cotion (3) - Designs using other topulated data

- Designs of sloping or benching systems will be selected from and be in accordance with tabulated data, such as tables and charts.
- The tabulated data will be in written form and will include all of the following:
 - A. Identification of the parameters that affect the selection of a sloping or benching system drawn from such data:
 - Identification of the limits of use of the data, to include the magnitude and configuration of slopes determined to be safe;
 - C. Explanatory information as may be necessary to aid the user in making a correct selection of a protective system from the data.
- 3. At least one copy of the tabulated data which identifies the registered professional engineer who approved the data, will be maintained at the jobsite during construction of the protective system. After that time the data may be stored off the jobsite, but a copy of the data will be made available to the Secretary upon request.

Option (4) - Design by a registered professional engineer

- Sloping and benching systems not utilizing Option (1) or Option (2) or Option (3) under paragraph 2 of this section will be approved by a registered professional engineer.
- Designs will be in written form and will include at least the following:
 A. The magnitude of the slopes that were determined to be safe for the particular project;
 - B. The configurations that were determined to be safe for the particular project; and
 - C. The identity of the registered professional engineer approving the design.
- 3. At least one copy of the design will be maintained at the jobsite while the slope is being constructed. After that time the design need not be at the jobsite, but a copy will be made available to OSHA upon request.
- Design of support systems, shield systems, and other protective systems. Designs of support systems shield systems, and other protective systems will be in accordance with the requirements of option 1; or, in the alternative, option 2; or, in the alternative, option 3; or, in the alternative, option 4 as follows:

3.

OBG TECHNICAL SERVICES. INC.	SECTION: 10.06
HEALTH AND SAFETY MANUAL	SUPERCEDES: 6/3/92
	DATE: 12/31.93
, <u> </u>	REVIEW: 12/31/95
	MARCH: 6 of 3

<u>Option (1) - Designs using appendices A. C. and D</u> Designs for imber shoring in trenches will be determined in accordance with the conditions and requirements set forth in appendices A and C to this subpart. Designs for aluminum hydraulic shoring will be in accordance with paragraph (c)(C) of this section, but if manufacturer's tabulated data cannot be utilized, designs will be in accordance with Appendix D.

Option (2) - Designs Using Manufacturer's Tabulated Data

- a. Design of support systems, shield systems, or other protective systems
- that are drawn from manufacturer's inbuisted data will be in accordance with all specifications, recommendations, and limitations issued or made by the manufacturer.
- b. Deviation from the specifications, recommendations, and limitations issued or made by the manufacturer will only be allowed after the manufacturer issues specific written approval.
- c. Manufacturer's specifications, recommendations, and limitations, and manufacturer's approval to deviate from the specifications, recommendations, and limitations will be in written form at the jobsite during construction of the protective system. After that time this data may be stored off the jobsite, but a copy will be made available to OSHA upon request.

Option (3) - Designs using other tabulated data

Ъ.

Ъ.

- Designs of support systems, shield systems, or other protective systems will be selected from and be in accordance with tabulated data, such as tables and charts.
- The tabulated data will be in written form and include all of the following:
 - Identification of the parameters that affect the selection of a protective system drawn from such data;
 - 2. Identification of the limits of use of the data;
 - 3. Explanatory information as may be necessary to aid the user in making a correct selection of a protective system from the data.
- c. At least one copy of the tabulated data, which identifies the registered professional engineer who approved the data, will be maintained at the jobsite during construction of the protective system. After that time the data may be stored off the jobsite, but a copy of the data will be made available upon request.

Option (4) - Design by a registered proressional engineer

- a. Support systems, shield systems, and other protective systems not utilizing Cotion 1, Option 2, or Option 3, above, will be approved by a registered professional engineer.
 - Designs will be in written form and will include the following:
 - 1. A plan indicating the sizes, types, and configurations of the materials to be used in the protective system; and
 - 2. The identity of the registered professional engineer approving the design.

SECTION: 10.06 -SUPERCEDES: 6/3/92 DATE: 12/31.93 REVIEW: 12/31/95 MARCH: 7 of 3

- c. At least one copy of the design will be maintained at the jobsite during construction of the protective system. After that time, the design may be stored off the jobsite, but a copy of the design will be made available to OSHA upon request.
- 4. Marerials and equipment.
 - a. Materials and equipment used for protective systems will be free from damage or defects that might impair their proper function.
 - b. Manufactured materials and equipment used and maintained in a manner that is consistent with the recommendations of the manufacturer, and in a manner that will prevent employee exposure to hazards.
 - c. When material or equipment that is used for protective systems is damaged, a competent person will examine the material or equipment and evaluate its suitability for continued use. If the competent person cannot assure the material or equipment is able to support the intended loads or is otherwise suitable for safe use, then such material or equipment will be removed from service, and will be evaluated and approved by a registered professional engineer before being remirned to service.
- 5. Installation and removal of support
 - a. General.
 - 1. Members of support systems will be securely connected together to prevent sliding, falling, kickouts, or other predictable failure.
 - Support systems will be installed and removed in a manner that protects employees from cave-ins, structural collapses, of from being struck by members of the support system.
 - 3. Individual members of support systems will not be subjected to loads exceeding those which those members were designed to withstand.
 - 4. Before temporary removal of individual members begins, additional precautions will be taken to ensure the safery of employees, such as installing other structural members to carry the loads imposed on the support system.
 - 5. Removal will begin at, and progress from, the bottom of the excavation. Members will be released slowly so as to note any indication of possible failure of the remaining members of the structure or possible cave-in of the sides of the excavation.
 - 6. Backfilling will progress together with the removal of support systems from excavations.
 - ь.
- Additional requirements for support systems for trench excavations.
 - Excavation of material to a level no greater than 2 feet (.61 m) below the bottom of the members of a support system will be permitted, but only if the system is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open of a possible loss of soil from behind or below the bottom of the support system.

SECTION: 10.05 SUPERCEDES: 5/3/92 DATE: 12/31.93 REVIEW: 12/31.95 MARCH: 8 of 3

Installation of a support system will be closely coordinated with the 2. excavation of trenches.

N. Sloping and benching systems

Employees will not be permitted to work on the faces of sloped or benched excavations at levels above other employees except when employees at the lower levels are adequately protected from the hazard of failing, foiling, or sliding material or equipment.

0. Shield systems

- 1. General
 - Shield systems will not be subjected to loads exceeding those which the system з. was designed to withstand.
 - Shield will be installed in a manner to restrict lateral or other hazardous Ъ. movement of the shield in the event of the application of sudden lateral loads.
 - Employees will be protected from the hazard of cave-ins when entering or exiting ¢.
 - the areas protected by shields. Employees will act be allowed in shield when shields are being installed, đ. removed, or moved vertically.
- 2. Additional requirement for shield systems used in trench excavations. Excavations of earth material to a level not greater than 2 feet (.61 m) below the bottom of a shield will be permitted, but only if the shield is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open of a possible loss of soil from behind or below the bottom of the shield.

CONSTRUCTION STANDARDS

(1) Option (1)—Designs using appendicat A. C. and D. Designs for under shorting in trenches shall be determined in accordance with the conditions and reduirements set forth in appendices A and C to this subpart. Designs for aluminum hydraulic shorting shall be in accordance with paragraph (c)(2) of this section, but if manufacturer's tabulated data cannot be utilized, designs shall be in accordance with appendix D.

(2) Option (2)—Designs Using Manufacturer's Tabulated Data. (i) Design of support systems, shield systems, or other protective systems that are drawn from manufacturer's tabulated data shall be in accordance with all specifications, recommendations, and limitations issued or made by the manufacturer.

(ii) Deviation from the specifications, recommendations, and ilmitations issued or made by the manufacturer shall only be allowed after the manufacturer issues specific written approval.

(ii) Manufacturer's specifications, recommendations, and limitations, and manufacturer's approval to deviate from the specifications, recommendations, and limitations shall be in written form at the jobsite during construction of the protective system. After that time this data may be stored off the jocsite, but a copy shall be made available to the Secretary upon request.

(3) Option (3)—Designs using other tabulated data. (i) Designs of support systems, shield systems, or other protective systems shall be selected from and be in accordance with tabulated data, such as tables and charts.

(ii) The tabulated data shall be in written form and include all of the following:

(A) Identification of the parameters that affect the selection of a protective system drawn from such data:

(B) Identification of the limits of use of the data:

(C) Explanatory information as may be necessary to aid the user in making a correct selection of a protective system from the data.

(iii) At least one copy of the tabulated data, which identifies the registered professional engineer who approved the data, shall be maintained at the jobsite during construction of the protective system. After that time the data may be stored off the jobsite, but a copy of the data shall be made available to the Secretary upon request.

(4) Option (4)—Design by a registered professional engineer. (i) Support systems, shield systems, and other protective systems not utilizing Option 1. Option 2 or Option 3, above, shall be approved by a registered professional engineer.

(ii) Designs shall be in written form and shall include the following:

(A) A plan indicating the sizes, types, and configurations of the materials to be used in the protective system; and

. .

(B) The identity of the registered proressional engineer approving the design.

(iii) At least one sopy of the design shall be maintained at the jobsite during construction of the protective system. After that time, the design may be stored off the jobsite, but a copy of the design shall be made available to the Secretary upon request.

(d) Materials and equipment. (1) Materials and equipment used for protective systems shall be free from damage or defects that might impair their proper function.

(2) Manufactured materials and equipment used for protective systems shall be used and maintained in a manner that is consistent with the recommendations of the manufacturer, and in a manner that will prevent employee exposure to harares.

(3) When material or equipment that is used for protective systems is damaged, a competent person shall examine the material or equipment and evaluate its suitability for continued use. If the competent person dannot assure the material or equipment is able to support the intended loads or is otherwise suitable for safe use, then such material or equipment shall be removed from service, and shall be evaluated and approved by a registered professional engineer before being returned to service.

(e) Installation and removal of support—(1) General. (i) Members of support systems shall be securely connected together to prevent slicing, failing, kickouts, or other predictable failure.

(ii) Support systems shall be installed and removed in a manner that protects employees from cave-ins, structural collapses, or from being struck by members of the support system.

(iii) Individual members of support systems shall not be subjected to loads excreding those which those members were designed to withstand.

(iv) Before temporary removal of individual members begins, additional precautions shall be taken to ensure the safety of employees, such as installing other structural members to carry the loads imposed on the support system.

(v) Removal shall begin at, and progress from, the bottom of the excavation. Members shall be released slowly so as to note any indication of possible failure of the remaining members of the structure or possible cave-in of the sides of the excavation.

(vi) Backfilling shall progress together with the removal of support systems from excavations.

(2) Additional requirements for support systems for trench excavations. (i) Excavation of material to a level no greater than 2 fest (.51 m) below the bottom of the members of a support system shall be permitted, but only if the system is designed to resist the forces calculated for the full depth of the trench, and there are no indications while the trench is open

of a possible loss of soil from behind or below the bottom of the support system. (ii) installation of a support system

(ii) installation of a support system shall be mosely coordinated with the encavation of trenenes.

(i) Slooing and benching systems. Employees shall not be permitted to work on the faces of sloped or benched exceptations at levels above other employees except when employees at the lower levels are adequately protocted from the hazard of failing, rolling, or sliding material or equipment.

(g) Shield systems ---(1) General. (i) Shield systems shall not be subjected to loads exceeding those which the system was designed to withstand.

(ii) Shields shall be installed in a manner to restrict lateral or other hanardous movement of the shield in the event of the application of sudden lateral loads.

(iii) Employees shall be protected from the bazard of cave-ins when entering or exiting the areas protected by shields.

(iv) Employees shall not be allowed in shields when shields are being installed, removed, or moved vertically.

(2) Additional requirement for shield systems used in trench excavations. Excavations of earth material to a level not greater than 2 feet (.51 m) below the bottom of a shield shall be permitted, but only if the shield is designed to resist the forces miculated for the full depth of the trench and there are no indications while the trench is open of a possible loss of soli from behind or below the bottom of the shield.

Appendix A to Subpart P

Soil Classification

(a) Scope and application-(1) Scope. This appendix describes a method of classifying soil and rock deposits based on site and environmental conditions, and on the structure and composition of the earth deposits. The appendix contains definitions, sets forth requirements, and describes acceptable visual and manual tests for use in classifying soils.

(2) Application. This appendix applies when a sloping or benching system is designed in accordance with the requirements set forth in §1925.552(b)(2) as a method of protection for employees from cave-ins. This appendix also applies when timber shoring for excavations is designed as a method of protection from cave-ins in accordance with appendix C to subpart P of part 1926, and waea sluminum bydrzulic shoring is designed in accordance with appendix D. This Appendix also applies if other protective systems are designed and selected for use from data prepared in accordance with the requirements set forth in §1926.652(c), and the use of the data is predicated on the use of the soil classification system set forth in this appendix.

(b) Definitions. The definitions and examples given below are based on, in whole or in part, the following: American Society for Testing Materials (ASTM) Standards D653-65 and D2438; The U.S. Department of Agriculture (USDA) Textural Classification

Seneme: and The National Bureau of Stanfarm Report BSS-(21)

Commented tott means a soil in which the particles are held together by a chemical agent, such as calcium carbonates such that a hand-sized sample cannot be crushed into powder or individual soil particles by inger pressure.

Cohesive soil means siny (fine grained soil), or soil with a high day content, which has cohesive strength. Cohesive soil does not crimble, can be excavated with vertical sidencoes, and is plastic when moist. Cohesive soil is hard to break up when dry, and entibits significant cohesion when submerged. Cohesive soils include single site sancy day, silty day, alay and organic day.

Dry soil means soil that does not exhibit visible signs of moisture content.

Fiscured means a soil materia, that has a tendency to break glong definite planes of fracture with little resistance, or a material that exhibits open cracks, such as tension cracks, in an exposed surface.

Granular soil means gravei, sand, or silt. (cearse grained soil) with little or no day content. Granular soil has no cohesive strength. Some moist granular soil sannot be moided when moist and crumbles easily when dry.

Lavered system means two it more distinony different soil or rock types arranged in layers. Micaceous seams or weakened planes in rock or shale are considered layered.

Moist soil means a condition in which a soil looks and feels damp. Moist cohesive soil can easily be shaped into a ball and rolled into small diameter threads before crumbling. Moist granular soil that contains some cohesive material will exhibit signs of cohesion between particles.

Plastic means 1 property of 2 soil which allows the soil to be deformed or molded without cracking, or appreciable volume change.

Saturated soil means a soil in which the voids are filled with water. Saturation does not require flow. Saturation, or near saturation, is necessary for the proper use of instruments such as a pocket penetrometer or sheer vane.

Soil classification system means, for the purpose of this subpart, a method of categorizing soil and rock deposits in a hierarchy of Stable Rock. Type A. Type B. and Type C. in decreasing order of stability. The categories are determined based on an analysis of the properties and performance characteristics of the deposits and the environmental conditions of exposure.

Stable rock means natural solid mineral matter that can be excavated with vertical sides and remain intact while exposed.

Submerged soil means soil which is underwater or is free seeping.

Type A means conesive soils with an unconfined compressive strength of 1.5 ton per square foot (151) (1 ± 4 kPa) or greater. Examples of cohesive soils are: ciay, silty clay, sandy clay, clay loam and, in some cases silty clay loam and sandy clay loam. Cemented soils such as callche and hardpan are also considered Type A. However, no soil is Type A if:

(i) The soil is assured: or

(8) The soil is judgen to vibration from heavy traffic, one inving, or similar effects; or

(iii) The soil ins teen previously dis-

(iv) The soil is part of a sloped, layered system where the layers dip into the excavation on a slope of four portiontal to one vertical (4H: 1V) or greater, or

 (v) The material is subject to other factors that would require it to be classified as a less stable material.

Type 3 means:

(i) Cohesive soil with an unconduct compressive strength greater than 0.5 af (48 kPa) but less that 1.5 af (144 kPa); or

(ii) Granular schemeniess solls including: ingular gravel (similar to crushed rock), suit silt learn, sandy learn ind, in some cases silty day learn ind sandy may learn.

(iii) Previously disturbed soils except those which would otherwise be classed is Type C soil.

(iv) Soil that meets the unconfined compressive strength or computation requirements for Type A, but is fissured or subject to vibration; or

(v) Dry rock that is not stable: or

(vi) Material that is part of a sloped, layered system where the layers dip into the excavation on a slope less steep than four horizontal to one verical (4H:1V), but only if the material would otherwise be classified as Type 3.

Type C means:

(i) Cobesive soil with an unconfined compressive strength of 0.5 (43 kPa) or less: or

(ii) Granular soils including gravel, sand, and loamy sand, or

(iii) Submerged soll or soll from which water is fracily sceping; or

(iv) Submerged rock that is not stable, or (v) Material in a sloped, layered system

where the layers dip into the excavation or a slope of four horizontal to one vertical (4H:1V) or steeper.

Unconfined compressive strength means the load per unit area at which a soil will fail in compression. It can be determined by laboratory testing, or estimated in the field using a pocket penetrometer, by thumb penetration tests, and other methods.

Wet soil means soil that contains significantly more moisture than moist soil, but in such a range of values that cohesive material will slump or begin to flow when vibrated. Granular material that would exhibit cohesive properties when moist will lose those cohesive properties when wet.

(c) Requirements—(1) Classification of soil and rock deposits. Each soil and rock deposit shall be classified by a competent person as Stable Rock. Type A. Type B, or Type C in accordance with the definitions set forth in paragraph (b) of this appendix.

(2) Basis of classification. The classification of the deposits shall be made based on the results of at least one visual and at least one manual analysis. Such analyses shall be conducted by a competent person using tests described in paragraph (d) below, or in other recognized methods of soil classification and testing such as those adopted by the America Society for Testing Materials, or the U.S. Department of Agriculture textural classification system.

(3) Visual and manual analyses. The visual and manual analyses, such as those not-

.

se is teng inceptible a paragraph (d) of this inpendia, that to ferigred and conducted to provide sufficient quantitative and qualitative information is may be necessary to identify property the properties, factors, and conditions infecting the answindation of the reconsts.

(4) Layered systems. Is a layered system, the system shall be classified in accordance with its weakert layer. However, each layer may be classified individually where a more stable layer lies under a less stable layer. (5) Reclassification. If, after classifying a

(5) Reclassification. If after classifying a deposit, the properties, factors, or conditions infecting its massification change in any way, the changes shall be evaluated by a competent perton. The isposit shall be reclassified as necessary to redect the changed directionstances.

(d) Accessible visual and manual lesiz—(i) Visual tests. Visual analysis is conducted to determine qualitative information regarding the encovation site in general, the soil idjacent to the encovation. the soil forming the sides of the open excavation, and the soil taken is samples from excavation ed maternal.

(i) Observe samples of soil that are excavated and soil in the sides of the encavation. Estimate the range of particle sizes and the relative amounts of the particle sizes. Soil that is primarily composed of fine-grained material is cohesive material. Soil composed primarily of mane-grained sand or gravel is granular material.

(ii) Observe soil as it is excavated. Soil that remains in slumps when excavated is cohesive. Soil that breaks up easily and does not stay in slumps is gradular.

(iii) Observe the side of the open excavation and the surface area adjacent to the excavation. Crack-like openings such as tension oracks could indicate Essured material. If chunks of soil spail of a vertical side, the soil could be Essured. Small spails are evidence of moving ground and are indications of potentially hazardous situations.

(iv) Observe the area adjacent to the excavation and the excavation itself for evidence of existing utility and other underground structures, and to identify previously disturbed soil.

(v) Observe the opened side of the excavation to identify layered systems. Examine layered systems to identify if the layers slope toward the excivation. Estimate the degree of slope of the layers.

(vi) Observe the area adjacent to the excavation and the sides of the opened excavation for evidence of surface water, water seering from the sides of the excavation. Or the location of the ievel of the water table.

(vii) Observe the area adjacent to the excavation and the area within the excavation for sources of vibration that may affect the stability of the excavation face.

(1) Manual tests. Manual analysis of soil samples is conducted to determine quantitative is well as qualitative properties of soil and to provide more information in order to classify soil property.

(i) Plassicity. Moid a moist or wet sample of soil into a ball and attempt to roll it into threads as thin as V-not in diameter. Cohesive material can be successfully rolled into threads without crambling. For example, if at least a two inch (50 cam) length of Vinch thread can be heid on one and without teering, the soil is cohesive.

CONSTRUCTION STANDARDS

suantifative 33 well as qualifative properties of 3011 and to provide more information in order to classify 3011 property

ill Plannany, Mord a moist or wet sample of soil into a bail and attempt to roll it into intrads as thin as monot in manater. Conserve material can be successfully rolled into threads without mumbung. For example, if at least a two inch (50 mm) length of Minch thread can be beid on one and without tearing the soil is concerve.

(ii) Dry strength. If the soil is dry and crumbles on its own or with moderate pressure into individual grains or fine powder, it is granular (any combination of gravel, sand, or silt). If the soil is dry and falls into clumps which break up into smaller strumps, but the smaller diamos can only be broken up with difficulty, it may be day in any combination with gravel, and or silt. If the dry soil breaks into clumps which do not break up into small diamos and which dan only be broken with difficulty and there is no visual indication the soil is fastured, the soil may be considered unfasture.

(iii) Thumb penetration. The chumb penetration test and be used to estimate the unconfined compressive strength of cohesive soils. This test is based on the mund penetration test described in American Society for Testing and Matemals (ASTM) Standard designation D2+35-"Standard Recommended Practice for Description of Souls (Visual-Manual Procedure),") Type A soils with an unconfined compressive strength of 1.5 is? can be readily indented by the thumb: however, they can be penetrated by the thumb only with very great affort. Type C soils with an unconfined compressive strangth of 0.5 isf can be easily penetrated several inches by the thumb, and can be moided by light finger pressure. This test should be conducted on an undisturbed soil sample, such as a large clump of spoil, as scon as practicable after excavation to keep to a minimum the effects of exposure to drying influences. If the excavation is later exposed to werting influences (rain, flooding). the classification of the soil must be changed accordingiv.

(iv) Other strangth tests. Estimates of unconfined compressive strongth of soils can also be obtained by use of a pocket penetrometer or by using a hand-operated shearvane.

(v) Organg test. The basic purpose of the drying test is to differentiate perween conesive material with fissures, unfissured conesive material, and granular material. The procedure for the drying test involves drying a sample of soil that is approximately one inch thick (2.54 cm) and six inches (15.24 cm) in diameter until it is thoroughly drying

(A) If the sample develops cracks as it dress significant fissures are indicated.

(B) Samples that dry without cracking are to be broken by hand. If considerable force is necessary to break a sample, the soil has significant cohesive material content. The soil can be classified as a unifseured concesive material and the unconfined compressive strength should be determined.

(C) If a sample breaks easily by hand, it is estner a fissured cohesive material or a granular material. To distury ush between the two, pulvence the dried diamps of the sample by hand or by stepping on them. If the diamps do not pulvence easily, the material is cohesive with fissures. If they pulvence easily into very small fragments, the material is granular.

Appendix 3 to Subpart P

Sloping and Senshing

(a) Scope and application. This appendix contains specifications for sloping and benaming when used as methods of protecting employees working in excavations from caveins. The requirements of this appendix apply when the design of sloping and benching protective systems is to be performed in accordance with the requirements set forth in § 1903.652(b)(2).

(b) Definitions.

Actual slope means the slope to which an excavation face is excavated.

Distress means that the soil is in a condition where a cave-in is imminent or is likely to occur. Distress is evidenced by such phenomena as the development of fissures in the face of or adjacent to an open excavanon: the subsidence of the edge of an excavanon; the subsidence of the edge of an excavanon; the subsidence of material from the buttom of an excavation; the spailing of (material from the face of an excavation; and raveiling its, small amounts of material such as peoples or little clumps of material suddenty separating from the face of an excavation and modiling or rolling down into the excavanon.

Maximum allowable slope means the steepest incline of an excavation face that is acceptable for the most favorable site conditions as protection against cave-ins, and is expressed as the rado of horizontal distance to vernaal new (HAV).

Short term expressive means a period of time less than or equal to 24 dours that an excavation is open.

(c) Recurrenter (1) Soul stassification. Soil and mox deposits shall be tlassified in accordance with appendix A to subpart P of part 1921.

(2) Meximum silowebie siose. The maximum silowable sloce for a soil or rock deposit shall be determined from Table 3-1 of this appendix.

(3) Activat stope: (3) The actual stope shall not be steeper than the maximum allowable slope.

(ii) The actual slope shall be less steep than the maximum allowable slope, when there are signs of fistness. If that situation occurs, the slope shall be cut back to an actual slope which is at least 'n homeontal to one vertical ('mfritt') less steep than the maximum allowable slope.

(iii) When sucharge loads from stored material or equipment, operating equipment, or traffic are present, a competent person shall determine the degree to which the actual slope must be reduced below the maximum allowable slope, and shall assure that such reduction is achieved. Surcharge loads from adjacent structures shall be evaluated in accordance with § 1925.851(i).

(4) Configurations. Configurations of sloping and benching systems shall be in accordance with Figure 3-1.

12-3-39

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, G.C. 20027

95

[Appendix 3]

31:3182.4

TABLE B-1 MAXIMUM ALLOWABLE SLOPES

SOIL OR ROCK TYPE	MAXIMUM ALLOWABLE SLOPES (H:V) [] FOR EXCAVATIONS LESS THAN 20 FEET DEEP []]
STABLE ROCK	VERTICAL (90 ⁻)
TYPE A [2]	3/4:1 (53 ^c)
TYPE B	1:1 (45)
TYPE C	1 ¹ ₂ :1 (34)

NOTES:

•

:

- 1. Numbers shown in parentheses next to maximum allowable slopes are angles expressed in degrees from the horizontal. Angles have been rounded off.
- A short-term maximum allowable slope of 1/2H:1V (63°) is allowed in excavations in Type A soil that are 12 feet (3.67 m) or less in depth. Short-term maximum allowable slopes for excavations greater than 12 feet (3.67 m) in depth shall be 3/4H:1V (53°).
- Sloping or benching for excavations greater than 20 feet deep shall be designed by a registered professional engineer.

Figure B-1

Slope Configurations

(All slopes stated below are in the horizontal to vertical ratio)

B-1.1 Excavations made in Type A soil.

1. All simple slope excavation 20 feet or less in depth shall have a maximum allowable slope of %:1.



Simple Slope-General

Exception: Simple slope excavations which are open 24 hours or less (short term) and which are 12 feet or less in depth shall have a maximum allowable slope of %:1.

[Sec. 1926.552, Table B-1]

Occupational Safety & Health Reporter

70

CONSTRUCTION STANDARDS

e

S-2+2 31:3182.5



Simple Slope-Short Term

Z. All benched excavations 22 feet or less in depth shall have a maximum allowable slope of % to 1 and maximum bench dimensions as follows:



Simple Bench



Multiple Bench

۰.

÷.,

3. All excavations 5 feet or less in depth which have unsupported vertically sided lower portions shall have a maximum vertical side of 3% feet.

7:

12-6-89

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

31:3182.5

.

REFERENCE FILE



Unsupported Vertically Sided Lower Portion-Maximum 8 Feet in Depth

All excavations more than 3 feet but not more than 12 feet in depth which unsupported vertically sided lower portions shall have a maximum allowable slope of 1:1 and a maximum vertical side of 3% feet.



Unsupported Vertically Sided Lower Portion-Maximum 12 Feet in Depth

All excavations 20 feet or less in depth which have vertically sided lower portions that are supported or shielded shall have a maximum allowable slope of %:1. The support or shield system must extend at least 18 inches above the top of the vertical side.



Suported or Shielded Vertically Sided Lower Portion

4. All other simple slope, compound slope, and vertically sided lower portion excavations shall be in accordance with the other options permitted under \$ 1928.652(b).

B-1.2 Excavations Made in Type B Soil

1. All simple slope excavations 20 feet or less in depth shall have a maximum allowable slope of 1:1.

[Sec. 1925.552, Table B-1.2]

Occupational Safety & Health Reporter

72 -

CONSTRUCTION STANDARDS

\$-240 31:3132.7



Simple Slope

2. All benched excavations 20 feet or less in depth shall have a maximum allowable slope of 1:1 and maximum bench dimensions, as follows:







Multiple Beach

3. All excavations 20 feet or less in depth which have vertically sided lower portions shall be shielded or supported to a height at less 13 inches above the top of the vertical side. All such excavations shall have a maximum allowable slope of 1:1.

[Sec. 1926.652, Figure B-1.3]

12-6-89

•••

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

72

.



Vertically Sided Lower Portion

4. All other sloped excavations shall be in accordance with the other options permitted in § 1978.852(b).

31:3182.8

B-1.3 Excavations Made in Type C Soil

1. All simple slope excavations 20 feet or less in depth shall have a maximum allowable slope of 1567.



Simple Slope

2. All excavations 20 feet or less in depth which have vertically sided lower portions shall be shielded or supported to a height at least 13 inches above the top of the vertical side. All such excavations shall have a maximum allowable slope of 1%:1.



Vertical Sided Lower Portion

3. All other sloped excavations shall be in accordance with the other options permitted in § 1925.652(b).

B-1.4 Excevations Made in Layered Soils

1. All excavations 20 feet or less in depth made in layered soils shall have a maximum allowable slope for each layer as set forth below.

[Sec. 1925.552, Figure B-1.4]

Occupational Safety & Health Reporter

74





2. All other sloped excavations shall be in accordance with the other options permitted in § 1923.652(b).

Appendix C to Subpart P

Timber Shoring for Trenches

(a) Scope. This appendix contains information that can be used timber shoring is provided as a method of protection from cave-ins in trenches that do not exceed 20 feet [6.1 m] in depth. This appendix must be used when design of timber shoring protective systems is to be performed in accordance with § 1928.852(c)[1]. Other timber shoring configurations: other systems of support such as hydraulic and pneumatic systems: and other protective systems such as sloping, beaching, shielding, and freezing

Occupational Safety & Health Reporter

systems must be designed in accordance with the requirements set forth in § 1923.852(b) and § 1923.852(c).

(b) Soil Classification. In order to use the data presented in this appendix, the soil type or types in which the excavation is made must first be determined using the soil

[Appendix C]

,75
CONSTRUCTION STANCARDS

classification method set form in appendix A of succart P of this part.

(c) Presentation of information. Information is presented in several forms as follows:

(1) information is presented in tabilat form in Tables C-1.3. C-1.2, and C-1.3, and Tables C-2.1. C-2.2 and C-2.2 following paragraph (2) of the appendix. Each table presents the minimum stress of timber members to use in a shoring system, and each table contains data only for the particular soil type in which the excavation of portion of the excavation is made. The data are arranged to allow the user the flexibility to select from among several acceptable configurations of members based on varying the horizontal spacing of the trossbraces. Stable rock is exempt from shoring requirements and therefore, no data are presented for this condition.

(2) Information concerning the basis of the tabular data and the limitations of the data is presented in paragraph (d) of this appendix. and on the tables themselves.

(3) information explaining the use of the tabular data is presented in paragraph (e) of this appendix.

(4) information illustrating the use of the tabular data is presented in paragraph (f) of this appendix.

(3) Misocilaneous notations regarding Tables C-1.1 through C-1.3 and Tables C-21 through C-2.2 are presented in paragraph (g) of this Appendix.

(d) Bosis and limitations of the dota.--(1) Dimensions of timber members. (i) The sizes of the timber members listed in Tables C-1.1 through C-1.3 are taken from the National Bureau of Standards (NES) report. "Recommended Technical Provisions for Construction Practice in Shoring and Sloping of Trenches and Excavations." In addition, where NBS did not recommend specific sizes of members, member sizes are based on an analysis of the sizes required for use by existing codes and on empirical practice.

(ii) The required dimensions of the members listed in Tables C-1.1 through C-1.3 refer to actual dimensions and not nominal dimensions of the timber. Employers wanting to use nominal size shoring are directed to Tables C-2.1 through C-2.3, or have this choice under § 1925.3521cl(3), and are referred to The Corps of Engineers. The Burseu of Reclamation or data from other acceptable sources. (2) Limitation of apolication. (i) It is not intended that the under shoring specification apply to every situation that may be expenienced in the field. These data were developed to apply to the situations that are most commonly expensed on current tenching practice. Shoring systems for use in situations that are not covered by the data in this appendix must be designed as specified in § 1928;452(c).

(ii) When any of the following conditions are present, the members specified in the tables are not considered adequate. Either an alternate timber shoring system must be designed or another type of protective system designed in accordance with § 1928.852.

(A) When loads imposed by structures or by stored material adjacent to the trench weigh in excess of the load imposed by a two-foot soil surcharge. The term "adjacent" as used here means the area within a horizontal distance from the edge of the wench equal to the depth of the trench.

(B) When vertical loads imposed on cross braces exceed a 240-pound gravity load distributed on a one-foot section of the center of the mossbrace.

(C) When surcharge loads are present from equipment weighing in excess of 20.000 pounds.

(D) When only the lower portion of a trench is shored and the remaining portion of the trench is sloped or benched unless: The sloped portion is sloped at an angle less steep than three horizontal to one vertical: or the members are selected from the tables for use at a depth which is determined from the top of the overall trench, and not from the top of the sloped portion.

(e) Use of Tables. The members of the scoring system that are to be selected using this information are the cross braces, the uprights, and the wales, where wales are required. Minimum sizes of members are specified for use in different types of sail. There are six tables of information, two for each soil type. The soil type must first be determined in accordance with the soil classification system described in appendix A to subpart P of part 1925. Using the appropriate table, the selection of the size and spacing of the members is then made. The selection is based on the depth and width of the mench where the members are to be installed and, in most instances, the selection is also based on the horizontal

spacing of the crossbraces. Instances where a choice of horizontal spacing of crossbracing is available, the horizontal spacing of the crossbraces must be chosen by the user before the size of any member can be determined. When the soil type, the width and depth of the trench, and the horizontal spacing of the crossbraces are known, the sure and vertical spacing of the crossbraces, the size and vertical spacing of the wales, and the size and borizontal spacing of the uprights can be read from the appropriate table.

(1) Examples to Illustrate the Use of Tables C-1.1 through C-1.2.

(1) Example 1.

A rench dug in Type A soil is 13 feet deer and five feet wide.

From Tuble G-1.1. for acceptable arrangements of timber can be used.

Arrangement #1

Space 4 × 4 crossbraces at six feet homeontaily and four feet vertically. Wales are not required.

Space 2×3 uprights at six feet horizontally This arrangement is commonly called "skip shoring."

Arrangement =2

Space 4×8 crossbraces at eight feet, horizontally and four feet vertically. Space 3×3 wales at four feet vertically.

Space 1x 5 uprights at four feet horizontally.

Arrongement =3

Space 3×3 crossbraces at 10 feet horizontally and four feet vertically. Space 3×10 wales at four feet vertically.

Space 1×8 uprights at five faet horizontally.

Arrongement 🛋

Space 5×6 crossbraces at 12 feet horizontally and four feet vertically.

Space 10×10 wales at four fact vertically. Spaces 3×8 uprights at six feet

horizontally.

(2) Example 2.

A trench dug in Type B soil in 13 feet deep and five feet wide. From Table C-1.2 three acceptable arrangements of members are listed.

Arrangement ≠1

12-8-89

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

77

Space 5% 5 crossbraces at six feet homeonially and five feet vertically. Space 3% 3 wales at five feet vertically. Space 2% 5 uprights at two feet homeonially.

Arrangement =2

Space 5×3 crossoraces at eight left homeonially and five left vertically.

Space 10 × 10 waies at five feet vertically. Space 2×6 uprights at two feet

homzontaily.

Arrangement =J

Scace 3×8 crossbraces at 10 (eet

homionially and five feet vertically. Space 10×12 wales at five feet vertically. Space 2×3 comptis at two feet vertically. (3) Example 3.

A trench dug in Type C soil is 13 feet deep and five feet wide.

From Table C-1.3 two acceptable arrangements of members can be used.

Arranzement #1

Space 3×5 mosscraces at six (eet

homiontally and five feet vertically. Space 10×12 wales at five feet vertically. Position 2×3 uprights as closely together

as possible. If water must be retained use special

tangue and groove uprights to form tight sheeting.

Arrengement #2

Space 8 x 10 crossbraces at eight feet homeontaily and five feet vertically.

Space 12×12 wales at five feet vertically. Position 2×5 uprights in a close sheeting configuration unless water pressure must be resisted. Tight sheeting must be used where water must be retained.

(4) Example 4.

A trench dug in Type C soil is 20 feet deep and 11 feet wide. The size and spacing of members for the section of trench that is over 15 feet in depth is determined using Table C-1.3. Only one arrangement of members is provided. Space 3 × 10 crossbraces at six feet homeontally and five feet vertically. Space 12× 12 wates at five feet vertically. Use 2× 5 tight sheeting. Use of Tables C-21 through C-23 would (ollow the same procedures. (g) Notes for all Tables. 1. Member sizes at spacings other than

indicated are to be determined as specified in § 1925.652(c), "Design of Protective Systems."

L When conditions are saturated or submerged use Tight Sheeting. Tight Sheeting refers to the use of specially-edged timber planks (e.g., tongue and groove) at least three inches thick, steel sheet piling, or similar construction that when driven or placed in position provide a tight wall to resist the lateral pressure of water and to prevent the loss of backfill material. Close Sheeting refers to the placement of planks side-by-side allowing as little space as possible between them.

3. All spacing indicated is measured center to center.

4. Wales to be installed with greater dimension horizontal

5. If the vertical distance from the center of the lowest crossbrace to the bottom of the trench exceeds two and one-half fast, uprights shall be firmly embedded or a mudsill shall be used. Where uprights are embedded, the vertical distance from the center of the lowest crossbrace to the bottom of the trench shall not exceed 26 inches. When mudsills are used, the vertical distance shall not exceed 42 inches. Mudsills are wales that are installed at the toe of the trench side.

6. Trench jacks may be used in lieu of or in combination with timber crossbraces.

7. Placement of crossbraces. When the vertical spacing of crossbraces is four feet, place the top crossbrace no more than two feet below the top of the trench. When the vertical spacing of crossbraces is five feet, place the top crossbrace no more than 2.5 feet below the top of the trench.

Occupational Salaty & Health Reporter

[Appendix C]

TABLE C-1.1

۲ TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS

P_a = 25 X H + 72 psf (2 ft Surcharge) SOIL TYPE A

	· · · · · · · · · · · · · · · · · · ·				512	(ACTU	NL) NIO	SPACTIC.	OL NUMBU	11				
WI d 30			CRO	S BRACI	5			IVI	5		IN IN	'KIGHTS_		-
	110812.	IN	DTH OF	TREACH	(LET)		VERT.		VÍRT.	TIME YAN	I ALLOUAL	U.E. 110R1	ZORTAL S	PAC 111G
(FFFT)	SPACING	IIP TO	UP 10	UP T0	UP 10	UP 10	SPAC 186	S12E	SPACING			(FEET)		
1 1	(FEET)	4	9	6	12	15	(1.11)	÷ E	(TEET)	CLOSE		5		
	NP 10				2		•	hot						
2	9	4 X 4	-171-	476	<u>070</u>	979	*	1.0.1					4 <u>7</u>	
10	0, 10 0	4 X 1	4X1	4 X 5	6X6	6Хб	4	Reg ^t d	-					2 X B
01	UP T0	4 X K	4 X 6	4 X G	6 X 6	6X6		AXA	V		-	2 X 6		
	UP 10 12	4 X 6	4 X 6	6X6	6X6	6X6	4	8X0					2 X 6	
	01 JU	0 Y 0	AYA	A Y.G	6 X K	616	. 	Not Rea'd					17.8	
2 2	UP 10 8	4 X 6	4 X 6	6X6	6X6	6X6		8X8	ŀ		2X6		N. 1.	
	UP 10	6X6	6X5	6X6	6XB	UX9	-	01X8				2X6		
•	UP TO 12	6X6	- 6X6	6X6	6X8	6 X B	Ł	01X01	4				3X0	
-	0F 10 6	6 X 6	6X6	6X6	6X8	6 X B	~	6XB	ŀ	3X6				
2. OI	01 10 8	6X6	6X6	6X6	6X0	6X0	~	BXB	4	9X6 -				
20	up T0 10	8X8	8X8	8X8	0X0	0110	Ł	01X8	٢	3X6				
	up 10 12	BXB	BXB	BXB	BXB	01X8	4	01X01	ł	3X6			r	
OVER 20	SEE NOT	E 1												
	* Mixed ** Manuf	oak or actured	equiva member	lent wi 's of er	lth a bo jutvaler	ending at stree	strength ngth may	not les by subs	s than B Effuted	50 pst. for vood	_			

CONSTRUCTION STANEARDS

[Sec. 1926.652, Table C-1.1]

.

12-6-89

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

5-242 31:3182.13

.

79

TABLE C-1, 2

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

- 45 X H + 72 psf (2 ft. Surcharge) SOIL TYPE B P a

ПСОТИ					<u>\$12</u> E	(ACT1//	L)_AND_S	SPACING_	OF_HEMBEI	15**	·····			······································
06611			CROS	<u>5 BRACE</u>	5			<u></u> !/AL	<u>65</u>	·····	<u> </u>	RIGHTS		
TRENCH	HOR12.	<u>V</u>]]	TB_OF_	<u>FRENCH</u>	(FEET)_	,	VERT.		VERT.	иххтнон	ALLOUAR	a.E. nora z	OUTAL SI	'ACHIG
(FEET)	SPACING	UP TO	UP TO	UP TO	UP TO	UP TO	SPACING	SIZE	SPACING		r	(FEET)		·
	(1 5 5 1)	4	6	9	12	15	(FEET)	(11)	(FEET)	CLOSE	2	3		
5	ир то <u>6</u>	4×6	4x6	6X6	6X6	6X6	5	6X8	5		·	286	[
то	UP TO 	<u>6x6</u>	_6X6	<u>6X6</u>	<u>6X8</u>	<u>6X8</u>	5	<u>8x10</u>	5			_286	[
10	UP TO 10	6X6	6X6	6X6	6X8	6X8	5	10x10	5			2×6		
	See Nore 1		-											
10	UP TO 6	_6X6_	<u>6X6</u>	6X6	<u>6x8</u>	<u>6x8</u>	5	<u>8x8</u>	5		286			
то	UP TO 8	<u>6X8</u>	<u>6x8</u>	6X8	<u>8x8</u>	888	<u></u> 5	10x10	5		286			
15	UP TO 10	_ <u>8x8</u> _	8X8_	_8 <u>x</u> 8	<u>8x8</u>	8X1Ω	5	<u>10x12</u>	5		286			
	See Note 1										····			
15	UP TO 6	6X8	6XB	6X8	8X8	BXB	5	8X10	5	3X6			 	
то	UP TO 8	вха	8x8	8x8	вха	<u>8x10</u>	5	10X12	5	<u> 3x6</u>				
20	UP TO 10	<u>8x10</u>	<u>8x10</u>	8X10	<u>8x10</u>	<u>10X10</u>	5	<u>12X12</u>	5	<u>3x6</u>				<u></u>
	See Note l					<u> </u>								
OVER 20	SEE NOT	e 1								•				

Mixed oak or equivalent with a bending strength not less than 850 ps1.
 Manufactured members of equivalent strength may by substituted for wood.

Occupational Salary & Health Reporter

....

[Sec. 1925.552. Table C-1.2]

80

31:2182.14

TABLE C-1.)

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS A

SOLL TYPE C P = 80 X 11 + 72 psf (2 ft. Surcharge) CONSTRUCTION STANCARDS

S-242 31:3182.15

DEPTH		SIZE (AC						SPACING	OF MEMD	ERSAA				
OF			<u>C</u> (t)	SS BRAC	ES				.		()	PRIGHTS		
TRENCH	HOR1Z.	14	DTH OF	TREBCH	(FEET)		WEDE		11111	HAXIBUU	ALLOWA	BLE HORE	ZORTAL SI	PACING
(FEET)	SPACING	up to	UP TO	UP TO	UP TO	UP TO	SPACING	SIZE	SPACING		r	(FEET)	(See Hot	<u>s_2)</u>
	(FEET)	4	6	.9	12	_15_	(FEET)	(11)	(FEET)	CLOSE				
5	UP TO 6	6X8	6X8	6X8	BXB	8X8	5	8x10	5	2X6			1	1
TO	Ú́Р ТО 8	8X8	8x8	8X8	8x8	8x10	5	10X12	5	2X6				
	UP TO	8210	8x10	8x10	8x10	10X10	5	12X12	5	216				
	See Note 1													
	UP TO	888	8x8	888	axa	8×10	5	10112	5	286		-		
10	UP TO	810	AX10	8X10	810	10110	 S	17812	5	286				
70	548										·			
15	See										·			
[Note I	[[[·	[
15	ир то <u>6</u>	8x10	exio	8X10	8x1,0	10X10	5	12212	5	386				[
то	See Note 1													
20	See Note 1		· ·											
	See Note l												,	
OVER 20	SEE NOTE	. 1					₹	· · · · · · · · · · · · · · · · · · ·		•		· · · · · · · · · · · · · · · · · · ·		

* Mixed Oak or equivalent with a bending strength not less than 850 ps]. ** Manufactured members of equivalent strength may be substituted for wood.

12-6-39

8

TABLE C-2.1

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

SOIL TYPE A P - 25 X H t 72 paf (2 ft. Surcharge)

DEPTH	1				\$12	<u>E (\$45</u>	_AUD_SPA	PACING_OF_MENDERS_**						
OF			CROS	SS BRAC	ES	<u></u>	r=		LES		U	PRICHTS		
TRENCI	RORIZ.	<u> </u>	DTH OF	TRENCH	(FEET)		VERT.	•	VERT.	пахани	L ALLOVA	BLE HORI	ZOHTAL S	PACING
(FEET)	SPACING	WP TO	OF TO	ир то	or au	UP TO	SPACING	SIZE	SPACING		ı — —	(FEET)	······	
	(FEET)		6	2	_12		(FEET)	<u> </u>	(FEET)	<u>CLOSE</u>	<u> </u>		<u> </u>	<u>8</u>
5	UP TO 6	4X4	<u>4x4</u>	<u>4x4</u>	4x4	<u>4x6</u>	4	Not Req'd	Not Reg'd			 	<u>4x6</u>	
ТО	UР ТО 8	4X4	4x4	4X4	4X6	4x6	4 ·	Not Req d	Not Req ⁺ d					4x8
	UP TO	4X6	4x6	4x6	6X6	6X6	6	8X8	4			4X6		
	UP TO	4x6	4x6	4X6	6X6	6X6	4	8X8	4				486	
10	UP TÒ 6	4X4	4x4	4X4	6X6	6X6	4	Nor Req ¹ d	Nor Reg d				4X10	
то	UP TO 8	486	4X6	4x6	6X6	6X6	4	6X8	_4		<u>4x6</u>			
15	UP TO 10	6X6	6X6	686	6X6	- 6X6	4	8X8	4	-		4X8		
	UP TO 12	6X6	6X6	6X6	6X6	6X6	4	8X10	4		4X6	·		- <u></u>
15	UP TO 6	6X6	6X6	6X6	6X6	6X6	4	6X8	4	386				
то	UP TO 8	6X6	6X6	6X6	6X6	6X6	4	4X8	- 4	386	<u>4x12</u>			
20	UP TO	6X6	6X6	6X6	6X6	6X8	4	8X10	4	<u>3x6</u>				·
	UP TO 12	6X6	6X6	6X6	6x8	6X9	4	8X12	4	3x6	4X12			
OVER 20	SEE NOTE	2 1							·				ı	

A Douglas fir or equivalent with a bending strength not less than 1500 psi, ** Manufactured members of equivalent strength may be substituted for wood.

31:3122.16

REFERENCE FLE

ß

TABLE C-2.2

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

CONSTRUCTION STANDARDS

SOIL TYPE B P = 45 X II * 72 psf (2 ft. Surcharge)

БЕРТИ					SIZĘ	(848) /	ND SPACE	UC OF 1	EHBERS A		·····		··········	
05				SS_BRACI	<u>'S</u>				,ES		<u> </u>	PRIGHTS_		
TRENCH	HORIZ.	<u> \}</u>	<u>DTH_OF</u>	TRENCH	(FEET)		VERT.		VERT.	нахтни	1 ALLOUA	BLE HORI	ZOUTAL	SPACIEG
(FEET)	SPACING	UP TO	UP TO	UP TO	UP TO	ир то	SPAC 111G	SIZE	SPACING			(FEET)		
	(FEET)	4	6	9	12	15	(FEET)	(111)	_(FEET)	CLOSE	2	3	4	6
5	UP TO 6	4x6	4X6	4X6	6X6	6X6	5	6X8	5			3X12 4X8	, 	4812
то	ир то 8	4X6	4X6	6X6	6X6	6X6	5	8X8	5		388		4X8	
10	UP TO 10	4x6	4x6	6X6	6X6	6X8	5	8X10	5			4X8		
	See Note 1	 	<u> </u>			<u> </u>								
10	UP TO 6	6X6	6X6	6X6	6X8	6X8	5	8X8	5	<u>3x6</u>	4X10			
то	UP TO 8	6X8	6X8	6X8	8X8	8X8	5	10X10	5	3X6	4x10			
15	UP TO 10 ·	6X8	6X8	8X8	8X8	8x8	5	10X12	5	3X6	4x10			
	See Note 1	}												
15	UP TO 6	6X8	6X8	6X8	6XB	BX8	5	8X10	5	4x6				
то	UP TO 8	6XB	6X8	6X8	8x8	8X8	5	10x12	5	- 4X6				
20	ир то 10	8X8	8x8	8X8	8X8	8x8	5	12X12	5 .	486				
	See Note 1												-	
OVER	SEE NOTE	: I												

* Douglas fir or equivalent with a bending strength not less than 1500 psi,

** Manufactured members of equivalent strength may be substituted for wood.

•

44 1-1

TABLE C-2.3

TIMBER TRENCH SHORING -- MINIMUM TIMBER REQUIREMENTS *

SOIL TYPE C P = 80 X II + 72 psf (2 fL. Surcharge)

BT93G					<u>S17E</u>	(\$45)	AND_SPAC	ACTING_OF_HEUBERS_FA						
OF			CRO	<u>SS BRAC</u>	ES				ĽS			PRICHTS		
	HORIZ.	<u> </u>	<u>oth of</u>	TRENCIL	(FEET)		VERT.		VERT.	HAXEDUN	ALLOVAE	LE RORL2	ONTAL SI	PACING
TRENCH	SPACING	UP TO	UP TO	UP TO	UP TO	UP TO	SPACING	512E	SPACING			(FEET)		
	(FEET)	4	_6	2	_12	15	(FEET)	<u>_(116)</u>	(FEET)	CLOSE				-
	UP TO						_						1	
5	6	6X6	6X6	6X6	6X6	888	5	8X8	5	<u> </u>				
	UP TO													
10	8	6X0	6X6	6X6	UXU	BXB	<u> </u>	10X10	<u> </u>	3X0	l	<u> </u>		
	UP TO											i		
10	10	6X6	6X6	8X8	8X8	8X8	5	10X12	5	<u>3x6</u>		<u> </u>		
	See													
	Note 1]				
	טוי דס	, inc			0.110									
10	6	080	<u>688</u>	<u>6X8</u>	<u> 888</u>	<u></u>		<u>10X10</u>	·	480				
[]	UP TO	ava	กงอ	040	040	0.4.6	د ¹	1 2 2 1 2	۲.	1.86				
то								12/12	ļ	470			ł ———	
	See										ĺ		ĺ	
15	NOCE L													
	See									1				
	note 1 ·													
	UP TO	8X8	888	AXA	8X10	- 8X10	5	10X12	5	4x6				
15	<u> </u>											} ───→		
	See									1				
TO								<u> </u>						
	Note I													· .
20	See					·								
	Note 1									1	ļ			
OVER		••••••••••••••••••••••••••••••••••••••	.	II	J			•	.		·····		B	•
20	SEE NOTE	E 1												

* Douglas fir or equivalent with a bending strength not less than 1500 pst.

** Manufactured members of equivalent strength may be substituted for wood.

BILLING CODE 4810-28-C

ĉ

REFERENCE FILE

Appendix D to Subpart P

Aluminum Hydraullo Sharing for Trenches

(a) Secre This appendix contains information that can be used when aluminum hydrautic snorng is provided as a method of protection against cave-ins in trenches that do not exceed 20 feet (6.1m) in depth. This appendix must be used when design of the aluminum hydrautic protective system cannot be performed in accordance with 1 1923:6521cl(2).

(b) Soil Classification. In order to use data presented in this appendix, the soil type or types in which the excavation is made must first be determined using the soil classification method set forth in appendix A of subpart P of part 1928.

(a) Presentation of Information.

information is presented in several forms as follows:

(1) information is presented in tabular form in Tables D-1.1, D-1.2, D-1.3 and E-1.4, Each table presents the maximum vertical and horizontal spacings that may be used with various aluminum member sizes and various hydraulic cylinder sizes. Each table contains data only for the particular soil type in which the excavation or portion of the excavation is made. Tables D-1.1 and D-1.2 are for vertical shores in Types A and B soil. Tables D-1.3 and D1.4 are for horizontal water systems in Types B and C soil.

(2) information concerning the basis of the tabular data and the limitations of the data is presented in paragraph (d) of this appendix.

(3) information explaining the use of the

tabular data is presented in paragraph (e) of this appendix.

(4) Information illustrating the use of the tabular data is presented in paragraph [I] of this appendix.

(5) Miscellaneous notations (footnotes) regarding Table D-1.1 through D-1.4 are presented in parsgraph (g) of this appendix.

(6) Figures, illustrating typical installations of hydraulic shoring, are included just prior to the Tables. The illustrations page is entitled "Aluminum Hydraulic Shoring: Typical Installations."

(c) Jasis and limitations of the data.

(1) Vertical shore rails and horizontal wales are those that meet the Section Modulus requirements in the D-1 Tables. Aluminum material is 8061-T6 or material of equivalent strength and properties.

(2) Hydraulic cylinders specifications. (1) 2inch cylinders shall be a minimum 2-inch inside diameter with a minimum safe working capacity of no less than 18.000 pounds axial compressive load at maximum extension. Maximum extension is to include full range of cylinder extensions as recommended by product manufactors.

(3) Limitation of application.
 (i) It is not intended that the aluminum hydraulic specification apply to every situation that may be experienced in the

field. These data were developed to apply to the situations that are most commonly experienced in current trenching practice. Shoring systems for use in situations that are not covered by the data in this sependix must be otherwise designed as specified in § 1928.352(c).

(ii) When any of the following conditions are present, the members specified in the Tables are not considered adequate. In this case, an alternative aluminum hydraulic shoring system or other type of protective system must be designed in accordance with § 1928.652.

(A) When vertical loads imposed on cross braces exceed a 100 Pound gravity load distributed on a one foot section of the center of the hydraulic cylinder.

(B) When surcharge loads are present from equipment weighing in excess of 20.000 pounds.

(C) When only the lower portion or a trench is shored and the remaining portion of the trench is sloped or benched unless: The sloped portion is sloped at an angle less steep than three horizontal to one vertical; or the members are selected from the tables for use at a dopth which is determined from the top of the averall brench, and not from the top of the sloped portion.

(e) Use of Tables 2-1.1, 2-1.2, 2-1.3 and D-1.4. The members of the shoring system that are to be selected using this information are the hydraulic cylinders, and either the vertical shores or the horizontal wales. When a water system is used the vertical timber sheeting to be used is also selected from these tables. The Tables D-1.1 and D-1.2 for vertical shores are used in Type A and B soils that do not require sheeting. Type B soils that may require sheeting, and Type C soils that always require sheeping are found in the horizontal wale Tables D-1.3 and D-1.4. The soil type must first be determined in accordance with the soil diassification system described in appendix A to subpart P of part 1925. Using the appropriate table, the selection of the size and spacing of the

[Appendix 0]

Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

85

members is made. The selection is based on the depth and width of the trench where the members are to be installed. In these tables the vertical spacing is held constant at four feet on center. The tables show the maximum horizontal spacing of cylinders allowed for each size of wale in the water system tables, and in the vertical shore tables, the hydrsulic cylinder horizontal spacing is the same as the vertical shore spacing.

(f) Example to Illustrate the Use of the Tables:

(1) Example 1:

A trench dug in Type A soil is 8 feet deep and 3 feet wide. From Table D-1.1: Find vertical shores and 2 inch diameter cylinders spaced 8 feet on center (0.0.) horizontally and 4 feet on center (0.0.) vertically. (See Figures 1 & 3 for typical installations.)

(2) Example 2

A trench is dug in Type B soil that does not require sheeting, 13 feet deep and 5 feet wide. From Table D-1.2: Find vertical shores and 2 inch diameter cylinders spaced 6.5 feet o.c. homionially and 4 feet o.c. vertically. (See Figures 1 & 3 for typical installations.)

(3) A trench is dug in Type B soil that does not require sheeting, but does experience some minor raveling of the trench face. The trench is 18 lest deep and 9 lest wide. From Table D=1.2: Find vertical shores and 2 linch diameter cylinder (with special oversleeves as designated by footnote =2) spaced 5.5 feet c.c. horizontally and 4 feet c.c. vertically, plywood (per footnote (§)(7) to the D=1 Table) should be used behind the shores. (See Figures 2 & 3 for typical installations.)

(4) Example 4: A trench is dug in previously disturbed Type B soil, with characteristics of a Type C soil, and will require sheeting. The trench is 13 feet deep and 12 feet wide, 8 foot horizontal spacing between cylinders is desired for working space. From Table D-1.3: Find horizontal wale with a section modulus of 14.0 spaced at 4 feet o.c. vertically and 3 inch diameter cylinder spaced at 9 feet maximum o.c. horizontally. 3 x 12 timber sheeting is required at close spacing vertically. (See Figure 4 for typical installation.)

(5) Example 5: A trench is dug in Type C soil, 9 feet deep and 4 feet wide. Horizontal cylinder spacing in excess of 9 feet is desired for working space. From Table D-1.4: Find horizontal wale with a section modulus of 7.0 and 2 inch diameter cylinders spaced at 6.5 feet a.c. horizontally. Or, find horizontal wale with a 14-2 section modulus and 2 inch diameter cylinder spaced at 10 feet a.c. horizontally. Both wales are spaced 4 feet a.c. vertically. 3 < 12 timber sheeting is required at close spacing vertically. (See Figure 4 for typical installation.)

(5) Footnotes, and general notes, for Tables -+ .D=1.1, D=1.2, D=1.3, and D=1.4.

(1) For applications other than those listed in the tables, refer to § 1926.652(c)(2) for use of manufacturer's tabulated data. For trench depths in excess of 20 feet, refer to § 1926.652(c)(2) and § 1928.652(c)(3).

(2) 2 inch diameter cylinders, at this width, shall have structural steel tube $(3.5 \times 3.5 \times 0.1375)$ oversieeves, or structural

oversidences of manufacturer's specification. extanding the full collapsed length.

(3) Hydraulie cylinders capacities. (i) 2 inch cylinders shall be a minimum 2-inch inside diameter with a safe working capacity of not less than 13.000 pounds axial compressive load at maximum extension. Maximum extension is to include full range of cylinder extensions as recommended by product manufacturer.

(ii) 34nch cylinders shall be a minimum 3inch inside diameter with a safe work capacity of not less than 50,000 pounds axial compressive load at maximum extension. Maximum extension is to include full range of cylinder extensions as recommended by product manufacturer.

(4) All spacing indicated is measured center to center.

(3) Vertical shoring rails shall have a minimum section modulus of 0.40 inch.

(6) When vertical shores are used, there must be a minimum of three shores spaced equally, horizontally, in a group.

(7) Plywood shall be 1.125 in thick softwood or 0.75 inch. thick. 14 ply, arctic white birch (Finland form). Please note that plywood is not intended as a structural member, but only for prevention of local raveling (sloughing of the trench face) between shores.

(8) See appendix C for timber specifications.

(9) Wales are calculated for simple span conditions.

(10) See appendix D. item (d), for basis and Umitations of the data.

Occupational Safety & Health Reporter

CONSTRUCTION STANDARDS

5-240 31:3182.21



TABLE D - 1.1 ALUMINUM HYDRAULIC SHORING VERTICAL SHORES FOR SOIL TYPE A

		HYDRAULIC	CYLINDERS		
DEPTH	ΜΑΧΙΜΙΙΜ	ΜΑΧΙΜΙΙΜ	wir	OTH OF TRENCH (FI	EET)
OF TRENCH (FEET)	HORIZONTAL SPACING (FEET)	VERTICAL SPACING (FEET)	UP TO 8	OVER 8 UP TO 12	OVER 12 UP TO 15
OVER 5 UP TO 10	8	· ·			
OVER 10 UP TO 15	8	4	2 INCH DIAMETER	2 INCH DIAMETER NOTE (2)	3 INCH DIAMETER
OVER 15 UP TO 20	7				
OVER 20		NOTE (I)	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	

Footnotes to tables, and general notes on hydraulic shoring, are found in Appendix D, Item (g)

Note (1): See Appendix D, Item (g) (1)

Note (2): See Appendix D, Item (g) (2)

Occupational Safety & Health Reporte

[Sec. 1925.552. Table D-1.1]

8

31:3182.22

REFERENCE FILE

TABLE D - 1.2
ALUMINUM HYDRAULIC SHORING
VERTICAL SHORES
FOR SOIL TYPE B

		HYDRAULIC	CYLINDERS		
DEPTH	махімим	MAXIMUM	wii	OTH OF TRENCH (FF	EET)
OF TRENCH (FEET)	HORIZONTAL SPACING (FEET)	VERTICAL SPACING (FEET)	UP TO 8	OVER 8 UP TO 12	OVER 12 UP TO 15
OVER 5 UP TO 10	8				
OVER 10 UP TO 15	6.5	4	2 INCH DIAMETER	2 INCH DIAMETER NOTE (2)	3 INCH DIAMETER
OVER 15 UP TO 20	5.5				
· OVER 20		NOTT! (1)	······································	9	1

CONSTRUCTION STANDARDS

Footnotes to tables, and general notes on hydraulic shoring, are found in Appendix D, Item (g)

Note (1): See Appendix D, Item (g) (1)

Note (2): See Appendix D, Item (g) (2)

58

[Sec. 1925.552, Table D-1.2]

12-5-39

The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

S-240 31:3182.00 TABLE D - 1.3 ALUMINUM HYDRAULIC SHORING WALER SYSTEMS FOR SOIL TYPE B

	WA1	LES		HŶ	(DRAULIC	CYLINDI	ERS		TIMBI	R UPRI	GITES
DEPTH				WI	отн ог ті	ÈENCH (FI	HED)		MAX.H (O)	ORIZ.SP	ACING ER)
OF TRENCH	VERTICAL SPACING	SECTION MODULUS	UP	TO 8	OVER 8	UP TO 12	OVER 12	UP TO15	SOLID	2 FF.	3 FT.
(FEET)	(FEET)	(IN ³)	HORIZ, SPACING	CYLINDER DIAMETER	HORIZ. SPACING	CYLINDER DIAMETER	BORIZ. SPACING	CYLINDER DIAMETER	SHEET		
OVER		3.5	8.0	2 IN	8.0	2 IN NOTE(2)	8.0	3 IN			
5 UP TO	4	7.0	9.0	2 IN	9.0	2 IN NOTE(2)	9.0	3 IN		-	3×12
10	1	14.0	12.0	3 IN	12.0	3 IN	12.0	3 IN			
OVER		3.5	6.0	2 IN	6.0	2 IN NOTE(2)	6.0	3 1N			
10 UP TO	4	7.0	8.0	3 IN	8.0	<u>3 IN</u>	8.0	<u>3 IN</u>		3x12	
15		14.0	- 10.0	3 IN	10.0	3 IN	10.0	3 IN			
OVER		3.5	5,5	2 IN	5.5	2 IN NOTE(2)	5.5	<u>3 IN</u>	,		
15 UP TO	4	7.0	6.0	3 IN	6.0	<u>3 IN</u>	6.0	<u>3 IN</u>	3x12		
20		14.0	9.0	3 IN	9.0	3 IN	9.0	3 IN			
OVER 20		÷	NOTE (I))							

31:3192.24

Footnotes to tables, and general notes on hydraulic shoring, are found in Appendix D, Item (g)

Notes (1): See Appendix D, item (g) (1)

Notes (2): See Appendix D, Item (g) (2)

* Consult product manufacturer and/or qualified engineer for Section Modulus of available wales.

REFERENCE FILE

CONSTRUCTION STANDARDS

TIMBLE UPRIGHTS MAX.HORIZ SPACING (ON CENTER)

[Sec. 1925.55] Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, O.C. 20017

12-5-89

[Sec. 1926.652. Table D-1.4]

Footnotes to tables, and	general notes on h	ydraulic shoring.	are found in A	ppendix D, Item (g)
--------------------------	--------------------	-------------------	----------------	---------------------

Notes (1): See Appendix D, item (g) (1)

Notes (2): See Appendix D, Item (g) (2)

* Consult product manufacturer and/or qualified engineer for Section Modulus of available wales.

	TAB ALUMINUM HY WALE FOR SO	LE D - 1.4 DRAULIC SHORING R SYSTEMS DIL TYPE C
WALES		HYDRAULIC CYLINDERS
	۷	WIDTH OF TRENCH (FEET)
VERTICAL SECTION		OVER 8 UP TO 12 OVER 12 UP TO

TRENCH	SPACING	MODULUS	UP '	ro 8	OVER 8	UP TO 12	OVER 12	UP TO 15	SOLID	2 IFT.	3 FT.
(FEET)	(FEET)	(IN')	HORIZ, SPACING	CYLINDER DIAMETER	HORIZ, SPACING	CYLINDER DIAMETER	HORIZ, SPACING	CYLINDER DIAMETER	SHEEL		
OVER		3.5	6.0	2 IN	6.0	2 IN NOTE(2)	6.0	3 IN		·	
5 UP TO	4	7.0	6.5	2 IN	6.5	2 IN NOTE(2)	6.5	3 IN	3x12	·	
10		14.0	10.0	3 IN	10.0	3 IN	10.0	3 IN			
OVER		3.5	4.0	2 IN	4.0	2 IN NOTE(2)	4.0	3 IN	3x12		
10 10	4	7.0	5.5	3 IN	5.5	3 IN	5.5	<u>3 IN</u>		`	
15		14.0	8.0	3 IN	8.0	3 IN	8.0	3 IN		1	
OVER		3.5	3.5	2 IN	3.5	2 IN NOTE(2)	3.5	3 IN	3x12		
15 ПР ТО	4	7.0	5.0	<u>3 IN</u>	5.0	<u>3 IN</u>	5.0	<u></u>			
20		14.0	6.0	3 IN	6.0	3 IN	6.0	3 IN			
OVER 20			NOTE (1)						1		

5-242 31:3182.25

DEPTH OF

Appendix E to Subpart P-Alternatives to Timber Shoring

Figure 1. Aluminum Hydraulic Shoring -



Figure 2. Pneumatic/hydraulic Shoring





· ·



[Appendix E]

92

Figure 3. Trench Jacks (Screw Jacks)



Published by The BUREAU OF NATIONAL AFFAIRS, INC., Washington, D.C. 20037

93

[Appendix E]

31:3182.28

Appendix F to Subpart P-Selection of Protective Systems

The following figures are a graphic summary of the requirements contained in subpart 9 for excavations 20 feet or less in depth. Protective systems for use in excavations more than 20 feet in depth must be designed by a registered professional engineer in accordance with § 1928.652 (b) and (c).



FIGURE 1 - PRELIMINARY DECISIONS

[Appendix F]

94

CONSTRUCTION STANDARDS



[Appendix F]

REFERENCE FILE



FIGURE 3 - SHORING AND SHIELDING OPTIONS

[Appendix F]



SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 1 of 5

4.04 Hazard Communication

A. Introduction

OBC Technical Services. Inc. is committed to providing a safe and healthy work environment for all employees through compliance with the Federal Hazard Communication Standards. It is recognized that certain work activities and procedures require the use of chemicals which have potentially hazardous properties. When using such chemicals, it is important that workers are aware of the identity and potentially hazardous properties of the chemicals. Taus, this section represents the Written Program and has been established to summarize the components of the Hazard Communication Program instituted by OBC Tech.

This Written Hazard Communication Program is available to all employees, their designated representatives, and representatives of the Federal Occupational Safety and Health Administration. This Program has been developed in compliance with the federal regulations contained in 29 CFR 1910.1200 and 29 CFR 1926.59. Further information on the program may be obtained from the Safety Coordinator in the Syracuse office.

B. <u>Materials Exempted</u>

The following materials are exempt from ALL aspects of the Federal Hazard Communication regulations:

- 1. Tobacco or tobacco products;
- Wood or wood products;
- 3. Articles (defined as manufactured items, intended for use in specific shape or design, which do not release, or otherwise result in exposure to a hazardous chemical);
- 4. Food, drugs, cosmetics, or alcoholic beverages in a retail establishment packaged for sale to consumers;
- 5. Any consumer product or hazardous substance (as governed by the Consumer Product Safety Commission) which is used in the workplace in the same manner as normal consumer use, which results in a frequency and duration of exposure which is not greater than exposures experienced by consumers;
- Drugs in solid, final form for direct administration to the patient (i.e., tablets or pills); and
 Hazardous wastes, as defined under the Solid Waste Disposal Act and the Resource Conservation and Recovery Act.

C' Labeling program

ii.

1. No hazardous chemicals or chemical products will be used or accepted for use by OBG Tech, unless labeled with at least the following information:

i.	Identity of the hazardous chemical(s);	
----	--	--

- Appropriate hazard warnings;
- iii. Name and address (and phone number, if available) of the chemical manufacturer, importer, distributer, supplier, or other responsible party.

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 2 of 5

All OBC Tech employees are responsible for compliance with these requirements.

- 2. The identity of the chemical that appears on the manufacturer's label will be the same name to identify the chemical on the List of Hazardous Chemicals, and the MSDS for that substance.
- 3. All labels will be legible, in English, and prominently displayed on the container.
- 4. In the situation of a stationary container, the label may be replaced by a sign, placard, process sheet, batch ticket, or other means to convey the identity of the hazardous chemical and the appropriate hazard warnings. If these other forms of warning are used, they will be readily accessible to employees in their work area throughout each work shift.
- 5. The Standard does not require a label to be placed on portable containers into which hazardous chemicals are transferred from labeled containers, and which are intended only for the immediate use of the employee who performs the transfer. However, by labeling the portable container appropriately, accidental misuse of the chemical by other employees may be prevented.
- 6. No label is to be defaced or removed unless the container is immediately marked with the required information. Any container without an appropriate label should be reported immediately to the job-site Supervisor.
- 7. Exemptions from these labeling requirements are as follows:
 - i. Pesticides subject to the Federal Insecticide. Fungicide and Rodenticide Actu ii. Food, food additives, color additives, drugs, cosmetics, or medical or
 - veterinary devices, as governed by the Food and Drug Administration:
 - iii. Alcoholic beverages, wine or malted beverages intended for non-industrial use;
 - iv. Any consumer product or hazardous substance as defined by the Consumer Product Safety Commission;

D. <u>Comprehensive List of Hazardous Chemicals</u>

Every chemical or product mixture of chemicals in any of the OBG Tech jobsites which is included in any of the four following lists of hazardous chemicals is listed in the List of Hazardous Chemicals for each jobsite. This List will be kept and maintained by the job-site Supervisor in a central location at each jobsite, such as the field office trailer. The four defining lists of hazardous chemicals are as follows:

- 1. Those compounds with designated OSHA Permissible Exposure Limits, as listed in 29 CFR 1910, Subpart Z ("Toxic and Hazardous Substances");
- 2. Those compounds listed in "Threshold Limit Values for Chemical Substances in the Work Environment" as published in the latest edition by the American Conference of Governmental Industrial Hygienists;
- 3. The latest edition of the "Annual Report on Carcinogens", published by the National Toxicology Program;
- 4. The Monographs published by the International Agency for Research on Cancer.

If a product contains a mixture of chemicals, it is considered to be hazardous if it contains 1% or more of any of the chemicals contained in any of the four lists referenced above. In the case of

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 3 of 5

carcinogens, a mixture is considered to be hazardous if it contains 0.1% or more of any of the carcinogens within the lists referenced above.

The identity of each chemical, as noted in the List of Hazardous Chemicals, will be the same as the identity that appears on the container label and on the MSDS. A copy of the List of Hazardous Chemicals will be located at each Field Office along with a corresponding MSDS file for each compound on the list. Updating of the Comprehensive List of Hazardous Chemicals will be done regularly, upon acquisition of new materials and/or elimination of "old" materials. Each job-site Supervisor or his designate within OBG Tech will be responsible for updating each jobsite's list and MSDS file.

E. <u>Material Safery Data Sheets</u>

Copies of Material Safety Data Sheets for all hazardous chemicals to which employees may be exposed will be kept at a central location at each jobsite, such as the field office trailer and are readily accessible to employees in the work area during each work shift. The job-site Supervisor or his designate is responsible for maintaining and updating the file of Material Safety Data Sheets.

Each Material Safety Data Sheet shall be in English and shall contain the following information:

- 1. The identity used on the label.
- Physical and chemical characteristics of the hazardous chemical.
- 3. Physical and health hazards of the hazardous chemical.
- 4. The primary routes of entry.
- 5. The OSHA permissible exposure limit, ACGIH threshold limit value, and any other exposure limit used or recommended by the chemical manufacturer, importer, or employer preparing the material safety data sheet, where available.
- 6. Whether the chemical is found to be a potential carcinogen.
- 7. Precautions for safe handling.
- 8. Control measures, such engineering controls, work practice, or appropriate personal protective equipment.
- Emergency and first aid procedures.
- 10. The date of the preparation of the material safety data sheet or the last change to it; and
- 11. The name, address, and telephone number of the responsible party for preparation or distribution of the material safety data sheet.

F. Employee Information and Training

Employees working with or potentially exposed to hazardous chemicals appropriately informed and trained concerning the potential hazards of the materials to which they may be exposed at the time of their initial work assignment. Training will be appropriately supplemented whenever a new hazard is introduced into the work area.

Employee information includes:

- 1. The requirements of the Hazard Communication Standard;
- 2. Any job-related activities involving hazardous chemicals;

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 4 of 5

The locations and availability of the Written Hazard Communication Program, the List(s) 3. of Hazardous Chemicals, and the Material Safety Data Sheet File.

Employee training includes:

- Methods and observations that may be used to detect the presence or release of a hazardous 1. chemical in the work area. Such procedures include monitoring, visual appearance and/or odor recognition, symptoms, etc.;
- 2. The physical and health hazards of the chemicals in the work area.
- Measures employees can take to protect themselves from such hazards. This includes 3. specific procedures instituted by OBG Tech to protect employees from exposure to hazardous chemicals such as appropriate work practices, emergency procedures, and personal protective equipment suitable for use.
- The details of the Hazard Communication Program developed by OBG Tech, including an 4. explanation of the labelling system and the Material Safety Data Sheets, and how employees can obtain and use the appropriate hazard information.

The Safety Coordinator is responsible for overseeing the employee information and training program. All training is documented and is filed in the Syracuse office.

G. Provisions for Non-routine Tasks and Sub-Contractors

Before any non-routine task is performed that could involve exposure to hazardous chemicals, the supervisor for the area involved will evaluate the task for appropriate work practice procedures, potential chemical hazards of the task, safety and protective measures are reviewed and understood by the employee.

Prior to beginning work on OBG Tech jobsite, all sub-contractors will be given a copy of the Written Hazard Communication Program. The particular hazards associated with the work area(s) will be identified, along with all appropriate MSDSs for the materials to be used and/or encountered. Sub-Contractors are responsible for training their own employees. The job-site Supervisor or Foreman is responsible for verifying that training has been performed by the sub-contractors.

H. Multi-Employer Workplaces

OBG Technical Services, Inc. employees may use or store hazardous chemicals at a workplace in such a way that the employees of another employer or employers may be exposed. If this occurs, then the following steps must be taken.

- Prior to the beginning of work on a multi-employer site, the Supervisor will notify all of 1. the other contractors and sub-contractors where the MSDSs for all the hazardous chemicals, brought on-site by OBG Tech, are available.
- The Supervisor will inform the other employers of precautionary measures which need to 2. be taken to protect employees from hazardous chemicals used by OBG Tech employees during normal operating conditions and in foreseeable emergencies. This communication will also include a brief description of the container labeling system used by OBG Tech. 3.
 - OBG Technical Services, Inc. will request similar information from all the other contractors

SECTION: 4.04 SUPERSEDES: None DATE: May 14, 1992 REVIEW: January 3, 1993 PAGE: 5 of 5

working on this site.

4.

OBG Tech employees will receive training in the hazards of new chemicals that they will use, and the chemicals to be used by other contractors at a tool box safety meeting to be held prior to the beginning of work.

Appendix I

:

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 1 of 6

3.05 Confined Spaces

A. Introduction

OSHA has a standard, 29 CFR 1910.146, for permit-required confined spaces. The standard includes provisions for testing and entering confined spaces. The employee who enters a confined space may be subject to multiple hazards. The purpose of this section is to outline procedures to reduce these hazards.

B. Definitions

"Permit-required confined space" means a confined space (see definition below) that has one or more of the following characteristics:

- 1. Contains or has a potential to contain a hazardous atmosphere (see definition below).
- 2. Contains a material that has the potential for engulfing an entrant.
- 3. Has an internal configuration such that an entrant could be trapped or asphyxiated by inwardly converging walls, or by a floor which slopes downward and tapers to a smaller cross-section, or
- 4. Contains any other recognized serious safety or health hazard.

"Confined Space" means a space that:

- 1. Is large enough and so configured that an employee can bodily enter and perform assigned work; and
- 2. Has limited or restricted means for entry or exit (for example, tanks, vessels, silos, storage bins, hoppers, vaults, and pits are spaces that may have limited means of entry): and
- 3. Is not designed for continuous employees occupancy.

"Hazardous Atmosphere" means an atmosphere that may expose employees to the risk of death, incapacitation, impairment of ability to self-rescue, injury, or acute illness from one or more of the following causes:

- 1. Flammable gas, vapor, or mist in excess of 10 percent of its lower flammable limit (LFL);
- 2. Airborne combustible dust at a concentration that meets or exceeds its LFL (Note: this concentration may be approximated as a condition in which dust obscures vision at a distance of 5 feet (1.52 meters) or less);
- 3. Atmospheric oxygen concentration below 19.5% or above 23.5%;
- 4. Atmospheric concentration of any substance for which a dose or a permissible exposure limit is published in 29 CFR 1910 Subparts G or Z and which could result in employee exposure in excess of its dose or permissible exposure limit (Note: An atmospheric concentration of any substance that is not capable of causing death, incapacitation, impairment or ability to self-rescue, injury, or acute illness due to its health effects is not covered).
- 5.
- Any other atmospheric condition that is immediately dangerous to life or health.

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 2 of 6

Note: For air contaminants for which OSHA has not determined a dose or permissible exposure limit, other sources of information, such as Material Safery Data Sheets that comply with the Hazard Communication Program, published information, and internal documents can provide guidance in establishing acceptable atmospheric conditions.

"Non-permit confined space" means a confined space that does not contain or, with respect to atmospheric hazards, have the potential to contain any hazard tapable of causing death or serious physical harm.

- C. Confined Space Entry Program
 - Coordination with Host Employer (Owner) 1.

The Supervisor/Foreman in charge of entry will:

- Obtain any available information regarding permit space hazards and entry operations а. from the host employer (Owner);
- Coordinate entry operations with the host employer (Owner), when both host employer Ъ. personnei and OBG Tech personnei will be working near permit spaces; inform the host employer (Cwner) of the permit space program that OBG Tech personnel
- c. will follow and of any hazards confronted or created in permit spaces, either through a debriefing or during the entry operation.
- Foilow the Host Employers (Owners) confined space entry program whenever possible. d.
- 2. Determine classification of Confined Space

Should OBG Tech employees need to enter a confined space that has not been classified, then the following procedures will be used:

- Determine whether the space meets the definition of a confined space as defined in Part 2 B of this Section.
- If the space meets the definition of a confined space, then determine whether it meets Ъ. the definition of a permit-required confined space. If the confined space does not meet the definition of a permit-required confined space, employees may enter the space after the appropriate safety pre-cautions have been taken.
- If the confined space requires a permit, refer to Part 3 of this section for the procedures c. to be followed under these conditions.
- Preparations for Permit-Required Confined Spaces 3.

The Supervisor/Foreman in charge of entry is responsible for:

- Implementing the measures necessary to prevent unauthorized entry. 2.
- Ъ.
- Identifying and evaluating the hazards before employees enter the permit space. Specifying acceptable entry conditions on the "Safety, Work, Flame, Tank Entry, & Gas c. Test Permit" (An example is attached. Four-copy pressure sensitive forms are available from the Syracuse office).
- Isolating the permit space and following lock-out/tag-out procedures as necessary. d.
- Purging, inerting, flushing or ventilating the permit space as necessary to eliminate or e. control atmospheric hazards.
- f. Providing signs and pedestrian, vehicle, or other barriers as necessary to protect entrants from external hazards.
- Verifying that conditions in the permit space are acceptable for entry throughout the g. duration of an authorized entry.
- Confirming that the necessary equipment is available and that employees are trained to **b**. use it as outlined on the permit.
- Performing testing as described in Section 6. i.
- Verifying that at least one attendant is stationed outside the permit space into which entry Ĵ∙ j

OBG TECHNICAL SERVICES, INC.SECTION: 3.05HEALTH AND SAFETY MANUALSUPERSEDES: 5/14/92DATE: April 12, 1993DATE: April 12, 1993REVIEW: March 31, 1994PAGE: 3 of 6

is authorized for the duration of the entry.

- Venifying that the emergency rescue team has been notified (see Section 10).
- Coordinating entry operations when employees of more than one employer are simultaneously as authorized attendants in a permit space, so that the employees of one employer do act endanger the employees of any other employers.
- m. Prior to entry, signing the permit to authorize entry.
- a. Posting the completed permit at the entry portal or by any other equally effective means, so that the entrants can confirm that pre-entry preparations have been completed.
- o. When a ladder is required to enter the confined space, the ladder must be tied securely and must not be removed while anyone is in the confined space.
- p. Adequate lighting of an approved safety type must be provided.
- q. All necessary safety equipment to be used by the person entering the confined space as well as for the safety watch will be checked.
- r. Emergency procedures will be reviewed.
- s. Smoking is prohibited inside of and within twenty feet of the confined space.
- t. Spark proof hand tools and explosion proof equipment will be used.
- u. If weiding is to be performed in the confined space that previously or now contain combusticies, all residues, including dry scale or sediment must be removed. If it is not possible to remove all combustible materials, they must be covered with a noncombustible blanket.
- At least one 20 lb. ABC multi-purpose fire extinguisher must be available for instant use in a confined space containing flammable gases or vapors.
- w. Each person involved shall be trained on the hazards, as outlined in Section 4, how to recognize the hazards, and how to protect themselves from the hazards. Additionally, the attendants will be training on the use of rescue equipment and other duties outlined Section 3.

4. Training

k. 1.

Each employee who is required to enter a permit-required confined space will be trained in the safe performance of duties.

a. Training will be provided to each affected employee:

(1) before an employee is assigned duties which require entry into permit-required confined spaces,

(2) whenever there is a change in permit space operations that presents a hazard about which an employee has not previously been trained, or

(3) whenever there is reason to believe either that there are deviations from the permitrequired confined space entry procedures or that there are inadequacies in the employee's knowledge or use of these procedures.

- b. The training will be documented by the employees name, the signatures or initials of the trainers, and the dates of training. This documentation will be filed in the employees safety file and is available for inspection.
- 5. Procedures during entry
 - a. The duration of the permit may not exceed the time required to complete the assigned task or job identified on the permit.
 - b. The Foreman/Supervisor in charge of entry will terminate and cancel the entry permit when:

(1) Entry operations covered by the entry permit have been completed; or

(2) A condition that is not allowed under the entry permit arises in or near the permit space.

6. Procedures for Atmospheric Testing

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 4 of 6

a. Before an employee enters that space, the internal atmosphere will be tested, with a calibrated direct-reading instrument to determine if acceptable entry conditions exist, for the following conditions in the order given:

(1) Oxygen content

- (2) Flammable gases and vapors
- (3) Potential toxic air contaminants
- b. If isolation of the space is infeasible because the space is large or is part of a coatinuous system, such as a sewer, pre-entry testing will be performed to the extent feasible before entry is authorized and, if entry is authorized, entry conditions will be continuously monitored in the areas where authorized entrants are working.
- c. Test or monitor the permit space as necessary to determine if acceptable entry conditions are being maintained during the course of the entry operations.
- d. Evaluation and interpretation of data should be done or reviewed by the Safety Coordinator.
- e. When monitoring for entries involving a descent into atmospheres that may be stratified, the atmospheric envelope should be tested a distance of 4 feet in the direction of travel and to each side. If a sampling probe is use, the entrant's rate of progress should be slowed to accommodate the sampling speed and detector response.

7. Authorized Eatrants

The Foreman/Supervisor in charge of entry is responsible for confirming that all authorized entrants:

- a. Know the hazards they face during entry, including information on the mode, signs or symptoms, and consequences of exposure;
- b. Properly use equipment as required by this section;
- c. Communicate with the attendant as necessary to enable the attendant to monitor entrants status and to enable the attendant to alert entrants of the need to evacuate the space;
- d. Alert the attendant whenever:

(1) The entrant recognizes any warning sign or symptom of exposure to a dangerous situation;

(2) The entrant detects a prohibited condition;

Exit from the permit space as quickly as possible whenever:

(1) An order to evacuate is given by the attendant or the entry supervisor,

(2) The entrant recognizes any warning sign or symptom of exposure to a dangerous situation

- (3) The entrant recognizes a prohibited condition, or
- (4) An evacuation alarm is activated.

8. Attendants

ĉ.

The Foreman/Supervisor is charge of entry is responsible for confirming that attendants:

- a. Know the hazards that may be faced during entry, including information on the mode, signs or symptoms, and consequences of exposure;
- b. Is aware of possible behavioral effects of hazard exposure in authorized entrants;
- c. Initial entry will be accomplished with an attendant stationed outside for the purpose of immediate assistance.
- d. The designated attendant will never enter into the permit space for the purpose of an attempt to rescue the entrants.

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 5 of 6

- e. Attendants will use any rescue equipment provided for their use and perform other rescue and emergency duties, without entering the permit space.
- f. The attendant will be responsible for continuously maintaining an accurate count of all authorized entrants in the space.
- g. Communication between attendants and entrants will be maintained continuously during entry.
- h. Attendants will order authorized entrants to evacuate a space immediately when:
 - (1) The attendant observes a prohibited condition.
 - (2) The attendant detects behavioral effects hazard exposure.
 - (3) The attendant detects a situation outside the space which could endanger entrants.
 - (4) The attendant detects an uncontrolled situation within the permit space.

(5) The attendant is monitoring entry in more than one permit space and must focus attention on the rescue of entrants from one of those spaces.

(6) The attendant must leave the work station.

(7) If the attendant cannot effectively and safely perform all the duties required under this section.

- i. Attendants will summon rescue and other emergency services as soon as the attendant determines that the authorized entrants may need assistance to escape from the confined space hazards.
- j. Attendants will warn unauthorized persons away from the space, request that unauthorized persons exit immediately if they have entered the space, and inform authorized entrants if unauthorized persons have entered the space.
- k. Attendants will perform no duties that might interfere with the amendant's primary duty to monitor and protect the authorized entrants.
- 9. Supervisor/Foreman

Employees of OBG Tech authorizing or in charge of entry will:

- a. Know the hazards that may be faced during entry, including information on the mode, signs or symptoms, and consequences of exposure;
- b. Determine that the entry permit is properly filled out.
- b. Determine that the entrants and attendants are properly trained.
- c. Determine that necessary procedures, practices and equipment for safe entry are in place,
- including verifying rescue services are available and that the means of summoning them are operable.
- d Remove unauthorized individuals who enter or attempt to enter the permit space during entry operations.
- e. Periodically monitor to determine that confined space operations remain consistent with the terms of the entry permit and that acceptable entry conditions are present.
- f. Cancel authorization and terminate entry whenever entry conditions are not present.
- g. Take the necessary measures for concluding an entry operation, such as closing off a
- permit space and cancelling the permit, once the authorized work has been completed.
- h. The supervisor/foreman in charge of authorizing the confined space entry may also be an entrant or an attendant.
- 10. Rescue and Emergency Services
 - a. Arrangements must be made prior to entry under which a rescue team will respond to a request for rescue activities. The outside rescue team will be informed of the hazards they may confront when called to the rescue.
 - b. To facilitate non-entry rescue, retrieval systems or methods will be used whenever an authorized entrant enters a permit-required confined space, unless the retrieval equipment would increase the overall risk of entry or would not contribute to the rescue of the

SECTION: 3.05 SUPERSEDES: 5/14/92 DATE: April 12, 1993 REVIEW: March 31, 1994 PAGE: 6 of 6

entrant. Retrieval systems will meet the following requirements:

(1) Each authorized entrant will use a chest or full body harness, with a retrieval line attached at the center of the entrants back near shoulder level, or above the entrant's head. Wristlets may be used in lieu of the chest or full body harness if it can be demonstrated that the use of a chest or full body harness is infeasible or creates a greater hazard and that the use of wristlets is the safest and most effective alternative.

(2) The other end of the retrieval line will be attached to a mechanical device or fixed point outside the permit space in such a manner that rescue can begin as soon as the rescuer becomes aware that rescue is necessary. A mechanical device will be available to retrieve personnel from vertical type permit-required confined spaces more than 5 feet deep.

If an injured entrant is exposed to a substance for which a Material Safety Data Sheet (MSDS) or other similar written information is required to be kept at the worksite, that MSDS or written information will be made available to the medical facility treating the exposed entrant.

D. <u>Records</u>

Permits will be filed with other pertinent project health and safery information, as well as with the Safery Coordinator.

F. Evaluation of Program

c. -

The Permit-Required Confined Space Entry Program will be reviewed once per year, using the cancelled permits and revise the program as necessary to confirm that employees participating in entry operations are protected from permit space hazards.

G. Exception to Permit Requirements

If the only hazard posed by the confined space is an actual or potential hazardous atmosphere and it can be demonstrated that continuous forced air ventilation alone is sufficient to maintain that confined space for safe entry, then a permit is not required under the regulations. However, it is OBG Tech's policy to require a permit for entries that meet the definition of a "permit-required confined space" (see Section B). If a project requires regular entry into a confined space that does not require a permit under the regulations, the Safety Coordinator must be contacted and the permit waiver will be considered on a project by project basis. Special procedures to be followed under these conditions will be developed by the Safety Coordinator.

OEG TECHNICAL SERVICES INC. SAFETY, WORK, FLAME, TANK ENTRY & GAS TEST PERMIT

PROJECT

IPECJ.NUMEER

LOCATION OF WORK: PLANT

IBUILDING

FLOOR

DESCRIPTION OF WORK TO BE PERFORMED:

SPECIAL SAFETY PRECAUTIONS TO BE OBSERVED:

HAZARD COMMUNICATION Work crew supervisor and all work crew personel have been notified of chemicals ans physical hazards in the work area. MSOS location, safety equipment required and emergency instructions. Work personnel have been instructed to report any unsale or unusual conditions.

Salety Equipment required: :	penuper http://wowno		_
Gozgies	Ventilation Equip.	Salety Beit	
Face Shield	Water -use	Plastic Aoron	
Nitroe or vinvi Gloves	Vinvi Suit	Uncoated Twee Suit	
Ruzder Boots	Acic Suit I	Coated Tyvek Suit	_
Hait-lace Resolvator	Air-ine Respirator	Ladeer/Scafford	
Full-face Respirator	Cartridge Type	Non-sparking Tools	
GR	Fire Exanguisher	Vaporproof Ext. Light	
Leexeut	Grounding Equipment)	jCther	-
Communication Equip:	Barriers & Signs	Cther	

CHECK EACH QUESTION: YES, NO CRINCT APPLICABLE	I YES	NC	I NA
* Have valves been properly set. locked and tagged?	1	ł	i
Has electric been locked boen? Test local starter button.	I	i	i
Has bloing been disconnected or blanked off?	r t	!	i
* Has vessel/bibling been drained oteaned or purged?	1	1	1
Is Gas or Air test necessary? Cont? Every hrs.	1	1	1
* Can sparks ignite material in vicinity sewers, lower floors?	1		1
* Is adiacent equipment sale?	1	1	
* is a stand-by fire watch required? If so, present?	1	1	4
Tests have been conducted for Flammadie Vapors?			1
Tests have been conduted for Cxygen?	1		1
* Tests for Toxic Gases have been conducted and approved?	1	1	<u> </u>
* Has a pre-entry briefing been conducted?	1	1	
* Have Safety Belt and Hamess been provided for work?	Ī	1	
* Has OSHA approved shoring been provided for work?	<u> </u>	1	1

Contined Scace Entry
Potential mazaros:
Entry Supervisor:
Amendantis):
<u></u>
Rescue Procedures (Equipment to use and
Numpers to Cail):
Monitoring Procedures:
Communication Procedures:
Communication Freesedies.

# Tags	1	Air Tests	PEL	Required?	PreEntry	Initial	1 nour	2 hours	4 hours	6 hours	8 hours
T	-	Oxygen	19.5 - 23%	4		1		1	1	1	I
ī		LEL	10 45	1		1		1		1	1
1	i	00	35 opm	1	1	1	1	1		1	1
1	-	H2S	10 opm	1		1	1	J	1	!	1
1		Voiatiles	500mSTE	<u>1</u>		T -		1	1	1	1
1		Cthers	1	1		1	1	1	1	1	
	# Tacs 	# Tags 	# Tags Air Tests I Oxygen I I I I I I I I I I I Voiatiles I Cthers	# Tags Air Tests PEL I I Oxygen 19.5 - 23% I I IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO % I IO % IO % IO %	# Tags Air Tests PEL Required? I Cxygen 19.5 - 23% I I I CO 35 opm I I I CO IS opm I I H2S 10 opm I I I Voiatiles IsomsTEU I I	# Tags Air Tests PEL Fequired? PreEntry I Oxygen 19.5 - 23% I I LEL 10 % I I CO 35 opm I I F2S 10 opm I Voiatiles 5pomSTEU I	# Tags Air Tests PEL Required? PreEntry Initial I Oxygen 19.5 - 23% Image: Second Secon	# Tags Air Tests PEL Required? PreEntry Initial 1 hour I Oxygen 19.5 - 23% I	# Tags Air Tests PEL Required? PreEntry Initial 1 hour 2 hours I Oxygen 19.5 - 23% I <td># Tags Air Tests PEL Required? PreEntry Initial 1 hour 2 hours 4 hours I Oxygen 19.5 - 23% Image: Second Sec</td> <td># Tags Air Tests PEL Required? PreEntry Initial 1 hour 2 hours 4 hours 6 hours I Oxygen 19.5 - 23% I I I I I I IO. 10.% I I I I I I I IO. 10.% I I I I I I I IO. 10.% I I I I I I I IO. 10.0pm I I I I I I I Voiatiles 5pomSTEU I I I I I I IO. IO. IO. IO. IO. IO. IO.</td>	# Tags Air Tests PEL Required? PreEntry Initial 1 hour 2 hours 4 hours I Oxygen 19.5 - 23% Image: Second Sec	# Tags Air Tests PEL Required? PreEntry Initial 1 hour 2 hours 4 hours 6 hours I Oxygen 19.5 - 23% I I I I I I IO. 10.% I I I I I I I IO. 10.% I I I I I I I IO. 10.% I I I I I I I IO. 10.0pm I I I I I I I Voiatiles 5pomSTEU I I I I I I IO. IO. IO. IO. IO. IO. IO.

Authorizations							
Owner:					PERMIT VALIDATION	(Valid for one shift only)	:
Agent:					FROM: DATE	TIME	
Subcontractor:					TO: DATE	TIME	
Work Completed?	iVee	1NIG	INFA	Commenter			/-:

work Completed?	IYes	INO	INA	Comments:		
Lockout/Tagout Protec	tion has	been rei	noved a	nd equipment (can be placed into service	Yes
Protection Removed B	y:			Time:	Date:	

END

OF

DOCUMENT